

Bayesian control of Markov chains

Citation for published version (APA): Hee, van, K. M. (1978). *Bayesian control of Markov chains*. [Phd Thesis 1 (Research TU/e / Graduation TU/e), Mathematics and Computer Science]. Stichting Mathematisch Centrum. https://doi.org/10.6100/IR12881

DOI: 10.6100/IR12881

Document status and date:

Published: 01/01/1978

Document Version:

Publisher's PDF, also known as Version of Record (includes final page, issue and volume numbers)

Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

 The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.tue.nl/taverne

Take down policy

If you believe that this document breaches copyright please contact us at:

openaccess@tue.nl

providing details and we will investigate your claim.

BAYESIAN CONTROL OF MARKOV CHAINS

K.M. VAN HEE

BAYESIAN CONTROL OF MARKOV CHAINS

BAYESIAN CONTROL OF MARKOV CHAINS

PROEFSCHRIFT

TER VERKRIJGING VAN DE GRAAD VAN DOCTOR IN DE TECHNISCHE WETENSCHAPPEN AAN DE TECHNISCHE HOGESCHOOL EINDHOVEN, OP GEZAG VAN DE RECTOR MAGNIFICUS, PROF. DR. P. VAN DER LEEDEN, VOOR EEN COMMISSIE AANGEWEZEN DOOR HET COLLEGE VAN DEKANEN IN HET OPENBAAR TE VERDEDIGEN OP VRIJDAG 10 MAART 1978 TE 16.00 UUR

DOOR

KEES MAX VAN HEE

GEBOREN TE DEN HAAG

Ť.a

1978

Mathematisch Centrum, Amsterdam

Dit proefschrift is goedgekeurd door de promotoren Prof. dr. J. Wessels en

Dr. F.W. Steutel

CONTENTS

1.	INTRODUCTION	
1.1	Historical perspective	1
1.2	Informal description of the model	4
1.3	Summary of the following chapters	8
1.4	Notations, conventions and prerequisites	10
2.	THE MODEL AND THE PROCESS OF POSTERIOR DISTRIBUTIONS	
2.1	The Bayesian control model	19
2.2	Posterior distributions	26
2.3	Limit behaviour of the posterior distributions	33
з.	THE EQUIVALENT DYNAMIC PROGRAM AND OPTIMAL REWARD OPERATORS	
3.1	Transformation into a dynamic program	43
3.2	A class of optimal reward operators	53
3.3	Miscellaneous results for the Bayesian control model	65
4.	BAYESIAN EQUIVALENT RULES AND THE AVERAGE-RETURN CRITERION	
4.1	Bayesian equivalent rules and other approaches	71
4.2	Optimal strategies for the average-return criterion	74
5.	BAYESIAN EQUIVALENT RULES AND THE TOTAL-RETURN CRITERION	
5.1	Preliminaries and the independent case	91
5.2	Linear system with quadratic costs	98
5.3	A simple inventory control model	108
6.	APPROXIMATIONS	
6.1	Bounds on the value function and successive approximations	121
6.2	Discretizations	138
7.	COMPUTATIONAL ASPECTS AND EXAMPLES	
7.1	Algorithm for models where I is a singleton	149
7.2	Algorithm for models with known transition law	157
	except for one state	
7.3	Numerical examples	161

CONTENTS (continued)

,

APPENDIX A. RESULTS FROM ANALYSIS	175
APPENDIX B. REMARKS ON THE MINIMAX CRITERION	179
REFERENCES	183
SAMENVATTING	189
CURRICULUM VITAE	193

1. INTRODUCTION

In this monograph we study the control of Markov chains with incompletely known transition law. The Bayes criterion, which is used explains the name of the monograph. We start this chapter with a short historical overview of the problem field (section 1.1), In section 1.2 we give an informal description of the model we are dealing with.

Then we summarize the contents of the following chapters (section 1.3). We conclude this chapter with a summary of notations and prerequisites (section 1.4).

1.1 Historical perspective

After A. Wald founded (statistical) sequential analysis, it was R. Bellman who recognized that the technique of backward induction, which is frequently used in sequential analysis, is also applicable to a wide range of non-statistical sequential decision problems (cf. [Wald (1947)], [Bellman (1957)]). Bellman formalized the technique and called it *dynamic programming*. In [Howard (1960)] the first extensive treatment is found on the relations between dynamic programming and the control of Markov chains. Independently, in [Shapley (1953)] sequential control problems concerning Markov chains are studied, using a game theoretic formulation. Later on, in [Blackwell (1965)] and [Derman (1966)] the results of Howard are refined and extended for the criterion of *expected total rewards* and the criterion of *expected average rewards*, respectively. Blackwell and Derman started an explosive development of the theory of control of Markov chains.

Before enclosing the problem field we first specify what is meant by a dynamic program or a Markov decision process. A dynamic program is a system that is determined by a state space, an action space, a reward function and a transition law, such that for each pair (state, action) a probability distribution on the state space is specified. At discrete points in time, called moments or stages, the controller or decision maker chooses an action from the action space. Then, according to the transition law, the system moves to a new state and an immediate reward is obtained, depending on the state before the transition, on the action itself and on the new state. A recipe for choosing an action at each stage, is called a strategy. To apply the results of dynamic programming in practice, one has to know the transition law. Unfortunately it seldom happens that these probability distributions are known. So the controller has to estimate the transition

law during the course of the process. Therefore, apart from the control problem, there is an estimation problem.

From now on we assume that the transition law depends on an unknown parameter, which belongs to some parameter set. Therefore the expected return at each stage depends on the unknown parameter and so we have to choose a criterion to measure the return at each stage. In literature the Bayes criterion is mainly used (cf. section 1.2 for a definition). The first attempts in the field of dynamic programming with an incompletely known transition law have been made by Bellman (see [Bellman (1961)]). He used the term adaptive control of Markov chains. Bellman noticed that, if the Bayes criterion is used, the problem can be transformed into an equivalent dynamic program with a completely known transition law and with a state space which is the Cartesian product of the original one and the set of all probability distributions on the parameter set. This transformation is also suggested in [Shiryaev (1964)], [Dynkin (1965)] and [Aoki (1967)] for models, which allow unobservability of the states, and in [Wessels (1971), (1972)]. In [Hinderer (1970)] the first systematic proof is given for the case that the state and action spaces are both countable, and afterwards in [Rieder (1972), (1975)] the transformation is given for completely separable metric state and action spaces. In fact it is shown that, for the Bayes criterion, the posterior distributions of the unknown parameter are sufficient statistics. In [Wessels (1968)], among other things, the problem of sufficient statistics is studied in connection with several other criteria, such as the minimum criterion. Almost all other authors considered only the Bayes criterion and studied the equivalent dynamic program, mentioned above. In [Martin (1967)], [Rieder (1972)], [Satia and Lave (1973)], and [Waldmann (1976)] the method of successive approximations for the equivalent dynamic program is studied. Only Satia and Lave tried to exploit the special structure of this dynamic program. In [Fox and Rolph (1973)], [Mandl (1974), (1976)], and in [Rose (1975)] optimal strategies are constructed for the criterion of expected average return. Here it is possible to construct strategies which are at least as good as all other strategies, for all parameter values, hence it is not necessary to work with the Bayes criterion or anything like it. Special models, arising in control theory are studied in [Sworder (1966)] and [Aoki (1967)]. Inventory control models with an incompletely known demand distribution are studied in [Scarf (1959)], [Iglehart (1964)], [Wessels (1971), (1972)], [Rieder (1972)], [Zacks and Fennel (1973)] and in

[Waldmann (1976)]. A number of other problems can be found in the literature. The most famous one is the two-armed bandit problem. We will return to most of the contributions of the above-mentioned authors in the other chapters of this monograph. The number of publications in the field of dynamic programming with an incompletely known transition law is very small compared with the overwhelming amount of literature on dynamic programming with a known transition law.

We conclude this section with a sketch of the problems we examine in this monograph. We choose the Bayes criterion too. From a mathematical point of view this criterion has the advantage, as compared with the minimax criterion, that the model can be transformed into the so-called equivalent dynamic program. Further it has the nice property that the decision maker may express his opinion on the importance of the various parameter values, which characterize the unknown transition law, by a weight function. Even if the model with known transition law has finite state and action spaces, the equivalent dynamic program has a state space which is essentially infinite. However, the method of successive approximations to determine the optimal expected total return is workable, since in order to determine the n-th approximation we have to consider all possible paths through n stages of which there are a finite number, if the state and action spaces are finite. The effort needed to obtain good approximations proved to be very large in the studies of Martin and Satia and Lave (in [Martin (1967)] examples with only two states and two actions turn out to be very time-consuming and in [Satia and Lave (1973)] examples with four states and two actions are considered to be of "moderate-size"). One of the objectives of our study is to show that the method of successive approximations can be applied successfully to rather large models, that have a suitable parameter structure. Our analysis is based on the construction of special scrap-vectors for the successive approximation method and on the exploitation of the convergence of the posterior distributions. We note that some results of our analysis are also interesting for the problem of robustness of the model under variations in the parameter value. In section 1.3 we specify the approximation methods we advocate, in an informal way.

Another objective of our study is to show that there are easy-to-handle optimal strategies for maximizing the average expected return, and also for some practical examples of our model for maximizing the expected total return. At the end of section 1.2 we consider this matter in more detail.

1.2 Informal description of the model

We start this section with a motivation of the choice of the model we study in this monograph: the *Bayesian control model*.

Consider a dynamic program with finite state and action spaces. It sometimes happens that a transition is affected by a random variable which is observable for the decision maker, but the value of which cannot be reconstructed from the state values of the process. For example consider a waiting-line model in discrete time, where Y_{n+1} is the number of arrivals in the time period [n, n+1) and where X_n is the number of customers in the system at time n. Then it is obvious that the value of Y_{n+1} is not determined by X_n and X_{n+1} , if the number of services completed in each time interval is random. If the distribution of the random variable Y_n is incompletely known, then it is useful to keep this random variable as a supplementary state variable. Confining ourselves to the state values of the original process only, means that we throw away information concerning the transition law. In our model we assume that for each state and action the transition may be affected by a random variable, the value of which is observed by the decision maker immediately after the transition. The value of this random variable is obtained by a random drawing from a distribution, depending only on the actual state and action. There are at most countably many different distributions from which is sampled. Further we assume that only these distributions are incompletely known. We call these random variables supplementary state variables. In case the transition, for some state and action, is not affected by a supplementary state variable we may consider the next state variable itself as a supplementary state variable. We return to this point in chapter 2.

We now continue with the model description. For simplicity, we assume here that all considered sets are finite. Let the state space be denoted by X and the action space by A. Further let the random variables X_n and A_n denote the state and action at stage n, respectively. The transition to state X_{n+1} , given X_n and A_n is also affected by the outcome of the supplementary state variable Y_{n+1} which is observed at stage n + 1 and which takes on values in the set Y. This works in the following way. The conditional probability of X_{n+1} , given $X_n = x$, $A_n = a$ and $Y_{n+1} = y$, is

$$\mathbb{P}[X_{n+1} = x' | X_n = x, A_n = a, Y_{n+1} = y] = \mathbb{P}(x' | x, a, y)$$

where the function P is assumed to be known.

However, the random variables Y_{n+1} , X_n and A_n are dependent, while the conditional distribution of Y_{n+1} , given X_n and A_n depends on some unknown parameter θ , which belongs to a given parameter set θ , i.e. we have

$$\mathbb{P}_{\theta}[\mathbf{Y}_{n+1} = \mathbf{y} \mid \mathbf{X}_{n} = \mathbf{x}, \mathbf{A}_{n} = \mathbf{a}] = \sum_{\mathbf{i} \in \mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x}, \mathbf{a}) \mathbb{P}_{\mathbf{i}}(\mathbf{y}|\theta)$$

where {K_i, i \in I} is a partition of X × A, and I is some index set. Hence the distribution in the set {p_i(•|θ), i \in I} from which the random variable Y_{n+1} is sampled depends on the state and action at stage n. Further, if X_n = x , A_n = a and Y_{n+1} = y there is an immediate, possibly negative, reward: r(x,a,y).

Although the model may seem to be rather artificial, there are many wellknown models which fit into this framework. For example, inventory control models, where X_n is the inventory level at time n and Y_{n+1} is the demand during the interval [n, n + 1). Here we always sample Y_n from the same distribution, hence I is a singleton. Also the ordinary dynamic program with finite state and action spaces and all transition probabilities unknown, is included in our model. We return to this matter in chapter 2. We note that, if the parameter θ is known, we are dealing with a dynamic program with state space X, action space A, transition law:

$$\mathbb{P}[X_{n+1} = x' \mid X_n = x, A_n = a] = \widetilde{\mathbb{P}}(x' \mid x, a) := \sum_{i \in I} \mathbb{1}_{K_i}(x, a) \sum_{y \in Y} \mathbb{P}(x' \mid x, a, y) \mathbb{P}_i(y \mid \theta),$$

and reward function:

$$\widetilde{r}(\mathbf{x},\mathbf{a}) := \sum_{\mathbf{i}\in\mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x},\mathbf{a}) \sum_{\mathbf{y}\in\mathbf{Y}} \mathbf{p}_{\mathbf{i}}(\mathbf{y}|\boldsymbol{\theta}) \mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y}) \ .$$

In this monograph X,Y,A and θ are complete separable metric spaces, but the index set I is at most countable. Hence we do not allow more than countably many unknown distributions $p_i(\cdot | \theta)$, $i \in I$ and $\theta \in \theta$.

A strategy π is a procedure which chooses at each stage n an action, based on the *history* of the process, i.e. $X_0, A_0, Y_1, X_1, A_1, \dots, Y_n, X_n$. Each strategy π , each parameter value θ , and each starting state x together determine a probability on the *sample space* of the process. The expectation with respect to this probability of the immediate reward at stage n is denoted by:

$$\mathbb{E}_{\mathbf{x},\theta}^{\pi}[r(\mathbf{x}_{n},\mathbf{A}_{n},\mathbf{Y}_{n+1})] .$$

The expected total discounted return $v(x, \theta, \pi)$ is:

$$\mathbf{v}(\mathbf{x},\boldsymbol{\theta},\boldsymbol{\pi}) := \mathbf{I} \mathbf{E}_{\mathbf{x},\boldsymbol{\theta}}^{\boldsymbol{\pi}} \left[\sum_{n=0}^{\infty} \boldsymbol{\beta}^{n} \mathbf{r} \left(\mathbf{X}_{n}, \mathbf{A}_{n}, \mathbf{Y}_{n+1} \right) \right]$$

where $\beta \in [0,1)$ is called the *discount factor*. Only in trivial situations there is a strategy π^* such that $v(\mathbf{x}, \theta, \pi^*) \ge v(\mathbf{x}, \theta, \pi)$ for all $\mathbf{x} \in \mathbf{X}, \theta \in \theta$ and all strategies π . So it is unwise to use this as a criterion for a strategy to be optimal. Criteria for which there are always (nearly) optimal strategies, are the already mentioned minimax and Bayes criteria. A strategy π^* is called ε -optimal, $\varepsilon \ge 0$, for the minimax criterion, if

$$\min_{\boldsymbol{v}} \mathbf{v}(\mathbf{x}, \boldsymbol{\theta}, \boldsymbol{\pi}^{\star}) \geq \min_{\boldsymbol{v}} \mathbf{v}(\mathbf{x}, \boldsymbol{\theta}, \boldsymbol{\pi}) \sim \varepsilon \quad \text{for all } \mathbf{x} \in \mathbf{X}, \ \boldsymbol{\theta} \in \boldsymbol{\theta}$$

and all strategies π . We do not use this criterion. In appendix B we consider an example, which shows that the use of this criterion has some odd implications. We use the Bayes criterion. So, we fix some probability distribution q on the parameter set θ and we call a strategy $\pi^* \epsilon$ -optimal, $\epsilon \ge 0$, if

$$\sum_{\boldsymbol{\theta} \in \boldsymbol{\Theta}} \mathbf{q}(\boldsymbol{\theta}) \mathbf{v}(\mathbf{x}, \boldsymbol{\theta}, \boldsymbol{\pi}^*) \geq \sum_{\boldsymbol{\theta} \in \boldsymbol{\Theta}} \mathbf{q}(\boldsymbol{\theta}) \mathbf{v}(\mathbf{x}, \boldsymbol{\theta}, \boldsymbol{\pi}) - \boldsymbol{\varepsilon}$$

for all $x \in X$ and all strategies π . If a strategy is 0-optimal we call it *optimal*. We note again that the so-called *prior distribution* q can be considered as a weight function, expressing the importance of the various parameter values in the opinion of the decision maker.

In chapter 4 we consider the average expected return instead of the expected total discounted return. We call a strategy $\pi^* \epsilon$ -optimal, $\epsilon \ge 0$, with respect to this criterion, if

$$\underset{N \to \infty}{\operatorname{liminf}} \frac{1}{N} \sum_{\theta \in \Theta} q(\theta) \sum_{n=0}^{N-1} \mathbb{E}_{x,\theta}^{\pi} [r(x_n, A_n, Y_{n+1})] \geq$$

$$\geq \underset{N \to \infty}{\operatorname{liminf}} \frac{1}{N} \sum_{\theta \in \Theta} q(\theta) \sum_{n=0}^{N-1} \mathbb{E}_{x,\theta}^{\pi} [r(x_n, A_n, Y_{n+1})] - \varepsilon$$

for all $x \in X$ and all strategies π (again, a 0-optimal strategy is called optimal).

The Bayes criterion allows us to consider another interpretation of the Bayesian control model. In this interpretation we consider the unknown para-

meter as a random variable with distribution q. The posterior distributions of this random variable, or in other words the conditional distributions of this random variable, given the history of the process, play an important role in this monograph. It is well-known that the name of Bayes is connected with the criterion since he suggested to consider the unknown parameter of a distribution as a random variable itself in statistical inference. It turns out that the Bayesian control model is equivalent to a dynamic program with a known transition law and with a compound state space X × W, where W is the set of all probability distributions on θ . For each starting state and each strategy, we are dealing with a stochastic process (X_n, Q_n, A_n) where Q_n is the actual posterior distribution of the random variable that represents the parameter.

It is desirable to have good strategies that are easy to handle, i.e. to have a formula or a simple recipe which yields an action as a function of the actual state $x \in X$ and the actual posterior distribution $q \in W$. A way of deriving easy to handle strategies is based on the following idea. If the parameter is known to be θ and if there is an optimal strategy then an optimal action in state $x \in X$ often is a maximizer of $F(x, \theta, \cdot)$ where F : X × θ × A \rightarrow IR. Note that the action depends on the parameter θ and that the function F is assumed to be known. Now let the parameter be unknown. Then we may use an action a which maximizes the function $a \rightarrow \int q(d\theta) F(x,\theta,a)$ if the actual state is x and the actual posterior distribution is q (provided that integration is possible and the maximum exists). Such a rule is called a Bayesian equivalent rule. It will be proved that such a rule yields an optimal strategy, if we are maximizing the average expected return, under conditions which guarantee that in the long run the decision maker obtains enough information about the unknown parameter, i.e. the sets K_i have to be recurrent. For maximizing the expected discounted total return we do not know a Bayesian equivalent rule that is optimal in general, however for some special models, such as the linear system with quadratic cost and a simple inventory control model, there is an optimal Bayesian equivalent rule. For the linear system with quadratic cost this rule can be considered as a generalization of the well-known certainty equivalent rule.

1.3 Summary of the following chapters

In chapter 2 we start with a formal description of the Bayesian control model and we consider some examples. Then we study the process of posterior distributions. The main result is the convergence of the posterior distributions to a degenerate distribution, under each strategy which assures the number of visits to each set K_i , $i \in I$ to be infinite, with probability one. This result is used in several places in chapters 4 and 6. In chapter 3 we deal with two rather technical points. First we show that the Bayesian control model is equivalent to a dynamic program (see section 1.2) and after that we study a class of optimal reward operators for dynamic programs in general. Here we consider optimal reward operators based on stopping times, for dynamic programs as introduced by Wessels (cf. [Van Nunen and Wessels (1977)]). We generalize the operators for dynamic programs with complete separable state and action space and we derive some new properties of these operators. These operators determine the maximal expected total return until some stopping time, and with a terminal reward at the stopping time, depending on the state at the time. Successive applications of these operators yield a sequence of funtions on the state space, which converges to the function of optimal values. We use these operators in chapter 6 where we consider the method of successive approximations for the equivalent dynamic program.

In chapter 4 we first introduce the Bayesian equivalent rules. Then we construct optimal strategies in order to maximize the average expected re-ward.

Chapter 5 is devoted to the study of optimal strategies for the expected total-return criterion. For three examples of our model we show that a Bayesian equivalent rule provides an optimal strategy. The first example we call the *independent case* since the rewards are independent of the state, i.e. r is constant in the first coordinate. In all examples it is assumed that the index set I is a singleton, so the random variables Y_n , $n \in \mathbb{N}$ are sampled from the same (unknown) distribution at each stage. The second example is the linear system with quadratic cost and the last one is a simple inventory control model. For this inventory model the Bayesian equivalent rule is not always optimal. However, we give an upper bound for the loss we incur by using this rule when it is not optimal. In chapter 6 we consider approximations for the "function of optimal values"

when maximizing the expected discounted total return. This function is called

the value function and is defined on $X \times W$ by:

$$\mathbf{v}(\mathbf{x},\mathbf{q}) := \sup_{\pi} \sum_{\theta} \mathbf{q}(\theta) \mathbf{v}(\mathbf{x},\theta,\pi)$$

where the supremum is taken over all strategies. We first indicate an upper bound on v and several lower bounds. These bounds have simple interpretations and are computable if the parameter set is finite or equivalently, if the prior distribution is concentrated on a finite set. We study the use of these bounds for successive approximations of the value function. We also give a lower bound on the expected discounted total return if a special Bayesian equivalent rule is used and we construct an other easy-to-handle strategy which is not a Bayesian equivalent rule but which behaves nicely. Further we specialize the parameter structure as follows: there is a subset B of the state space X with the property that, if $X_n \in B$ then Y_{n+1} is sampled from the same unknown distribution for all actions chosen, for $X_n \in X \setminus B$ the distribution of Y_{n+1} is known (hence $K_1 = B \times A$ and $\theta_2, \theta_3 \dots$ are singletons). A special example of this structure arises in the model where B = X, e.g. the models studied in chapter 5. Here we use an optimal reward operator as studied in chapter 3, with the entrance time in the set B as stopping time. In fact, this operator allows us to consider the process which is embedded on the set B. For this parameter structure we use the convergence of the posterior distributions to a degenerate distribution, and also the upper and lower bounds, to compute in advance an error estimate on the n-th successive approximation, starting with a fixed prior distribution. If the error estimate for the n-th approximation is small enough, then we may compute the value function for this prior distribution by backward induction. The effort needed for the computation of the n-th error estimate is small compared with the backward induction procedure. Since usually the computed quantities to determine the n-th approximation cannot be used to compute the n + 1-st approximation, it is nice to know in advance whether the n-th approximation is sufficiently accurate. We also consider in this chapter another type of approximations, namely

discretizations of the parameter set. Here we split up the parameters set into a finite partition, and in each set of the partition we choose a representative point. We give bounds for the error caused by replacing the given prior distribution q by the discrete prior distribution which attributes probabilities to the representative points equal to the given probabilities of the corresponding sets. In [Fox (1973)] and [Whitt (1976)] also discretizations of dynamic programs are studied. To apply their method, we would have to split up the set of all distributions on the parameter set into a finite partition and, in the equivalent dynamic program, the process would then jump between representative points in these partition sets. However, we then loose the nice property that the second state-coordinate of the process (i.e. Q_n) is the posterior distribution of the unknown parameter, at every stage.

Our discretizations are of interest, since in general we can compute the upper and lower bounds, mentioned above, only if the prior distribution is concentrated on a finite set of parameters. As a byproduct of our analysis of discretizations we obtain a bound for the difference between the value function of the Bayesian control model and the model that is obtained by replacing in advance the distributions $p_i(\cdot | \theta)$ by their Bayes estimates based on the prior distribution and considering these estimates as the true distributions. This last model is used very frequently in practice, instead of the Bayesian model.

Finally, in chapter 7 we construct algorithms, based on the approximations of chapter 6, which compute the value function v(x,q) for a fixed prior distribution, and which also determine ε -optimal strategies. We illustrate the quality of the algorithms by numerical data for some examples. In appendix A we collect some results of measure theory which are used in chapter 3. In appendix B we illustrate the odd implications of the minimax criterion by an example.

We note that it is possible to start reading at chapter 4 after reading the model description in chapter 2 and the assertions of the theorems and corollaries of chapters 2 and 3.

1.4 Notations, conventions and prerequisites

We start with some conventions. A numbered sentence indicates a definition a result or a formula. Such a sentence may occupy several lines, each one of which is indicated by an indentation. Symbols used for objects, which are defined in a numbered sentence have a *global* meaning, i.e. if we use a symbol without defining it in the theorem proof, example or comment where it is used, then it has the meaning given in the numbered sentence where it is defined. References to lemmas, theorems, corollaries, examples, sections and chapters are preceded by the words "lemma", "theorem", etc. Each chapter has its own numbering, for example 2.4 is the fourth numbered sentence in

chapter 2. References to appendix A are preceded by the capital: A. The end of a proof is indicated by: []. If there is no ambiguity concerning the domain of some index or variable, we omit the domain in the notations. We continue with a list of notations.

1.1 IN := {0,1,2,...}, \overline{IN} := $IN \cup \{\infty\}$, IN^* := {1,2,3,...}, \overline{IN}^* := $IN^* \cup \{\infty\}$. 1.2 IR is the set of real numbers, \overline{IR} := $R \cup \{-\infty,\infty\}$.

1.3 $\delta(\cdot, \cdot)$ is the Kronecker symbol, i.e. $\delta(i, i) = 1$ and $\delta(i, j) = 0$ if $i \neq j$.

1.4 # A is the cardinality of the set A.

1.5
$$x^{\top} := \max(x, 0), x := -\min(x, 0).$$

- 1.6 Let (X_i, X_i) be measurable spaces for $i \in I$, where I is a countable set then $X := \prod X_i$ is the Cartesian product and $X := \bigotimes X_i$ the $i \in I^{-1}$ product- σ -field on X. If μ_i is a probability on X_i then $\mu := \bigotimes \mu_i$ is the product measure on X, if I is finite and μ_i a σ -finite measure on X, then μ is also the product measure on X.
- Let A, X and Y be sets, such that $A \subset X \times Y$ then

1.7 proj (A) :=
$$\{x \in X \mid \text{ there is some } y \in Y \text{ with } (x,y) \in A\}$$
.

1.8 i.i.d. means "independent and identically distributed", iff means "if and only if" and a.s. means "almost surely".

Let (X, X) and (Y, Y) be measurable spaces and let $f : X \rightarrow Y$ be measurable then

1.9 $\sigma(f)$ is the sub- σ -field of X induced by f, i.e.

 $\sigma(f) := \{ A \in X \mid A = f^{-1}(B), B \in Y \}, \text{ where } f^{-1}(B) := \{ x \in X \mid f(x) \in B \}.$

1.10 P(X) is the set of all probabilities on a measurable space (X, X).

Let f be a function on a set X then

1.11 $x \rightarrow f(x)$, $x \in X$ is a notation for this function.

1.12 \emptyset is the empty set.

Let x_1, x_2, x_3, \ldots be a sequence of real numbers, then

1.13 $\inf\{x_i \mid i \in \emptyset\} := \infty, \sum_{i \in \emptyset} x_i := 0 \text{ and } \Pi x_i := 1.$

Let (X, X) be a measurable space and let q be a measure on X and f a non-negative Borel measurable function on (X, X), then

1.14 f(x)q(dx) is a notation for the measure μ defined by

$$\mu(A) := \int f(x)q(dx), A \in X.$$

1.15 Let f and g be functions on some set X with range \mathbb{R} and let $y \in \mathbb{R}$, f \leq g if and only if f(x) \leq g(x) for all x \in X, f \leq y if and only if f(x) \leq y for all x \in X. The analogous convention is used if \leq is replaced by \langle , \rangle , \rangle or = .

We continue with some pertinent facts on transition probabilities and conditional expectations. Let (Ω, F, \mathbb{P}) be a probability space, (A, A) a measurable space, and let $Y : \Omega \rightarrow A$ be measurable. Then we call Y a *random variable* and we write

1.16 (i)
$$\mathbb{IP}[Y \in B] := \mathbb{IP}[\{\omega \in \Omega \mid Y(\omega) \in B\}], B \in A.$$

(ii) $\mathbb{IE}[Y] := \int Y(\omega) \mathbb{IP}(d\omega)$.

A real-valued function on Ω is called *F-measurable* or simply measurable, if it is measurable with respect to the Borel σ -field on \mathbb{R} . The following lemma is well-known (cf. [Bauer (1968) lemma 55.1]).

Lemma 1.1

Let (Ω, F) and (A, A) be measurable space, and let $f : \Omega \rightarrow A$ be measurable. Then a real-valued function g on Ω is $\sigma(f)$ -measurable iff there is a realvalued measurable function h on A such that g = h(f). If f is a surjection then the function h is unique.

- 1.17 A measurable space (A,A) is called *Borel space* if A is a non-empty Borel subset of a complete separable metric space and A is the Borel σ -field on A (note that in [Hinderer (1970) page 187] such a space is called a standard Borel space and in [Blackwell (1965)] a Borel set).
- 1.18 The topological product of at most countably many Borel spaces which, because of the separability of the spaces, coincides with the measure theoretic product, is again a Borel space (cf.[Parthasarathy (1967) p. 135]).

Let (Ω, F) and (A, A) be measurable spaces, then a function P from $\Omega \times A$ to [0,1] is called a *transition probability from* (Ω, F) to (A, A), or simply from Ω to A, if

1.19 (i) P(B ·) is F-measurable for each B ∈ A.
(ii) P(· | ω) is a probability on A, for each ω ∈ Ω.

Let (Ω, F, \mathbb{P}) be a probability space, let B be a sub- σ -field of F and let X be a real-valued measurable function on Ω , with $\mathbb{E}[X^+] < \infty$.

- 1.20 (i) The conditional expectation of X given B is denoted by $\mathbf{E}[\mathbf{X}|B]$ and defined as a real-valued B-measurable function on Ω such that $\mathbf{E}[\mathbf{X}]_B = \mathbf{E}[\mathbf{E}[\mathbf{X}|B]]_B$ for all $B \in B$. (Here 1, is the indicator function of the set B.)
 - (ii) If Y is another a real-valued measurable function on Ω we define $\mathbb{E}[X|Y] := \mathbb{E}[X|\sigma(Y)]$.
 - (iii) For every $A \in F$ we define the conditional probability of A given B, respectively the conditional probability of A given Y by $P[A|B] := E[1_a|B]$, respectively $P[A|Y] := E[1_a|Y]$.

Note that the conditional expectation is not uniquely defined, however two versions of it are equal P-a.s.

Theorem 1.2

Let (Ω, F) be a Borel space and let \mathbb{P} be a probability on F. Then for every sub- σ -field B of F the conditional probability is *regular*, i.e. there exists a transition probability P from (Ω, B) to (Ω, F) such that for every realvalued F-measurable function X that is bounded from above, we have $\omega \rightarrow \int X(\widetilde{\omega})P(d\widetilde{\omega}|\omega)$ is a version of $\mathbb{E}[X|B]$. If P' is another transition probability from (Ω, B) to (Ω, F) with this property, then $\mathbb{P}[\{\omega|P(\cdot|\omega) \neq P'(\cdot|\omega)\}] = 0$.

For a proof cf. [Bauer (1968) th. 56.5].

We sometimes need the following corollary of th. 1.2.

Corollary 1.3

Let (Ω, F) be a Borel space, let \mathbb{P} be a probability on F, let (A, A) be a measurable space and let Y be a measurable map from Ω to A. The probability Q on A is defined by $Q(B) := \mathbb{P}[Y \in B]$, $B \in A$. Then there is a

transition probability P from (A, A) to (Ω, F) such that

(*)
$$\mathbb{IP}[D \cap Y^{-1}(B)] = \int_{B} P(D|\tilde{y})Q(dy)$$
 for all $B \in A$ and $D \in F$.

If P' is another transition probability from (A,A) to (Ω,F) with this property, then

$$Q[\{y \mid P(\cdot \mid y) \neq P'(\cdot \mid y)\}] = 0$$
.

P is called a *regular conditional probability given* Y = y and we usually write $\mathbb{P}[\cdot|Y = y]$ for $\mathbb{P}(\cdot|y)$.

Proof.

By th. 1.2 there is a transition probability \widetilde{P} from $(\Omega, \sigma(\mathbf{Y}))$ to (Ω, F) such that for all $D \in F$ and $B \in A$:

$$\mathbf{IP}[\mathbf{D} \cap \{\mathbf{Y} \in \mathbf{B}\}] = \int \widetilde{\mathbf{P}}(\mathbf{D} | \boldsymbol{\omega}) \mathbf{IP}(\mathbf{d}\boldsymbol{\omega}) .$$
$$\mathbf{Y}^{-1}(\mathbf{B})$$

By lemma 1.1 there is for each $D \in \mathcal{F}$ a real-valued measurable function on A, denoted by $P(D|\cdot)$ such that

$$P(D|Y(\omega)) = \widetilde{P}(D|\omega)$$
, for $\omega \in \Omega$.

It is easy to verify that P, considered as a function on $A \times F$ is a transition probability from (A, A) to (Ω, F) with property (*). Let P' be another transition probability on $A \times F$ with property (*), and define N := { $y \in A | P(\cdot | y) \neq P'(\cdot | y)$ }. Then

п

$$\mathbf{Y}^{-1}(\mathbf{N}) = \{ \boldsymbol{\omega} \in \boldsymbol{\Omega} | \mathbf{P}(\cdot | \mathbf{Y}(\boldsymbol{\omega})) \neq \mathbf{P}^{*}(\cdot | \mathbf{Y}(\boldsymbol{\omega})) \}.$$

By th. 1.2 $\mathbb{IP}[Y^{-1}(N)] = 0$. Hence Q[N] = 0.

Let the assumptions of corollary 1.3 hold and let X be a real-valued measurable function on Ω , bounded from above. Then we define

1.21 IE
$$[X|Y = Y] := f(Y) := \int X(\omega) IP [d\omega|Y = Y]$$
.

It is easy to verify that f(Y) is a version of the conditional expectation of X given Y.

We frequently use the following theorem of Ionescu Tulcea (cf. [Neveu (1965) page 165]).

Theorem 1.4

Let (X_n, X_n) , $n \in \mathbb{N}$ be measurable spaces and let Q_{n+1} be a transition probability from $(\prod_{t=0}^n X_t, \otimes_{t=0}^n X_t)$ to (X_{n+1}, X_{n+1}) , $n \in \mathbb{N}$. Further let $(X, X) := (\prod_{t=0}^{\infty} X_t, \otimes_{t=0}^{\infty} X_t)$ and let ξ_0, ξ_1, \ldots be the coordinate functions on X i.e. $\xi_n(x) := x_n, x = (x_0, x_1, \ldots) \in X$. Then

(i) for all $n \in \mathbb{N}$ there is an unique transition probability P from $(\prod_{t=0}^{n} X_{t}, \otimes_{t=0}^{n} X_{t})$ to (X, X) denoted by $P(B|_{X_{0}}, \dots, X_{n})$, $B \in X$, $x_{i} \in X_{i}$ $i = 0, \dots, n$, such that for cylinder sets of the form $B := A_{0} \times \dots \times A_{m} \times X_{m+1} \times X_{m+2} \times \dots$ and $m \ge n$:

$$P(\mathbf{B} | \mathbf{x}_0, \dots, \mathbf{x}_n) = \mathbb{1}_{A_1 \times \dots \times A_n} (\mathbf{x}_0, \dots, \mathbf{x}_n) \int_{A_{n+1}} \mathbb{Q}_{n+1} (d\mathbf{x}_{n+1} | \mathbf{x}_0, \dots, \mathbf{x}_n) \dots$$
$$\dots \int_{A_m} \mathbb{Q}_m (d\mathbf{x}_m | \mathbf{x}_0, \dots, \mathbf{x}_{m-1}) .$$

(ii) for every probability ρ on X_0 there is a unique probability \mathbb{P}_{ρ} on X given by $\mathbb{P}_{\rho}[B] = \int_{X_0} \rho(dx_0) P(B|x_0)$, $B \in X$ and for any measurable function Y on X that is bounded from above, $\int P(dx|\xi_0, \dots, \xi_n) Y(x)$ is a version of the conditional expectation of Y given the σ -field $\sigma(\xi_0, \dots, \xi_n)$. Hence one may define: (cf. lemma 1.1)

$$\mathbb{E}_{\rho}[Y \mid \xi_{0} = x_{0}, \dots, \xi_{n} = x_{n}] := \int P(dx \mid x_{0}, \dots, x_{n}) Y(x)$$

or

$$\mathbb{IE}_{\rho}[\mathbf{Y} \mid \boldsymbol{\xi}_{0}, \dots, \boldsymbol{\xi}_{n}] := \int P(d\mathbf{x} \mid \boldsymbol{\xi}_{0}, \dots, \boldsymbol{\xi}_{n}) \mathbf{Y}(\mathbf{x}),$$

Finally we summarize some pertinent facts concerning the set P(X) of all probabilities on a Borel space (X, X).

1.22 On P(X) we have the topology of weak convergence; this is the

coarsest topology such that for functions $f \in C(X)$ the map $\mu \Rightarrow \int f(x) \mu(dx)$ is continuous, $\mu \in \mathcal{P}(X)$, where C(X) is the set of bounded real-valued continuous functions on X (cf. [Parthasarathy (1967)]).

Lemma 1.5

Let E be the topology of weak convergence on P(X) and F the σ -field generated by E. Then F is also:

- (i) the smallest σ -field such that the functions $\mu \rightarrow \mu$ (B) are measurable, $\mu \in \mathcal{P}(X)$, B $\in X$.
- (ii) the smallest σ -field such that the functions $\mu \rightarrow \int f(x)\mu(dx)$ are measurable, $\mu \in P(X)$, $f \in C(X)$.

The proof of statement (i) can be found in [Rieder (1975) lemma 6.1]. Note that this implies that F is also the smallest σ -field such that $\mu \neq \int f d\mu$, $\mu \in P(X)$ are measurable for all real-valued bounded measurable functions f on X.

Proof of statement (ii). Let B be the smallest σ -field in P(X) such that $\mu \neq \int f(\mathbf{x})\mu(d\mathbf{x})$ is measurable, for $f \in C(X)$. For each Borel subset $D \subseteq \mathbb{R}$ and every $f \in C(X)$ we have $\{\mu\} = \int f(\mathbf{x})\mu(d\mathbf{x}) \in D\} \in \mathcal{B}$, for $f \in C(X)$. This is true in particular for all open sets of \mathbb{R} . Hence the topology E is contained in B, i.e. $F \subseteq B$. On the other hand, since for all open subsets $D \subseteq \mathbb{R}$ $\{\mu\} = \int f(\mathbf{x})\mu(d\mathbf{x}) \in D\} \in F$ and since the Borel σ -field on \mathbb{R} is generated by the open sets, we have $\{\mu\} = \int f(\mathbf{x})\mu(d\mathbf{x}) \in D\} \in F$ for all Borel subsets $D \subseteq \mathbb{R}$. Hence $B \subseteq F$.

In lemma 1.6 we collect some miscellaneous results.

Lemma 1.6

- (i) Let (X,X) be a Borel space and F the σ -field on P(X), generated by the topology of weak convergence, then (P(X),F) is a Borel space.
- (ii) The identification of elements of X with the point measures in P(X) is a homeomorphism.
- (iii) Let (X,X) and (Y,Y) be Borel spaces and f a nonnegative measurable function on $X \times Y$, then the function

 $(x,q) \rightarrow \int f(x,y)q(dy) , \quad x \in X, q \in \mathcal{P}(\mathcal{Y})$ is measurable.

The proof of (i) is found in [Hinderer (1970) th. 12.13], the proof of part (ii) in [Parthasarathy (1967) lemma 6.1 page 42] and part (iii) is an immediate consequence of lemma 1.5 (i) (cf. [Rieder (1975) lemma 6.2]).

2. THE MODEL AND THE PROCESS OF POSTERIOR DISTRIBUTIONS

In section 2.1 we define the *Bayesian control model*, the model we study in this monograph, and we present some examples. In section 2.2 the posterior distributions of the random variable, which represents the unknown parameter, are defined and some properties are derived. Finally, in section 2.3 the limit behaviour of the posterior distributions is studied and also the differences of successive posterior distributions.

2.1 The Bayesian control model

Our model is similar to models described in [Shiryaev (1964), (1967)], [Dynkin (1965)], [Martin (1967)] and [Hinderer (1970)]. In fact, it is a special case of the model considered in [Rieder (1975)], which will be shown later on in this section. In this monograph several models are considered, which are special cases of the Bayesian control model, we describe now.

model 1: Bayesian control model The modelconsists of the following objects.

- 2.1 (a) (X,X) a Borel space. X is called the state space.
 - (b) (Y, Y) a Borel space. Y is called the supplementary state space.
 - (c) (A, A) a Borel space. A is called the action space.
 - (d) D, a function from X to the non-empty subsets of A such that
 K := {(x,a) | x ∈ X, a ∈ D(x)} is an element of X ⊗ A.
 D(x) is called the set of *admissible actions* in state x. It is assumed that K contains the graph of some measurable function from X to A.
 - (e) I is a countable set, called the *index set*.
 - (f) For all i ∈ I there is a Borel space (θ_i, T_i) and θ_i is called the parameter space of index i. The Borel space (θ, T) is defined by θ := Π_{i∈I} θ_i, T := ⊗_{i∈I} T_i. The set θ is called the parameter space.
 (g) {K_i, i ∈ I} is a measurable partition of X × A.
 - (h) P is a transition probability from $X \times A \times Y$ to X (cf. 1.19).
 - (i) v is a σ -finite measure on Y. If Y is countable then v is assumed to be the counting measure.
 - (j) p_i is a nonnegative measurable function on $Y \times \theta_i$, for all $i \in I$ such that $\int_{Y} p_i(Y|\theta_i) v(dY) = 1$ for all $\theta_i \in \theta_i$ and $i \in I$.

For all $i \in I$ and $\theta_i, \tilde{\theta}_i \in \theta_i, \theta_i \neq \tilde{\theta}_i$ we assume

$$v(\{y \in Y \mid p_i(y|\theta_i) \neq p_i(y|\tilde{\theta}_i)\}) > 0.$$

This property is called: the separation property.

(k) r is a real-valued measurable function on $X \times A \times Y$, bounded from above, and called the *reward function*.

We continue with some definitions which clarify the meaning of the objects defined in 2.1. Each $\theta \in \theta$ can be described by $\theta = (\theta_i)_{i \in I}$ where $\theta_i \in \theta_i$ is called the i-th coordinate of θ . For each $\theta \in \theta$ we define a *transition probability* \bar{P}_{θ} from X × A to Y × X, by

2.2
$$\vec{P}_{\theta}(E \times F | x, a) := \sum_{i \in I} \mathbf{1}_{K_{i}}(x, a) \int_{E} \nabla(dy) \mathbf{p}_{i}(y|\theta_{i}) \int_{F} P(dx'|x, a, y)$$

where $E \in Y$, $F \in X$, $x \in X$, $a \in A$ and θ_i the i-th coordinate of $\theta \in \theta$. (Note that \overline{P}_{θ} satisfies all requirements for a transition probability (cf. 1.19)).

2.3 The set of *histories* H at stage n is defined by

- (i) $H_0 := X$, $H_n := X \times (A \times Y \times X)^n$, $n \in \overline{\mathbb{IN}}^*$. (ii) H_n is the product- σ -field on H_n induced by X, A and Y for $n \in \mathbb{IN}$.
- 2.4 A strategy π is a sequence: $\pi = (\pi_0, \pi_1, ...)$ where π_n is a transition probability from (H_n, H_n) to (A, A) such that $\pi_n (\cdot | \mathbf{x}_0, \mathbf{a}_0, \mathbf{y}_1, \mathbf{x}_1, \mathbf{a}_1, ..., \mathbf{y}_n, \mathbf{x}_n)$

is concentrated on the set $D(x_n)$. The set of all possible strategies is denoted by II.

It is easy to verify, by the condition on K (cf. 2.1 (d)), that ${\rm I\!I}$ is non-empty.

2.5 The sample space of the Bayesian control process is $\Omega := \theta \times H_{\infty}$, and on Ω we have the product- σ -field $H := T \otimes H_{\infty}$.

Note that (θ, T) and (Ω, H) are Borel spaces (cf. 1.18). On Ω we define the coordinate functions Z, X_n, Y_n, A_n, n $\in \mathbb{N}$, also called random variables:

2.6
$$Z(\omega) := \theta, X_n(\omega) := x_n, Y_n(\omega) := y_n, A_n(\omega) := a_n \text{ for } \omega = (\theta, x_0, a_0, y_1, x_1, a_1, \dots) \in \Omega.$$

According to the Ionescu Tulcea theorem (cf. th. 1.4) we have for each socalled starting distribution $\rho \in P(X)$, each so-called prior distribution $q \in P(T)$ and each strategy $\pi \in \Pi$, a probability $\mathbb{P}_{\rho,q}^{\pi}$ on (Ω, \mathcal{H}) , defined by

2.7
$$\mathbf{P}_{\rho,q}^{\pi}[\mathbf{Z} \in \mathbf{B}, \mathbf{x}_{0} \in \mathbf{C}, \mathbf{A}_{0} \in \mathbf{D}_{0}, (\mathbf{Y}_{1}, \mathbf{x}_{1}) \in \mathbf{E}_{1}, \dots, (\mathbf{Y}_{n}, \mathbf{x}_{n}) \in \mathbf{E}_{n}] :=$$
$$\int_{\mathbf{B}} \mathbf{q}(\mathbf{d}\theta) \int_{\mathbf{C}} \rho(\mathbf{d}\mathbf{x}_{0}) \int_{\mathbf{D}_{0}} \pi_{0}(\mathbf{d}\mathbf{a}_{0}|\mathbf{x}_{0}) \int_{\mathbf{E}_{1}} \mathbf{\tilde{P}}_{\theta}(\mathbf{d}(\mathbf{y}_{1}, \mathbf{x}_{1})|\mathbf{x}_{0}, \mathbf{a}_{0}) \dots$$

$$\cdots \int_{\substack{D_{n-1} \\ D_{n-1}}} \pi_{n-1} (da_{n-1} | x_0, a_0, y_1, x_1, a_1, \dots, y_{n-1}, x_{n-1}) \int_{E_n} \overline{P}_{\theta} (d(y_n, x_n) | x_{n-1}, a_{n-1})$$

where $\mathbf{B} \in \mathcal{T}$, $\mathbf{C} \in X$, $\mathbf{D}_{n} \in \mathbf{A}$ and $\mathbf{E}_{n} \in \mathcal{Y} \otimes X$, $n \in \mathbb{N}$.

2.8 The expectation with respect to
$$\mathbb{IP}^{\pi}$$
 is denoted by $\mathbb{IE}^{\pi}_{\rho,q}$.

2.9 Define W := P(T) and let W be the σ -field on W generated by the weak topology (cf. 1.22).

We identify each $\theta \in \theta$ with the element of W which is *degenerate* in θ , i.e. θ represents the probability that is concentrated on $\{\theta\}$. (By lemma 1.6(ii) this identification is a homeomorphism). And similarly we identify each $x \in X$ with the degenerate distribution in P(X). Hence, for $\pi \in \Pi$, $x \in X$, $\theta \in \theta$ the probability $\mathbb{P}_{x,\theta}^{\pi}$ is well-defined. Using th. 1.4 and the identification we easily derive:

2.10 The conditional probability may be chosen as:

$$\mathbf{P}_{\rho,q}^{\pi}[\cdot | \mathbf{Z} = \theta] = \mathbf{P}_{\rho,\theta}^{\pi}[\cdot]$$

or

$$\mathbb{P}_{\rho,q}^{\pi}[\cdot | \mathbf{Z}] = \mathbb{P}_{\rho,\mathbf{Z}}^{\pi}[\cdot] .$$

Note that the difference in these expressions is that the first one is a function on θ , while the second one is a function on Ω , depending on the first coordinate only.

Using 2.10 we find, for $B \in T$ and $C \in H_m$:

2.11
$$\mathbb{P}_{\rho,q}^{\pi}[\mathbb{Z} \in \mathbb{B}, (X_0, A_0, Y_1, X_1, A_1, \ldots) \in \mathbb{C}] =$$
$$\int_{\mathbb{B}} q(d\theta) \mathbb{P}_{\rho,\theta}^{\pi}[(X_0, A_0, Y_1, X_1, A_1, \ldots) \in \mathbb{C}].$$

Further we define *criterion functions* for the discrimination of strategies. 2.12 (i) The *Bayesian discounted total return* v is a real-valued function on $X \times W \times \Pi$:

$$\mathbf{v}(\mathbf{x},\mathbf{q},\pi) := \mathbb{E}_{\mathbf{x},\mathbf{q}}^{\pi} \left[\sum_{n=0}^{\infty} \beta^{n} \mathbf{r}(\mathbf{X}_{n},\mathbf{A}_{n},\mathbf{Y}_{n+1}) \right]$$

where $\beta \in [0,1)$ is the discount factor. (ii) The value function v is a real-valued function on X × W:

$$v(x,q) := \sup_{\pi \in \Pi} v(x,q,\pi).$$

Note that we use the symbol v for two different, but related functions, and note that we use the name "value function" only in connection with the discounted total return.

2.13 The Bayesian average return g is a real-valued function on $X \times W \times \Pi$:

$$g(\mathbf{x},\mathbf{q},\pi) := \liminf_{\mathbf{N}\to\infty} \frac{1}{\mathbf{N}} \mathbb{E}_{\mathbf{x},\mathbf{q}}^{\pi} \begin{bmatrix} \sum_{n=0}^{\mathbf{N}-1} \mathbf{r}(\mathbf{X}_{n},\mathbf{A}_{n},\mathbf{Y}_{n+1}) \end{bmatrix} .$$

Finally we define (nearly) optimal strategies. Let $\varepsilon \ge 0$.

- 2.14 (i) A strategy π is called ε -optimal for the total return criterion in $x \in X$ and $q \in W$, if $v(x,q,\pi) \ge v(x,q) - \varepsilon$.
 - (ii) A strategy π is called ϵ -optimal for the average return criterion in $\mathbf{x} \in X$ and $\mathbf{q} \in W$, if $\mathbf{g}(\mathbf{x}, \mathbf{q}, \pi) \ge \sup_{\widetilde{\pi} \in \Pi} \mathbf{g}(\mathbf{x}, \mathbf{q}, \widetilde{\pi}) - \epsilon$. $\widetilde{\pi} \in \Pi$ A 0-optimal strategy is simply called optimal.

Now the Bayesian control model has been described completely. Note that for each starting distribution $\rho \in P(X)$, each prior distribution $q \in W$ and each strategy $\pi \in \Pi$ the probability $\mathbb{P}_{\rho,q}^{\pi}$ and the stochastic process $(Z, X_0, A_0, Y_1, X_1, A_1, \ldots)$ are completely described. Only in chapter 4 we shall consider the average-return criterion, everywhere else we consider the total-return criterion.

The Bayesian control model is an example of the so-called *Bayesian decision model* studied in [Rieder (1975)]. This relationship is not used in our monograph. However, it simplifies comparisons of our results with the literature. To substantiate this we introduce the following notations.

ation of our model the starting distribution ρ^* on Y × X is required. However, only the marginal distribution of ρ^* on X plays a role, since the transition probability P^*_{θ} has the property: $y \rightarrow P^*_{\theta}(B \mid (y,x),a)$ is constant, by 2.15(ii).

We conclude this section with some examples, illustrating the applicability of our model.

Example 2.1

If the parameter set θ is a singleton, or equivalently if the prior distribution $q \in W$ is degenerate in $\theta \in \theta$, the Bayesian control model is an ordinary dynamic program, with state space (X, X), action space (A, A) and transition probability \widetilde{P}_{α} , given by

$$\widetilde{P}_{\theta}(B \mid x, a) := \sum_{i \in I} \mathbf{1}_{K_{i}}(x, a) \int v(dy) p_{i}(y \mid \theta_{i}) \int_{B} P(dx' \mid x, a, y), \quad B \in X$$

and reward function $\widetilde{r}_{_{\!\!H}}$:

$$\widetilde{r}_{\theta}(\mathbf{x},\mathbf{a}) := \sum_{\mathbf{i} \in \mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x},\mathbf{a}) \int v(d\mathbf{y}) \mathbf{p}_{\mathbf{i}}(\mathbf{y} | \boldsymbol{\theta}_{\mathbf{i}}) \ .$$

Example 2.2

Each dynamic program with countable state space \tilde{X} , countable action space \tilde{A} and incompletely known transition probability \tilde{P} from $\tilde{X} \times \tilde{A}$ to \tilde{X} and realvalued reward function \tilde{r} on $\tilde{X} \times \tilde{A}$ can be transformed into a Bayesian control model. To verify this define $X := \tilde{X}$, X is the power set of \tilde{X} , $A_{\underline{i}} := \tilde{A}$, A is the power set of \tilde{A} , Y := X, Y := X and $r(x,a,y) := \tilde{r}(x,a)$ for all $x \in X$, $a \in A$ and $y \in Y$. Further define $I := X \times A$, $K_{\underline{i}} := \{i\}$, $i \in I$ and $\theta_{\underline{i}} := P(X)$. Note that I is countable and that $(\theta_{\underline{i}}, T_{\underline{i}})$ is a Borel space if $T_{\underline{i}}$ is the σ -field on $\theta_{\underline{i}}$ generated by the weak topology (cf. lemma 1.6). Finally define $P(\{x'\}|x,a,y) := \delta(x',y)$, $x,x' \in X$, $a \in A$, $y \in Y$ and $p_{\underline{i}}(\cdot|\theta_{\underline{i}}) := \theta_{\underline{i}}$, $\theta_{\underline{i}} \in \theta_{\underline{i}}$, $i \in I$. It is straightforward to verify that all assumptions of 2.1 are satisfied. If, for some pair $x,a \in X \times A$, $\tilde{P}(\cdot|x,a)$ is known, then the marginal distribution on $\theta_{x,a}$ of $q \in W$ has to be degenerate in $\tilde{P}(\cdot|x,a)$. Similarly, if $\tilde{P}(\cdot|x,a)$ is unknown but belongs to some family of probabilities on (X, X)then the marginal on $\theta_{x,a}$ of $q \in W$ has to be concentrated on this family.

[Rose (1975)] can be regarded as special cases of our model.

Example 2.3

The class of models considered here is specified by Euclidean spaces X, Y and A, and a measurable function F from X × A × Y to X. The state X_n at time n is a function of the action A_{n-1} at time n - 1, the state X_{n-1} at time n - 1, and a random variable Y_n such that

Consequently the models described in [Martin (1967)], [Wessels (1968)],

 $X_n = F(X_{n-1}, A_{n-1}, Y_n)$, $n \in \mathbb{N}^*$

where $X_n \in X$, $A_n \in A$ and $Y_n \in Y$. The random variables $\{Y_n, n \in \mathbb{N}^+\}$ are i.i.d. and cannot be *controlled* by the decision maker, however they can always be *observed* by him. For that reason the sequence $\{Y_n, n \in \mathbb{N}^+\}$ is called the *external process*. The external process can be considered as a nuisance process. It is assumed that the distribution of Y_n is not completely known: $p(\cdot | \theta)$ is the probability density of Y_n with respect to the σ -finite measure v on Y for all $\theta \in \theta$ where (θ, T) is a Borel space. We also assume $v(\{y \in Y \mid p(y|\theta) \neq p(y|\tilde{\theta})\}) > 0$ for $\theta \neq \tilde{\theta}$. It is easy to transform these models into our framework. To this end let $P(\{F(x,a,y)\}|x,a,y) = 1$ for $x \in X$, $a \in A$ and $y \in Y$, and let X be the Borel σ -field on X, and let \tilde{A} and

Y be the Borel σ -fields on A and Y respectively. Further let I be a singleton, i.e. I := {1} and K₁ := X × A. At each stage Y_n is sampled from the distribution with density $p(\cdot | \theta)$, $\theta \in \theta$. Let there be a reward function satisfying 2.1 (k). Then all conditions of 2.1 are satisfied. Examples of this class are the *linear system* with unknown disturbance distribution as studied in [Aoki (1968)], and *inventory models* with unknown demand distribution with or without backlogging (in chapter 5 we study such a model extensively). Another example of these models is the *replacement model* with *additive damage* as considered in [Taylor (1975)] where the distribution of the so-called *shocks* is not completely known (in chapter 7 we consider this model too).

Example 2.4

A model that satisfies all conditions 2.1(a) - 2.1(j), but for which the reward function is not bounded from above, can sometimes be transformed into a model satisfying all conditions of 2.1. For this purpose we replace 2.1(k)by another condition which is due to Wessels (cf. [Wessels (1977]), who assumes the existence of a so-called *bounding function* b, i.e. a positive measurable function on X, and a positive number M such that for all $x \in X$, $a \in A$ and $y \in Y$:

(i)
$$\int P(dx' | x, a, y) b(x') \leq b(x)$$

(ii) $r(x,a,y) \leq Mb(x)$.

We shall carry out this transformation for the case where X is countable. It is easy to extend the argument to the general case. Define:

$$P^{*}(x'|x,a,y) := P(\{x'\}|x,a,y)b(x')b(x)^{-1}$$

$$r^{*}(x,a,y) := r(x,a,y)b(x)^{-1}, \text{ for } x,x' \in X, a \in A, y \in Y.$$

As it may happen that $\sum_{x' \in X} p^*(\{x'\} | x, a, y) < 1$ we add a state x^* to X and let $x^* := X \cup \{x^*\}$. Further we define for $x \in X$, $a \in A$ and $y \in Y$:

$$P^{*}(x^{*}|x^{*},a,y) := 1, P^{*}(x^{*}|x,a,y) := 1 - \sum_{x^{*} \in X} P(\{x^{*}\}|x,a,y),$$

 $r^*(x^*,a,y) := 0$ and $b(x^*) := 1$. Each strategy for the original model is also a strategy for the new model (except in state x^*). We denote the expectation for the transformed model by \mathbf{I}_{\cdot}^* . Note that for $x_j \in X$, $a_j \in A$, $y_j \in Y$:

$$b(x_0)^{-1} \{ \begin{array}{l} \prod_{j=0}^{n-1} P(\{x_{j+1}\} | x_j, a_j, y_{j+1}) \} b(x_n) = \\ = \prod_{j=0}^{n-1} \{ P(\{x_{j+1}\} | x_j, a_j, y_{j+1}) b(x_{j+1}) b(x_j)^{-1} \} = \prod_{j=0}^{n-1} P^*(x_{j+1} | x_j, a_j, y_{j+1}) . \end{cases}$$

And therefore

$$b(x_0)^{-1} \{ \prod_{j=0}^{n-1} P(\{x_{j+1}\} | x_j, a_j, y_{j+1}) \} r(x_n, a_n, y_{n+1}) = \\ = \{ \prod_{j=0}^{n-1} P^*(x_{j+1} | x_j, a_j, y_{j+1}) \} r^*(x_n, a_n, y_{n+1}) .$$

Now it is straightforward to verify that for $x \in X$, $q \in W$ and $\pi \in \Pi$:

$$b(x)^{-1} \mathbb{E}_{x,q}^{\pi} [r(x_n, A_n, Y_{n+1})] = \mathbb{E}_{x,q}^{*\pi} [r^*(x_n, A_n, Y_{n+1})].$$

This shows the equivalence of both models.

2.2 Posterior distributions

As already announced, the *posterior distribution* of the random variable Z, which represents the unknown parameter, plays an important role in this monograph. We define random variables on (Ω, H) with range the set W, the set of distributions on (θ, T) and afterwards we show that these random variables are versions of the conditional distribution of Z, given the observed histories of the process. This property justifies calling these random variables the *posterior distributions*.

We start with some definitions.

2.16 On
$$\Omega$$
 we define, for $i \in I$, the function $Z_i: Z_i(\omega) = \theta_i$ where $\omega = (\theta, x_0, a_0, y_1, x_1, a_1, \ldots) \in \Omega$ and where $\theta = (\theta_i)_{i \in I}$.

Hence $Z = (Z_i)_{i \in I}$ and we may interpret the random variable Z_i , $i \in I$ as the parameter of the distribution from which Y_n is sampled, if

Note that the n-th observation from the distribution determined by $p_i(\cdot | \theta_i)$, $\theta_i \in \theta_i$ occurs at stage $\tau(i,n)$ and note also that for each $\omega \in \Omega$ and each $k \in \mathbb{N}^*$ there is exactly one pair (i,n), with $i \in I$, $n \in \mathbb{N}^*$ such that

$$\tau(\mathbf{i},\mathbf{n})(\omega) = \mathbf{k} \ .$$

In the rest of this chapter the sub- σ -fields in H, induced by the observable random variables, are used frequently, therefore we introduce the notation:

2.18
$$F_n := \sigma(X_0, A_0, Y_1, X_1, A_1, \dots, Y_n, X_n, A_n) , \quad n \in \mathbb{N}$$

For the stopping times $\tau(i,n)$ we define the usual σ -fields $F_{\tau(i,n)}$:

2.19
$$F_{\tau(i,n)} := \{ B \in H \mid B \cap \{\tau(i,n) = k\} \in F_k \text{ for all } k \in \mathbb{N} \}.$$

Note that $\{\tau(i,n) = k\} \in F_{k-1}$ for $n, k \in \mathbb{N}^*$. Since (θ, T) is a product space we define, for each $q \in W$ the marginal distributions q_i on (θ_i, T_i) , for $i \in I$:

2.20 Let $B \in T$, then

$$q_{i}(B) := \int_{\{\theta \mid \theta_{i} \in B\}} q(d\theta) .$$

It seems to be quite natural to work with prior distributions q that are product-measures, i.e. $q = \mathfrak{O} q_i$. However most results of this monograph are valid without this assumption. Note that the assumption that $q = \mathfrak{O} q_i$ is equivalent with the assumption that Z_i , $i \in I$ are independent. In $i \in I$ th. 2.1 we return to this matter.

In order to define the posterior distributions we define, for $n \in \mathbb{N}^*$, the functions α_n on Ω with range the set of measures on the parameter space (θ, T) and for $i \in I$ the random variables $\alpha_{i,n}$ on Ω with range the set of measures on (θ_i, T_i) :

2.21 (i)
$$\alpha_{n}(B) := \int_{B}^{n} \prod_{j=1}^{n} \sum_{i \in I} \mathbf{1}_{K_{i}} (\mathbf{x}_{j-1}, \mathbf{A}_{j-1}) \mathbf{p}_{i} (\mathbf{x}_{j} | \boldsymbol{\theta}_{i}) \mathbf{q}(d\boldsymbol{\theta})$$

(ii)
$$\alpha_{i,n}(B_i) := \int_{B_i}^{n} \frac{1}{j=1} \left\{ 1_{K_i}(X_{j-1}, A_{j-1}) p_i(Y_j | \theta_i) + 1 - 1_{K_i}(X_{j-1}, A_{j-1}) \right\} q_i(d\theta_i)$$

where $B \in T$, $B_i \in T_i$ and θ_i the i-th coordinate of $\theta \in \theta$ (for notational convenience we have omitted the dependence on $\omega \in \Omega$ in 2.21). The integrand of 2.21 (i) may be considered as the *likelihood function* of the parameter θ at time n and similarly the integrand of 2.21 (ii) as the likelihood function of the parameter-coordinate θ_i , at time n. The following equality clarifies this. It is easy to verify that on Ω we have

2.22
$$\prod_{j=1}^{n} \{1_{K_{i}}(x_{j-1}, A_{j-1}) p_{i}(y_{j} \mid \theta_{i}) + 1 - 1_{K_{i}}(x_{j-1}, A_{j-1})\}$$

 $= \prod_{\{k>0 \mid \tau(i,k) \le n\}} p_i(Y_{\tau(i,k)} \mid \theta_i), \quad i \in I.$

Here we used the convention:

2.23 For any real-valued function f on Y and a stopping time τ

$$f(Y_{\tau(\omega)}(\omega)) := 0$$
 if $\tau(\omega) = \infty$, for $\omega \in \Omega$.

Finally, we are ready to define the *posterior distribution* Q_n for the *prior distribution* $q \in W$, as a random variable on Ω with range the set W: 2.24 Let $B \in T$, then $Q_n(B) := q(B)$ and for $\omega \in \Omega$ and $n \in \mathbb{N}^*$:

$$Q_{n}(B)(\omega) := \alpha_{n}(B)(\omega) \{\alpha_{n}(\theta)(\omega)\}^{-1}$$
, if $\alpha_{n}(\theta)(\omega) > 0$

:= q(B) otherwise.

(In th. 2.1 it turns out that $\alpha_{n}(\theta) > 0$, $\mathbb{P}_{\rho,q}^{\pi}$ -a.s.). And similarly we define the *posterior distributions* $Q_{i,n}$ for $i \in I$, $n \in \mathbb{N}$: 2.25 Let $B \in T_{i}$, then $Q_{i,0}(B) := q_{i}(B)$ and for $\omega \in \Omega$ and $n \in \mathbb{N}^{*}$

$$Q_{i,n}(B)(\omega) := \alpha_{i,n}(B)(\omega) \{\alpha_{i,n}(\theta_i)(\omega)\}^{-1}, \quad \text{if } \alpha_{i,n}(\theta_i)(\omega) > 0$$
$$:= q_i(B) \text{ otherwise.}$$

Note that $Q_n(\cdot)(\omega)$ and $Q_{i,n}(\cdot)(\omega)$ are probabilities for all $\omega \in \Omega$. The measurability of Q_n and $Q_{i,n}$ is a direct consequence of lemma 1.5 (i). The name "posterior distribution" is justified in th. 2.1.

In th. 2.1 we collect some obvious properties of the random variables Q_n and $Q_{i,n}$. Throughout this chapter we fix a starting distribution $\rho \in \mathcal{P}(X)$, a prior distribution $q \in W$ and a strategy $\pi \in \Pi$, and for notational convenience we write \mathbb{P} and \mathbb{E} instead of $\mathbb{P}_{\rho,q}^{\pi}$ and $\mathbb{E}_{\rho,q}^{\pi}$.

Theorem 2.1

- Let $B \in T$ and $B_i \in T_i$, for $i \in I$. Then:
- (i) $\mathbb{IP}[Z \in B | F_n] = Q_n(B)$, \mathbb{IP} -a.s.
- (ii) if $q = \bigotimes q_i$ then $Q_n = \bigotimes Q_{i,n}$. $i \in I$ $i \in I$
- (iii) if $q = \bigotimes_{i \in I} q_i$ then $\mathbb{P}[\mathbb{Z}_i \in B_i | \mathcal{F}_n] = Q_{i,n}(B_i)$, \mathbb{P} -a.s.
- (iv) if $q = \bigotimes_{i \in I} q_i$ then $\mathbb{P}[\mathbb{Z}_i \in \mathbb{B}_i | \mathcal{F}_{\tau(i,n)}] = Q_{i,\tau(i,n)}(\mathbb{B}_i)$ on $\{\tau(i,n) < \infty\}$ \mathbb{P} -a.s.

(v)
$$Q_{n+1}(B) = \sum_{i \in I} {}^{1}K_{i}(X_{n}, A_{n}) \frac{\int_{B} p_{i}(Y_{n+1} | \theta_{i})Q_{n}(d\theta)}{\int_{\theta} p_{i}(Y_{n+1} | \theta_{i})Q_{n}(d\theta)}$$

(on the subset of Ω where the denominator is positive).

(vi)
$$\operatorname{IE}[Q_n(B)|F_m] = Q_m(B)$$
 if $n > m$, IP-a.s.

Proof.

t

Let $C := \Theta \times E_0 \times F_0 \times D_1 \times E_1 \times F_1 \times \dots \times D_n \times E_n \times F_n \times (Y \times X \times A)^{IN}$ where $D_i \in Y$, $E_i \in X$ and $F_i \in A$ for $i \in IN$. Then $C \in F_n$ and

$$\int \mathbf{IP} \left[\mathbf{Z} \in \mathbf{B} \middle| \mathbf{F}_n \right] d\mathbf{IP} = \mathbf{IP} \left[\mathbf{Z} \in \mathbf{B}, \mathbf{X}_0 \in \mathbf{E}_0, \mathbf{A}_0 \in \mathbf{F}_0, \dots, \mathbf{Y}_n \in \mathbf{D}_n, \mathbf{X}_n \in \mathbf{E}_n, \mathbf{A}_n \in \mathbf{F}_n \right] = \mathbf{C}$$

$$= \int_{B} q(d\theta) \int_{E_{0}} \rho(dx_{0}) \int_{E_{0}} \pi_{0}(da_{0} | x_{0}) \int_{D_{1}} \nu(dy_{1}) \int_{E_{1}} P(dx_{1} | x_{0}, a_{0}, y_{1}) \dots$$

$$\cdots \int_{D_{n}}^{V} (dy_{n}) \int_{E_{n}}^{P} (dx_{n} | x_{n-1}, a_{n-1}, y_{n}) \int_{E_{n}}^{\pi} (da_{n} | x_{0}, a_{0}, y_{1}, x_{1}, \dots, y_{n}, x_{n}) \cdot \\ \prod_{j=1}^{n} (\sum_{i \in I}^{1} x_{i} (x_{j-1}, a_{j-1}) p_{i} (y_{j} | b_{i})) = \\ = \int_{E_{0}}^{\rho} (dx_{0}) \int_{F_{0}}^{\pi} (da_{0} | x_{0}) \int_{D_{1}}^{V} (dy_{1}) \int_{E_{1}}^{P} (dx_{1} | x_{0}, a_{0}, y_{1}) \cdots \\ \cdots \int_{D_{n}}^{V} (dy_{n}) \int_{E_{n}}^{P} (dx_{n} | x_{n-1}, a_{n-1}, y_{n}) \int_{F_{n}}^{\pi} (da_{n} | x_{0}, a_{0}, y_{1}, x_{1}, \dots, y_{n}, x_{n}) \cdot \\ \cdot \int_{Q}^{V} (dy_{n}) \int_{E_{n}}^{P} (dx_{n} | x_{n-1}, a_{n-1}, y_{n}) \int_{F_{n}}^{\pi} (da_{n} | x_{0}, a_{0}, y_{1}, x_{1}, \dots, y_{n}, x_{n}) \cdot \\ \cdot \int_{Q}^{Q} (d\theta) \prod_{j=1}^{n} \{\sum_{i \in I}^{1} x_{i} (x_{j-1}, a_{j-1}) p_{i} (y_{j} | b_{i})\} \frac{\int_{Q}^{Q} (d\theta) \prod_{i=1}^{n} \{\sum_{i \in I}^{1} x_{i} (x_{j-1}, a_{j-1}) p_{i} (y_{j} | b_{i})\}}{\int_{Q}^{Q} (d\theta) \prod_{j=1}^{n} \{\sum_{i \in I}^{1} x_{i} (x_{j-1}, a_{j-1}) p_{i} (y_{j} | b_{i})\}} = \\ = \mathbb{E} [1_{C} \rho_{n} (B)] = \int_{C}^{Q} \rho_{n} (B) d\mathbb{P} \cdot$$

The second equality is a consequence of changing the order of integration. (Note that we used the convention $\frac{0}{0} = 1$) Hence

(*)
$$\int_{C} \mathbb{IP} \left[\mathbf{Z} \in \mathbf{B} \middle| \mathcal{F}_{n} \right] d\mathbb{IP} = \int_{C} \mathcal{Q}_{n} (\mathbf{B}) d\mathbb{IP}$$

By standard arguments we have (*) for all $C \in F_n$. Since, by 2.21(i) and 2.24 $Q_n(B)$ is F_n -measurable the assertion (i) is proved. We proceed with assertions (ii) and (iii). It is easy to verify that on Ω

2.26
$$\prod_{j=1}^{n} \sum_{i \in I} (X_{j-1}, A_{j-1}) p_i(Y_j | \theta_i) = \prod_{i \in I} \prod_{\{k>0 \mid \tau(i,k) \le n\}} p_i(Y_{\tau(i,k)} | \theta_i)$$

(remember that for each $m \in \mathbb{N}^{*}$ there is exactly one pair (i,k), $i \in I$, $k \in \mathbb{N}^{*}$ such that $\tau(i,k) = m$ and remember that an empty product equals one). Since $q = \bigotimes q$, we have, by 2.21 and 2.26, $i \in I^{*}$
$$\alpha_{n} \{ \{ \theta | \theta_{i} \in B_{i}, i \in I \} \} = \prod_{i \in I} \alpha_{i,n}(B_{i})$$

for $B_i \in T_i$, $i \in I$ and therefore

$$Q_{n} \{ \{ \theta \mid \theta_{i} \in B_{i}, i \in I \} \} = \pi Q_{i,n}(B_{i})$$

In particular we have for $B = \prod_{i \in I} B_i$ with $B_i = \theta_i$ for all $i \neq j$ and with $B_j \in T_j$: $Q_n(B) = Q_{j,n}(B_j)$. Hence $IP[Z_j \in B_j | F_n] = Q_{j,n}(B_j)$. To prove (iv) note that

$$\mathbb{P}[\mathbf{z}_{i} \in \mathbf{B}_{i} | \mathbf{F}_{\tau(i,n)}] = \mathbb{P}[\mathbf{z}_{i} \in \mathbf{B}_{i} | \mathbf{F}_{i}]$$

on the set $\{\tau(i,n) = k\}$ for $k \in \mathbb{N}$ (cf. [Neveu (1972) th. III-1-3]). Next we prove (v). It is easy to verify that, for $B \in \mathcal{T}$:

$$\alpha_{n+1}(B) = \int_{B} \sum_{i \in I} \frac{1}{K_{i}} (X_{n}, A_{n}) p_{i}(Y_{n+1} | \theta_{i}) \{ \prod_{j=1}^{n} \sum_{i \in I} \frac{1}{K_{i}} (X_{j-1}, A_{j-1}) p_{i}(Y_{j} | \theta_{i}) \} q(d\theta) =$$
$$= \int_{B} \sum_{i \in I} \frac{1}{K_{i}} (X_{n}, A_{n}) p_{i}(Y_{n+1} | \theta_{i}) \alpha_{n}(d\theta) .$$

Hence on the subset of Ω where $\alpha_{n+1}(\theta) > 0$ we have

$$Q_{n+1}(B) = \frac{\alpha_{n+1}(B)}{\alpha_{n+1}(\theta)} = \sum_{i \in I} \mathbf{1}_{K_i}(X_n, A_n) \frac{\int_{B} \mathbf{p}_i(Y_{n+1} | \theta_i) \alpha_n(d\theta)}{\int_{\Theta} \mathbf{p}_i(Y_{n+1} | \theta_i) \alpha_n(d\theta)},$$

and dividing numerator by $\alpha_n(\theta)$ we obtain the desired result, since

$$\frac{1}{\alpha_n(\theta)} \alpha_n(d\theta) = Q_n(d\theta)$$
, on Ω

The proof of assertion (vi) is an immediate consequence of (i).

We introduce some notations which are useful in the following chapters. On $Y \times W$ we define real-valued functions p_i , $i \in I$:

2.27
$$p_{i}(y,q) := \int q(d\theta)p_{i}(y|\theta_{i}) \quad y \in Y, q \in W$$
.

Notice that these functions may be considered as extensions of the functions defined in 2.1(j), by the embedding of θ in W, in fact $p_i(y, \theta) = p_i(y|\theta_i)$, $\theta \in \theta$.

It is a consequence of lemma 1.6(iii) that the function $(y,q) \rightarrow p_i(y,q)$ is measurable, i ϵ I.

Note that $p_i(\cdot,q)$ is a probability density with respect to the measure v, for $i \in I$ and $q \in W$.

Further we define functions T, on Y \times W with range the set W, for i \in I:

2.28
$$T_{i,y}(q)$$
 is a probability on θ such that, for $B \in \theta$

$$T_{i,Y}(q)(B) := \int_{B} p_i(y|\theta_i)q(d\theta) \{p_i(y,q)\}^{-1}, \text{ if } p_i(y,q) > 0$$
$$:= q(B), \text{ otherwise.}$$

Again, using lemma 1.6(iii) we find the measurability of

$$(y,q) \rightarrow T_{i,y}(q)(B)$$
, $B \in T, i \in I$.

Hence T_i is a transition probability from $Y \times W$ to θ . We may interpret $T_{i,y}(q)$ as the posterior distribution, if q is the prior distribution and $y \in Y$ is observed from the distribution belonging to the set K_i . The following formula is easily verified:

2.29
$$Q_{n+1} = \sum_{i \in I} {}^{1}\kappa_{i} (X_{n}, A_{n}) T_{i, Y_{n+1}} (Q_{n})$$
.

For $q \in W$ we define the functions q_0, q_1, q_2, \dots recursively: 2.30 (i) $q_0 : W \neq W$, $q_0(q) := q$.

(ii)
$$q_n : W \times (X \times A \times Y)^n \neq W$$
, $n \in \mathbb{N}^3$

such that

$$q_{n}(q,x_{0},a_{0},\ldots,y_{n}) := \sum_{i \in I}^{1} K_{i}(x_{n-1},a_{n-1}) T_{i,y_{n}}(q_{n-1}(q,x_{0},a_{0},\ldots,y_{n-1},x_{n-1}))$$

for $q \in W$, $x_{\ell} \in X$, $y_{\ell} \in Y$ and $a_{\ell} \in A$, $\ell \in \mathbb{N}$. It is straightforward to verify that, for $B \in T$

2.31
$$q_{n}(q,x_{0},a_{0},\ldots,y_{n}) (B) = \frac{\int_{i=0}^{n-1} \{\sum_{i \in I} {}^{1}K_{i}(x_{j},a_{j})p_{i}(y_{j+1}|\theta_{i})\}q(d\theta)}{\int_{i=0}^{n-1} \{\sum_{i \in I} {}^{1}K_{i}(x_{j},a_{j})p_{i}(y_{j+1}|\theta_{i})\}q(d\theta)}$$

provided that the denominator is positive. Using the notations above, we may write:

$$Q_n = q_n(Q_0, X_0, A_0, \dots, Y_n)$$
, on Ω .

We conclude this section with a few remarks:

Remarks .

(i) If ξ and η are independent random variables, defined on some probability space, and B is a sub- σ -field, then in general $\mathbb{E}[\xi \cdot \eta | B] \neq \mathbb{E}[\xi | B] \mathbb{E}[\eta | B]$.

However in th. 2.1(ii) we proved that equality holds \mathbb{P} -a.s., if $\xi := f(Z_i)$, $n := g(Z_j)$ and $B := F_n$ for $i \neq j$, i,j, $\in I$, f and g non-negative measurable functions on θ_i and θ_j respectively, and if $q = \otimes q_i$.

- (ii) Instead of defining the posterior distributions Q_n by 2.25 we could define them directly as conditional distributions (cf. th. 2.1(i)). However the conditional distribution $\mathbb{P}[z \in \cdot | F_n]$ is undetermined on some set with \mathbb{P} -measure zero.
- (iii) If the prior distribution q is concentrated on a set of finitely many points then all posterior distributions are concentrated on this set.

2.3 Limit behaviour of the posterior distributions

The main result of this section is the convergence of the posterior distributions Q_n of Z (cf. 2.16 and 2.25) to a degenerate distribution, i.e. $Q_n(B)$ converges almost surely to $1_B(Z)$ for all $B \in T$, provided that the

strategy, by which the system is controlled, ensures that the number of visits to the set $K_1 \subset X \times A$ is infinite, with probability one. Our proof of this statement is similar to the proof of a theorem in [Doob (1949)]. In fact, our result extends Doob's result to a more abstract setting. Another result of this section concerns the expected differences of the successive posterior distributions. In the proofs of the above mentioned results, elementary martingale theory plays an important role. As a corollary of the next lemma we shall show that given Z, the sequence $\{Y_{\tau(i,n)}, n \in \mathbb{N}^*\}$ forms a sequence of *conditionally independent and identically distributed* random variables (cf. [Neveu (1965) page 129]), provided that $\mathbb{P}[\tau(i,n) < \infty] = 1$ for all $n \in \mathbb{N}^*$. This lemma is also used in chapter 6 (remember that $\rho \in P(X)$, $q \in W$ and $\pi \in \mathbb{I}$ are fixed).

Lemma 2.2

Let f be a nonnegative measurable function on Yⁿ. Then

$$\mathbb{IE}\left[f(\mathbb{Y}_{\tau(i,1)},\ldots,\mathbb{Y}_{\tau(i,n)})\right] \leq \int q_{i}(d\theta_{i})\left\{\int \ldots \int v(dy_{1})\ldots v(dy_{n})\prod_{j=1}^{n} p_{i}(y_{j}|\theta_{i})f(y_{1},\ldots,y_{n})\right\}$$

with equality if

 $\mathbb{IP}[\{\tau(i,n) < \infty\}] = 1$

(we use convention 2.23, note that $\tau(i,k) < \tau(i,n)$, k < n).

Proof.

It suffices to consider functions f of the form:

$$f(y_1, \dots, y_n) = \prod_{j=1}^n \prod_{i=1}^n (y_j), \qquad E_j \in \mathcal{V}.$$

It is easy to verify that for $E \in Y$ we have \mathbb{P} -a.s. (cf. th. 1.4):

(*)
$$\mathbb{E} \begin{bmatrix} 1_{E}(Y_{k}) | Z, X_{0}, A_{0}, \dots, Y_{k-1}, X_{k-1}, A_{k-1} \end{bmatrix} = \\ = \int v(dy) \sum_{i \in I} 1_{K_{i}} (X_{k-1}, A_{k-1}) p_{i}(Y|Z_{i}) 1_{E}(Y) .$$

Note that $\{\tau(i, n) = k\} \in F_{k-1} \subset \sigma(Z, X_{0}, A_{0}, \dots, Y_{k-1}, X_{k-1}, A_{k-1})$

Let h be a nonnegative measurable function on $\boldsymbol{\theta}_i$. Consider

$$\begin{split} \mathbf{E} \left[\mathbf{h} \left(\mathbf{Z}_{i} \right)_{j=1}^{n} \mathbf{1}_{\mathbf{E}_{j}} \left(\mathbf{Y}_{\tau} \left(\mathbf{i}, j \right) \right) \right] &= \sum_{k=n}^{\infty} \mathbf{E} \left[\mathbf{h} \left(\mathbf{Z}_{i} \right)_{j=1}^{n-1} \mathbf{1}_{\mathbf{E}_{j}} \left(\mathbf{Y}_{\tau} \left(\mathbf{i}, j \right) \right)^{1} \left\{ \tau \left(\mathbf{i}, n \right) = k \right\}^{1} \mathbf{E}_{n} \left(\mathbf{Y}_{k} \right) \right] \\ &= \sum_{k=n}^{\infty} \mathbf{E} \left[\mathbf{h} \left(\mathbf{Z}_{i} \right)_{j=1}^{n-1} \mathbf{E}_{j} \left(\mathbf{Y}_{\tau} \left(\mathbf{i}, j \right) \right)^{1} \left\{ \tau \left(\mathbf{i}, n \right) = k \right\}^{1} \mathbf{E} \left[\mathbf{1}_{\mathbf{E}_{n}} \left(\mathbf{Y}_{k} \right) \left| \mathbf{Z}, \mathbf{X}_{0}, \mathbf{A}_{0}, \dots, \mathbf{Y}_{k-1}, \mathbf{X}_{k-1}, \mathbf{A}_{k-1} \right] \right] = \\ &= \sum_{k=n}^{\infty} \mathbf{E} \left[\mathbf{h} \left(\mathbf{Z}_{i} \right)_{j=1}^{n-1} \mathbf{E}_{j} \left(\mathbf{Y}_{\tau} \left(\mathbf{i}, j \right) \right)^{1} \left\{ \tau \left(\mathbf{i}, n \right) = k \right\}^{1} \int_{\mathbf{V}} \left(d\mathbf{y} \right) \mathbf{P}_{i} \left(\mathbf{y} \right| \mathbf{Z}_{i} \right)^{1} \mathbf{E}_{n} \left(\mathbf{y} \right) \right] \\ &\leq \mathbf{E} \left[\mathbf{h} \left(\mathbf{Z}_{i} \right)_{j=1}^{n-1} \mathbf{E}_{j} \left(\mathbf{Y}_{\tau} \left(\mathbf{i}, j \right) \right)^{1} \int_{\mathbf{V}} \left(d\mathbf{y} \right) \mathbf{P}_{i} \left(\mathbf{y} \right| \mathbf{Z}_{i} \right)^{1} \mathbf{E}_{n} \left(\mathbf{y} \right) \right] \\ &\leq \mathbf{E} \left[\mathbf{h} \left(\mathbf{Z}_{i} \right)_{j=1}^{n-1} \mathbf{E}_{j} \left(\mathbf{Y}_{\tau} \left(\mathbf{i}, j \right) \right)^{1} \int_{\mathbf{V}} \left(d\mathbf{y} \right) \mathbf{P}_{i} \left(\mathbf{y} \right| \mathbf{Z}_{i} \right)^{1} \mathbf{E}_{n} \left(\mathbf{y} \right) \right] \end{aligned}$$

with equality if $\mathbb{P}[\{\tau(i,n) < \infty\}] = 1$. Note that the third equality is a consequence of (*) and the fact that $(X_{k-1}, A_{k-1}) \in K_i$ on $\{\tau(i,n) = k\}$. Now we may repeat the argument for $\tau(i,n-1)$ with the function $\widetilde{h}(Z_i) := h(Z_i) \int v(dy) p_i(y|Z_i) 1_{E_n}(y)$. So we find putting h = 1

$$\mathbb{E}\left[\prod_{j=1}^{n} \mathbb{1}_{E_{j}}(\mathbb{Y}_{\tau(i,j)})\right] \leq \mathbb{E}\left[\prod_{j=1}^{n} \int \mathcal{V}(d\mathbf{y}_{j}) \mathbb{P}_{i}(\mathbf{y}_{j} | \mathbf{z}_{i}) \mathbb{1}_{E_{j}}(\mathbf{y}_{j})\right]$$

which proves the lemma.

The following corollary is immediate.

Corollary 2.3

Let $\mathbb{P}[\tau(i,n) < \infty] = 1$. Then for $\mathbb{E}_1, \dots, \mathbb{E}_n \in \mathcal{Y}$ we have q-a.s.

$$\mathbb{P}_{\rho,\theta}^{\pi}[\mathbb{Y}_{\tau(i,1)} \in \mathbb{E}_{1}, \dots, \mathbb{Y}_{\tau(i,n)} \in \mathbb{E}_{n}] = \prod_{j=1}^{n} \int_{\mathbb{E}_{j}} \mathbb{P}_{i}(y|\theta_{i}) \vee (dy)$$

or similarly as functions on $\Omega\,$ we have IP-a.s.

$$\mathbb{P}_{\rho,\mathbf{Z}}^{\pi}[\mathbb{Y}_{\tau(i,1)} \in \mathbb{E}_{1}, \dots, \mathbb{Y}_{\tau(i,n)} \in \mathbb{E}_{n}] = \prod_{j=1}^{n} \int_{\mathbb{E}_{j}} \mathbb{P}_{i}(y|\mathbf{Z}_{j}) \vee (dy)$$

Hence

$$Y_{\tau(i,1)}, \dots, Y_{\tau(i,n)}$$
 are, conditionally given Z, i.i.d.

Theorem 2.4

If $\mathbb{IP}[\cap_{\tau \in \mathbb{N}} \{\tau(i,n) < \infty\}] = 1$, then for all bounded real-valued measurable $n \in \mathbb{N}$ functions f on θ_i :

$$\lim_{n\to\infty} \int f(\theta_i)Q_n(d\theta) = f(Z_i) \qquad \text{IP-a.s.}$$

Remark.

This holds in particular for all bounded continuous functions f. Hence Q_n converges weakly to the distribution which is degenerate in Z_i , **P**-a.s.

Proof. The proof is divided into five parts.

a) We first reduce the problem.

It is easy to verify that $\int f(\theta_i) Q_n(d\theta) = \mathbb{E}[f(Z_i)|F_n] \quad \mathbb{P}\text{-a.s.}$ (cf. th. 2.1) by considering indicator functions first. Since $\{F_n, n \in \mathbb{N}\}$ is an increasing sequence of σ -fields with limit F_{∞} , and since the sequence $\{\mathbb{E}[f(Z_i)|F_n], n \in \mathbb{N}\}$ is a martingale with respect to $\{F_n, n \in \mathbb{N}\}$, we have (cf.[Neveu (1972) th. II-2-11]):

$$\lim_{n \to \infty} \left| f(\theta_i) Q_n(d\theta) = \operatorname{IE}[f(Z_i) | F_{\infty}], \quad \operatorname{IP-a.s.} \right|$$

Let F_{∞}^{\star} be the completion of F_{∞} in H, i.e. $F_{\infty}^{\star} := \{A \land N | A \in F_{\infty}, N \in H, \mathbb{P}[N] = 0\}$. Obviously, it is sufficient to prove that $f(Z_{1})$ is F_{∞}^{\star} -measurable, since $\mathbb{IE}[f(Z_{1}) | F_{\infty}] = \mathbb{IE}[f(Z_{1}) | F_{\infty}^{\star}]$, \mathbb{IP} -a.s.

So we proceed with proving that $f(Z_i)$ is F_{m}^* -measurable.

b) We define on $\Omega^* := \cap \{\tau(i,n) < \infty\}$ the *empirical distributions* on $(Y,Y): n \in \mathbb{N}$

$$F_{n}(E) := \frac{1}{n} \sum_{j=1}^{n} \mathbf{1}_{E} (\mathbf{Y}_{\tau(i,j)}) , \quad E \in \mathcal{V} \quad (\text{note that } \Omega^{*} \in F_{\infty})$$

First we verify that $F_n(\cdot)$ is a measurable function from Ω^* to P(Y), where P(Y) is endowed with the σ -field generated by the weak topology.

This σ -field is generated by sets of the form (cf. lemma 1.5)

$$\{\mu \in \mathcal{P}(\mathcal{Y}) | \mu(\mathbf{E}) \leq \mathbf{a}\}, \quad \mathbf{E} \in \mathcal{Y}, \quad \mathbf{a} \in \mathbb{IR}.$$

Since

$$\{\omega \in \Omega^* | F_n(E) \in \{\mu \in \mathcal{P}(\mathcal{Y}) | \mu(E) \le a\}\} = \{\omega \in \Omega^* | F_n(E) \le a\} \in \mathcal{F}_{\omega},$$

the measurability of $F_n(\cdot)$ has been verified.

c) Next we consider the limit behaviour of \mathbf{F}_{n} for $n \to \infty$, as functions on Ω^{*} . Let \widetilde{F}_{∞} be the restriction of F_{∞} on Ω^{*} , i.e. $\widetilde{F}_{\infty} := \{\mathbf{A} \cap \Omega^{*} | \mathbf{A} \in F_{\infty}\}$. Since $\Omega^{*} \in F_{\infty}$ we have $\widetilde{F}_{\infty} \subset F_{\infty}$. By corollary 2.3 we have, for q-almost all $\theta: \Upsilon_{\tau(\mathbf{i},1)}, \ldots, \Upsilon_{\tau(\mathbf{i},n)}$ are i.i.d. on $(\Omega^{*}, \widetilde{F}_{\infty}, \mathbb{P}_{\rho,\theta}^{\pi})$ for $n \in \mathbb{N}^{*}$ with common distribution $\mathbb{P}_{\theta_{i}}$, where $\mathbb{P}: \theta_{i} \to \mathcal{P}(Y)$ is defined by

$$P_{\theta_{\mathbf{i}}}(\mathbf{E}) := \int_{\mathbf{E}} p_{\mathbf{i}}(\mathbf{y} \mid \theta_{\mathbf{i}}) \vee (d\mathbf{y}) , \quad \text{for } \mathbf{E} \in \mathcal{Y} .$$

Let C(Y) be the set of all bounded continuous functions on Y. Since (Y, V) is a Borel space we may use a generalization of the Glivenko-Cantelli lemma (cf. [Parthasarathy (1967) th. 7.1 page 53]). This lemma states that F_n converges weakly to P_{θ_i} , $\mathbb{P}_{\rho,\theta}^{\pi}$ -a.s. Hence, for q-almost all θ , we have

$$\mathbb{P}_{\rho,\theta}^{\pi}\left[\{\omega \in \Omega^{*} | \lim_{n \to \infty} \int g(y) F_{n}(dy) = \int g(y) P_{\theta_{i}}(dy) \text{ for all } g \in C(Y)\}\right] = 1.$$

Since $\mathbf{IP}_{\rho,\theta}^{\pi}[\mathbf{Z}_{i} = \theta_{i}] = 1$, we have for

$$\Omega^{**} := \{ \omega \in \Omega^* | \lim_{n \to \infty} \int g(y) F_n(dy) = \int g(y) P_{Z_i}(dy) \text{ for all } g \in C(Y) \}$$

that $\mathbf{IP}_{\rho,\theta}^{\pi} [\Omega^{**}] = 1$ and using 2.10 we find $\mathbf{IP}[\Omega^{**}] = 1$. Hence, since $\mathbf{IP}[\Omega^{**}] = 1$ we have $\Omega^{**} \in F_{\infty}^{*}$.

d) Further we prove that the function P_{Z_i} from Ω to P(Y) is F_{∞}^* measurable. Remember that the σ -field on P(Y) is also generated by sets of the form (cf. lemma 1.5):

$$\{ \mu \in \mathcal{P}(\mathcal{Y}) \mid \int g(y) \mu(dy) \le a \} , \qquad g \in C(\mathcal{Y}), a \in \mathbb{R} .$$
 Since $\Omega^{**} \in F_{\infty}^{*}$ we have

$$\{\omega \in \Omega^{**} | \int g(y) P_{Z_{i}}(dy) \leq a\} = \{\omega \in \Omega^{**} | \lim_{n \to \infty} \int g(y) F_{n}(dy) \leq a\} \in F_{\infty}^{*}.$$

Hence $\{\omega \in \Omega \mid \int g(y) P_{Z_{\underline{i}}}(dy) \le a\} \in F_{\infty}^{*}$ and therefore

$$P_{Z_i}: \Omega \rightarrow P(Y)$$
 is F_{∞} -measurable.

e) Finally we show that Z_i is F_{∞}^* -measurable.

By the separation property (cf. 2.1(j)) we have $P : \theta_i \neq P(Y)$ is a oneone map into P(Y). Since this mapping is measurable we have by Kuratowski's theorem (cf. A7) that P^{-1} is also measurable. Hence, since the function $P^{-1}(P_{Z_i}) : \Omega \rightarrow \theta_i$ is F_{∞}^* -measurable and since $P^{-1}(P_{Z_i}) = Z_i$ we have that Z_i is F_{∞}^* -measurable on Ω^{**} . Therefore, by part (a), the theorem is proved.

```
Corollary 2.5
```

Let
$$i_1, i_2, \dots, i_k \in I$$
 and let f be a bounded measurable function on $e_1 \times \dots \times e_i$, $k \in \overline{\mathbb{N}}^*$. Then:
 i_1 k

(*)
$$\mathbb{IP}\left[\bigcap \bigcap \{\tau(i_j, n) < \infty\} \right] = 1$$

 $j=1 n \in \mathbb{IN}$

implies

$$\lim_{n \to \infty} \int f(\theta_{i_1}, \dots, \theta_{i_k}) Q_n(d\theta) = f(Z_{i_1}, \dots, Z_{i_k}) \quad \mathbb{P} \text{-a.s.}$$

Proof.

We extend f to a function on θ by defining $\tilde{f}(\theta) := f(\theta_1, \dots, \theta_k)$ for $\theta \in \theta$ with θ_{i_j} as i_j -coordinate. Let (*) hold. As in part (a) of the proof of th. 2.4 we have:

$$\lim_{n\to\infty}\int \widetilde{f}(\theta)Q_n(d\theta) = \operatorname{IE}\left[\widetilde{f}(Z)\middle|F_{\infty}\right] = \operatorname{IE}\left[\widetilde{f}(Z)\middle|F_{\infty}^{\star}\right] \qquad \operatorname{IP-a.s.}$$

It suffices to consider functions f of the form

(**)
$$f(\theta_{i},\ldots,\theta_{j}) = \prod_{i=1}^{m} (\theta_{i}) \text{ where } E_{j} \in \mathcal{T}_{i}, m \leq k, m \in \mathbb{N}^{\star}.$$

In the proof of th. 2.4 we showed that (*) implies that Z_{j} is F_{∞}^{*} -measurable, $1 \le j \le k$. Hence $\tilde{f}(Z)$ is F_{∞}^{*} -measurable here, which proves the statement.

Corollary 2.6

Let q be concentrated on a countable subset of θ . Then $\mathbb{P}\left[\begin{array}{c} 0 & \pi \in \mathbb{N} \\ n \in \mathbb{N} \\ \end{array} \\ \left\{\tau(i,n) < \infty\right\}\right] = 1$ implies $\lim_{n \to \infty} \int f(\theta_i) Q_n(d\theta) = f(\theta_i)$, $\mathbb{P}_{\rho,\theta}^{\pi}$ -a.s. for $\theta \in \theta$ with $q(\{\theta\}) > 0$ and for any bounded measurable function f on θ_i . To prove this, note that for $B = \theta \times C$, $C \in H_{\infty}$: $\mathbb{P}_{\rho,q}^{\pi}[B] = 1$ implies $\mathbb{P}_{\rho,\theta}^{\pi}[B] = 1$ for all $\theta \in \theta$ with $q(\{\theta\}) > 0$ (cf. 2.11). The countability condition in corollary 2.6 is essential. In [Freedman (1963),(1965)] and [Fabius (1964)] this problem is studied for the situation of real-valued i.d.d. random variables.

In th. 2.7 we consider a slight extension of th. 2.4, to be used in section 4.2.

Theorem 2.7

Let $i_1, \ldots, i_k \in I$, with $k \in \overline{\mathbb{N}}^*$ and let $\{\sigma_n, n \in \mathbb{N}\}$ be a sequence of stopping times, such that for $m \in \mathbb{N}$, $\{\tau_n = m\} \in F_m$ and $\sigma_0 := 0$, $\sigma_{n+1} > \sigma_n$. Let the σ -field F_{σ} be defined as in 2.19. Assume: $\mathbb{P}\left[\begin{array}{cc} n & n \in [\tau, n] < \infty \end{bmatrix}\right] = 1$. Then, for all bounded measurable functions f on $\theta_1 \times \ldots \times \theta_i$ we have, on $n \in [\sigma_n] < \infty$:

$$\lim_{n \to \infty} \int Q_{\sigma_n} (d\theta) f(\theta_{i_1}, \dots, \theta_{i_k}) = f(Z_{i_1}, \dots, Z_{i_k}) \quad \mathbb{P} \text{-a.s.}$$

(by convention Q_{m} is the zero-measure).

,

Proof. a) We first consider the σ -fields F_{σ_n} in more detail. Let $B \in F_n$, then $B \cap {\sigma_n = k} = \emptyset$ if k < n since ${\sigma_n = k} = \emptyset$. If $k \ge n$ then $B \cap {\sigma_n = k} \in F_k$. Hence $F_n \subseteq F_{\sigma_n}$, $n \in \mathbb{N}$. Now, let $B \in F_{\sigma_n}$. Then

 $B \cap \{\sigma_{n+1} = k\} = B \cap (\cup \{\sigma_n = \ell\}) \cap \{\sigma_{n+1} = k\} \in F_k \text{, since}$

B \cap (\cup { $\sigma_n = l$ }) $\in F_{k-1}$. Hence the sequence { F_{σ_n} , $n \in \mathbb{N}$ } is increasing. Since $F_n \subseteq F_{\sigma_n} \subseteq F_{\infty}$, $n \in \mathbb{N}$ we have $\cup F_n \subseteq \cup F_{\sigma_n} \subseteq F_{\infty}$. Therefore f_{∞} is the smallest σ -field containing $\cup F_{\sigma_n}$. $n \in \mathbb{N}$ σ_n

b) Further we consider $\int_{Q_0} Q_0$ (d0) f(0) where f is a bounded measurable function on 0. Let $B \in T$. Then:

$$\int Q_{\sigma_n}(d\theta) \mathbf{1}_{B}(\theta) = Q_{\sigma_n}(B) = \sum_{m=n}^{\infty} \mathbf{1}_{\{\sigma_n=m\}} Q_m(B) =$$
$$= \sum_{m=n}^{\infty} \mathbf{1}_{\{\sigma_n=m\}} \mathbb{P}[z \in B|F_m] = \mathbb{P}[z \in B|F_{\sigma_n}] \mathbf{1}_{\{\sigma_n < \infty\}} , \quad \mathbb{P}\text{-a.s.}$$

(for the last equality cf. [Neveu (1972) prop. II 1-3]). Hence, using standard arguments, we find for each bounded measurable function f:

(*)
$$\int Q_{\sigma_n}(d\theta) f(\theta) = \mathbb{I} \mathbb{E} [f(Z) | F_{\sigma_n}]_{\{\sigma_n < \infty\}}.$$

c) Note that, by the conclusion of part (a): $\lim_{n \to \infty} \mathbb{E}[f(Z) | F_{\sigma}] = \mathbb{E}[f(Z) | F_{\omega}]$ IP-a.s. Since $\{\sigma_n < \infty\}$, $n \in \mathbb{N}$ is a nonincreasing sequence with limit $\cap \{\sigma_n < \infty\}$ we have, IP-a.s.: $n \in \mathbb{N}$

$$\lim_{n \to \infty} \operatorname{IE} [f(Z) | F_{\sigma}]_{1}^{1} \{ \sigma_{n} < \infty \} = \operatorname{IE} [f(Z) | F_{\sigma}]_{n \in \mathbb{IN}}^{1} \{ \sigma_{n} < \infty \}.$$

In exactly the same way as in the proof of corollary 2.5 we find $\mathbb{E}[f(Z)|F_{m}] = f(Z)$, P-a.s., which proves the theorem.

We conclude this section with a theorem concerning the expected quadratic differences of successive posterior distributions. This result, which might be used to obtain approximations of the value function (cf. chapters 6 and 7) is of some interest in its own right.

Theorem 2.8

Let B_1, B_2, \ldots be a measurable partition of θ_i , let $q = \bigotimes q_i$ and assume $IP[\cap \{\tau(i,n) < \infty\}] = 1.$ $n \in \mathbb{N}$ Then

$$\sum_{n=m}^{\infty} \mathbb{E}\left[\sum_{j=1}^{\infty} \{Q_{i,n+1}(B_j) - Q_{i,n}(B_j)\}^2 \middle| F_m\right] = 1 - \sum_{j=1}^{\infty} Q_{i,m}^2(B_j)$$

in particular, for m = 0

$$\sum_{n=0}^{\infty} \mathbb{IE}\left[\sum_{j=1}^{\infty} \{Q_{i,n+1}(B_{j}) - Q_{i,n}(B_{j})\}^{2}\right] = 1 - \sum_{j=1}^{\infty} q_{i}^{2}(B_{j})$$

Proof.

According to th. 2.1 $Q_{i,n}(B_j) = \mathbb{P}[Z_i \in B_j | F_n], \mathbb{P} \text{-a.s. For } n \ge m \ge 0$: $\mathbb{E}[\{Q_{i,n+1}(B_j) - Q_{i,n}(B_j)\}^2 | F_m] =$ $\mathbb{E}[Q_{i,n+1}^2(B_j) + Q_{i,n}^2(B_j) - 2\mathbb{E}[Q_{i,n+1}(B_j)Q_{i,n}(B_j) | F_n] | F_m] =$ $\mathbb{E}[Q_{i,n+1}^2(B_j) - Q_{i,n}^2(B_j) | F_m], \text{ since } \mathbb{E}[Q_{i,n+1}(B_j) | F_n] = Q_{i,n}(B_j)$ $\mathbb{P} \text{-a.s. (cf. th. 2.1 (vi)). Hence}$ $\sum_{n=m}^{N} \mathbb{E}[\{Q_{i,n+1}(B_j) - Q_{i,n}(B_j)\}^2 | F_m] = \mathbb{E}[Q_{i,N+1}^2(B_j) | F_m] - Q_{i,m}^2(B_j) .$ By th. 2.4 we have $\lim_{n \to \infty} Q_{i,n}(B_j) = {}^1_{B_j}(Z_i), \mathbb{P} \text{-a.s. Hence by the dominated}$

Π

convergence theorem, for conditional expectations

$$\lim_{n \to \infty} \mathbb{IE}\left[Q_{i,n}^{2}(B_{j}) \middle| F_{m}\right] = \mathbb{IE}\left[1_{B_{j}}(Z_{i}) \middle| F_{m}\right] = Q_{i,m}(B_{j}) .$$

Consequently, by changing the order of summation

$$\sum_{n=m}^{\infty} \mathbb{IE} \left[\sum_{j=1}^{\infty} \{ Q_{i,n+1}(B_{j}) - Q_{i,n}(B_{j}) \}^{2} \middle| F_{m} \right] = \sum_{j=1}^{\infty} \{ Q_{i,m}(B_{j}) - Q_{i,m}^{2}(B_{j}) \}$$

which proves the theorem.

Remark .

The quantity $\sum_{j=1}^{\infty} q_{j}^{2}(B_{j})$ is a measure of degeneration for the distribution q_{j} . In fact,

$$2\{1 - \sum_{j=1}^{\infty} q_{i}^{2}(B_{j})\}$$

is the parabolic entropy of q_i with respect to the partition $B_1, B_2, ...$ if only a finite number of sets B_k are non-empty, see [Behara and Nath (1973)]. It is easy to verify that if N is the number of non-empty sets in the partition, then $1 - \sum_{j=1}^{\infty} q_i^2(B_j) \le 1 - \frac{1}{N}$ with equality if $q_i(B_j) = \frac{1}{N}$ for the non-empty sets B_j , and $1 - \sum_{j=1}^{\infty} q_i(B_j) = 0$ if q_i is concentrated on one.

3. THE EQUIVALENT DYNAMIC PROGRAM AND OPTIMAL REWARD OPERATORS

In section 3.1 we show that the Bayesian control model is equivalent to a dynamic program with a state space that is the Cartesian product of the original state space X and the set W of all distributions on the parameter set. This property is used frequently in the remaining chapters. In section 3.2 we study a class of optimal reward operators based on stopping times, as introduced by Wessels (cf. [Van Nunen and Wessels (1977)]). Here we consider general dynamic programs and therefore the section may be of some independent interest. However in section 3.3 we return to the Bayesian control model and we specialize the results of section 3.2 for the equivalent dynamic program. Further we give some useful properties of the value function (cf. 2.12). We note that the results of section 3.2 are used in chapters 6 and 7.

3.1 Transformation into a dynamic program

Before discussing in detail the methods and results of this section, we define the dynamic program, that turns out to be equivalent to the Bayesian control model.

model 2: Equivalent dynamic program

The model is defined in terms of the objects of model 1 (cf. 2.1).

- 3.1 (a) $X \times W$ is the state space endowed with the σ -field $X \otimes W$.
 - (b) A is the action space endowed with the σ -field A.
 - (c) D is a function from X × W to the non-empty subsets of A such that $\widetilde{D}((x,q)) := D(x), x \in X, q \in W$ (the sets of *admissible actions*).
 - (d) $\stackrel{\sim}{P}$ is a transition probability from X \times W \times A to X \times W such that

$$\widetilde{P}(B \times C | \mathbf{x}, \mathbf{q}, \mathbf{a}) := \sum_{i \in I} \mathbf{1}_{K_i} (\mathbf{x}, \mathbf{a}) \int \mathcal{V}(d\mathbf{y}) \mathbf{p}_i(\mathbf{y}, \mathbf{q}) \left| P(d\mathbf{x}' | \mathbf{x}, \mathbf{a}, \mathbf{y}) \right| \\ \{\mathbf{y} \in \mathbf{y} | \mathbf{T}_{i \in \mathbf{y}}(\mathbf{q}) \in C\} = \mathbf{B}$$

 $(x \in X, q \in W, a \in A, B \in X, C \in W)$ (cf. 2.28).

(e) the reward function $\tilde{\mathbf{r}}$: X × W × A + IR is defined by

$$\widetilde{\mathbf{r}}(\mathbf{x},\mathbf{q},\mathbf{a}) := \sum_{i \in \mathbf{I}} \mathbf{i}_{\mathbf{K}_{i}}(\mathbf{x},\mathbf{a}) \int v(\mathrm{d}\mathbf{y}) \mathbf{p}_{i}(\mathbf{y},\mathbf{q}) \mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y}) .$$

As in section 1.2 we define the sets of histories, the strategies, the random variables and the probabilities on the sample space.

3.2 The set of *histories* at stage n, is defined by: (i) $\widetilde{H}_0 := X \times W, \widetilde{H}_n := (X \times W \times A)^n \times X \times W, n \in \mathbb{N}^*,$ $\widetilde{H}_{\infty} := \widetilde{\Omega} := (X \times W \times A)^{\mathbb{N}}.$

(ii) H
_n is the σ-field on H
_n induced by X, W and A, n ∈ N. Let H := H
_∞.
3.3 A strategy π is a sequence π = (π
₀, π
₁, π
₂,...), where π
_n is a transition probability from H
_n to A such that π
_n(•|x₀,q₀,a₀,...,x_n,q_n) is concentrated on D
(x_n,q_n). The set of all strategies is dentoed by Π.
3.4 On Ω we define the coordinate functions or nonder variables X
₀ = X
₀.

3.4 On
$$\Omega$$
 we define the coordinate functions or random variables x_n , Q_n and \widetilde{A}_n by $\widetilde{X}_n(\omega) := x_n$, $\widetilde{Q}_n(\omega) := q_n$, $\widetilde{A}_n(\omega) := a_n$ where
 $\omega = (x_0, q_0, a_0, x_1, q_1, a_1, \dots) \in \widetilde{\Omega}$.

3.5 For each
$$\rho \in P(X)$$
, each $q \in W$ and each $\pi \in \Pi$ there is a probability
 $\widetilde{\mathbf{P}}_{\rho,q}^{\pi}$ on $\widetilde{\mathbf{H}}$ determined by (cf. th. 1.4):
 $\widetilde{\mathbf{P}}_{\rho,q}^{\pi}[\widetilde{\mathbf{X}}_{0} \in \mathbf{B}_{0}, \widetilde{\mathbf{Q}}_{0} \in \mathbf{C}_{0}, \widetilde{\mathbf{A}}_{0} \in \mathbf{E}_{0}, \dots, \widetilde{\mathbf{X}}_{n} \in \mathbf{B}_{n}, \widetilde{\mathbf{Q}}_{n} \in \mathbf{C}_{n}, \widetilde{\mathbf{A}}_{n} \in \mathbf{E}_{n}] :=$
 $1_{\mathbf{C}_{0}}(\mathbf{q}_{0}) \int_{\mathbf{B}_{0}} \rho(\mathbf{d}\mathbf{x}_{0}) \int_{\mathbf{E}_{0}} \widetilde{\pi}_{0}(\mathbf{d}\mathbf{a}_{0} | \mathbf{x}_{0}, \mathbf{q}_{0}) \dots$
 $\dots \int_{\mathbf{B}_{0}} \widetilde{\mathbf{P}}(\mathbf{d}(\mathbf{x}_{n}, \mathbf{q}_{n}) | \mathbf{x}_{n-1}, \mathbf{q}_{n-1}, \mathbf{a}_{n-1}) \int_{\mathbf{E}_{n}} \widetilde{\pi}_{n}(\mathbf{d}\mathbf{a}_{n} | \mathbf{x}_{0}, \mathbf{q}_{0}, \mathbf{a}_{0}, \dots, \mathbf{x}_{n}, \mathbf{q}_{n})$
for $\mathbf{B}_{i} \in X$, $\mathbf{C}_{i} \in W$ and $\mathbf{E}_{i} \in \mathbf{A}$, $i \in \mathbb{N}$ and $\mathbf{q}_{0} := \mathbf{q}$.

We introduce a sequence of transformations $t_n : W \times H_n \rightarrow \widetilde{H}_n$ which relate histories for model 1 to histories for model 2.

3.6 $t_n(q,x_0,a_0,y_1,x_1,\ldots,y_n,x_n) := (x_0,q,a_0,x_1,q_1,\ldots,x_n,q_n)$

where $q_i := q_i(q, x_0, a_0, \dots, y_n)$ (cf. 2.30(ii)), for $i = 1, \dots, n$.

Hence, if $q \in W$ is the prior distribution of Z, then $t_n(q,h_n)$ is the history at time n if we only observe the states x_i , the actions a_i and the posterior distributions q_i ($i \le n, h_n \in H_n$).

Further we define the subset Π_{Ω} of Π by:

3.7 $\pi \in \Pi_0$ iff there is for each $q \in W \ge \pi \in \widetilde{\Pi}$ such that for all $h_n \in H_n$, $n \in \mathbb{I}N$:

$$\pi_n(\cdot | \mathbf{h}_n) = \widetilde{\pi}_n(\cdot | \mathbf{t}_n(\mathbf{q}, \mathbf{h}_n)) .$$

The strategy $\tilde{\pi}$ is called the *corresponding strategy* of π (with respect to q).

Notice that the strategies $\pi \in \Pi_0$ base the choice of the action at time n only on $X_0, Q_0, A_0, \dots, X_n, Q_n$.

Moreover we shall show that for each $\pi \in \Pi_0$ and its corresponding $\widetilde{\pi} \in \widetilde{\Pi}$. the following equality is valid for all $\rho \in \mathcal{P}(X)$ and $q \in W$:

 $\mathbb{E}_{\rho,q}^{\pi}[r(X_{n},A_{n},Y_{n+1})] = \mathbb{E}_{\rho,q}^{\widetilde{\pi}}[\tilde{r}(X_{n},\widetilde{Q}_{n},\widetilde{A}_{n})] \quad \text{for all } n \in \mathbb{N}.$

This implies that we may apply techniques of dynamic programming to the equivalent dynamic program in order to determine or to approximate the value function v and nearly optimal strategies.

Transformations of this type are well-known, see for example [Martin (1967)], [Wessels (1968)], [Hinderer (1970)], [Furukawa (1970)], [Yushkevich (1976)]. As far as we know the most general result is proved in [Rieder (1975)]. Translated to our situation Rieder's result implies that the Bayesian control model is equivalent to a dynamic program with state space $X \times Y \times W$, in other words that the process $\{(X_n, Y_n, Q_n, A_n), n \in \mathbb{N}\}$ is a Markov decision process. To this end Rieder transforms the Bayesian equivalent model into a so-called non-Markovian decision model, as defined in [Hinderer (1970)] and afterwards he shows, using Hinderer's concept of sufficiency, that the non-Markovian decision model is equivalent to the dynamic program with state space $X \times Y \times W$. However, we need the equivalence of the Bayesian control model to model 2, a dynamic program with state space $X \times W$. Therefore we prefer a direct proof. Our approach employs the same idea in [Strauch (1966) th. 4.1], which is also the basis of Hinderer's sufficiency concept.

We start with some preliminaries.

Note that, according to th. 1.4, we have a "natural" regular conditional distribution $\mathbb{P}_{\rho,q}^{\pi}[\cdot|Z,X_{0},A_{0},Y_{1},\ldots,Y_{n},X_{n},A_{n}], \rho \in P(X), q \in W, \pi \in \mathbb{I}$. We always choose this version without comment. For real-valued measurable functions f on Ω that are bounded form above we always define $\mathbb{E}_{\rho,q}^{\pi}[f|Z,X_{0},A_{0},Y_{1},\ldots,Y_{n},X_{n},A_{n}]$ as in th. 1.4(ii).

Note that we are working in model 1 until th. 3.5. Recall that in 2.24 we defined the mapping $Q_n : \Omega \to W$. On W we have the Borel σ -field W, generated by the weak topology, which is the smallest σ -field on W such that the maps $B \to \mu(B)$ are measurable (cf. lemma 1.5) for every $B \in T$. Since for every $B \in T$ the mapping $\omega \to Q_n(B)(\omega)$ is measurable it follows that $\mathbb{E} \begin{bmatrix} \pi \\ \rho, q \end{bmatrix} [Q_n(B) | Q_n] = Q_n(B), \mathbb{P} \begin{bmatrix} \pi \\ \rho, q \end{bmatrix} -a.s.$

Lemma 3.1

Fix $\rho \in P(X)$, $q \in W$, $\pi \in \Pi$ and $n \in \mathbb{N}$.

(i) Let f be a real-valued measurable function on $\theta \times X \times W \times A$, bounded from above. Then

$$\mathbb{E}_{\rho,q}^{\pi}[f(Z,X_n,Q_n,A_n)|X_n,Q_n,A_n] = \int f(\theta,X_n,Q_n,A_n)Q_n(d\theta) , \mathbb{P}_{\rho,q}^{\pi}-a.s.$$

(ii) Let f be a real-valued measurable function on $\theta \times (X \times W \times A)^{n+1}$, bounded above. Then, for $V := (X_0, Q_0, A_0, \dots, X_n, Q_n, A_n)$, we have

$$\mathbb{E}_{\rho,q}^{\pi}[f(\mathbb{Z},\mathbb{V}) | \mathbb{V}] = \int f(\theta,\mathbb{V}) Q_n(d\theta) , \quad \mathbb{P}_{\rho,q}^{\pi}-a.s.$$

Proof.

We start with assertion (i). All equalities hold $\mathbb{IP}_{\rho,q}^{\pi}$ -a.s. We may suppose $f(\theta, x_n, q_n, a_n) := 1_B(\theta) 1_C(x_n, q_n, a_n)$ with B ϵ T and C ϵ X \odot W \odot A

$$\mathbb{E}_{\rho,q}^{\pi}[f(Z,X_n,Q_n,A_n)|X_n,Q_n,A_n] = \mathbb{1}_{C}(X_n,Q_n,A_n)\mathbb{P}_{\rho,q}^{\pi}[Z \in B|X_n,Q_n,A_n].$$

Note that $\sigma(x_n, Q_n, A_n) \in F_n$. Hence

$$\mathbb{P}_{\rho,q}^{\pi}[z \in B | x_n, Q_n, A_n] = \mathbb{E}_{\rho,q}^{\pi}[\mathbb{P}_{\rho,q}^{\pi}[z \in B | F_n] | x_n, Q_n, A_n] =$$
$$= \mathbb{E}_{\rho,q}^{\pi}[Q_n, B | x_n, Q_n, A_n] = Q_n(B) .$$

Therefore we have

$$\mathbb{E}_{\rho,q}^{\pi} [f(z, x_n, Q_n, A_n) | x_n, Q_n, A_n] = \mathbb{1}_C (x_n, Q_n, A_n) Q_n (B) =$$
$$= \int \mathbb{1}_B (\theta) \mathbb{1}_C (x_n, Q_n, A_n) Q_n (d\theta) = \int f(\theta, x_n, Q_n, A_n) Q_n (d\theta) .$$

The $\sigma(X_n, Q_n, A_n)$ -measurability of $\int f(\theta, X_n, Q_n, A_n)Q_n(d\theta)$ follows from lemma 1.6(iii). The proof of assertion (ii) is analogous.

In lemma 3.2 we employ symbols we used before, but here we do not use their interpretation.

Lemma 3.2

Let (Ω, \mathcal{H}) , (U, X) and (V, Γ) be Borel spaces, let $X : \Omega \to U$ and $Y : \Omega \to V$ be measurable. Let \mathbb{P} be a probability on \mathcal{H} . Further let $\mathbb{P}[\cdot | X = x]$ be a regular conditional probability given X = x (cf. corollary 1.3) and let f be a real-valued measurable function on $U \times V$, that is bounded above. Define $m_{X}(C) := \mathbb{P}[Y \in C | X = x]$ for all $x \in U$ and $C \in \Gamma$. Then $\int f(X, y) m_{Y}(dy)$ is a version of $\mathbb{E}[f(X, Y) | X]$.

Proof.

First let $f(x,y) := 1_A(x)1_B(y)$ with $A \in X$, $B \in \Gamma$. By corollary 1.3 and by 1.21 we have $\mathbb{E}[1_B(Y)|X] = \int 1_B(y)m_y(dy)$, \mathbb{P} -a.s. Hence

$$\mathbb{E}\left[f(X,Y) \mid X\right] = \mathbf{1}_{A}(X) \int \mathbf{1}_{B}(Y) \mathbf{m}_{X}(dy) = \int f(X,Y) \mathbf{m}_{X}(dy) \ .$$

By standard arguments the statement can be proved in general.

In th. 3.4 we shall show that we may restrict attention to the subset of strategies $\mathbb{H}_0 \subset \mathbb{I}$, defined in 3.7. In fact we shall show more: the only interesting strategies are those, where for all $n \in \mathbb{N}$ the choice of the distribution of the action at time n depends only on the values of X_n and Q_n . Here we use the same construction as in [Strauch (1966) th. 4.1]. The idea of this construction can also be found in [Derman and Strauch (1966)], [Wessels (1968) th. 7.4 and th. 7.5] and in [Hinderer (1970) th. 18.1]. We start with a lemma where this construction is carried out.

Lemma 3.3

Fix $\rho \in P(X)$, $q \in W$ and $\pi \in \mathbb{I}$. For all $n \in \mathbb{N}$ we fix a regular conditional probability $\mathbb{P}_{\rho,q}^{\pi} [\cdot | X_n = x_n, Q_n = q_n]$, such that $\mathbb{P}_{\rho,q}^{\pi} [\mathbb{A}_n \in D(x_n) | X_n = x_n, Q_n = q_n] = 1$ for $x_n \in X$ and $q_n \in W$. We define

the strategy $\pi^* \in \Pi$ by

3.8
$$\pi_n^*(B|x_0,a_0,y_1,x_1,a_1,\ldots,y_n,x_n) := \mathbb{IP}_{\rho,q}^{\pi}[A_n \in B|x_n = x_n, Q_n = q_n]$$
,
B $\in A$, where $q_n := q_n(q_1,x_0,a_0,y_1,\ldots,y_n)$ (cf. 2.30).

Then for each real-valued measurable function f on $X \times W \times A \times X \times W$ that is bounded above, we have simultaneously for all $n \in \mathbb{N}$:

$$\mathbb{E}_{\rho,q}^{\pi}[f(X_{n},Q_{n},A_{n},X_{n+1},Q_{n+1})] = \mathbb{E}_{\rho,q}^{\pi^{*}}[f(X_{n},Q_{n},A_{n},X_{n+1},Q_{n+1})]$$

Proof.

The existence of a regular conditional probability $\mathbb{P}_{\rho}^{\pi} \begin{bmatrix} \cdot & x_n \\ \cdot & \cdot \\$

$$\pi^{\star}(\mathbf{B}|\mathbf{x}_{0}) = \mathbf{IP}_{\rho,q}^{\pi}[\mathbf{A}_{0} \in \mathbf{B}|\mathbf{X}_{0} = \mathbf{x}_{0}] = \pi(\mathbf{B}|\mathbf{x}_{0}) .$$

Hence the statement is proved for n = 0. Assume it holds for n - 1 and for all functions f satisfying the assumptions of the lemma. Define for notational convenience the function F on Ω by F := $f(X_n, Q_n, A_n, X_{n+1}, Q_{n+1})$. First we show that for all $\pi' \in \mathbb{I}$ we have $\mathbb{P}^{\pi}_{\rho,q}$ -a.s.:

(a)
$$\mathbb{E}_{\rho,q}^{\pi'} [F|X_n, Q_n, A_n] = \sum_{i \in I} i_{K_i} (X_n, A_n) \int v (dy)$$

$$\cdot \int P(dx|X_n, A_n, y) p_i(y, Q_n) f(X_n, Q_n, A_n, x, T_{i,y}(Q_n)) =: g(X_n, Q_n, A_n)$$

To prove this note that (according to th. 1.4):

$$\mathbb{E}_{\rho,q}^{\pi^{*}}[F|Z,X_{0},A_{0},Y_{1},...,Y_{n},X_{n},A_{n}] = \sum_{i \in I}^{i} K_{i}(X_{n},A_{n}) \int v(dy)$$

$$\cdot \int P(dx|X_{n},A_{n},y)P_{i}(y|Z_{i})f(X_{n},Q_{n},A_{n},x,T_{i},y(Q_{n})) =:h(Z,X_{n},Q_{n},A_{n}) .$$

$$\pi^{*}$$

Hence,
$$\mathbb{P}_{\rho,q}^{\pi'}$$
-a.s.
 $\mathbb{E}_{\rho,q}^{\pi'}[\mathbb{F}|\mathbf{x}_{n},\mathbf{Q}_{n},\mathbf{A}_{n}] = \mathbb{E}_{\rho,q}^{\pi'}[\mathbb{h}(\mathbf{Z},\mathbf{x}_{n},\mathbf{Q}_{n},\mathbf{A}_{n})|\mathbf{x}_{n},\mathbf{Q}_{n},\mathbf{A}_{n}] =$
 $= \int \mathbb{h}(\theta,\mathbf{x}_{n},\mathbf{Q}_{n},\mathbf{A}_{n})\mathbf{Q}_{n}(\mathbf{d}\theta)$,

where the last equality follows from lemma 3.1. This proves (a). Next we prove that there are versions of the conditional expectations such that

(b)
$$\operatorname{IE}_{\rho,q}^{\pi}[F|X_{n},Q_{n}] = \operatorname{IE}_{\rho,q}^{\pi}[F|X_{n},Q_{n}].$$

Note that, by (a), $\mathbb{P}_{\rho,q}^{\pi}$ -a.s.

$$\mathbb{E}_{\rho,q}^{\pi}[\mathbb{F}|\mathbf{x}_{n},\mathbf{Q}_{n}] = \mathbb{E}_{\rho,q}^{\pi}[g(\mathbf{x}_{n},\mathbf{Q}_{n},\mathbf{A}_{n})|\mathbf{x}_{n},\mathbf{Q}_{n}] .$$

Let $m_{X_n,q_n}(B) := \mathbb{IP}_{\rho,q}^{\pi} [A_n \in B | X_n = x_n, Q_n = q_n]$. Hence, by lemma 3.2, we have $\mathbb{IP}_{\rho,q}^{\pi}$ -a.s.

$$\mathbb{E}_{\rho,q}^{\pi}[\mathbf{F}|\mathbf{X}_{n},\mathbf{Q}_{n}] = \int g\{\mathbf{X}_{n},\mathbf{Q}_{n},\mathbf{a}\} \mathbf{m}_{\mathbf{X}_{n},\mathbf{Q}_{n}}(\mathbf{d}\mathbf{a}) .$$

Remember that by definition we have for all B ϵ B

$$\mathbb{P}_{\rho,q}^{\pi^{*}}[A_{n} \in B | F_{n}] = \pi_{n}^{*}(B | X_{0}, A_{0}, Y_{1}, \dots, Y_{n}, X_{n}) = \mathbb{P}_{X_{n},Q_{n}}^{(B)}$$

Hence $\mathbf{IP}_{\rho,q}^{\pi}$ -a.s., by lemma 3.2:

$$\mathbb{E}_{\rho,q}^{\pi^{*}}[F|X_{n},Q_{n}] = \int g(X_{n},Q_{n},a)m_{X_{n}}Q_{n}(da)$$

This proves that $\widetilde{g}(X_n, Q_n) := \int g(X_n, Q_n, a) m_{X_n, Q_n}(da)$ is a version of $\mathbb{E}_{\rho,q}^{\pi}[F|X_n, Q_n]$ and also of $\mathbb{E}_{\rho,q}^{\pi^*}[F|X_n, Q_n]$. We proceed with the final step. $\mathbb{E}_{\rho,q}^{\pi}[F] = \mathbb{E}_{\rho,q}^{\pi}[\mathbb{E}_{\rho,q}^{\pi}[F|X_n, Q_n]] = \mathbb{E}_{\rho,q}^{\pi}[\widetilde{g}(X_n, Q_n)] =$ $= \mathbb{E}_{\rho,q}^{\pi^*}[\widetilde{g}(X_n, Q_n)] = \mathbb{E}_{\rho,q}^{\pi^*}[\mathbb{E}_{\rho,q}^{\pi^*}[F|X_n, Q_n]] = \mathbb{E}_{\rho,q}^{\pi^*}[F]$

where the third equality follows from the induction assumption, if we define $\tilde{f}(x_{n-1}, Q_{n-1}, A_{n-1}, x_n, Q_n) := \tilde{g}(x_n, Q_n)$.

Theorem 3.4

(ĩ

Let $\rho \in \mathcal{P}(X)$, $q \in W$, $\pi \in \mathbb{I}$, $n \in \mathbb{I}$ and let π^* be as in 3.8. Then

$$\mathbb{E}_{\rho,q}^{\pi}[r(X_n,A_n,Y_{n+1})] = \mathbb{E}_{\rho,q}^{\pi}[\tilde{r}(X_n,Q_n,A_n)] = \mathbb{E}_{\rho,q}^{\pi^*}[\tilde{r}(X_n,Q_n,A_n)]$$

is defined in 3.1(e)).

Proof.

The second equality follows directly from lemma 3.3. We proceed with the proof of the first equality. According to th. 1.4 we have

$$\mathbb{E}_{\rho,q}^{\pi} [r(X_{n}, A_{n}, Y_{n+1}) | z, X_{0}, A_{0}, Y_{1}, \dots, Y_{n}, X_{n}, A_{n}] = \sum_{i \in I} 1_{K_{i}} (X_{n}, A_{n})$$

$$\int v(dy) r(X_{n}, A_{n}, y) p_{i}(y | Z_{i}) =: f(Z, X_{n}, A_{n}) .$$

By taking conditional expectations with respect to $\sigma(X_n, Q_n, A_n)$ we obtain $\mathbb{P}_{\sigma, \sigma}^{\pi}$ -a.s.

$$(*) \qquad \mathbb{E}_{\rho,q}^{\pi} [r(\mathbf{X}_{n}, \mathbf{A}_{n}, \mathbf{Y}_{n+1}) | \mathbf{X}_{n}, \mathbf{Q}_{n}, \mathbf{A}_{n}] = \mathbb{E}_{\rho,q}^{\pi} [f(\mathbf{Z}, \mathbf{X}_{n}, \mathbf{A}_{n}) | \mathbf{X}_{n}, \mathbf{Q}_{n}, \mathbf{A}_{n}] = = \int_{\mathbf{I}} f(\theta, \mathbf{X}_{n}, \mathbf{A}_{n}) \mathbf{Q}_{n} (d\theta) = \sum_{\mathbf{i} \in \mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}} (\mathbf{X}_{n}, \mathbf{A}_{n}) \int v(d\mathbf{y}) r(\mathbf{X}_{n}, \mathbf{A}_{n}, \mathbf{y}) \mathbf{p}_{\mathbf{i}} (\mathbf{y}, \mathbf{Q}_{n}) = = \widetilde{r} (\mathbf{X}_{n}, \mathbf{Q}_{n}, \mathbf{A}_{n}) .$$

Hence integration of the first and the last member of (*) with respect to $\mathbf{p}_{\alpha,\sigma}^{\pi}$ yields the desired result.

In th. 3.5 we prove the announced correspondence between the strategies of Π_0 for model 1 and the strategies of $\widetilde{\Pi}$ for model 2. Note that according to th. 1.4, we have for model 2 a "natural" regular conditional probability $\mathbb{P}_{\rho,q}^{\pi}[\cdot|\widetilde{X}_0,\widetilde{Q}_0,\widetilde{A}_0,\ldots,\widetilde{X}_n,\widetilde{Q}_n,\widetilde{A}_n]$.

Theorem 3.5

Let $\rho \in P(X)$, $q \in W$, $\pi \in \Pi_0$ and let $\widetilde{\pi} \in \widetilde{\Pi}$ be the corresponding strategy (cf. 3.7). Then, for all $n \in \mathbb{N}$ and all measurable functions $f : (X \times W \times A)^{n+1} \to \mathbb{R}$ that are bounded from above, we have

$$\mathbb{E}_{\rho,q}^{\pi}[f(X_0,Q_0,A_0,\ldots,X_n,Q_n,A_n)] = \mathbb{E}_{\rho,q}^{\pi}[f(\widetilde{X}_0,\widetilde{Q}_0,\widetilde{A}_0,\ldots,\widetilde{X}_n,\widetilde{Q}_n,\widetilde{A}_n)] .$$

Proof.

Let n = 0. In this case the statement is valid, since $\pi_0(\cdot | \mathbf{x}) = \widetilde{\pi}_0(\cdot | \mathbf{x}, q)$ for all $\mathbf{x} \in \mathbf{X}$. Assume the statement is valid for n and all admissible functions f. It is straightforward to verify that

$$\begin{split} & \mathbb{E}_{\rho,q}^{\pi} [f(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n+1}, \varrho_{n+1}, A_{n+1}) | z, x_{0}, A_{0}, x_{1}, \dots, x_{n}, x_{n}, A_{n}] = \\ & = \sum_{i \in I} \mathbf{1}_{K_{i}} (x_{n}, A_{n}) \int v(dy) \int \mathbb{P}(dx | x_{n}, A_{n}, y) \int \pi_{n+1} (da | x_{0}, A_{0}, x_{1}, \dots, x_{n}, x_{n}, A_{n}, y, x) \\ & \cdot \mathbb{P}_{i}(y | z_{1}) f(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}, x, T_{i,y}(\varrho_{n})) = \\ & = \sum_{i \in I} \mathbf{1}_{K_{i}} (x_{n}, A_{n}) \int v(dy) \int \mathbb{P}(dx | x_{n}, A_{n}, y) \\ & \cdot \int \widetilde{\pi}_{n+1} (da | x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}, x, T_{i,y}(\varrho_{n})) \\ & \cdot \mathbb{P}_{i}(y | z_{1}) f(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}, x, T_{i,y}(\varrho_{n})) = : h(z, x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}) \\ & \text{where the second equality is a consequence of 3.7. \\ & \text{By lemma 3.1(ii) we have } \mathbb{P}_{\rho,q}^{\pi} - a.s. \\ & \mathbb{E}_{\rho,q}^{\pi} [f(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n+1}, \varrho_{n+1}, A_{n+1}) | x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}] = \\ & = \sum_{i \in I} \mathbf{1}_{K_{i}} (x_{n}, A_{n}) \int v(dy) \int \mathbb{P}(dx | x_{n}, A_{n}, y) \\ & \cdot \int \widetilde{\pi}_{n+1} (da | x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}, x, T_{i,y}(\varrho_{n})) \\ & \cdot \mathbb{P}_{i} (y, \varrho_{n}) f(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}, x, T_{i,y}(\varrho_{n})) \\ & \cdot \mathbb{P}_{i} (y, Q_{n}) f(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}, x, T_{i,y}(\varrho_{n})) = : g(x_{0}, \varrho_{0}, A_{0}, \dots, x_{n}, \varrho_{n}, A_{n}) \\ & \text{For model 2 we have, according to th. 1.4: } \end{split}$$

$$\begin{split} \widetilde{\mathbf{m}}_{\rho,q}^{\pi}[f(\widetilde{\mathbf{x}}_{0},\widetilde{\mathbf{Q}}_{0},\widetilde{\mathbf{A}}_{0},\ldots,\widetilde{\mathbf{x}}_{n+1},\widetilde{\mathbf{Q}}_{n+1},\widetilde{\mathbf{A}}_{n+1})|\widetilde{\mathbf{x}}_{0},\widetilde{\mathbf{Q}}_{0},\widetilde{\mathbf{A}}_{0},\ldots,\widetilde{\mathbf{x}}_{n},\widetilde{\mathbf{Q}}_{n},\widetilde{\mathbf{A}}_{n}] &= \\ &= g(\widetilde{\mathbf{x}}_{0},\widetilde{\mathbf{Q}}_{0},\widetilde{\mathbf{A}}_{0},\ldots,\widetilde{\mathbf{x}}_{n},\widetilde{\mathbf{Q}}_{n},\widetilde{\mathbf{A}}_{n}) \end{split}$$

Hence, using the induction hypothesis we have

$$\mathbb{E}_{\rho,q}^{\pi}[f(X_{0},Q_{0},A_{0},\ldots,X_{n+1},Q_{n+1},A_{n+1})] = \mathbb{E}_{\rho,q}^{\pi}[g(X_{0},Q_{0},A_{0},\ldots,X_{n},Q_{n},A_{n})] =$$
$$=\mathbb{E}_{\rho,q}^{\pi}[g(\widetilde{X}_{0},\widetilde{Q}_{0},\widetilde{A}_{0},\ldots,\widetilde{X}_{n},\widetilde{Q}_{n},\widetilde{A}_{n})] = \mathbb{E}_{\rho,q}^{\pi}[f(\widetilde{X}_{0},\widetilde{Q}_{0},\widetilde{A}_{0},\ldots,\widetilde{X}_{n+1},\widetilde{Q}_{n+1},\widetilde{A}_{n+1})] \cdot [$$

Remark.

In fact we proved by th. 3.5 the following result. Let $q \in W$ and let $F : \Omega + \widetilde{\Omega}$ be defined by

$$F(\theta, x_0, a_0, y_1, x_1, a_1, \ldots) := (x_0, q_0, a_1, x_1, q_1, a_1, \ldots)$$

where $q_n := q_n(q, x_0, a_0, y_1, \dots, y_n)$. Then for all $B \in \tilde{H}$:

$$\mathbb{IP}_{\rho,q}^{\pi}[\{\omega \mid F(\omega) \in B\}] = \mathbb{IP}_{\rho,q}^{\pi}[B]$$

The following corollary is an immediate consequence of theorems 3.4 and 3.5. Corollary 3.6

Let $\rho \in \mathcal{P}(X)$ and $q \in W$. Then for all $\pi \in \Pi_{\rho}$ and its corresponding $\widetilde{\pi}$ (cf. 3.7)

(i) $\int \rho(dx) \mathbf{v}(x,q,\pi) = \mathbf{E} \widetilde{\mathbf{p}}_{\rho,q} \begin{bmatrix} \sum_{n=0}^{\infty} \beta^n \widetilde{\mathbf{r}}(\widetilde{\mathbf{x}}_n, \widetilde{\boldsymbol{\varrho}}_n, \widetilde{\mathbf{a}}_n) \end{bmatrix} \quad (cf. 2.12)$ and

(ii)
$$\lim_{N\to\infty} \inf \frac{1}{N} = \prod_{\rho,q}^{n-1} \sum_{n=0}^{N-1} r(X_n, A_n, Y_{n+1}) = \lim_{N\to\infty} \inf \frac{1}{N} = \prod_{\rho,q}^{n-1} \sum_{n=0}^{N-1} r(X_n, Q_n, A_n)].$$

Moreover, the supremum over all $\pi \in \Pi_0$ on the left hand side equals the supremum over all $\tilde{\pi} \in \widetilde{\Pi}$ on the right hand side, in (i) and also in (ii). (To verify this, note that for each $\tilde{\pi} \in \widetilde{\Pi}$ there is a $\pi \in \Pi_0$ such that $\tilde{\pi}$ is the corresponding strategy for π).

Remark.

In case $q \in W$ is concentrated at $\theta \in \theta$ then all posterior distributions Q_n are degenerate in θ (cf. the remarks at the end of section 2.2). Hence, in this situation the Bayesian control model is equivalent to a dynamic program with state space $x^* := \{ (x, \theta) \mid x \in X \}$. So we have shown here that observation of the supplementary state variables Y_1, Y_2, Y_3, \ldots of the system is superfluous, in case the transition law is completely known, i.e. all information needed to control the system is contained in the state variables X_0, X_1, X_2, \ldots

Since model 1 and model 2 are equivalent we shall omit the tilde in the notations for model 2 and we shall switch between these models without comment.

We conclude this section with the introduction of some terminology. In the class of strategies Π_0 we shall consider two nested subsets, that are of special interest in the remaining chapters.

- 3.9 Each measurable function $f : X \rightarrow A$ such that $f(x) \in D(x)$ for all $x \in X$, is called a *Markov policy* and the strategy $\pi \in \Pi_0$, defined by $\pi_n(\{a\}|x_0,a_0,y_1,x_1,\ldots,y_n,x_n) := 1$ if $f(x_n) = a$, $n \in \mathbb{N}$ is called a stationary strategy.
- 3.10 Each measurable function $f : X \times W \rightarrow A$ such that $f(x,q) \in D(x)$ for all $x \in X$ and $q \in W$, is called a *Bayesian Markov policy* and the strategy $\pi \in \Pi_0$ defined by $\pi_n(\{a\}|x_0,a_0,y_1,x_1,\ldots,y_n,x_n\} := 1$ if $f(x_n,q_n) = a$, $n \in \mathbb{N}$ where $q_i = q_i(q_0,x_0,a_0,y_1,\ldots,y_i)$, is called a *Bayesian stationary strategy*.

Remarks.

- (i) Each stationary strategy is also a Bayesian stationary strategy.
- (ii) It is easy to verify that, under each Bayesian stationary strategy, the process $\{(X_n, Q_n), n \in \mathbb{N}\}$ forms a Markov chain. This is well-known if we are considering model 2, however for model 1 it is a consequence of the equivalence between the two models.
- (iii) As a consequence of the equivalence between the models 1 and 2 we may apply the numerous results for dynamic programs to model 1. We only mention one of these results: if r is bounded then the supremum over all Bayesian stationary strategies of the Bayesian discounted total return equals the optimal value (cf. [Blackwell (1965)]). In other words it suffices to consider only the Bayesian stationary strategies.
- (iv) If the action space A is a finite set then any Bayesian Markov policy f such that, f(x,q) is a maximizer in the set D(x) of

$$a \rightarrow \sum_{i \in I} \mathbf{1}_{K_{i}}(\mathbf{x}, \mathbf{a}) \int \forall (d\mathbf{y}) \mathbf{p}_{i}(\mathbf{y}, q) \{ \mathbf{r}(\mathbf{x}, \mathbf{a}, \mathbf{y}) + \beta \int \mathbf{P}(d\mathbf{x}' | \mathbf{x}, \mathbf{a}, \mathbf{y}) \forall (\mathbf{x}', \mathbf{T}_{i, \mathbf{y}}(q)) \}$$

for $(x,q) \in X \times W$ is optimal (cf. [Blackwell (1965) th. 7]).

3.2 A class of optimal reward operators

In this section we study optimal reward operators for dynamic programs with complete separable metric state and action spaces. These operators are based on stopping times. They generalize the well-known optimal reward operator introduced in [Blackwell (1965)]. In [Wessels (1974)] these operators have been studied for dynamic programs with finite state and action spaces and they have been generalized for models with a countable state space and an arbitrary action space in [Van Nunen and Wessels (1977)]. Van Nunen and Wessels show that a number of well-known approximation methods for the value function in discounted dynamic programming, such as the Gauss - Seidel iteration are equivalent to successive applications of an optimal reward operator corresponding to a suitable stopping time. We prove some new results on these optimal reward operators. First we show that if such an operator is applied to a function on the state space that is upper semi-analytic and bounded from above, then the result of the operation is again an upper semi-analytic function which is bounded from above. This generalizes a rather theoretical result in [Blackwell, Freedman and Orkin (1974)] and [Shreve (1977)] for the optimal reward operators of the class we consider.

Further we show that successive applications of two of these operators, possibly for different stopping times, have the same result as one application of the optimal reward operator which belongs to the *composed stopping time* (cf. 3.14 for a definition). This property has some interesting consequences, one of which is that we can generalize results proved by Van Nunen and Wessels for discounted dynamic programs, using the fixed point theorem for contraction mappings, to more general models.

In chapter 6 we use another consequence of this property for the equivalent dynamic program (model 2). There we study the optimal reward operator, corresponding to the entrance time in a subset of the state space. Using this operator is equivalent to transforming the model into a dynamic program with this subset as a state space.

Since we are dealing with a general dynamic program here, we have to introduce some new notations. (Symbols used in this section do not have the interpretation, given in the foregoing part of the monograph).

model 3: General dynamic program.

3.11 (a) (S,S) is a Borel space, called the state space.

- (b) (A,A) is a Borel space, called the action space.
- (c) D is a function from S to the non-empty subsets of A such that $K := \{(s,a) | s \in S, a \in D(s)\}$ is an element of $S \otimes A$, and it is assumed that K contains the graph of some measurable function. from S to A.
- (d) P is a transition probability from S × A to S.
- (e) r is a real-valued measurable function on S × A, that is bounded from above. $\beta \in [0,1)$ is the discount factor.

The sets of histories, the strategies, the random variables and the probabilities on the sample space are defined analogous to model 2 (cf. 3.2-3.5). We even use the same notations, with the exception that we omit the tilde and that the coordinate functions on the state space are denoted by S_n for $n \in \mathbb{N}$.

We start with some definitions:

- 3.12 A stopping time τ is a measurable function from Ω to $\overline{\mathbf{M}}$ such that $\{\tau = n\} \in H_n$.
- 3.13 The shift operator ψ is a function from Ω to Ω such that $\psi(s_0, a_0, s_1, a_1, \ldots) := (s_1, a_1, s_2, a_2, \ldots)$ for $(s_0, a_0, s_1, a_1, \ldots) \in \Omega$. The iterates of ψ are defined by: $\psi^0(\omega) := \omega$ and $\psi^n(\omega) := \psi(\psi^{n-1}(\omega))$ for $\omega \in \Omega$ and $n \in \mathbb{N}^*$.
- 3.14 The set of all stopping times is denoted by \sum , and on \sum we define the operation: by

$$(\tau_1 \circ \tau_2)(\omega) := \tau_1(\omega) + \tau_2(\psi^{\tau_1}(\omega)) \quad \text{if } \tau_1(\omega) < \infty$$
$$:= \infty \text{ if } \tau_1(\omega) = \infty, \text{ for } \omega \in \Omega \text{ and } \tau_1, \tau_2 \in \Sigma'.$$

The function $\tau_1 \circ \tau_2$ on Ω is called the *composed stopping time*.

It is easy to verify that $\tau_1 \circ \tau_2 \in \sum$ (cf. [Revuz (1976) page 22]).

- 3.15 (i) $B_{m}(S)$ is the set of real-valued measurable functions on S, which are bounded from above.
 - (ii) $B_a(S)$ is the set of upper semi-analytic (u.s.a.) functions on S, which are bounded from above.

In appendix A we give a definition of u.s.a.functions and there we also collect some useful properties of these functions. Note that $B_m(S) \subset B_a(S)$. Finally we define for each $\tau \in \sum$ the corresponding optimal reward operator. 3.16 The optimal reward operator U_{τ} is defined for functions $b \in B_a(S)$ by:

$$(\mathbf{U}_{\tau}\mathbf{b})(\mathbf{s}) := \sup_{\pi \in \Pi} \mathbb{I}\mathbf{E}_{\mathbf{s}}^{\pi} \begin{bmatrix} \tau^{-1} \\ \sum_{n=0}^{n} \beta^{n} \mathbf{r} (\mathbf{S}_{n}, \mathbf{A}_{n}) + \beta^{\tau} \mathbf{b} (\mathbf{S}_{\tau}) \end{bmatrix}$$

(we use the convention $b(S_{\tau}) = 0$ on $\{\tau = \infty\}$ (cf. 2.23)). The usual optimal reward operator U, introduced by Blackwell, can be defined by 3.17 U := U₁ where 1 is the stopping time identically one on Ω . 56

Note that, (Ub) (s) = sup {r(s,a) +
$$\beta \int P(ds'|s,a)b(s')$$
}, for b $\in B_a(S)$.
 $a \in D(s)$
It is well-known that Ub need not to be an element of $B_m(S)$ if $b \in B_m(S)$
(cf. [Blackwell (1965)]). However in [Strauch (1966)] it is shown that the
value function is u.s.a. and in [Blackwell, Freedman and Orkin (1974)] it
has been proved that Ub $\in B_a(S)$ if $b \in B_a(S)$. In [Shreve (1977)] the same
result is obtained. We show this property for all operators U_{τ} , $\tau \in \Sigma$.
Theorem 3.7
Let $\tau \in \Sigma$ and $b \in B_a(S)$. Then $(U_{\tau}b) \in B_a(S)$.
Proof.
Define on Ω $Y := \sum_{n=0}^{\tau-1} \beta^n r(S_n A_n) + \beta^{\tau}b(S_{\tau})$. It is straightforward to verify
that $Y \le M(1 - \beta)^{-1}$, where $M := \sup r(s,a) + \sup b(s)$.
 $(s,a) \in S^{\times}A \qquad s \in S$
(We divide the proof into three parts.)
(a) We first show that Y is u.s.a. on Ω . Define for $c \in \mathbb{R}$:
 $E_c := \{\omega \in \Omega | \beta^T b(S_{\tau}) > c\}$. Let $c > 0$. Then:
 $E_c := \{\omega \in \Omega | \beta^T b(S_{\tau}) > c\beta^{-n}\}$). Note that $b(S_n)$ is u.s.a. (cf. A9).
Consequently E_c is analytic (cf. A2). Let $c \le 0$. Then:
 $E_c = \bigcup_{n \in \mathbb{N}} (\{\tau = n\} \cap \{b(S_n) > c\beta^{-n}\}) \cup \{\tau = \infty\}$. Hence in this case E_c
is also analytic. Therefore $\beta^T b(S_{\tau})$ is u.s.a. on Ω and so Y is u.s.a.
on Ω (cf. A 8).
(b) Consider the function on $P((A \times S)^{\mathbb{N}}) \times S$ defined by:
 $(m, s) + \int Y(s, \omega') P(d\omega')$, $(\omega' \in (A \times S)^{\mathbb{N}})$. We show that this function
 $is u.s.a. (P(i = n) = P(i = 1)$.

is u.s.a. (Note that $P((A \times S)^{\mathbb{N}})$ is endowed with the topology of weak convergence.) To this end we define the function \widetilde{Y} on $P((A \times S)^{\mathbb{N}}) \times \Omega$ by: $\widetilde{Y}(\mathbb{IP}, \omega) := Y(\omega)$. To show that \widetilde{Y} is u.s.a. note that, for $c \in \mathbb{R}$:

$$\{(\mathbf{IP}, \omega) | \widetilde{\mathbf{Y}}(\mathbf{IP}, \omega) > c\} = \mathcal{P}((\mathbf{A} \times \mathbf{S})^{\mathbb{IN}}) \times \{\omega \in \Omega | \mathbf{Y}(\omega) > c\}$$

Since $P((A \times S)^{\mathbb{N}})$ is a measurable set and by part (a) $\{\omega \in \Omega | \Psi(\omega) > c\}$ is analytic we have by A2 that $\widetilde{\Psi}$ is u.s.a. Further we define a transition probability p from $P((A \times S)^{\mathbb{N}}) \times S$ to $(A \times S)^{\mathbb{N}}$ by

 $p(d\omega' | \mathbb{I}P, s) := \mathbb{I}P(d\omega')$.

To verify that p is indeed a transition probability, note that

$$\{(\mathbf{IP}, \mathbf{s}) | \mathbf{p}(\mathbf{B} | \mathbf{IP}, \mathbf{s}) \le \mathbf{c}\} = \{\mathbf{IP} \in \mathcal{P}((\mathbf{A} \times \mathbf{S})^{\mathbb{IN}}) | \mathbf{IP}(\mathbf{B}) \le \mathbf{c}\} \times \mathbf{s}\}$$

for $B \in (A \otimes S)^{\mathbb{I}N}$ and $c \in \mathbb{I}R$. Hence, by lemma 1.5(i) this set is measurable. Finally we note that, by A10 the function on $P((A \times S)^{\mathbb{I}N}) \times S$:

$$(\mathbf{IP}, \mathbf{s}) \rightarrow \int \widetilde{\mathbb{Y}}(\mathbf{IP}, \mathbf{s}, \omega') \mathbf{p}(d\omega' | \mathbf{IP}, \mathbf{s}) = \int \mathbb{Y}(\mathbf{s}, \omega') \mathbf{IP}(d\omega')$$

is u.s.a.

(c) Introduce the set Δ := {(IP,s) | s ∈ S, IP ∈ P((A × S)^{IN}) such that for a π ∈ Π : IP[B] = IP^π_S[S × B] for all B ∈ (A ⊗ S)^{IN}}. It has been proved in [Hinderer (1970) lemma 13.1] that Δ is a Borel subset of P((A × S)^{IN}) × S. (Note that model 3 is an example of Hinderer's model.) Hence it is straightforward to verify that

$$(\mathbf{IP}, \mathbf{s}) \neq \mathbf{F}(\mathbf{IP}, \mathbf{s}) := \int \mathbf{Y}(\mathbf{s}, \boldsymbol{\omega}') \mathbf{IP}(\mathbf{d}\boldsymbol{\omega}') \mathbf{1}_{\Delta}(\mathbf{IP}, \mathbf{s}) - \mathbf{1}_{\Delta^{\mathbf{C}}}(\mathbf{IP}, \mathbf{s}) \cdot \boldsymbol{\omega}$$

is u.s.a. on $\mathcal{P}((\mathbf{A} \times \mathbf{S})^{\mathbb{N}}) \times \mathbf{S}$.

Finally we remark that $(U_{\tau}b)(s) = \sup F(\mathbb{I}P, s)$, where the supremum is taken over all $\mathbb{I}P \in P((\mathbb{A} \times s)^{\mathbb{I}N})$. Hence, since $\{s \in S \mid (U_{\tau}b)(s) > c\} = \operatorname{proj}_{S} \{(\mathbb{I}P, s) \mid F(\mathbb{I}P, s) > c\}$ for $c \in \mathbb{I}R$, we have by A4 : $U_{\tau}b$ is u.s.a.

It has been shown in [Blackwell (1965) example 1] that even if $b \in \mathcal{B}_{m}(S)$ it is not necessarily true that for every $\varepsilon > 0$ there is a strategy $\pi \in \mathbb{I}$ such that

$$\mathbb{E}_{s}^{\pi}[r(S_{0},A_{0}) + \beta b(S_{1})] \geq (Ub)(s) - \varepsilon \quad \text{for all } s \in S.$$

However, in [Blackwell, Freedman and Orkin (1974)] it has been shown that there always exists a "universally measurable strategy" π with this property (i.e. $\pi_n(B|\cdot)$ is universally measurable for all $B \in A$, see appendix A). Moreover in [Shreve (1977)] the same property is proved for the stopping time that is identically infinite. It should be possible to establish a similar result for arbitrary stopping times in Σ . However, we do not need such a result, as we have the following lemma.

Lemma 3.8

If $b \in \mathcal{B}_{m}(S)$, $\tau \in \Sigma$, $\rho \in \mathcal{P}(S)$ and $\varepsilon > 0$, then there is a strategy $\pi \in II$ such that

3.18
$$\mathbb{E}_{\mathbf{s}}^{\pi} \left[\sum_{n=0}^{\tau-1} \beta^{n} \mathbf{r} (\mathbf{S}_{n}, \mathbf{A}_{n}) + \beta^{\tau} \mathbf{b} (\mathbf{S}_{\tau}) \right] \geq (\mathbf{U}_{\tau} \mathbf{b}) (\mathbf{s}) - \varepsilon.$$

Proof.

This statement is a simple consequence of th. 14.1 in [Hinderer (1970)], which is a generalization of th. 8.1 in [Strauch (1966)] to non-stationary models. To verify this, we note that the function $1_{\{\tau=n\}}$ on Ω is \mathcal{H}_n -measurable. Hence by lemma 1.1 there are real-valued measurable functions f_n and ℓ_n on (S × A)ⁿ × S such that:

3.19 (i)
$$f_n(s_0, A_0, \dots, s_n) = 1_{\{\tau=n\}}$$
 on Ω .
(ii) $\ell_n(s_0, A_0, \dots, s_n) = 1$ if $\sum_{m=0}^n f_m(s_0, A_0, \dots, s_m) = 0$
= 0 otherwise,

 $s_i \in S, a_i \in A$ and $i = 0, \dots, n$.

Hence $l_n(S_0, A_0, \dots, S_n) = 1$ if and only if $\tau > n$. Further we define, for $n \in \mathbb{N}$:

$$r_n(s_0,a_0,\ldots,s_n,a_n) := \beta^n r(s_n,a_n) \ell_n(s_0,a_0,\ldots,s_n) + \beta^n b(s_n) f_n(s_0,a_0,\ldots,s_n)$$

for $s_i \in S$, $a_i \in A$ and i = 0, ..., n. It is straightforward to verify that

$$\sum_{n=0}^{\infty} r_n(S_0, A_0, \dots, S_n, A_n) = \sum_{n=0}^{\tau-1} \beta^n r(S_n, A_n) + \beta^{\tau} b(S_{\tau}) .$$

Hence we are dealing with a total-return model in the sense of Hinderer and the assertion follows from the above mentioned result of Hinderer. $\hfill \Box$

The main result of this section is th. 3.11 which states that for each pair of stopping times $\tau_1, \tau_2 \in \Sigma$ and each function $b \in \mathcal{B}_m(S)$ the following identity is valid:

$$v_{\tau_1 \circ \tau_2} = v_{\tau_1} (v_{\tau_2})$$
.

To prove this we need some preparations.

3.20 For any pair of strategies
$$\pi^{(1)}, \pi^{(2)} \in \mathbb{I}$$
 and any $\tau \in \Sigma$ we define a new strategy $\pi^{\tau} \in \mathbb{I}$ by:

$$\pi_{n}^{\tau}(\cdot | s_{0}, a_{0}, \dots, s_{n}) := \sum_{k=0}^{n} \pi_{n-k}^{(2)}(\cdot | s_{k}, a_{k}, \dots, s_{n}) f_{k}(s_{0}, a_{0}, \dots, s_{k}) + \pi_{n}^{(1)}(\cdot | s_{0}, a_{0}, \dots, s_{n}) \ell_{n}(s_{0}, a_{0}, \dots, s_{n})$$

for $s_i \in S$, $a_i \in A$ and $i \in \mathbb{N}$ (f_k and l_n are defined in 3.19). Note that π^{τ} uses $\pi^{(1)}$ until time τ and $\pi^{(2)}$ afterwards. It is easy to verify that indeed $\pi^{\tau} \in I$, since $\pi_n^{\tau}(B|\cdot)$ is $(S \otimes A)^{n-1} \otimes S$ -measurable for all $B \in A$. 3.21 Let $\tau \in \Sigma$. The σ -field H_{τ} is defined as usual by:

 $H_{\tau} := \{ B \in H \mid B \cap \{ \tau = n \} \in H_n \} .$

Lemma 3.9

Let f be a real-valued measurable function on Ω , which is bounded from above. Let $\pi^{(1)}, \pi^{(2)} \in \Pi, \tau \in \Sigma$ and let π^{τ} be defined by 3.20. Then we have on $\{\tau < \infty\}$:

$$\mathbb{IE}_{s}^{\pi^{\mathsf{T}}}[f(\psi^{\mathsf{T}})|\mathcal{H}_{\tau}] = \mathbb{IE}_{s_{\tau}}^{\pi^{(2)}}[f], \qquad \mathbb{IP}_{s}^{\pi^{(1)}}-a.s.$$

(By convention the both sides vanish on $\{\tau = \infty\}$,)

Proof.

Let $n \in \mathbb{N}$, $B_1 \in \mathcal{H}_n$ and $B_2 \in \mathcal{H}$. For the stopping time n, which is identically equal to n, it is straightforward to verify, using th. 1.4:

(*)
$$\mathbb{P}_{s}^{\pi^{n}}[\mathbb{B}_{1} \cap \{\psi^{n} \in \mathbb{B}_{2}\}] = \int_{B_{1}}^{\pi^{n}}\mathbb{P}_{s}^{\pi^{n}}[\mathbb{B}_{2}]d\mathbb{P}_{s}^{\pi^{n}}$$

Let $f := {1 \choose B_2}$ and let $B \in H_{\tau}$. Then, since $\mathbb{P}_s^{\tau}[B] = \mathbb{P}_s^{\tau}[B]$ if $B \in \{\tau = n\}$, $B \in H$ we have

$$(**) \int_{B_1} \mathbb{E}_{S_{\tau}}^{\pi^{(2)}} [f] d\mathbb{P}_{S}^{\pi^{\tau}} = \sum_{n=0}^{\infty} \int_{B_1 \cap \{\tau=n\}} \mathbb{P}_{S_n}^{\pi^{(2)}} [B_2] d\mathbb{P}_{S}^{\pi^n}, \text{ for } s \in S.$$

On the other hand, by the definition of conditional expectations (cf. 1.20(i))

$$(***) \int_{\mathbf{B}_{1}} \mathbf{E}_{\mathbf{S}}^{\pi^{\mathsf{T}}} [f(\psi^{\mathsf{T}}) | \mathcal{H}_{\tau}] d\mathbf{P}_{\mathbf{S}}^{\pi^{\mathsf{T}}} = \sum_{n=0}^{\infty} \int_{\mathbf{B}_{1} \cap \{\tau=n\}} \mathbf{1}_{\mathbf{B}_{2}} (\psi^{n}) d\mathbf{P}_{\mathbf{S}}^{\pi^{\mathsf{n}}}$$

Hence, using (*) we conclude that the left-hand sides of (**) and (***) are identical. By standard arguments the assertion is proved in general. $\hfill \Box$

Remark.

In fact we proved here the strong Markov property for a special stopping time and a special non-Markovian process.

3.22 For each $\pi \in \mathbb{I}$ and each $(s_0, a_0, \dots, s_k, a_k) \in (S \times A)^{k+1}$ we define a new strategy $\widetilde{\pi}(s_0, a_0, \dots, s_k, a_k) = (\widetilde{\pi}_0, \widetilde{\pi}_1, \dots)$ by:

$$\widetilde{\pi}_{n}(\cdot | \mathbf{h}_{n}) := \pi_{n+k+1}(\cdot | \mathbf{s}_{0}, \mathbf{a}_{0}, \dots, \mathbf{s}_{k}, \mathbf{a}_{k}, \mathbf{h}_{n}), \quad \mathbf{h}_{n} \in \mathbf{H}_{n}, \quad n \in \mathbb{N}$$

Note that $\tilde{\pi}(s_0, a_0, \dots, s_k, a_k) \in \mathbb{R}$. This strategy $\tilde{\pi}(s_0, a_0, \dots, s_k, a_k)$ acts like the strategy π if the process has a "prehistory" $s_0, a_0, \dots, s_k, a_k$.

Lemma 3.10

Let $\pi \in \Pi$ and let $\widetilde{\pi}(s_0, a_0, \dots, s_k, a_k)$ be defined in 3.22 for each $(s_0, a_0, \dots, s_k, a_k) \in (S \times A)^{k+1}$. Further let f be a real-valued measurable function on Ω , which is bounded from above, and let $\tau \in \Sigma$. Then

$$\operatorname{IE}_{\mathbf{s}}^{\pi}[f(\psi^{\mathsf{T}})|\mathcal{H}_{\tau}] = \operatorname{IE}_{\mathbf{S}_{\tau}}^{\pi}(S_{0}, A_{0}, \dots, A_{\tau-1})[f], \quad \operatorname{IP}_{\mathbf{s}}^{\pi}-\operatorname{a.s.} \text{ on } \{\tau < \infty\}.$$

(By convention both expressions vanish on $\{\tau = \infty\}$,)

Proof.

The seasy to verify (cf. th. 1.4) that $\mathbb{IE}_{S_k}^{\pi(S_0, A_0, \dots, A_{k-1})}$ [f] is \mathcal{H}_k -measurable, $k \in \mathbb{N}$. Let $B_1 \in \mathcal{H}_n$ and $B_2 \in \mathcal{H}$. Again using the 1.4 one easily verifies that

(*)
$$\mathbb{IP}_{s}^{\pi}[B_{1} \cap \{\psi^{n} \in B_{2}\}] = \int_{B_{1}} \mathbb{IP}_{s}^{\pi}(S_{0}, A_{0}, \dots, A_{n-1})[B_{2}]d\mathbb{IP}_{s}^{\pi}$$

Now we let $B_1 \in H_{\tau}$ and $f := 1_{B_2}$. Then

$$(**) \qquad \int_{B_{1}} \mathbb{E}_{S_{\tau}}^{\widetilde{\pi}(S_{0},A_{0},\ldots,A_{\tau-1})} [f] d\mathbb{P}_{S}^{\pi} = \sum_{n=0}^{\infty} \int_{B_{1} \cap \{\tau=n\}} \mathbb{E}_{S_{n}}^{\widetilde{\pi}(S_{0},A_{0},\ldots,A_{n-1})} [B_{2}] d\mathbb{P}_{S}^{\pi}$$

and on the other hand (cf. 1.20(i)):

$$(\star\star\star) \int_{\mathbf{B}_{1}} \mathbb{I} \mathbb{E}_{\mathbf{s}}^{\pi} [f(\psi^{\tau}) | \mathcal{H}_{\tau}] d\mathbb{I} \mathbb{P}_{\mathbf{s}}^{\pi} = \sum_{n=0}^{\infty} \int_{\mathbf{B}_{1} \cap \{\tau=n\}} \mathbf{1}_{\mathbf{B}_{2}}(\psi^{n}) d\mathbb{I} \mathbb{P}_{\mathbf{s}}^{\pi} =$$
$$\sum_{n=0}^{\infty} \mathbb{I}_{\mathbf{s}}^{\pi} [\mathbf{B}_{1} \cap \{\tau=n\} \cap \{\psi^{n} \in \mathbf{B}_{2}\}] .$$

From (*) we conclude that (**) and (***) are identical. Hence the assertion has been verified for indicator functions, and it can be proved in general by standard arguments.

Now we are ready to prove th. 3.11. Note that $U_{\sigma} b \in B_{a}(S)$ if $\sigma \in \Sigma$ and $b \in B_{m}(S)$.

Theorem 3.11

Let $\tau, \sigma \in \Sigma$ and let $b \in \mathcal{B}_{m}(S)$. Then we have

$$\mathbf{U}_{\tau \circ \sigma} = \mathbf{U}_{\tau} (\mathbf{U}_{\sigma} \mathbf{b})$$

Proof.

(a) Fix $\varepsilon > 0$ and let $s_0 \in S$, $\pi \in \Pi$ and $b \in \mathcal{B}_m(S)$. First we assume: $\mathbb{P}_{0}^{\pi} [\tau = \infty] < 1$. Then it is easily verified that the set-function ρ on s_0 . S, defined by

is a probability on ρ . By lemma 3.8 there exists a strategy $\pi^* \in \Pi$ such that

(*)
$$\mathbb{IE}_{s}^{\pi} \begin{bmatrix} \sum_{n=0}^{\sigma-1} \beta^{n} r(S_{n}, A_{n}) + \beta^{\sigma} b(S_{\sigma}) \end{bmatrix} \geq (U_{\sigma} b)(s) - \varepsilon, \rho \text{-a.s. on S.}$$

Define: $\pi^{(1)} := \pi, \pi^{(2)} := \pi^{*}$. Let π^{T} be defined by 3.20.
Then we have

$$Y(s,\pi) := \mathbb{I}\!\!E_{s}^{\pi} \begin{bmatrix} \tau \circ \sigma - 1 \\ \sum_{n=0}^{\infty} \beta^{n} r(s_{n}, A_{n}) + \beta^{\tau \circ \sigma} b(s_{\tau \circ \sigma}) \end{bmatrix} =$$

61

$$= \mathbb{I}\!\!E_{\mathbf{s}}^{\pi} \begin{bmatrix} \sum_{n=0}^{\tau-1} \beta^{n} \mathbf{r} (\mathbf{s}_{n}, \mathbf{A}_{n}) + \mathbb{I}\!\!E_{\mathbf{s}}^{\pi} \begin{bmatrix} \sum_{n=\tau}^{\tau\circ\sigma-1} \beta^{n} \mathbf{r} (\mathbf{s}_{n}, \mathbf{A}_{n}) + \beta^{\tau\circ\sigma} \mathbf{b} (\mathbf{s}_{\tau\circ\sigma}) \Big| \mathcal{H}_{\tau} \end{bmatrix} \end{bmatrix}$$

Note that $S_{\tau \circ \sigma} = S_{\tau + \sigma(\psi^{\tau})} = S_{\sigma(\psi^{\tau})}(\psi^{\tau})$ on $\{\tau < \infty, \sigma < \infty\}$.

Using lemma 3.9 we find

$$(\star\star) \quad \Upsilon(\mathbf{s},\pi) = \operatorname{IE}_{\mathbf{s}}^{\pi} \begin{bmatrix} \sum_{n=0}^{\tau-1} \beta^{n} r(\mathbf{s}_{n},\mathbf{A}_{n}) + \beta^{\tau} \\ \operatorname{IE}_{\mathbf{s}}^{\pi} \begin{bmatrix} \sum_{r=0}^{\tau-1} \beta^{n} r(\mathbf{s}_{n},\mathbf{A}_{n}) + \beta^{\sigma} b(\mathbf{s}_{\sigma}) \end{bmatrix} \end{bmatrix} .$$

Note that $\mathbf{P}_{\mathbf{s}}^{\tau}[\mathbf{B}] = \mathbf{P}_{\mathbf{s}}^{\tau}[\mathbf{B}]$ for $\mathbf{B} \in \mathcal{H}_{\tau}$, $\mathbf{s} \in \mathbf{S}$. (To verify this note that $\mathbf{P}_{\mathbf{s}}^{\pi^{\mathsf{T}}}[\mathbf{B}] = \sum_{n=0}^{\infty} \mathbf{P}_{\mathbf{s}}^{\pi^{\mathsf{T}}}[\mathbf{B} \cap \{\tau = n\}] = \sum_{n=0}^{\infty} \mathbf{P}_{\mathbf{s}}^{\pi}[\mathbf{B} \cap \{\tau = n\}] = \mathbf{P}_{\mathbf{s}}^{\pi}[\mathbf{B}]$ since $\mathbf{B} \cap \{\tau = n\} \in \mathcal{H}_{\mathbf{n}}$). Therefore we also have $\mathbf{E}_{\mathbf{s}}^{\pi^{\mathsf{T}}}[\mathbf{f}] = \mathbf{E}_{\mathbf{s}}^{\pi}[\mathbf{f}]$ for a realvalued \mathcal{H}_{τ} -measurable function f, which is bounded from above. Using (*), (**) and the definition of ρ we find:

$$\Upsilon(\mathbf{s}_{0},\pi) \geq \operatorname{IE}_{\mathbf{s}_{0}}^{\pi} \left[\sum_{n=0}^{\tau-1} \beta^{n} r(\mathbf{s}_{n},\mathbf{A}_{n}) + \beta^{\tau}(\mathbf{U}_{\sigma}\mathbf{b})(\mathbf{s}_{\tau})\right] - \varepsilon \operatorname{IE}_{\mathbf{s}_{0}}^{\pi} \left[\beta^{\tau}\right].$$

Now we assume $\mathbb{IP}_{s_{0}}^{\pi} [\tau = \infty] = 1$. Then

$$Y(s_{0},\pi) = \mathbb{IE}_{s_{0}}^{\pi} [\sum_{n=0}^{\infty} \beta^{n} r(s_{n},A_{n})] = \mathbb{IE}_{s_{0}}^{\pi} [\sum_{n=0}^{\tau-1} \beta^{n} r(s_{n},A_{n}) + \beta^{\tau}(U_{0}b)(s_{\tau})].$$

Hence

Consider for $s \in S$:

$$\sup_{\pi \in \Pi} Y(s_0,\pi) \geq (U_{\tau}(U_{\sigma}b))(s_0) - \varepsilon$$

and since $s_0 \in S$ and ε are arbitrary we conclude:

(b) We show that (***) is valid with \leq instead of \geq . Let $\pi \in \Pi$ and let $\widetilde{\pi}(s_0, a_0, \dots, s_k, a_k)$ be defined by 3.22 for $(s_0, a_0, \dots, s_k, a_k) \in (S \times A)^{k+1}$. Note that $S_n = S_{n-\tau}(\psi^{\tau})$ on $\{\tau \leq n\}$.

$$\mathbb{E}_{s}^{\pi} \left[\sum_{n=0}^{\tau \circ \sigma - 1} \beta^{n} r(s_{n}, A_{n}) + \beta^{\tau \circ \sigma} b(s_{\tau \circ \sigma}) \right] =$$

$$= \mathbb{E}_{s}^{\pi} \left[\sum_{n=0}^{\tau - 1} \beta^{n} r(s_{n}, A_{n}) + \mathbb{E}_{s}^{\pi} \left[\sum_{n=\tau}^{\tau \circ \sigma - 1} \beta^{n} r(s_{n}, A_{n}) + \beta^{\tau \circ \sigma} b(s_{\tau \circ \sigma}) \right] H_{\tau} \right] =$$

$$= \mathbb{E}_{s}^{\pi} \begin{bmatrix} \tau^{-1} \\ \sum \\ n=0 \end{bmatrix}^{n} r(S_{n}, A_{n}) + \beta^{\tau} \mathbb{E}_{S_{\tau}} \begin{bmatrix} \sum \\ n=0 \end{bmatrix}^{n} r(S_{n}, A_{n}) + \beta^{\sigma} b(S_{\sigma}) \end{bmatrix} \leq \\ \leq \mathbb{E}_{s}^{\pi} \begin{bmatrix} \tau^{-1} \\ \sum \\ n=0 \end{bmatrix}^{n} r(S_{n}, A_{n}) + \beta^{\tau} (U_{\sigma} b)(S_{\tau}) \end{bmatrix} \leq (U_{\tau} (U_{\sigma} b))(s) .$$

The second equality is justified by lemma 3.10 and the inequality by the definition of U_{a} . This proves the theorem.

To be able to apply th. 3.11 we need the following lemma.

Lemma 3.12

The operation \circ on Σ is associative.

Proof.

Let $\tau, \sigma, \rho \in \Sigma$. By 3.14 we have: $\tau \circ \sigma = \tau + \sigma(\psi^{T})$ and so

$$(\tau \circ \sigma) \circ \rho = \tau + \sigma(\psi^{\tau}) + \rho(\psi^{\tau+\sigma}(\psi^{\tau}))$$

On the other hand

$$\tau \circ (\sigma \circ \rho) = \tau + (\sigma \circ \rho)(\psi^{\tau}) = \tau + \sigma(\psi^{\tau}) + \rho(\psi^{\sigma(\psi^{\tau})}(\psi^{\tau})) .$$

Since $\psi^{k}(\psi^{\ell}) = \psi^{k+\ell}$ we have $\psi^{\tau+\sigma(\psi^{\tau})} = \psi^{\sigma(\psi^{\tau})}(\psi^{\tau})$ which proves the lemma.

Hence $\tau_1 \circ \tau_2 \circ \ldots \circ \tau_n$ is defined now. As a consequence of th. 3.11 and lemma 3.12 we find:

Corollary 3.13 Let b $\epsilon \ B_{m}(s)$ and let $\tau_{1}, \tau_{2}, \dots, \tau_{n} \ \epsilon \ \Sigma$. Then

$$\mathbf{U}_{\tau_1 \circ \tau_2 \circ \cdots \circ \tau_n} \mathbf{b} = \mathbf{U}_{\tau_1} (\mathbf{U}_{\tau_2} (\dots \mathbf{U}_{\tau_n} \mathbf{b}) \dots) \cdot$$

To prove this, note that $\tau_1 \circ \tau_2 \circ \ldots \circ \tau_n = \tau_1 \circ (\tau_2 \circ \ldots \circ \tau_n)$. Hence $U_{\tau_1 \circ \tau_2} \circ \ldots \circ \tau_n = U_{\tau_1} (U_{\tau_2} \cdots \tau_n)$, and the assertion follows by iteration.

We continue with two definitions:

3.23 Let
$$\tau \in \Sigma$$
. The stopping time τ^n , $n \in \mathbb{N}^*$ is defined by:
 $\tau' := \tau, \tau^n := \tau^{n-1} \circ \tau, n = 2,3,...$

3.24 Let $\tau \in \Sigma$. Then U_{τ}^{n} is defined on $\mathcal{B}_{a}(S)$ by: $U_{\tau}^{*}b := U_{\tau}b$ for all $b \in \mathcal{B}_{a}(S)$ and $U_{\tau}^{n}b := U_{\tau}(U_{\tau}^{n-1}b)$ for all $b \in \mathcal{B}_{a}(S)$ and n = 2, 3, ...

In th. 3.14 we collect some consequences of th. 3.11.

Theorem 3.14

Let $b \in \mathcal{B}_{m}(S)$, $\tau \in \Sigma$. Then:

- (i) $U_n b = U_n^n b$ (cf. 3.23).
- (ii) The value function v for model 3 is u.s.a. and satisfies the optimality equation:

$$U_v = v$$
, on S.

(iii) If the reward function r and b are bounded and if $\tau(\omega) \ge 1$ for all $\omega \in \Omega$ then $\lim_{\tau \to \infty} U_{\tau}^{n} b = v$, on S.

Proof.

Note that (i) is an immediate consequence of 3.23, 3.24 and corollary 3.13. We proceed with (ii). Note that $v(s) = (U_{\infty}v)(s)$, $s \in S$ by definition, where (∞ represents the stopping time that is ∞ with probability one). Hence by th. 3.7 v is u.s.a. By th. 3.11we find, for all $b \in B_m(S)$

$$\mathbf{v} = \mathbf{U}_{\mathbf{v}\infty}\mathbf{b} = \mathbf{U}_{\mathbf{v}}(\mathbf{U}_{\infty}\mathbf{b}) = \mathbf{U}_{\mathbf{v}}\mathbf{v}$$

Finally we prove (iii). First let $0 \le r \le M$ and $0 \le b \le M$ for $M \in \mathbb{R}$. Note that $\tau^n \ge n$ on Ω . Then we have, for $s \in S$:

(*)
$$(U_1^n 0)(s) = (U_n 0)(s) \le (U_n b)(s) \le (U_n 0)(s) + \beta^n M \le v(s) + \beta^n M.$$

It is well-known that $\lim_{n\to\infty} (U_1^n 0)(s) = v(s)$ (see e.g. [Hinderer (1970), th. 14.5]). Hence we have, by (*):

$$\lim_{n\to\infty} ({\tt U}^n_{\tau}{\tt b})({\tt s}) = {\tt v}({\tt s}), \quad {\tt s} \in {\tt S} \ .$$

Further let $-M \leq r \leq M$ and $-M \leq b \leq M$ for $M \in \mathbb{R}$, and define $\tilde{r} := r + M$ and $\tilde{b} := b + M$. Let \tilde{U}_{τ} be the operator for the model with reward function \tilde{r} instead of r, and let \tilde{v} be the value function in this case. Hence we have

$$(U_{\tau}n^{b})(s) + M \frac{1-\beta^{n+1}}{1-\beta} \leq (\widetilde{U}_{\tau}n^{\widetilde{b}})(s) \leq (U_{\tau}n^{b})(s) + \frac{M}{1-\beta}$$

and we have also $\tilde{v}(s) = v(s) + \frac{n}{1-\beta}$. Hence we find:

$$\lim_{n \to \infty} (\mathbf{U}_{n} \mathbf{b})(\mathbf{s}) = \lim_{n \to \infty} (\widetilde{\mathbf{U}}_{n} \widetilde{\mathbf{b}})(\mathbf{s}) - \frac{\mathbf{M}}{1-\beta} = \widetilde{\mathbf{v}}(\mathbf{s}) - \frac{\mathbf{M}}{1-\beta} = \mathbf{v}(\mathbf{s}), \ \mathbf{s} \in \mathbf{S} . \square$$

We conclude this section with some remarks.

Remarks.

- (i) The theorems th. 3.11 and th. 3.14 can easily be generalized to models with weaker conditions on the reward functions. In fact th. 3.14(iii) is valid for models where a strong convergence condition is satisfied (cf. [Van Hee, Hordijk and Van der Wal (1977)]).
- (ii) If there is a (nonempty) subset $\mathcal{B}(S)$ of $\mathcal{B}_{m}(S)$ such that for all $\tau \in \Sigma$ $U_{\tau}b \in \mathcal{B}(S)$ if $b \in \mathcal{B}(S)$ then $\{U_{\tau}, \tau \in \Sigma\}$ is a *semi-group* of operators on $\mathcal{B}(S)$. If S is countable then the set of all real-valued functions on S that are bounded from above will do for $\mathcal{B}(S)$. In section 3.3 we show that there is such a set $\mathcal{B}(S)$ for the equivalent dynamic program (model 2).
- (iii) There need not be, for each $b \in \mathcal{B}_{a}(S)$, $\rho \in \mathcal{P}(S)$ and $\varepsilon > 0$ a strategy $\pi \in \mathbb{I}$ such that:

(*)
$$\rho\left(\{\mathbf{s} \in \mathbf{S} \mid \mathbf{E}_{\mathbf{s}}^{\pi} \begin{bmatrix} \tau^{-1} \\ \sum \\ n=0 \end{bmatrix}^{n} r\left(\mathbf{S}_{n}, \mathbf{A}_{n}\right) + \beta^{T} b\left(\mathbf{S}_{\tau}\right) \end{bmatrix} \ge \left(\mathbf{U}_{\tau} b\right) \left(\mathbf{s}\right) - \varepsilon\} = 1$$

However, if $b = U_{\sigma}\tilde{b}$ for some $\tilde{b} \in \mathcal{B}_{m}(S)$ and $\sigma \in \Sigma$ then for each $\rho \in \mathcal{P}(S)$ and each $\varepsilon > 0$ there is a $\pi \in \Pi$ such that (*) holds. To verify this, note that by th. 3.11 $U_{\tau}b = U_{\tau \circ \sigma}\tilde{b}$. Hence by lemma 3.9 we have the desired property.

(iv) Th. 3.14(ii) and (iii) are also proved in [Van Nunen and Wessels (1976)] by use of the fixed point theorem for contraction mappings, for dynamic programs with countable state space.

3.3 Miscellaneous results for the Bayesian control model

In this section we first study the optimal reward operators for the Bayesian control model (model 1). We show that these operators applied to functions that are lower semi-continuous (l.s.c.) in the second coordinate, i.e. l.s.c.

on W, yield functions that are again l.s.c. in the second coordinate. In the rest of this section we consider the value function (cf. 2.12) in more detail. We show that the value function v is convex in the second coordinate. Finally, we consider another consequence of the convexity of v, namely an upperbound on the value function. We note that the second part of this section is independent of the first part.

(Remember that the symbols used in section 3.2 have a local meaning only.)

3.25 The set of stopping times Σ for the Bayesian control model consists of all measurable functions τ from Ω to N such that

$$\{\tau = n\} \in \sigma(X_0, A_0, Y_1, X_1, \dots, Y_n, X_n), n \in \mathbb{N}$$
.

- 3.26 $B_{\ell}(X \times W)$ is the set of real-valued bounded measurable functions on $X \times W$, which are lower semi-continuous (l.s.c.) on W (cf. appendix A).
- 3.27 For each $\tau \in \Sigma$ we define the optimal reward operator U_{τ} on $\mathcal{B}_{\ell}(X \times W)$ by (cf. 3.16)

$$(\mathbf{U}_{\tau}\mathbf{b})(\mathbf{x},\mathbf{q}) := \sup_{\pi \in \Pi_{\Omega}} \mathbb{E}_{\mathbf{x},\mathbf{q}}^{\pi} \left[\sum_{n=0}^{\tau-1} \beta^{n} \mathbf{r}(\mathbf{X}_{n},\mathbf{A}_{n},\mathbf{Y}_{n+1}) + \beta^{\tau} \mathbf{b}(\mathbf{X}_{\tau},\mathbf{Q}_{\tau}) \right]$$

where $(x,q) \in \mathbf{X} \times \mathbf{W}$ and $\mathbf{b} \in \mathcal{B}_{g}(\mathbf{X} \times \mathbf{W})$.

Note that, if $\tau \in \Sigma$ satisfies the property:

$$\{\tau = n\} \in \sigma(X_0, A_0, X_1, A_1, \dots, X_n), n \in \mathbb{N}$$

then there is a stopping time $\tilde{\tau}$ for the equivalent dynamic program (model 2), such that for all $x_i \in X$, $y_i \in Y$, $a_i \in A$, $q_i \in W$ and $i \in N$:

$$\tilde{\tau}(x_0, q_0, a_0, x_1, q_1, a_1, \dots) = \tau(x_0, a_0, y_1, x_1, a_1, \dots)$$

Then the optimal reward operator $\widetilde{U}_{\widetilde{\tau}}$ for the equivalent dynamic program, defined by 3.16, is *equivalent* to U_{τ} in the following sense:

$$(U_{\tau}b)(x,q) = (\widetilde{U}_{\tau}b)(x,q)$$
 for all $x \in X$, $q \in W$ and $b \in \mathcal{B}_{\varrho}(X \times W)$.

Theorem 3.15

Let r be bounded and let $\theta_i \rightarrow p_i(y | \theta_i)$ be bounded and continuous for all $i \in I$ and $y \in Y$. Further let $\tau \in \Sigma$. Then $b \in B_{\ell}(X \times W)$ implies $U_{\tau} b \in B_{\ell}(X \times W)$.

66
Proof.

(a) Fix the prior distribution $q \in W$, and let Q_n be the posterior distribution distribution at stage n. Then we have: $r(X_n, A_n, Y_{n+1}) \mathbf{1}_{\{\tau > n\}} + b(X_n, Q_n) \mathbf{1}_{\{\tau = n\}}$ is measurable with respect to the σ -field $\sigma(X_0, A_0, Y_1, \dots, A_n, Y_{n+1})$. Hence there are measurable functions $F_n: W \times (X \times A \times Y)^{n+1} \neq \mathbb{R}$ such that on Ω :

$$F_{n}(q, X_{0}, A_{0}, Y_{1}, \dots, X_{n}, A_{n}, Y_{n+1}) = r(X_{n}, A_{n}, Y_{n+1}) \mathbf{1}_{\{\tau > n\}} + b(X_{n}, Q_{n}) \mathbf{1}_{\{\tau = n\}}$$

(cf. lemma 1.1). Hence we have

$$\sum_{n=0}^{\tau-1} \beta^{n} r(x_{n}, A_{n}, Y_{n+1}) + \beta^{\tau} b(x_{\tau}, Q_{\tau}) = \sum_{n=0}^{\infty} \beta^{n} F_{n}(q, X_{0}, A_{0}, Y_{1}, \dots, X_{n}, A_{n}, Y_{n+1}).$$

(b) Further we show that F_n is l.s.c. in the first coordinate. To this end we first prove that Q_n is a continuous function of q in the sence of weak convergence. Let the sequence $\{q_k, k \in \mathbb{N}\} \subset \mathbb{W}$ converge weakly to $q \in \mathbb{W}$ (cf. 1.22) (notation: $q_k \stackrel{\mathbb{W}}{\to} q$). Fix $(x_0, a_0, y_1, \dots, y_n, x_n) \in X \times A \times$ $(Y \times X \times A)^{n-1} \times Y \times X$ and let:

$$g_n(q) := q_n(q, x_0, a_0, y_1, \dots, y_n)$$
 (cf. 2.30(i)).

We have to show that $g_n(q_k) \stackrel{\Psi}{\to} g_n(q)$. Let f be a bounded and continuous function on θ . Notice that, by 2.30:

$$\int f(\theta) g_{n}(q_{k}) (d\theta) = \int f(\theta) \frac{n-1}{n} \sum_{j=0}^{n-1} \sum_{i \in I} \mathbf{1}_{K_{i}}(x_{j}, a_{j}) p_{i}(y_{j+1} | \theta_{i}) q_{k}(d\theta)$$
$$\cdot \{\Delta_{n}(q_{k}, x_{0}, a_{0}, y_{1}, \dots, y_{n})\}^{-1}$$

where

$$\Delta_{\mathbf{n}}(\mathbf{q}_{\mathbf{k}},\mathbf{x}_{0},\mathbf{a}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n}) := \int_{\mathbf{j}=0}^{\mathbf{n}-1} \sum_{\mathbf{i}\in\mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x}_{\mathbf{j}},\mathbf{a}_{\mathbf{j}})\mathbf{p}_{\mathbf{i}}(\mathbf{y}_{\mathbf{j}+1}|\theta_{\mathbf{i}})\mathbf{q}_{\mathbf{k}}(\mathbf{d}\theta).$$

Hence, since

$$\theta \rightarrow \prod_{j=0}^{n-1} \sum_{i \in I} (x_j, a_j) p_i(y_{j+1} | \theta_i)$$

is bounded and continuous, we have $\int f(\theta)g_n(q_k)(d\theta)$ tends to $\int f(\theta)g_n(q)(d\theta)$ if k tends to infinity, provided that $\Delta_n(q,x_0,a_0,y_1,\ldots,y_n) \ge 0$. Hence $q \ge b(x_n,g_n(q))$ is l.s.c. (cf. A 15) and so $q \ge F_n(q,x_0,a_0,y_1,\ldots,x_n,a_n,y_{n+1})$ is l.s.c. if $\Delta_n(q,x_0,a_0,y_1,\ldots,y_n) \ge 0$. Consequently:

$$q \neq F_n(q, x_0, a_0, y_1, \dots, x_n, a_n, y_{n+1})^{\Delta}_{n+1}(q, x_0, a_0, y_1, \dots, y_{n+1})$$

is l.s.c. since $\Delta_n(q, x_0, a_0, y_1, \dots, y_n) = 0$ implies $\Delta_{n+1}(q, x_0, a_0, y_1, \dots, y_{n+1}) = 0.$ (c) Next we show that

$$q \neq \mathbf{E}_{\mathbf{x},q}^{\pi} [\mathbf{F}_{n}(q, \mathbf{X}_{0}, \mathbf{A}_{0}, \mathbf{Y}_{1}, \dots, \mathbf{Y}_{n+1})]$$
 is l.s.c.

Note that

$$\begin{aligned} (\star) \qquad \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\mathbf{F}_{n}(\mathbf{q},\mathbf{x}_{0},\mathbf{A}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1})] &= \\ &= \int \pi_{0}(\mathrm{d}\mathbf{a}_{0} | \mathbf{x}_{0}) \int \vee (\mathrm{d}\mathbf{y}_{1}) \int \mathbf{P}(\mathrm{d}\mathbf{x}_{1} | \mathbf{x}_{0},\mathbf{a}_{0},\mathbf{y}_{1}) \ldots \int \mathbf{P}(\mathrm{d}\mathbf{x}_{n} | \mathbf{x}_{n-1},\mathbf{a}_{n-1},\mathbf{y}_{n}) \\ &\cdot \int \pi_{n}(\mathrm{d}\mathbf{a}_{n} | \mathbf{x}_{0},\mathbf{a}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n},\mathbf{x}_{n}) \int \vee (\mathrm{d}\mathbf{y}_{n+1}) \mathbf{F}_{n}(\mathbf{q},\mathbf{x}_{0},\mathbf{a}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1}) \\ &\cdot \quad \Delta_{n+1}(\mathbf{q},\mathbf{x}_{0},\mathbf{a}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1}) \,. \end{aligned}$$

Since F_n is bounded, it follows from Fatou's lemma, applied in (*) that

$$(**) \qquad \lim_{k \to \infty} \inf \mathbb{E}_{\mathbf{x},\mathbf{q}_{k}}^{\pi} [\mathbb{F}_{n}(\mathbf{q}_{k},\mathbf{x}_{0},\mathbf{A}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1})] \geq \mathbb{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\mathbb{F}_{n}(\mathbf{q},\mathbf{x}_{0},\mathbf{A}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1})]$$

(d) Finally we consider $q \neq (U_{\tau}b)(x,q)$. Let the sequence $\{q_k, k \in \mathbb{N}\}$ converge weakly to $q \in W$. Again by Fatou's lemma we have:

$$\underset{k \to \infty}{\lim \inf} \sum_{n=0}^{\infty} \beta^{n} \mathbf{E}_{\mathbf{x},\mathbf{q}_{k}}^{\pi} [\mathbf{F}_{n}(\mathbf{q}_{k},\mathbf{x}_{0},\mathbf{A}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1})] \geq$$

$$\geq \sum_{n=0}^{\infty} \beta^{n} \underset{k \to \infty}{\lim \inf} \mathbf{E}_{\mathbf{x},\mathbf{q}_{k}}^{\pi} [\mathbf{F}_{n}(\mathbf{q}_{k},\mathbf{x}_{0},\mathbf{A}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1})] \geq$$

$$\geq \sum_{n=0}^{\infty} \beta^{n} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\mathbf{F}_{n}(\mathbf{q},\mathbf{x}_{0},\mathbf{A}_{0},\mathbf{y}_{1},\ldots,\mathbf{y}_{n+1})] .$$

Note that

$$(\mathbf{U}_{\tau}\mathbf{b})(\mathbf{x},\mathbf{q}) = \sup_{\pi \in \Pi_{0}} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \left[\sum_{n=0}^{\infty} \beta^{n} \mathbf{F}_{n}(\mathbf{q},\mathbf{X}_{0},\mathbf{A}_{0},\mathbf{Y}_{1},\ldots,\mathbf{Y}_{n+1}) \right].$$

Hence, by A16, we have $q \neq (U_{\tau}b)(x,q)$ is l.s.c.

68

It is an immediate consequence of th. 3.15 that the function

 $q \neq v(x,q)$

is l.s.c. if r is bounded.

We conclude this section with two convexity properties of the value function.

Theorem 3.16

The value function of the Bayesian control model (cf. 2.12) is convex on W, i.e.

3.28
$$v(x,\lambda q_1 + (1 - \lambda)q_2) \le \lambda v(x,q_1) + (1 - \lambda)v(x,q_2), x \in X, q_1,q_2 \in W$$

and $\lambda \in (0,1) \cdot ((\lambda q_1 + (1 - \lambda)q_2)(B) := \lambda q_1(B) + (1 - \lambda)q_2(B)$ for $B \in T_{\bullet}$) Further v satisfies the inequality:

3.29
$$v(x,q) \leq \int q(d\theta)v(x,\theta) \text{ for } x \in X, q \in W$$
.

Proof.

Fix $\lambda \in (0,1)$ and $q_1, q_2 \in W$. Then we have for all $x \in X$ (cf. 2.12):

$$\begin{aligned} \mathbf{v}(\mathbf{x},\lambda\mathbf{q}_{1}+(1-\lambda)\mathbf{q}_{2}) &= \sup_{\pi\in\Pi} \int \left\{\lambda\mathbf{q}_{1}+(1-\lambda)\mathbf{q}_{2}\right\} (\mathrm{d}\theta)\mathbf{v}(\mathbf{x},\theta,\pi) \leq \\ &\leq \lambda \sup_{\pi\in\Pi} \int \mathbf{q}_{1} (\mathrm{d}\theta)\mathbf{v}(\mathbf{x},\theta,\pi) + (1-\lambda)\sup_{\pi\in\Pi} \int \mathbf{q}_{2} (\mathrm{d}\theta)\mathbf{v}(\mathbf{x},\theta,\pi) = \\ &= \lambda\mathbf{v}(\mathbf{x},\mathbf{q}_{1}) + (1-\lambda)\mathbf{v}(\mathbf{x},\mathbf{q}_{2}) \end{aligned}$$

We proceed with 3.29. Let $q \in W$ and $x \in X$. Then

$$\mathbf{v}(\mathbf{x},\mathbf{q}) = \sup_{\pi \in \Pi} \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta,\pi) \leq \int \mathbf{q}(\mathrm{d}\theta) \sup_{\pi \in \Pi} \mathbf{v}(\mathbf{x},\theta,\pi) = \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) \cdot \prod_{\pi \in \Pi} \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) = \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) \cdot \prod_{\pi \in \Pi} \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) = \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) \cdot \prod_{\pi \in \Pi} \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) = \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta) \cdot \prod_{\pi \in \Pi} \mathbf{v}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) = \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) = \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathrm{d}\theta) \mathbf{v}(\mathrm{$$

Remark.

The inequality 3.29 is a direct consequence of the convexity of $q \rightarrow v(x,q)$ in case this function is continuous. Namely, if F is a continuous and convex function on W which is bounded from

above then the following inequality can be proved:

$$F(q) \leq \int F(\theta)q(d\theta)$$

(remember that we have embedded $\boldsymbol{\theta}$ in W).

4. BAYESIAN EQUIVALENT RULES AND THE AVERAGE-RETURN CRITERION

In section 4.1 we consider procedures to construct good strategies. Special attention is given to strategies that are generated by so-called Bayesian equivalent rules. In section 4.2 we consider the average-return criterion and we give sufficient conditions for the existence of optimal strategies, based on such rules. In chapter 5 these rules will be considered in connection with the total-return criterion.

4.1 Bayesian equivalent rules and other approaches

We first consider the total discounted return criterion for models with finite sets X, Y and A. If the parameter value θ is known the usual technique to determine an optimal strategy is solving the optimality equation $v(x,\theta) = (Uv)(x,\theta)$ for all $x \in X$. Then each Markov policy for which the maximum in this equation is attained, is optimal (cf. [Blackwell (1965) th. 7]). It seldom happens that an analytic solution of the optimality equation can be found and that the value function $x \Rightarrow v(x,\theta)$, $x \in X$ is found in an explicit form. In chapter 1 we noted that even if θ is a finite set, the equivalent dynamic program (model 2) has an uncountable state space X × W and for each starting state $(x,q) \in X \times W$ there is a countable subset of $X \times W$ that can be reached in the long run. Hence it is even impossible to determine the value function v for all $(x,q) \in X \times W$. However there are rather complicated algorithms to determine v(x,q) for any fixed pair $(x,q) \in X \times W$ (cf. chapters 6 and 7).

Hence it is possible to determine in each state $(x,q) \in X \times W$ the action f(x,q), corresponding to an optimal Bayesian Markov policy f in the following way. First determine $v(x',T_{i,y}(q))$ for all $x' \in X$, $y \in Y$ and all $i \in I$ for which there is an $a \in D(x)$ with $(x,a) \in K_i$. Then f(x,q) is maximizing the function

4.1
$$a \neq \sum_{i \in I} \mathbf{1}_{K_i}(\mathbf{x}, a) \int v(dy) \mathbf{p}_i(\mathbf{y}, q) \{ \mathbf{r}(\mathbf{x}, a, y) + \beta \int P(d\mathbf{x}' | \mathbf{x}, a, y) \mathbf{v}(\mathbf{x}', \mathbf{T}_{i, y}(q)) \}$$

on the set D(x), $(x,q) \in X \times W$ (cf. the remarks at the end of section 3.1). Since this is in general a very complicated procedure, it would be preferable to have a simple recipe to determine in each state $(x,q) \in X \times W$ an action that corresponds to a good, not necessarily optimal, strategy. For example, in practice one often uses the following recipe: 4.2 At each stage estimate the unknown parameter θ using the available data, by $\hat{\theta}$. Then compute an (nearly) optimal strategy for the model where the parameter is known and equal to $\hat{\theta}$. Then use the action corresponding to this strategy in the actual state. Repeat this procedure at the next stage.

The computation of an optimal strategy for a fixed parameter value is carried out much faster than the determination of $v(x',T_{i,y}(q))$ for $x' \in X$, $y \in Y$ and the relevant $i \in I$. Hence the recipe 4.2 is simpler than the procedure given by 4.1. However, the strategy specified in 4.2 is not optimal in general. Under some conditions it is optimal when we are dealing with the average-return criterion (cf. section 4.2).

We consider the following method to construct simple recipes. Now we consider the average-return criterion too. If $\theta \in \theta$ is known, an action corresponding to an (nearly) optimal strategy is often found by maximizing some real-valued function F on X × θ × A, over all available actions a \in D(x). For example, if A is finite and r is bounded then we may define F by

4.3 a
$$F(x, \theta, a) :=$$

$$\sum_{i \in I} \mathbf{1}_{K_{i}}(\mathbf{x}, \mathbf{a}) \int v(d\mathbf{y}) \mathbf{p}_{i}(\mathbf{y} | \boldsymbol{\theta}_{i}) \{ \mathbf{r}(\mathbf{x}, \mathbf{a}, \mathbf{y}) + \beta \int \mathbf{P}(d\mathbf{x}' | \mathbf{x}, \mathbf{a}, \mathbf{y}) v(\mathbf{x}', \boldsymbol{\theta}) \}$$

in case we work with the total discounted return criterion (cf. 4.1). Sometimes there exist bounded measurable functions h and g such that

4.3 b
$$h(x,\theta) + g(\theta) =$$

$$\max_{\mathbf{a}\in D(\mathbf{x})} \sum_{i\in I}^{1} \sum_{\mathbf{x}_{i}}^{(\mathbf{x},\mathbf{a})} \int v(d\mathbf{y}) \mathbf{p}_{i}(\mathbf{y}|\boldsymbol{\theta}_{i}) \{r(\mathbf{x},\mathbf{a},\mathbf{y}) + \int P(d\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y})h(\mathbf{x}',\boldsymbol{\theta})\}$$

(cf.section 4.2).

Then each strategy that chooses in state x a maximizing action in the equation 4.3 b is optimal with respect to the average-return criterion (cf. section 4.2). Hence in this situation we have

 $F(x,\theta,a) :=$

$$\sum_{i \in I} \mathbf{1}_{K_{i}}(\mathbf{x}, \mathbf{a}) \int v(d\mathbf{y}) \mathbf{p}_{i}(\mathbf{y} \mid \boldsymbol{\theta}_{i}) \{ \mathbf{r}(\mathbf{x}, \mathbf{a}, \mathbf{y}) + \int P(d\mathbf{x}' \mid \mathbf{x}, \mathbf{a}, \mathbf{y}) \mathbf{h}(\mathbf{x}', \boldsymbol{\theta}) \} .$$

We now assume that such a function F is known. In the two examples above F can be computed by standard methods, if X,Y,A and θ are finite sets (cf. chapter 7). Using this function F we construct a Bayesian Markov policy f such that for some $\varepsilon > 0$:

4.4
$$\int q(d\theta)F(x,\theta,f(x,q)) \geq \sup_{a\in D(x)} \int q(d\theta)F(x,\theta,a) - \varepsilon,$$

for all $(x,q) \in X \times W$.

We call such a Bayesian Markov policy a *Bayesian equivalent rule* since we are maximizing the "Bayesian equivalent" of the function we have to maximize in case the parameter is known. Note that we may choose $\varepsilon = 0$ if there is a maximizer of $a \neq \int q(d\theta)F(x,\theta,a)$ for all $(x,q) \in X \times W$ (cf. the informal definition in section 1.2).

If $q \in W$ is degenerate, then the Bayesian equivalent rule is (nearly) optimal. But Bayesian equivalent rules are not optimal in general. However in section 4.2 we give sufficient conditions for optimality in case we are considering the average-return criterion, and in chapter 5 we consider examples of the Bayesian control model where a Bayesian equivalent rule is optimal for the total-return criterion.

Consider again the model with finite action space A and bounded reward function with respect to the total discounted return criterion. Then we may define a Bayesian equivalent rule using the function F defined in 4.3 a. This rule has the following interpretation. Consider a modified model where the decision-maker is told the true parameter value after one transition. It is easy to verify that this rule would be optimal in that situation. In th. 6.4 (chapter 6) we give a lower bound on the Bayesian discounted total return of this strategy. In th. 6.3 we consider another simple recipe to construct a good strategy for the total discounted return criterion. We conclude this section with an overview of procedures suggested by other authors for the average-return criterion. In [Mandl (1974), (1976)] the strategy described in 4.2 is studied. Mandl used minimum contrast estimators and in his model the parameter structure ensures the consistency of these estimators, under each strategy. Mandl considers the following average-return criterion: a strategy π is optimal if for all $\theta \in \theta$ and $x \in X$

$$\liminf_{N\to\infty} \mathbb{E}_{x,\theta}^{\pi} \left[\frac{1}{N} \sum_{n=0}^{N-1} r(x_n, A_n, Y_{n+1}) \right]$$

is maximal. Note that this criterion is stronger than ours and not depending on the choice of a prior distribution. Mandl shows that the strategy described in 4.2 is optimal in the model where X is a finite set, A a compact subset of a Euclidean space and where for each stationary strategy (cf. 3.9) the resulting Markov chain is irreducible. We show a similar result in section 4.2 for a Bayesian equivalent rule. In [Fox and Rolph (1973)] an optimal strategy is constructed for Markov renewal programs where also the recipe 4.2 is used. However, in their situation they have to ensure the consistency of the estimators for the unknown parameter. This problem is solved by socalled forced choice actions. These actions do not necessarily agree with the recipe of 4.2, but they are performed to get information. Fox and Rolph also use the stronger optimality criterion discussed above. In [Rose (1975)] another strategy is proposed. Rose assumes that for each parameter value an optimal Markov policy is known. At each stage an action is selected by randomizing over the actions belonging to some Markov policy that is optimal for some parameter value, according to the current posterior distribution. Rose also needs forced choice actions to ensure degeneration of the posterior distributions.

4.1 Optimal strategies for the average-return criterion

In this section we construct optimal strategies for the average-return criterion. Bayesian equivalent rules play an important role. We first consider an example showing that, even in case of finite state and action spaces, there need not be an optimal strategy.

Example 4.1

Consider the following model: $X = \{1,2,3,4,5,6\}$, $A = D(1) = D(2) = \{1,2,3\}$ $D(x) = \{1\}$ for $x \in \{3,4,5,6\}$. The transition probabilities $p(x^*|x,a)$ from x to x' if action a is chosen are:

p(3|3,1) = p(4|4,1) = p(5|5,1) = p(5|6,1) = p(1|1,1) = p(2|2,1) = 0 p(4|3,1) = p(3|4,1) = p(6|6,1) = p(6|5,1) = p(2|1,1) = p(1|2,1) = 1 - 0p(3|1,2) = p(5|1,3) = p(3|2,2) = p(5|2,3) = 1



Only in the states 3,4,5 and 6 a reward is obtained: r(3) = r(5) = 7 and r(4) = r(6) = 3. Let $\theta = (0,1)$. It is easy to transform this example into the framework of the Bayesian decision model.

The average return in the subchain $\{3,4\}$ is: $\frac{1}{2}(7+3) = 5$ and in the subchain $\{5,6\}$: $7\theta + 3(1 - \theta) = 4\theta + 3$. Consider a starting state $x \in \{1,2\}$. For fixed $\theta \in \theta$ the optimal action is a maximizer of $5\delta(2,a) + \{4\theta + 3\}\delta(3,a)$, $a \in \{2,3\}$. Hence, the corresponding Bayesian equivalent rule for the distribution q on (0,1) is the maximizer of $5\delta(2,a) + \{4\beta\thetaq(d\theta) + 3\}\delta(3,a)$, $a \in \{2,3\}$. It is easy to verify that if we have to choose one of the actions 2 or 3 and if q is the prior distribution, then this Bayesian equivalent rule is the best one. Let π^n be the strategy that chooses action 1 the first n times and in states 1 and 2 the maximizer of $5\delta(2,a) + \{4\int \theta Q_n(d\theta) + 3\}\delta(3,a)$, $a \in \{2,3\}$ thereafter, where Q_n is the posterior distribution at time n, if the system starts in state 1 with prior $q \in W$. Then the Bayesian average return in states 1 and 2 is:

$$\operatorname{IE}_{q}[\max \{5, 4 \int Q_{n}(d\theta) + 3\}]$$

(note that this expression does not depend on the starting state and the strategy). Note that:

$$\mathbb{E}_{q}[\max\{5, 4 \mid \theta \ Q_{n+1}(d\theta) + 3\} \mid Q_{1}, \dots, Q_{n}] \geq$$

$$\geq \max\{5,4 \mathbb{E}_{q} \left[\int_{Q_{n+1}} \left(d\theta \right) \left| Q_{1}, \dots, Q_{n} \right] + 3 \right\} = \max\{5,4 \int_{Q_{n}} \left(d\theta \right) + 3 \right\}$$

with equality if and only if $5 \ge 4 \int \theta \ Q_{n+1}(d\theta) + 3$, $\mathbb{P}_q^{-a.s.}$ However if q gives positive mass to the set $\{\theta \in \theta | \theta > \frac{1}{2}\}$ then equality never holds. Hence in this case the strategy π^n is worse than π^{n+1} and consequently there is no optimal strategy.

In this section we need the following assumptions:

Assumptions

- 4.5 (i) r is bounded on $X \times A \times Y$. (ii) D(x) = A for all $x \in X$.
- 4.6 There are bounded measurable functions g and h on θ and X \times θ respectively such that

 $h(x,\theta) + g(\theta) = \sup_{a \in A} L(x,\theta,a)$

where

$$\begin{split} L(x,\theta,a) &:= \sum_{i \in I} {}^{1}K_{i}(x,a) \int v(dy) p_{i}(y|\theta_{i}) \{r(x,a,y) + \\ &+ \int P(dx'|x,a,y)h(x',\theta) \} \text{ for } x \in X, a \in A \text{ and } \theta \in \theta \end{split}$$

4.7 For all $\varepsilon > 0$ there is a Bayesian Markov policy f such that

$$q(d\theta)L(x,\theta,f(x,q)) \geq \sup_{a \in A} \int q(d\theta)L(x,\theta,a) - \varepsilon \quad \text{for } (x,q) \in X \times W$$

Note that assumption 4.6 is identical to 4.3b. The assumption 4.5(ii) is not essential, but it makes things more transparant. Further it seems possible to weaken assumption 4.5(i). The only serious assumption is 4.6. For models with known parameter value θ and finite action space A, assumption 4.6 guarantees the existence of a stationary optimal strategy. This has been proved for finite X in [Derman (1966)] and for arbitrary X in [Ross (1968)].

In fact the strategy that chooses the maximizer of $a \rightarrow L(x,\theta,a)$ at each stage is optimal with average return $g(\theta)$. In [Ross (1968)] several situations are given where, for fixed parameter value θ a solution of 4.6 exists. For instance, if X and A are finite and for each stationary strategy the process is an irreducible Markov chain, then 4.6 is valid. The results of Derman and Ross follow from th. 4.1 below. Assumption 4.7 is a regularity condition to guarantee a measurable selection. In lemmas 4.4 and 4.5 we give some sufficient conditions guaranteeing 4.7. Note that a Bayesian Markov policy satisfying 4.7 is a Bayesian equivalent rule.

In th. 4.1 we derive a sufficient condition for a strategy to be optimal. In the rest of this section we consider model assumptions which guarantee this condition for strategies generated by the Bayesian equivalent rules of the form 4.7.

Remember that the functions g and h are easy to compute by standard methods (cf. [Derman (1970)]) if 4.6 holds and X and A are finite sets.

First we introduce some notations:

4.8 (i)
$$\phi(\mathbf{x},\theta,\mathbf{a}) := \mathbf{L}(\mathbf{x},\theta,\mathbf{a}) - \mathbf{h}(\mathbf{x},\theta) - \mathbf{g}(\theta)$$
, $\mathbf{x} \in \mathbf{X}$, $\theta \in \theta$ and $\mathbf{a} \in \mathbf{A}$.
(ii) $\phi(\mathbf{x},\mathbf{q},\mathbf{a}) := \int \mathbf{q}(\mathbf{d}\theta)\phi(\mathbf{x},\theta,\mathbf{a})$, $\mathbf{x} \in \mathbf{X}$, $\mathbf{q} \in \mathbf{W}$ and $\mathbf{a} \in \mathbf{A}$.
4.9 (i) $\mathbf{h}(\mathbf{x},\mathbf{q}) := \int \mathbf{q}(\mathbf{d}\theta)\mathbf{h}(\mathbf{x},\theta)$, $\mathbf{x} \in \mathbf{X}$ and $\mathbf{q} \in \mathbf{W}$.

The definitions 4.8(i) and (ii) are consistent, since we embedded
$$\theta$$
 in W, similarly the definitions of h and g in 4.9 are consistent.

Theorem 4.1

Assume 4.5, 4.6 and the existence of a strategy $\pi^* \in \mathbb{I}_{\cap}$ such that

4.10
$$\lim_{N \to \infty} \inf \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{E}_{x,q}^{\pi^{*}} [\phi(x_{n}, Q_{n}, A_{n})] = 0.$$

(ii) $g(q) := \left[q(d\theta)g(\theta) , q \in W \right]$

Then:

$$\sup_{\pi \in \Pi_{\Omega}} \liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{I} \mathbb{E}_{x,q}^{\pi} [r(x_n, A_n, Y_{n+1})] = g(q)$$

and π^* is optimal.

2

Proof.

First note that

$$\int q(d\theta) \{ \sum_{i \in I} \mathbf{I}_{K_{i}}(\mathbf{x}, \mathbf{a}) \mathbf{p}_{i}(\mathbf{y}|\theta_{i}) \mathbf{h}(\mathbf{x}', \theta) \} = \sum_{i \in I} \mathbf{I}_{K_{i}}(\mathbf{x}, \mathbf{a}) \mathbf{p}_{i}(\mathbf{y}, \mathbf{q}) \mathbf{h}(\mathbf{x}', \mathbf{T}_{i}, \mathbf{y}(\mathbf{q})) =$$

 $= \sum_{i \in T} \mathbf{1}_{K_{i}} (x, a) p_{i}(y, q) h(x', \sum_{i \in T} \mathbf{1}_{K_{i}} (x, a) T_{i, y}(q))$

since by 4.9(i) and 2.28

$$h(x',T_{i,y}(q)) = \{ \int q(d\theta) p_i(y|\theta_i) h(x',\theta) \} \{ p_i(y,q) \}^{-1} , \text{ if } p_i(y,q) > 0.$$

Hence, by definitions 4.8(ii) and 3.1(e) we have for all $\pi \in \Pi_{o}$: $\phi(\mathbf{X}_n, \mathbf{Q}_n, \mathbf{A}_n) = \widetilde{r}(\mathbf{X}_n, \mathbf{Q}_n, \mathbf{A}_n) + \mathbb{E}_{\mathbf{X}, \mathbf{Q}}^{\pi} [h(\mathbf{X}_{n+1}, \mathbf{Q}_{n+1}) | \mathbf{X}_n, \mathbf{Q}_n, \mathbf{A}_n] - h(\mathbf{X}_n, \mathbf{Q}_n) - g(\mathbf{Q}_n).$ Therefore, by first conditioning on $\sigma(X_n, Q_n, A_n)$, we have for all $N \in \mathbb{N}^*$ $\sum_{n=0}^{N-1} \mathbb{E}_{x,q}^{\pi} [\tilde{r}(x_n, Q_n, A_n) + h(x_{n+1}, Q_{n+1}) - h(x_n, Q_n) - g(Q_n) - \phi(x_n, Q_n, A_n)] = 0$ Since $g(Q_n) = \int Q_n(d\theta)g(\theta)$, we have (cf. th. 2.1) $\mathbb{E}_{q}^{\pi}[g(Q_{p})] = g(q) .$

Using the boundedness of h we find for $\pi \in \Pi_0$:

$$\liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{E}_{x,q}^{\pi} [\tilde{r}(x_n, Q_n, A_n)] = g(q) + \liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{E}_{x,q}^{\pi} [\phi(x_n, Q_n, A_n)].$$

Note that $\phi(x, \theta, a) \leq 0$ by 4.6. Hence, for all $\pi \in \mathbb{I}_{0}^{-}$, g(q) is an upperbound for the Bayesian average return. On the other hand, if 4.10 holds, then g(q)is the optimal value and π^* is optimal. Remarks .

(i) In [Mandl (1974) th. 3] a similar result has been obtained.In fact, Mandl's result, formulated in our terminology, reads:

$$\lim_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} r(X_n, A_n, Y_{n+1}) = g(\theta) , \quad \mathbb{P}_{x, \theta}^{\pi} - a.s.$$

if and only if

$$\lim_{N\to\infty}\frac{1}{N}\sum_{n=0}^{N-1}\phi(X_n,\theta,A_n)=0, \qquad \mathbb{IP}_{X,\theta}^{\pi}-a.s.$$

(note that one limit exists iff the other exists).

(ii) We conclude from th. 4.1 that if there is a ϵ > 0 such that for all $\pi \ \epsilon \ \Pi_0$

$$\liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{IE}_{x,q}^{\pi} [\phi(x_n, Q_n, A_n)] \geq \varepsilon ,$$

then the optimal Bayesian average return is at most equal to g(q) - ϵ , in state $(x,q)~\epsilon~X~\times$ W.

(iii) According to corollary 3.6 we may replace Π_0 by Π in th. 4.1. We need the following obvious lemma.

Lemma 4.2

Let $\{\varepsilon_n, n \in \mathbb{N}\}$ be a sequence of bounded real numbers such that $\lim_{n \to \infty} \varepsilon_n = 0$ then: $\lim_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \varepsilon_n = 0$.

The following corollary to th. 4.1 includes the already mentioned results of Derman and Ross.

Corollary 4.3

Let $\{\varepsilon_n, n \in \mathbb{N}\}$ be a non-increasing sequence of positive numbers such that $\lim_{n \to \infty} \varepsilon_n = 0$, and let f_n be a Markov policy such that for fixed $\theta \in \theta$: $n \to \infty$

$$L(x,\theta,f_n(x)) \ge \sup L(x,\theta,a) - \varepsilon_n, n \in \mathbb{N}$$
.
 $a \in A$

Then the strategy π^* that uses Markov policy f_n at stage n, n ϵ N, is optimal for this parameter value θ .

Proof.

Note that, if we start with a prior distribution which is degenerate at θ , then Q_n is degenerate in θ for $n \in \mathbb{N}$. Hence, using π^* we find: $\phi(X_n, Q_n, A_n) = L(X_n, \theta, f_n(X_n)) - h(X_n, \theta) - g(\theta) \leq \varepsilon_n$ for $n \in \mathbb{N}$. Therefore, using lemma 4.2

$$\liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{E}_{x,\theta}^{\pi} [\phi(x_n, Q_n, A_n)] \leq \liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \varepsilon_n = 0.$$

Now th. 4.1 applies.

We continue with some conditions guaranteeing 4.7. Note that if X is countable then 4.7 is fulfilled.

Lemma 4.4

If A is countable then 4.7 is valid.

Proof.

Fix $\varepsilon > 0$. Let a_1, a_2, \ldots be an enumeration of A and define

$$B_{1} := \bigcap_{a \in A} \{ (x,q) \in X \times W | \int q(d\theta) L(x,\theta,a_{1}) \geq \int q(d\theta) L(x,\theta,a) - \varepsilon \}$$

and, for k = 2, 3, ...

$$B_{k} := \bigcap \{ (x,q) \in X \times W | (x,q) \in \bigcup_{i=1}^{k-1} \int q(d\theta) L(x,\theta,a_{k}) \geq \int q(d\theta) L(x,\theta,a) - \varepsilon \}.$$

Note that B_k is measurable, for $k \in \mathbb{N}^*$ and $B_k \cap B_\ell = \emptyset$ if $k \neq \ell$. Further note that for each (x,q) there is at least one $k \in \mathbb{N}^*$ such that $(x,q) \in B_k$. Hence the function $f : X \times W \Rightarrow A$ defined by $f(x,q) := a_k$ iff $(x,q) \in B_k$ is a Bayesian Markov policy satisfying 4.7.

Lemma 4.5

Let the following assumptions hold:

80

4.11 (i) A is compact.

- (ii) $a \neq P(\cdot | x, a, y)$ is a continuous mapping from A to P(X), where P(X) is endowed with the weak topology, for all $x \in X$ and $y \in Y$.
- (iii) $a \rightarrow r(x,a,y)$ is continuous for all $x \in X$ and $y \in Y$.
- (iv) $x + h(x,\theta)$ is continuous for all $\theta \in \theta$.
- (v) $a \rightarrow 1_{K_{a}}$ (x,a) is continuous for all $x \in X$ and $i \in I$.

Then there is a Bayesian Markov policy f such that

$$\int q(d\theta)L(x,\theta,f(x,q)) = \sup_{a \in A} \int q(d\theta)L(x,\theta,a) , \text{ for } (x,q) \in X \times W .$$

Proof.

Since h and r are bounded, we have $(x,\theta,a) \rightarrow L(x,\theta,a)$ is bounded. And, since this mapping is measurable, we have $(x,q,a) \rightarrow \int q(d\theta)L(x,\theta,a)$ is bounded and measurable.

Further, since $a \neq \int v(dy) p_i(y|\theta_i) r(x,a,y)$ and $a \neq \int P(dx^*|x,a,y) h(x^*,\theta)$ are continuous, we have $a \neq \int q(d\theta) L(x,\theta,a)$ is continuous. Hence all conditions for Schäl's, selection theorem (cf. A17) are satisfied, which proves the lemma.

Remark.

The condition 4.11(v) is fulfilled in the following situation:

4.12 A := $\bigcup_{k=1}^{n} N_{k}$, where N_{k} is compact with $N_{k} \cap N_{\ell} = \emptyset$ if $k \neq \ell$, X := $\bigcup_{k=1}^{m} M_{k}$, where M_{k} is measurable and $M_{k} \cap M_{\ell} = \emptyset$ if $k \neq \ell$, and $K_{(i,j)} := M_{i} \times N_{j}$ and I := {(i,j) | i = 1,...,m, j = 1,...,n}.

If A is finite then 4.12 is valid, and therefore 4.11(i), (ii) and (iii). Another example of 4.11(v) is the situation where A is compact and $K_i = M_i \times A_i$, $i \in \mathbb{N}^+$ where M_1, M_2, \ldots is a measurable partition of X.

Theorem 4.6

Assume 4.5, 4.6 and 4.7. Let $\{\varepsilon_n, n \in \mathbb{N}\}$ be a nonincreasing sequence of positive numbers such that $\lim_{n \to \infty} \varepsilon_n = 0$. Let f_n be a Bayesian Markov policy for $n \in \mathbb{N}$, such that for $(x,q) \in X \times W$:

4.13
$$\int q(d\theta) L(x,\theta,f_n(x,q)) \geq \sup_{a \in A} \int q(d\theta) L(x,\theta,a) - \varepsilon_n \quad (cf. 4.7).$$

Further let π^* be the strategy that uses f_n at stage n, n ϵ N. Assume for fixed $(x_0,q_0) \in X \times W$:

4.14
$$\mathbb{IP}_{X_0, q_0}^{\pi^*} [\cap \cap_{\star} \{\tau(i, n) < \infty\}] = 1$$
.

Finally assume the existence of a finite set F of Markov policies such that $\theta \Rightarrow \inf \phi(x, \theta, f(x))$ is measurable for $f \in F$ and $x \in X$

4.15 max inf
$$\phi(x,\theta,f(x)) = 0$$
, for all $\theta \in \Theta$.
feF xeX

Then condition 4.10 holds and therefore π^* is optimal with Bayesian average return $g(q_0)$, in (x_0,q_0) (cf. 2.13).

Proof.

Note that $\pi^* \in \Pi_0$. Under the strategy π^* we have $A_n = f_n(X_n, Q_n)$ and therefore, by 4.6 and 4.8:

$$0 \ge \phi(X_n, Q_n, A_n) \ge \sup_{a \in A} \int Q_n(d\theta) \phi(X_n, \theta, a) - \varepsilon_n \ge$$
$$\ge \max_{f \in F} \int Q_n(d\theta) \phi(X_n, \theta, f(X_n)) - \varepsilon_n \ge \max_{f \in F} \int Q_n(d\theta) \inf_{x \in X} \phi(x, \theta, f(x)) - \varepsilon_n .$$

Using 4.14, the boundedness of ϕ , 4.15 and corollary 2.5 we find

$$\lim_{n\to\infty} \int Q_n (d\theta) \inf_{\mathbf{x}\in\mathbf{X}} \phi(\mathbf{x},\theta,f(\mathbf{x})) = \inf_{\mathbf{x}\in\mathbf{X}} \phi(\mathbf{x},\mathbf{Z},f(\mathbf{x})) = 0 , \quad \mathbb{IP}_{\mathbf{x}_0,\mathbf{q}_0}^{\pi^*} -a.s.$$

And since F is a finite set we have

$$\begin{split} \lim_{n \to \infty} \max_{f \in F} & \int \mathcal{Q}_n(d\theta) \inf_{x \in X} \phi(x, \theta, f(x)) = 0 , \ \mathbf{P}_{x_0, q_0}^{\pi^*} \text{-a.s.} \\ \text{Since } \lim_{n \to \infty} \varepsilon_n = 0 \text{ we finally have } \lim_{n \to \infty} \phi(x_n, \mathcal{Q}_n, \mathcal{A}_n) = 0 , \ \mathbf{P}_{x_0, q_0}^{\pi^*} \text{-a.s. And} \\ \text{therefore, by the boundedness of } \phi: \end{split}$$

82

$$\lim_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \mathbb{IE}_{x_0, q_0}^{\pi^*} [\phi(x_n, Q_n, A_n)] = 0.$$

This proves the theorem.

Although, at first glance the number of assumptions in th. 4.6 is overwhelming, only 4.14 and 4.15 are serious restrictions on the applicability of the theorem. In 4.14 it is required that the strategy π^* guarantees that we obtain enough "information" concerning the "true" parameter. If there is a finite collection of Markov policies, which contains an optimal one for all models with known parameter value, then 4.15 is fulfilled. In th. 4.8 we consider more appealing conditions guaranteeing all requirements of th. 4.6. We start with a lemma, the truth of which is intuitively clear.

Lemma 4.7

Let X and A be finite sets and assume that, for all $\theta \in \Theta$ and each stationary strategy, the resulting Markov chain $\{X_n, n \in \mathbb{N}\}$ is irreducible. Then, for all $\theta \in \Theta$, $x \in X$ and $\pi \in \Pi$, the number of visits to each state $x' \in X$ is infinite, $\mathbb{P}_{x,\theta}^{\pi}$ -a.s.

Proof.

Fix $\theta \in \Theta$ and x' $\in X$.

a) We first prove there is at least one visit, P^π_{x,θ}-a.s. To show this, transform the transition law in such a way that x' becomes absorbing:
 i.e. P({x'}|x',a,y) := 1 for all a ∈ A, y ∈ Y. Further consider the reward function:

$$r(\mathbf{x},\mathbf{a}) := \sum_{i \in I} \mathbf{1}_{K_{i}} (\mathbf{x},\mathbf{a}) \int v(d\mathbf{y}) P(\{\mathbf{x}'\} | \mathbf{x},\mathbf{a},\mathbf{y}) \mathbf{p}_{i}(\mathbf{y} | \theta) , \text{ if } \mathbf{x} \neq \mathbf{x}'$$
$$:= 0 \text{ if } \mathbf{x} = \mathbf{x}'$$

In this model the total expected return is defined as usual:

$$\mathbf{v}(\mathbf{x},\theta,\pi) := \sum_{n=0}^{\infty} \mathbf{I} \mathbf{E}_{\mathbf{x},\theta}^{\pi} [r(\mathbf{X}_{n},\mathbf{A}_{n})], \qquad \pi \in \mathbb{I} \text{ and } \mathbf{x} \in \mathbf{X}.$$

According to th. 3.4 there is a strategy $\pi^* \in \Pi_0$ such that $v(x,\theta,\pi) = v(x,\theta,\pi^*)$. However since we start with a degenerate prior distribution, all posterior distributions are degenerate. Hence Π_0 is the

Ω

set of strategies such that the action at time n only depends on X_0, A_0, \ldots, X_n (cf. the remark following corollary 3.6). Now we restrict ourselves to strategies in Π_0 . Note that $\mathbb{P}_{\mathbf{x},\theta}^{\pi}$ -a.s.

$$r(\mathbf{X}_n, \mathbf{A}_n) = \mathbb{P}_{\mathbf{X}, \theta}^{\pi} [\mathbf{X}_{n+1} = \mathbf{x}' | \mathbf{X}_n, \mathbf{A}_n] , \quad \text{if } \mathbf{X}_n \neq \mathbf{x}' .$$

Hence, for $x \neq x'$, $\pi \in \Pi_0$:

$$\mathbf{v}(\mathbf{x},\theta,\pi) = \sum_{n=0}^{\infty} \mathbf{E}_{\mathbf{x},\theta}^{\pi} [\mathbf{P}_{\mathbf{x},\theta}^{\pi} [\mathbf{X}_{n+1} = \mathbf{x}' | \mathbf{X}_{n}, \mathbf{A}_{n}]_{\mathbf{X} \setminus \{\mathbf{x}'\}}^{T} (\mathbf{X}_{n})] =$$
$$= \sum_{n=0}^{\infty} \mathbf{P}_{\mathbf{x},\theta}^{\pi} [\mathbf{X}_{n+1} = \mathbf{x}', \mathbf{X}_{n} \neq \mathbf{x}'] = \mathbf{P}_{\mathbf{x},\theta}^{\pi} [\mathbf{X}_{n} = \mathbf{x}' \text{ for some n}].$$

Since this last probability does not depend on the transition law in x', the probability $\mathbb{P}_{\mathbf{X},\theta}^{\pi}[\mathbf{X}_{\mathbf{n}} = \mathbf{x}'$ for some n] is not affected by the transformation of the model. We return to the transformed model. Minimizing $\mathbf{v}(\mathbf{x},\theta,\pi)$ over all $\pi \in \Pi_0$ is a *negative dynamic programming* problem with state space X and action space A. Therefore we have, by [Strauch (1966) th. 9.1] for all $\pi \in \Pi_0 \quad \mathbf{v}(\mathbf{x},\theta,\pi) \geq \min \quad \mathbf{v}(\mathbf{x},\theta,\pi)$, where the minimum has to be taken over all stationary strategies $\overline{\pi}$ in Π_0 . Since by assumption the Markov chain is irreducible under each stationary strategy, this minimum equals one, if $\mathbf{x} \neq \mathbf{x}'$. Therefore $\mathbf{v}(\mathbf{x},\theta,\pi) = 1$ for all $\pi \in \Pi$ and $\mathbf{x} \neq \mathbf{x}'$. So we have in the original model for $\mathbf{x} \neq \mathbf{x}'$: (*) $\mathbb{P}_{\mathbf{x},\theta}^{\pi}[\mathbf{X}_{\mathbf{n}} = \mathbf{x}'$ for some $\mathbf{n} > 0] = 1$ for all $\pi \in \Pi$. It is easy to verify that (*) is also valid for $\mathbf{x} = \mathbf{x}'$ in the original model.

b) Consider the original model. By conditioning on the first visit to x' we obtain:

$$\mathbb{P}_{\mathbf{x},\theta}^{\pi} [\mathbf{x} = \mathbf{x}' \text{ for at least two numbers } \mathbf{n} > 0] =$$

$$= \sum_{k=1}^{\infty} \sum_{\substack{x_{1}, \dots, x_{k-1} = a_{0}, \dots, a_{k-1} \\ x_{j} \neq x'}} \int \nu(dy_{1}) \dots \int \nu(dy_{k}) \prod_{j=0}^{k-1} \{P(\{x_{j+1}\} | x_{j}, a_{j}, y_{j+1}) \\ \sum_{i \in I} i_{K_{i}}(x_{j}, a_{j}) P_{i}(y_{j+1} | \theta_{i}) \pi_{j}(\{a_{j}\} | x_{0}, a_{0}, y_{1}, x_{1}, \dots, y_{j}, x_{j})\}$$

$$\cdot \mathbb{P}_{x, \theta}^{\pi} [X_{n+k} = x' \text{ for some } n > 0 | X_{0} = x, A_{0} = a_{0}, Y_{1} = y_{1}, X_{1} = x_{1}, \dots, Y_{k} = y_{k}, X_{k} = x']$$

Analogous to the construction in 3.22, there is for each

84

 $\begin{aligned} &(\mathbf{a}_{0}, \mathbf{y}_{1}, \mathbf{x}_{1}, \dots, \mathbf{y}_{k}) \in (\mathbf{A} \times \mathbf{Y} \times \mathbf{X})^{k-1} \times \mathbf{A} \times \mathbf{Y} \quad \text{a strategy } \widetilde{\pi} \in \mathbb{I} \text{ such that} \\ & \mathbb{P}_{\mathbf{x}, \theta}^{\pi} [\mathbf{x}_{n+k} = \mathbf{x}' \text{ for some } n > 0 | \mathbf{x}_{0} = \mathbf{x}, \ \mathbf{A}_{0} = \mathbf{a}_{0}, \ \mathbf{y}_{1} = \mathbf{y}_{1}, \ \mathbf{x}_{1} = \mathbf{x}_{1}, \dots, \mathbf{y}_{k} = \mathbf{y}_{k}, \ \mathbf{x}_{k} = \mathbf{x}'] = \\ & = \mathbb{P}_{\mathbf{x}', \theta}^{\widetilde{\pi}} [\mathbf{x}_{n} = \mathbf{x}' \text{ for some } n > 0] . \end{aligned}$

This last term is equal to 1, by part (a). Therefore there are, $\mathbb{P}_{\mathbf{x},\theta}^{\pi}$ -a.s., at least two visits to x'. Repeating the argument yields $\mathbb{P}_{\mathbf{x},\theta}^{\pi}[\mathbf{X}_{n} = \mathbf{x}' \text{ for at least } \mathbf{k} \text{ numbers } \mathbf{n} > 0] = 1 \text{ and therefore}$ $\mathbb{P}_{\mathbf{x},\theta}^{\pi}[\mathbf{X}_{n} = \mathbf{x}' \text{ infinitely often}] = 1$.

Theorem 4.8

Let X and A be finite sets and let the Markov chain $\{X_n, n \in \mathbb{N}\}$ be irreducible for each stationary strategy and all $\theta \in \Theta$. Let M_1, \ldots, M_m be a partition of X and $K_i := M_i \times A$, $i = 1, \ldots, m$. Then the strategy π^* , defined in th. 4.6, is optimal.

Proof.

We only have to verify the assumptions of th. 4.6. As already noted, it has been proved in [Ross (1970) corollary 6.20] that 4.6 is true, here. Further 4.7 is a consequence of lemma 4.4. Since for each known parameter value $\theta \in \theta$ the Bayesian control model reduces to a dynamic program with state space X and action space A (cf. the proof of lemma 4.7) we have for each $\theta \in \theta$ an optimal stationary strategy (cf. corollary 4.3). Since there are finitely many of these strategies 4.15 is fulfilled. Finally, by lemma 4.7, we have $\mathbb{P}_{x,\theta}^{\pi}[X_n \in M_i \text{ infinitely often}] = 1$. Hence 4.14 holds.

Remarks.

The conditions of th. 4.8 are satisfied in particular if, for
 X = {x₁,...,x_m} we have M_i = {x_i}, i = 1,...,m and if for all θ ∈ θ,
 a ∈ A and x_i,x_i ∈ X:

$$\nu(dy)P(\{x_j\}|x_i,a,y)P_i(y|\theta_i) > 0$$
.

ii) In the situation of th. 4.8 we may use at each stage a Bayesian equivalent rule maximizing $a \rightarrow \int q(d\theta) L(x, \theta, a)$ in $(x,q) \in X \times W$, since A is finite.

In th. 4.8 we assumed that, if we are in state $x \in M_i$, then the information we get after the next transition does not depend on the action chosen. In th. 4.9 we relax this assumption. We shall assume there, besides 4.6 and 4.7:

- 4.16 (i) A is a finite set and N₁,...,N_n is a partition of A.
 (ii) M₁,...,M_m is a measurable partition of X and K_{i,j} = M_i × N_j, (i,j) ∈ I = {(i,j) | i = 1,...,m, j = 1,...,n}.
 (iii) For each x ∈ X, θ ∈ θ and π ∈ Π₀ P^π_{x,θ}[X_n ∈ M_i infinitely often] = 1 , i = 1,...,m .
 (iv) There is a finite set F of Markov policies such that
 - $\theta \rightarrow \inf \phi(\mathbf{x}, \theta, f(\mathbf{x}))$ is measurable, for $f \in F$ and $\mathbf{x} \in X$ max $\inf \phi(\mathbf{x}, \theta, f(\mathbf{x})) = 0$ (cf. 4.15). $f \in F \mathbf{x} \in X$

First, we discuss the form of a reasonable strategy for this situation. Then, in th. 4.9, we prove the optimality of such a strategy and afterwards, in th. 4.10 we consider a practical situation where 4.16 is satisfied in a natural way.

Although 4.16(iii) guarantees that we return to M_i , i = 1, ..., m, infinitely often under the strategy π^* defined in th. 4.6, it is not sure that we return to each set $K_{i,j}$ infinitely often. Hence we have to modify the strategy π^* of th. 4.6.

The idea for the modification is found in [Mallows and Robbins (1964)]. In [Fox and Rolph (1973)] and in [Rose (1975)] this idea is worked out for Markov renewal programs and Markov decision processes respectively, in a way similar to our approach here. The idea is, that we make *forced choice actions* to guarantee that we return to all sets $K_{i,j}$ infinitely often. However, we do this with a frequency that is so low as not to influence the Bayesian average return.

We start with some preparations.

We define a (double) sequence of stopping times $\{\sigma(i,t) | t \in \mathbb{N}, i=1,...,m\}$:

4.17 $\sigma(i,0)(\omega) := 0$, $\sigma(i,t)(\omega) := \inf\{k > \sigma(i,t-1)(\omega) | X_k(\omega) \in M_i\}$ for $\omega \in \Omega$, $i = 1, \dots, m$, $t \in \mathbb{N}^*$.

Hence $\sigma(i,t)$ is the time of the t-th visit to set M_i , after stage zero.

4.18 An increasing sequence S = (s₁,s₂,s₃,...) of positive integers is said to be of *density zero* if

$$\limsup_{k \to \infty} \frac{1}{k} \max\{i \in \mathbb{N}^* | s_i \le k\} = 0$$

Examples of such sequences are: $s_i = 2^i$, $i \in \mathbb{N}^*$, since

and
$$\mathbf{s}_{\mathbf{i}} = \mathbf{i}^2$$
, since
 $\frac{1}{k} \max\{\mathbf{i} \in \mathbb{N}^* | 2^{\mathbf{i}} \le k\} \le \frac{2\log k}{k}$
 $\frac{1}{k} \max\{\mathbf{i} \in \mathbb{N}^* | \mathbf{i}^2 \le k\} \le k^{-\frac{1}{2}}$.

We define the strategy π^{**} , which will be considered in th. 4.9, in an informal way.

4.19 Fix in each set N_j an action a_j for j = 1, ..., n and fix some sequence $S = (s_1, s_2, s_3, ...)$ of density zero. If for $t \in \mathbb{N}$ there are $i \in \{1, ..., m\}$ and $k \in \mathbb{N}^*$ such that $t = \sigma(i, s_k)$ then determine $l \in \mathbb{N}$ such that k = bm + l with $1 \le l \le m$, $b \in \mathbb{N}$. In that case action a_l is chosen at stage t. If $t \ne \sigma(i, s_k)$ at stage t, for all i = 1, ..., m and all $k \in \mathbb{N}^*$ then the Bayesian Markov policy f_t , defined in 4.13 is used to select an action.

It is straightforward to make this definition more rigorous in a way similar to definition 3.20.

Note that π^{**} tries the actions a_1, \ldots, a_n successively at the forced choice stages: $\sigma(i,t)$, $i = 1, \ldots, m$, $t \in S$.

Since it is assumed in 4.16(iii), that the process visits each set M_i infinitely often, almost surely under every strategy, we have almost surely: $\sigma(i,s_k) < \infty$ for all i = 1, ..., m and $k \in \mathbb{N}^*$. Hence each set $K_{i,j}$ is visited infinitely often almost surely, under all strategies.

Theorem 4.9

Assume 4.5, 4.6 and 4.16. Then the strategy π^{**} defined in 4.19 is optimal with Bayesian average return g(q) in each starting state (x,q) $\in X \times W$.

Proof.

Fix $(x,q) \in X \times W$. For notational convenience we write \mathbb{P} instead of $\mathbb{P}_{x,q}^{\pi}$. a) As noted above, we have by 4.16(iii) and definition 4.19:

$$IP[n n \{\tau((i,j),k) < \infty\}] = 1 .$$
i,jeI keIN

Using th. 2.7 for the stopping times $\sigma(i,0)$, $\sigma(i,1)$,... we find in exactly the same way as in the proof of th. 4.6 that \mathbb{P} -a.s.

(*)
$$\lim_{k \to \infty} \phi(X_{\sigma(i,k)}, Q_{\sigma(i,k)}, A_{\sigma(i,k)}) = 0, \quad \text{for } i = 1, \dots, m.$$

$$k \notin S$$

b) For notational convenience we write B(i,k) instead of

 $\phi(X_{\sigma\{i,k\}}, Q_{\sigma(i,k)}, A_{\sigma(i,k)})$. The following assertion is easy to verify (cf. 4.17). For all $t \in \mathbb{N}^{*}$ there is exactly one pair i,k such that $\sigma(i,k) = t, i \in \{1, \ldots, m\}, k \in \{1, \ldots, t\}.$

Hence it is easy to verify that on Ω :

$$(**) \qquad \sum_{t=1}^{N} \phi(X_{t}, Q_{t}, A_{t}) = \sum_{i=1}^{m} \sum_{k=1}^{N} B(i, k) 1_{\{\sigma(i, k) \le N\}} =$$

$$= \sum_{i=1}^{m} \sum_{k=1}^{N} B(i,k) \mathbb{1}_{\{\sigma(i,k) \le N, k \in S\}} + \sum_{i=1}^{m} \sum_{k=1}^{N} B(i,k) \mathbb{1}_{\{\sigma(i,k) \le N, k \notin S\}}$$

Notice that $\sigma(i,k) \ge k$ for $k \in \mathbb{N}^{*}$. Hence we have

$$#\{k \in S | \sigma(i,k) \leq N\} \leq #\{k \in S | k \leq N\}.$$

Let M := inf $\phi(x,\theta,a)$. Then $-\infty < M \le 0$, since r is bounded and by 4.5. x, θ,a Therefore we have for i = 1,...,m:

$$\frac{1}{N}\sum_{k=1}^{N} B(\mathbf{i},k) \mathbf{1}_{\{\sigma(\mathbf{i},k) \leq N, k \in S\}} \geq \frac{M}{N} \# \{k \in S \mid k \leq N\}$$

which tends to zero as N tends to infinity by the definition of S (cf. 4.18). Now we consider the last term of (**). Since, by (*), B(i,k) tends \mathbb{P} -a.s. to zero, if k \notin S and if k tends to infinity, we have by lemma 4.2:

$$\lim_{N\to\infty} \frac{1}{N} \sum_{k=1}^{N} B(\mathbf{i},k) \mathbf{1}_{\{\sigma(\mathbf{i},k)\leq N, k\notin S\}} = 0 , \quad \mathbb{P}\text{-a.s.}$$

Finally we conclude from (**):

$$\lim_{N\to\infty}\frac{1}{N}\sum_{k=1}^{N}\phi(X_k,Q_k,A_k) = 0, \quad \text{IP-a.s.}$$

And therefore 4.10 is satisfied. Hence by th. 4.1 the theorem is proved.

In the following theorem we give more appealing conditions, which imply all conditions of th. 4.9. Hence the strategy π^{**} defined in 4.19 is optimal here.

Theorem 4.10

Let X and A be finite sets and let the Markov chain $\{X_n, n \in \mathbb{N}\}$ be irreducible for all $\theta \in \theta$ and each stationary strategy. Further let $I = X \times A$ and let $K_{x,a} = \{(x,a)\}$. Then the strategy π^{**} defined in 4.19 is optimal.

The proof of this theorem proceeds along the same lines as the proof of th. 4.8.

We conclude this section with some remarks.

Remarks.

- i) Consider the situation of th. 4.9.
- The strategy π^{**} is easy to handle. For each set M_i , i = 1, ..., m the decision maker has to keep count of the number of visits. If this number is equal to a number in the sequence S of density zero, then he has to select the next action from $\{a_1, ..., a_n\}$ in cyclical order (cf. 4.19). If the number of the visits does not belong to S, then the decision maker has to compute in state (x,q) an action a^* , such that $\phi(x,q,a^*) = \max \phi(x,q,a)$.
- ii) If we are dealing with a dynamic program with finite state space X and finite action space A and if for all x,x' \in X and all a \in A: P(x'|x,a) is positive but unknown, then we can transform this model into our Bayesian control model (cf. example 2.2), and by th. 4.10 the strategy π^{**} is optimal.
- 111) Th. 4.10 is more general then the results in [Rose (1975)], since we allow arbitrary prior distributions. Further, the strategy π^{**} is easier to handle than the strategy Rose proposes, if θ is finite.
- iv) It is not clear whether all situations considered in [Mandl (1974)] are covered by th. 4.6 or not. Mandl assumes that $\theta \Rightarrow \phi(\mathbf{x}, \theta, \mathbf{a})$ is continuous (cf. [Mandl (1974) th. 8]), moreover he assumes the existence of minimum contrast estimators. Although we conjecture that under the assumptions of th. 4.8 minimum contrast estimators exist, it is easy to show that under the assumptions of th. 4.10 they do not exist.

5. BAYESIAN EQUIVALENT RULES AND THE TOTAL-RETURN CRITERION

We do not know Bayesian equivalent rules that are optimal with respect to the discounted total-return criterion, in general. Example 5.1 shows that a "natural" Bayesian equivalent rule fails to be optimal. However, in section 5.1, we prove the optimality of a Bayesian equivalent rule for the so-called independent case, and in section 5.2 for the linear system with quadratic costs. Finally, in section 5.3, we study a simple inventory model for which a Bayesian equivalent rule is sometimes optimal. Here we also study the behaviour of this rule when it is not optimal.

5.1 Preliminaries and the independent case

In the models we study in this chapter, there is only one unknown parameter, i.e. the index set I is a singleton. This implies that the decision maker obtains information about the same parameter in each state $x \in X$, regardless of the action chosen. Since I is a singleton we shall omit the subscript i in the notations θ_i , $p_i(y|\theta_i)$, $p_i(y,q)$ and $T_{i,y}(q)$. Note that $\tau(1,n) = n$ (on Ω), for all $n \in \mathbb{N}$ and therefore, by lemma 2.2, the distribution of Y_1, \ldots, Y_n , $n \in \mathbb{N}$ only depends on the prior distribution and not on the starting state or the strategy. Hence the distribution of Q_n (cf. 2.24) depends only on the prior distribution and on Y_1, \ldots, Y_n . For that reason we shall write \mathbf{P}_q and \mathbf{E}_q instead of $\mathbf{P}_{x,q}^{\Pi}$ and $\mathbf{E}_{x,q}^{\Pi}$, when we are dealing with the random variables Y_n and Q_n .

We start with an example. In this example the Bayesian equivalent rule, based on the function:

$$\mathbf{F}(\mathbf{x},\boldsymbol{\theta},\mathbf{a}) := \int \mathbf{v}(d\mathbf{y})\mathbf{p}(\mathbf{y}|\boldsymbol{\theta}) \{\mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y}) + \boldsymbol{\beta} \int \mathbf{P}(d\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y})\mathbf{v}(\mathbf{x}',\boldsymbol{\theta})\}$$

turns out to be non-optimal.

We remark that this example has some similarity to example 4.1.

Example 5.1

Consider the following model. $X = \{1, 2, ..., 6\}, Y = \{0, 1\}, D(1) = A = \{1, 2, 3\}, D(x) := \{1\}, x \in \{2, 3, ..., 6\}, \theta := \{0, 1\}.$ The function $p(y|\theta)$ is given by: $p(1|\theta) = 1 - p(0|\theta) = \theta, \theta \in \theta$. And $P(\{x^i\}|x, a, y)$ is (we identify here x and $\{x\}$):

$$P(3|3,1,0) = P(3|4,1,0) = P(6|5,1,0) = P(6|6,1,0) = P(2|1,1,0) =$$
$$= P(1|2,1,0) = 1$$



Only in the states 3,4,5 and 6 a reward is obtained independent of $y \in Y$: r(3) = r(5) = c; r(4) = r(6) = b, c > b ≥ 0. The prior distribution q is: q(0) = q(1) = ½. The discounted total return $v(x,\theta)$ for discount factor β is:

$$v(3,0) = v(5,1) = \frac{c}{1-\beta}; v(3,1) = v(5,0) = c + \beta \frac{b}{1-\beta} (< \frac{c}{1-\beta}).$$

Hence $v(3,\theta) = (c + \frac{\beta_D}{1-\beta})\theta + \frac{c}{1-\beta}(1-\theta)$ and $v(5,\theta) = \frac{c}{1-\beta}\theta + (c + \frac{\beta_D}{1-\beta})(1-\theta)$. Further $v(2,\theta) = \frac{\beta(1-\theta)}{1-\beta\theta}v(1,\theta)$, hence

$$\mathbf{v}(1,\theta) = \max\{\beta\theta\mathbf{v}(1,\theta) + \beta(1-\theta)\frac{\beta(1-\theta)}{1-\beta\theta}\mathbf{v}(1,\theta), \beta\mathbf{v}(3,\theta), \beta\mathbf{v}(5,\theta)\}.$$

The first term equals:

$$\beta \frac{\theta + \beta - 2\beta\theta}{1 - \beta\theta} v(1, \theta) < v(1, \theta) \quad \text{for } \beta \in (0, 1), \ \theta \in \Theta .$$

The Bayesian equivalent rule is based on the function F specified by:

$$\mathbf{F}(1,\theta,1) = \beta \ \frac{\theta + \beta - 2\beta\theta}{1 - \beta\theta} \ \mathbf{v}(1,\theta), \ \mathbf{F}(1,\theta,2) = \beta \mathbf{v}(3,\theta)$$

anđ

$$F(1,\theta,3) = \beta v(5,\theta)$$

Hence the Bayesian equivalent rule in state (1,q) chooses action 2 or 3, with equal Bayesian discounted total return: $\frac{1}{2}\left\{c + \frac{\beta b}{1-\beta} + \frac{c}{1-\beta}\right\}$. Now we consider another strategy for starting in state 1.

At stage 1 take action 1 and thereafter take the best of actions 2 and 3, in state 1. Note that under this strategy the system remains in state 1 at stage 1, or it returns to state 1 at stage 2. The discounted total return becomes: $\beta^3 \frac{c}{1-\beta}$ if $\theta = 0$ and $\beta^2 \frac{c}{1-\beta}$ if $\theta = 1$. Hence the Bayesian dis-

counted total return is: $\frac{1}{2} \frac{c}{1-\beta} \beta^2 (\beta + 1)$. So, if for instance c = 2, b = 1 and $\beta = 0.9$, then this strategy is better than the Bayesian equivalent rule. It is straightforward to show that the latter strategy is optimal, in this case.

We proceed with a theorem. In this theorem we show that the process of posterior distributions is a Markov chain.

Theorem 5.1

(i) Let f be a real-valued measurable function on $W \times Y$ that is bounded from above. Then, for $n \in \mathbb{N}^*$:

$$\mathbb{E}_{q}[f(Q_{n}, Y_{n+1}) \mid Y_{1}, \dots, Y_{n}] = \int f(Q_{n}, y) p(y, Q_{n}) v(dy), \mathbb{P}_{q}^{-a.s.}$$

(ii) The process $\{\varrho_n,\ n\in\mathbb{N}\}$ is a (homogeneous) Markov chain.

Proof.

(i) Define

$$\mathbf{q}_{\mathbf{n}}(\mathbf{d}\boldsymbol{\theta}) := \mathbf{q}(\mathbf{d}\boldsymbol{\theta}) \prod_{j=1}^{n} \mathbf{p}(\mathbf{y}_{j} | \boldsymbol{\theta}) \cdot \left\{ \int \mathbf{q}(\mathbf{d}\boldsymbol{\theta}) \prod_{j=1}^{n} \mathbf{p}(\mathbf{y}_{j} | \boldsymbol{\theta}) \right\}^{-1}$$

if the denominator is non-zero. Let B \in y^n . Using lemma 2.2 (T = 1) we have

(*)
$$\int_{\{(Y_1,\ldots,Y_n)\in B\}} \mathbf{E}_q[f(Q_n,Y_{n+1}) \mid Y_1,\ldots,Y_n)] d\mathbf{I}_q =$$
$$= \int_{\{(d\theta)\}} \left\{ \int_{\{(Y_1,\ldots,Y_{n+1})\in B\times Y\}} f(q_n,Y_{n+1}) \prod_{j=1}^{n+1} p(y_j \mid \theta) \lor (dy_1) \ldots \lor (dy_{n+1}) \right\}$$

Note that

$$\int p(\mathbf{y}_{n+1} \mid \boldsymbol{\theta}) \prod_{j=1}^{n} p(\mathbf{y}_{j} \mid \boldsymbol{\theta}) q(d\boldsymbol{\theta}) \left\{ \int \prod_{j=1}^{n} p(\mathbf{y}_{j} \mid \boldsymbol{\theta}) q(d\boldsymbol{\theta}) \right\}^{-1} = p(\mathbf{y}_{n+1}, q_{n}) ,$$

if the factor between braces is non-zero. Hence (*) equals:

$$\int \dots \int f(\mathbf{q}_n, \mathbf{y}_{n+1}) p(\mathbf{y}_{n+1}, \mathbf{q}_n) f(\mathbf{y}_{n+1}, \mathbf{q}_n)$$

}.

$$\cdot \left\{ \int_{j=1}^{n} p(y_{j} | \theta) q(d\theta) \right\} \vee (dy_{1}) \dots \vee (dy_{n+1}) =$$

$$= \mathbf{E}_{q} \left[\mathbf{1}_{\{(Y_{1}, \dots, Y_{n}) \in \mathbf{B}\}} \int f(Q_{n}, y) p(y, Q_{n}) \vee (dy) \right]$$

which proves part (i).

(11)

$$\begin{split} \mathbf{E}_{\mathbf{q}}[f(\mathbf{Q}_{n+1}) \mid \mathbf{Q}_{0}, \dots, \mathbf{Q}_{n}] &= \mathbf{E}_{\mathbf{q}}[\mathbf{E}_{\mathbf{q}}[f(\mathbf{Q}_{n+1}) \mid \mathbf{Y}_{1}, \dots, \mathbf{Y}_{n}] \mid \mathbf{Q}_{0}, \dots, \mathbf{Q}_{n}] = \\ &= \mathbf{E}_{\mathbf{q}}[\int f(\mathbf{T}_{\mathbf{Y}}(\mathbf{Q}_{n})) \mathbf{p}(\mathbf{Y}, \mathbf{Q}_{n}) \mathbf{v}(d\mathbf{Y}) \mid \mathbf{Q}_{0}, \dots, \mathbf{Q}_{n}] = \int f(\mathbf{T}_{\mathbf{Y}}(\mathbf{Q}_{n})) \mathbf{p}(\mathbf{Y}, \mathbf{Q}_{n}) \mathbf{v}(d\mathbf{Y}) = \\ &= \mathbf{E}_{\mathbf{Q}_{n}}[f(\mathbf{Q}_{1})] = \mathbf{E}_{\mathbf{q}}[f(\mathbf{Q}_{n+1}) \mid \mathbf{Q}_{n}], \ \mathbf{P}_{\mathbf{q}}^{-\mathbf{a.s.}} \end{split}$$

The first equality follows from the fact that Q_m is a function of q and Y_1, \ldots, Y_m . The second equality is a consequence of part (i) and the equality $Q_{n+1} = T_{Y_{n+1}} (Q_n)$. The other equalities are obvious.

In th. 5.3 we prove the optimality of a Bayesian equivalent rule in the independent case. Here the reward function r is constant in the first coordinate, i.e. at each stage the reward only depends on the chosen action and the value of the supplementary state variable. Further it is assumed that all actions are available in every state, i.e. D(x) = A, for $x \in X$. Since, given Z, the sequence $\{Y_n, n \in \mathbb{N}^*\}$ is a sequence of i.i.d. random variables (cf. lemma 2.2) we call this case the *independent case*. It will play an important role in section 5.3. We start with a lemma.

Lemma 5.2

Let G be an upper semi-continuous (u.s.c.) function on $A \times Y$, that is bounded from above. Let A be compact. Then there is a measurable function f: $W \rightarrow A$ such that:

$$G(f(q),y)p(y,q)\nu(dy) = \max_{a \in A} \int G(a,y)p(y,q)\nu(dy) dx$$

Proof.

We show that all conditions of Schäl's selection theorem (cf. A17) are satisfied. Let G be u.s.c. and bounded above by $M \in \mathbb{R}$. Then there are bounded continuous functions G_k on $A \times Y$, such that the sequence $\{G_k, k \in \mathbb{N}\}$ is nonincreasing and $\lim_{k \to \infty} G_k = G$ (see A12).

Without loss of generality we may assume that $G_k \leq M$, $k \in \mathbb{N}$, since otherwise we may define $\widetilde{G}_k := \min\{G_k, M\}$ and then $\{\widetilde{G}_k, k \in \mathbb{N}\}$ is also a nonincreasing sequence of bounded continuous functions with limit G. Hence, by the monotone convergence theorem we have

$$\lim_{k\to\infty}\int \{-G_k(a,y) + M\}p(y,q)\nu(dy) = \int \{-G(a,y) + M\}p(y,q)\nu(dy)\}$$

and so { $\int G_k(a,y)p(y,q)\nu(dy)$, $k \in \mathbb{N}$ } is a nonincreasing sequence with limit $\int G(a,y)p(y,q)\nu(dy)$.

By the dominated convergence theorem the function

$$(a,q) \rightarrow \int G_k(a,y)p(y,q)v(dy)$$

is continuous in a, and for fixed k it is bounded since G_k is bounded. Using lemma 1.6 (iii) we find that this function is measurable, since

$$(\mathbf{a}, \mathbf{\theta}) \rightarrow \int G_{\mathbf{k}}(\mathbf{a}, \mathbf{y}) p(\mathbf{y} | \mathbf{\theta}) v(\mathbf{d}\mathbf{y})$$

is measurable. Hence we proved that $\int G_k(a,y)p(y,q)\nu(dy) \in L(W \times A)$ and therefore $\int G(a,y)p(y,q)\nu(dy) \in \hat{L}(W \times A)$ (cf. A17). Hence all conditions of A17 are satisfied. This proves the lemma.

Theorem 5.3

Let I be a singleton, let A be compact and let D(x) = A, $x \in X$. Further let $x \neq r(x,a,y)$ be constant for all $a \in A$, $y \in Y$ and let $(a,y) \neq r(x,a,y)$ be u.s.c. (We write r(a,y) := r(x,a,y), $a \in A$, $y \in Y$.) Then there is a strategy $\pi^* \in \Pi_0$ that choses a maximizer of $a \neq \int r(a,y)p(y,q)v(dy)$ in each state $(x,q) \in X \times W$. This strategy is optimal, and

$$\mathbf{v}(\mathbf{x},\mathbf{q}) = \sum_{n=0}^{\infty} \beta^{n} \mathbf{E}_{\mathbf{q}} [e(Q_{n})]$$

where;

5.1
$$e(q) := \max_{a \in A} \int r(a,y)p(y,q)v(dy), q \in W$$
.

Remember that r is bounded from above. Let $x \in X$, $q \in W$ and $\pi \in \Pi_{\alpha}$. We have:

$$\mathbf{E}_{x,q}^{\pi} [r(\mathbf{A}_{n}, \mathbf{Y}_{n+1})] = \mathbf{E}_{x,q}^{\pi} [\mathbf{E}_{x,q}^{\pi} [r(\mathbf{A}_{n}, \mathbf{Y}_{n+1}) | \mathbf{X}_{0}, \mathbf{Q}_{0}, \mathbf{A}_{0}, \dots, \mathbf{X}_{n}, \mathbf{Q}_{n}]] .$$

Since $\pi \in \Pi_0$, there is a corresponding strategy $\tilde{\pi} \in \tilde{\Pi}$ (cf. 3.7). Therefore we have $\mathbf{P}_{\mathbf{x},\mathbf{q}}^{\pi}$ -a.s.:

$$\begin{split} & \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\mathbf{r}(\mathbf{A}_{n},\mathbf{Y}_{n+1}) \mid \mathbf{X}_{0},\mathbf{Q}_{0},\mathbf{A}_{0},\ldots,\mathbf{X}_{n},\mathbf{Q}_{n})] = \\ & = \int \widetilde{\pi}_{n} (\mathrm{da} \mid \mathbf{X}_{0},\mathbf{Q}_{0},\mathbf{A}_{0},\ldots,\mathbf{X}_{n},\mathbf{Q}_{n}) \int \vee (\mathrm{dy}) \mathbf{p}(\mathbf{y},\mathbf{Q}_{n}) \mathbf{r}(\mathbf{a},\mathbf{y}) \leq \\ & \leq \sup_{\mathbf{a} \in \mathbf{A}} \int \vee (\mathrm{dy}) \mathbf{p}(\mathbf{y},\mathbf{Q}_{n}) \mathbf{r}(\mathbf{a},\mathbf{y}) \ . \end{split}$$

By lemma 5.2 there is a measurable function f from W to Y such that

$$\max_{a \in A} \int v(dy) p(y,Q_n) r(a,y) = \int v(dy) p(y,Q_n) r(f(Q_n),y) = e(Q_n) .$$

Note that the distribution of $e(Q_n)$ does not depend on x and π . Hence

$$\mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi}[\mathbf{\hat{r}}(\mathbf{A}_{n},\mathbf{Y}_{n+1})] \leq \mathbf{E}_{\mathbf{q}}[\mathbf{e}(\mathbf{Q}_{n})] \quad \text{for all } \mathbf{x} \in \mathbf{X}, \ \pi \in \Pi_{\mathbf{0}}$$

with equality if the strategy π^* is used. This proves the theorem.

The strategy π^* defined in th. 5.3 uses a Bayesian equivalent rule. To verify this, note that an optimal strategy for the model with known parameter θ is obtained by using a maximizer of $a \rightarrow F(x, \theta, a)$ at each stage, where

$$F(x,\theta,a) := \int v(dy)p(y|\theta)r(a,y)$$
.

Hence a Bayesian equivalent rule may be defined as a maximizer of $a \neq \int q(d\theta)F(x,\theta,a)$, in each state $(x,q) \in X \times W$. Hence π^* uses a Bayesian equivalent rule at each stage. Note that each maximizer of $a \neq \int q(d\theta)F(x,\theta,a)$ is also a maximizer of

$$\mathbf{a} \neq \int \mathbf{q}(\mathbf{d}\theta) \mathbf{p}(\mathbf{y}|\theta) \{ \mathbf{r}(\mathbf{a},\mathbf{y}) + \beta \int \mathbf{P}(\mathbf{d}\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y}) \mathbf{v}(\mathbf{x}',\theta) \}$$

since $x \neq v(x, \theta)$ is constant for all $\theta \in \theta$. Hence π^* uses a "natural" Bayesian equivalent rule.

In th. 5.4 we give an upper and a lower bound for the value function of the model. These provide a measure for the loss of return, due to the lack of information concerning the "true" parameter value.

Theorem 5.4

Under the conditions of th. 5.3 we have:

5.2
$$\frac{e(q)}{1-\beta} \leq v(x,q) \leq \frac{\int q(d\theta)e(\theta)}{1-\beta}.$$

Proof.

The right hand inequality follows from th. 3.16. To prove the left hand inequality, note that:

$$\mathbf{E}_{\mathbf{q}}[\mathbf{e}(\mathbf{Q}_{\mathbf{n}})] \geq \sup_{\mathbf{a}\in\mathbf{A}} \mathbf{E}_{\mathbf{q}}[\int \mathbf{Q}_{\mathbf{n}}(\mathbf{d}\theta) \{\int \mathbf{v}(\mathbf{d}\mathbf{y})\mathbf{p}(\mathbf{y}|\theta)\mathbf{r}(\mathbf{a},\mathbf{y})\}] = \sup_{\mathbf{a}\in\mathbf{A}} \int \mathbf{q}(\mathbf{d}\theta) \{\int \mathbf{v}(\mathbf{d}\mathbf{y})\mathbf{p}(\mathbf{y}|\theta)\mathbf{r}(\mathbf{a},\mathbf{y})\} = \mathbf{e}(\mathbf{q}) .$$

The first equality follows from the fact that $\mathbb{E}_{q}\left[\int f(\theta)Q_{n}(d\theta)\right] = \int f(\theta)q(d\theta)$, for real-valued measurable functions on θ , which are bounded from above.

We conclude this section with an example which has some relationship with the inventory control model we study in section 5.3. The model we consider in this example can be transformed into the model we called the independent case.

Example 5.2

Let I be a singleton, D(x) = A for all $x \in X$ and let A be compact. Further let r(x,a,y) = b(x) + c(a), $x \in X$ and $a \in A$, where b and c are u.s.c. and bounded from above on X and A respectively. Finally let $P({G(a,y)} | x,a,y) = 1$ for all $x \in X$, $a \in A$ and $y \in Y$ where G is a continuous function from $A \times Y$ to X.

For each $x \in X$, $q \in W$ and $\pi \in \Pi_{\Omega}$ we have

$$\mathbf{v}(\mathbf{x},\mathbf{q},\pi) = \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \begin{bmatrix} \sum_{n=0}^{\infty} \beta^{n} \mathbf{x} (\mathbf{X}_{n}, \mathbf{A}_{n}, \mathbf{Y}_{n+1}) \end{bmatrix} =$$
$$= \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \begin{bmatrix} \mathbf{b}(\mathbf{X}_{0}) + \sum_{n=0}^{\infty} \beta^{n} \{ \mathbf{c}(\mathbf{A}_{n}) + \beta \mathbf{b}(\mathbf{G}(\mathbf{A}_{n}, \mathbf{Y}_{n+1})) \} \end{bmatrix} =$$
$$= \mathbf{b}(\mathbf{x}) + \sum_{n=0}^{\infty} \beta^{n} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \begin{bmatrix} \mathbf{c}(\mathbf{A}_{n}) + \beta \mathbf{b}(\mathbf{G}(\mathbf{A}_{n}, \mathbf{Y}_{n+1})) \end{bmatrix} .$$

Define $\bar{r}(a,y) := c(a) + \beta b(G(a,y))$, $a \in A$, $y \in Y$. Then

$$v(x,q,\pi) = b(x) + \mathbf{E}_{x,q}^{\pi} \left[\sum_{n=0}^{\infty} \beta^{n} \ \bar{r}(A_{n},Y_{n+1}) \right].$$

Note that $(a,y) \rightarrow \bar{r}(a,y)$ is u.s.c. (cf. A15). Hence, by th. 5.3, we find that an optimal strategy is obtained by choosing in each state $(x,q) \in X \times W$ a maximizer of

$$a \rightarrow \{c(a) + \beta \int b(G(a,y))p(y,q)v(dy)\}$$

anđ

$$v(\mathbf{x},\mathbf{q}) = b(\mathbf{x}) + \sum_{n=0}^{\infty} \beta^{n} \mathbf{E}_{\mathbf{q}} [\tilde{\mathbf{e}}(\mathbf{Q}_{n})],$$

where

$$\widehat{e}(q) := \max\{c(a) + \beta \int b(G(a,y))p(y,q)v(dy)\} .$$

a < A

As in th. 5.4 we have the inequalities:

$$\mathbf{b}(\mathbf{x}) + \widetilde{\mathbf{e}}(\mathbf{q}) (1-\beta)^{-1} \leq \mathbf{v}(\mathbf{x},\mathbf{q}) \leq \mathbf{b}(\mathbf{x}) + \int \widetilde{\mathbf{e}}(\theta) \mathbf{q}(\mathrm{d}\theta) (1-\beta)^{-1}, \quad (\mathbf{x},\mathbf{q}) \in \mathbf{X} \times \mathbf{W}.$$

In the following sections we study models which have some practical relevance. As this is more natural, in these models we shall minimize costs rather than maximize rewards.

Note that all results up to here carry over if we define

5.3 (i)
$$(\mathbf{U}_{\tau}\mathbf{b})(\mathbf{x},\mathbf{q}) := \inf_{\boldsymbol{\pi} \in \Pi_{\Omega}} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\sum_{n=0}^{\tau-1} \beta^{n} c(\mathbf{X}_{n},\mathbf{A}_{n}) + \beta^{\tau} b(\mathbf{X}_{\tau},\boldsymbol{Q}_{\tau})]$$

for real-valued measurable functions b, that are bounded from below. (ii) $v(x,q) = (U_m 0)(x,q)$.

5.2 Linear system with quadratic costs

In this chapter we consider a linear system with quadratic costs and with a disturbance process of i.i.d. random variables with an incompletely known distribution. We show the optimality of a Bayesian equivalent rule. In fact this rule can also be considered as a so-called *certainty equivalent rule*. We generalize results of M. Aoki on this topic in several ways: first we allow other disturbance processes than normal processes, secondly we allow general prior distributions. Finally we allow the costs to be a quadratic

98

function of the control variable (cf. [Aoki (1967), page 94]. The concepts and techniques we use here, are familiar in the theory of linear systems (cf. [Kushner (1971), chapter 9] and [Bertsekas (1976)]). We remark that the greater part of this section appeared in [van Hee (1976)]. We shall use symbols that we used before. However, they loose their previous interpretation here. We start with the specifications of the model. We proceed with some preliminary results, and in th. 5.9 we obtain one of the main results of this section: an explicit expression for the optimal strategy and also for the value function.

In this section x' means the transpose of x, where x is a column vector or a matrix.

Model 4: the linear system

5.4 (i)
$$X := Y := \mathbb{R}^{n_1}, n_1 \in \mathbb{N}^*,$$

- (ii) $D(x) := A := \mathbb{R}^{n_2}, n_2 \in \mathbb{N}^*$ for all $x \in X$,
- (iii) c(x,a) := x'Rx + a'Sa where R is a nonnegative definite $n_1 \times n_1^{-1}$ matrix and S a positive definite $n_2 \times n_2^{-matrix}$,
- (iv) $P(\{Cx + Ba + y\} | x,a,y) = 1, x \in X, a \in A, y \in Y$ where C is a $n_1 \times n_1$ -matrix and B a $n_1 \times n_2$ -matrix satisfying the controllability assumption

 n_1^{-1} rank[B,CB,...,C B] = n_1 ,

(v) $\int v(dy) | y_i y_j | p(y|\theta)$ is bounded on θ where y_i is the i-th component of $y \in Y$, for all i, $j \in \{1, ..., n_i\}$.

For $\mathbf{q} \in \mathbf{W}$ we define the vector $\mathbf{m}_{_{\mathbf{G}}}$ and the matrices $\mathbf{M}_{_{\mathbf{G}}}$ and $\boldsymbol{\Sigma}_{_{\mathbf{G}}}$:

5.5 (i)
$$m_{q}(i) := \int y_{i} p(y,q) v(dy), i \in \{1,...,n_{1}\}$$
.
(ii) $M_{q}(i,j) := \int m_{\theta}(i) m_{\theta}(j) q(d\theta), i,j \in \{1,...,n_{1}\}$.
(iii) $\Sigma_{q}(i,j) := \int y_{i} y_{j} p(y,q) v(dy), i,j \in \{1,...,n_{1}\}$.

Note that $\Sigma_{q} - M_{q}$ is the *covariance matrix* of Y_{n} averaged over θ with q. By assumption 5.4 (v) m_{q} , M_{q} and Σ_{q} are bounded on W.

In lemma 5.5 we give some properties of m_{d} and M_{d} .

Lemma 5.5

For $q \in W$ we have

(i)
$$\int m_{T_{y}(q)}(i)p(y,q)v(dy) = m_{q}(i), \quad i \in \{1,...,n_{1}\}.$$

(ii)
$$\int y_{j}m_{T_{y}(q)}(i)p(y,q)v(dy) = M_{q}(i,j), \quad i,j \in \{1,...,n_{1}\}.$$

Proof.

$$\begin{split} \mathfrak{m}_{\mathrm{T}_{\mathbf{Y}}(\mathbf{q})}(\mathbf{i}) \mathfrak{p}(\mathbf{y},\mathbf{q}) &= \mathfrak{p}(\mathbf{y},\mathbf{q}) \int \widetilde{\mathbf{y}}_{\mathbf{i}} \left\{ \int \frac{\mathfrak{p}(\widetilde{\mathbf{y}}|\theta) \mathfrak{p}(\mathbf{y}|\theta)}{\mathfrak{p}(\mathbf{y},\mathbf{q})} \mathbf{q}(\mathrm{d}\theta) \right\} \mathfrak{v}(\mathrm{d}\widetilde{\mathbf{y}}) &= \\ &= \int \left\{ \int \widetilde{\mathbf{y}}_{\mathbf{i}} \mathfrak{p}(\widetilde{\mathbf{y}}|\theta) \mathfrak{p}(\mathbf{y}|\theta) \mathfrak{v}(\mathrm{d}\widetilde{\mathbf{y}}) \right\} \mathbf{q}(\mathrm{d}\theta) \quad . \end{split}$$

Hence

and

$$\int m_{\mathbf{T}_{\mathbf{Y}}(\mathbf{q})} (\mathbf{i}) p(\mathbf{y},\mathbf{q}) v(d\mathbf{y}) = \int \{ \int \widetilde{\mathbf{y}}_{\mathbf{i}} p(\widetilde{\mathbf{y}}|\theta) v(d\widetilde{\mathbf{y}}) \}_{\mathbf{q}} (d\theta) = m_{\mathbf{q}}(\mathbf{i})$$

$$\int \mathbf{y}_{\mathbf{j}} m_{\mathbf{T}_{\mathbf{y}}(\mathbf{q})} (\mathbf{i}) p(\mathbf{y},\mathbf{q}) v(d\mathbf{y}) = \iiint \mathbf{y}_{\mathbf{j}} \widetilde{\mathbf{y}}_{\mathbf{i}} p(\widetilde{\mathbf{y}}|\theta) p(\mathbf{y}|\theta) v(d\widetilde{\mathbf{y}}) v(d\mathbf{y}) q(d\theta) =$$

$$= \int \{ \int \mathbf{y}_{\mathbf{j}} p(\mathbf{y}|\theta) v(d\mathbf{y}) \}_{\mathbf{q}} \{ \int \widetilde{\mathbf{y}}_{\mathbf{j}} p(\widetilde{\mathbf{y}}|\theta) v(d\widetilde{\mathbf{y}}) \}_{\mathbf{q}} (d\theta) = M_{\mathbf{q}}(\mathbf{i},\mathbf{j}) .$$

Note that all changes of integration order are allowed by 5.4 (v). $\hfill \square$

Lemma 5.6 states that the optimal reward operator U (cf. 5.3(i)) maps the set of functions f on $X \times W$ of the form given in 5.6 below, into itself. The proof proceeds in a familiar way (cf. [Kushner (1971), section 9.2.2]).

Lemma 5.6

Let the real-valued function f on $X \times W$ be defined by:

5.6
$$f(x,q) := x'Kx + x'Im_q + H(q), x \in X, q \in W$$
,

where K is a nonnegative definite matrix, L an arbitrary $n_1 \times n_1$ -matrix and H a bounded and measurable function on W. Then:

$$(Uf)(x,q) := x'\widetilde{K}x + x'\widetilde{I}m_{q} + \widetilde{H}(q), \quad x \in X, q \in W,$$

where

5.7 (i)
$$\widetilde{K} := G_1(K) := R + \beta C'KC - \beta^2 C'KB(S + \beta B'KB)^{-1}B'KC$$
.
(ii) $\widetilde{L} := G_2(L,K) := 2\beta C'K + \beta C'L - \beta^2 C'KB(S + \beta B'KB)^{-1}(2B'K + B'L)$.

(iii)
$$\widetilde{H}(q) := G_3(q,H,K,L) := -\frac{1}{2}\beta^2 m_q^* (2KB + L'B) (S + \beta B'KB)^{-1} (2B'K + B'L)m_q + \beta \int H(T_y(q))p(y,q)v(dy) + \beta trace(K\Sigma_q) + \beta trace(LM_q)$$

and the minimizing action a(x,q) in (x,q) is:

5.8
$$a(x,q) = -\beta (S + \beta B'KB)^{-1} B'KCx - \beta (S + \beta B'KB)^{-1} (B'K + \frac{1}{2}B'L)m_{q}$$

Further: \widetilde{K} is nonnegative definite and $\widetilde{H}(\cdot)$ is bounded and measurable on W.

Proof.

(1) By some straightforward calculations, using lemma 5.5 we get:

$$(\mathrm{Uf})(\mathbf{x},\mathbf{q}) = \inf_{\mathbf{a} \in \mathbf{A}} \{\mathbf{a}^{*}(\mathbf{S} + \beta \mathbf{B}^{*}\mathbf{K}\mathbf{B})\mathbf{a} + (2\beta\mathbf{x}^{*}\mathbf{C}^{*}\mathbf{K}\mathbf{B} + 2\beta\mathbf{m}^{*}_{\mathbf{q}}\mathbf{K}\mathbf{B} + \beta\mathbf{m}^{*}_{\mathbf{q}}\mathbf{L}^{*}\mathbf{B})\mathbf{a}\} + \mathbf{a} \in \mathbf{A}$$

+ $\mathbf{x}^{*}(\mathbf{R} + \beta\mathbf{C}^{*}\mathbf{K}\mathbf{C})\mathbf{x} + \beta\mathbf{x}^{*}(2\mathbf{C}^{*}\mathbf{K} + \mathbf{C}^{*}\mathbf{L})\mathbf{m}_{\mathbf{q}} + \beta \int_{\mathbf{H}} \mathbf{H}(\mathbf{T}_{\mathbf{y}}(\mathbf{q}))\mathbf{p}(\mathbf{y},\mathbf{q})^{\vee}(d\mathbf{y}) + \beta \operatorname{trace}(\mathbf{K}\Sigma_{\mathbf{q}}) + \beta \operatorname{trace}(\mathbf{L}\mathbf{M}_{\mathbf{q}}) .$

Since K is nonnegative definite and S is positive definite we have $S + \beta B'KB$ is positive definite and therefore $(S + \beta B'KB)^{-1}$ exists and is positive definite. Hence by standard arguments for the minimization of quadratic forms we find 5.7 and 5.8.

(2) We shall prove that \widetilde{K} is nonnegative definite again. Note that the value of \widetilde{K} does not depend on L, H or $p(y|\theta)$, $y \in Y$, $\theta \in \theta$. Hence, to prove this, we may assume that H vanishes and that $\int |y_i y_j| p(y|\theta) \vee (dy) = 0$, for all $i, j \in \{1, \ldots, n_i\}$ and $\theta \in \theta$. In that case (Uf) $(x,q) = x^* \widetilde{K} x$, since \sum_q , M_q and m_q contain only zeros for all $q \in W$. By the definition of (Uf) (x,q) we have

$$(Uf) (x,q) = \inf \left\{ x'Rx + a'Sa + \beta \right\} (Cx + Ba + y)'K(Cx + Ba + y)p(y,q)v(dy) \right\}$$

a ϵA

and therefore (Uf) $(x,q) \ge 0$ for all $(x,q) \in X \times W$ since R, S and K are nonnegative definite. Hence $x'\widetilde{K}x \ge 0$ for all $x \in X$. It is easy to verify that \widetilde{K} is symmetric. Hence \widetilde{K} is nonnegative definite.

(3) Finally we consider the function $q \rightarrow \widetilde{H}(q)$. Using lemma 1.6 (iii) we have $q \rightarrow m_q(i), q \rightarrow M_q(i,j)$ and $q \rightarrow \Sigma_q(i,j)$ are bounded and measurable. So all terms in 5.7 (iii), except the second one, are bounded and measurable on W. We consider the second term separately. To show that

 $(y,q) \xrightarrow{\rightarrow} T_{y}(q)$ is measurable, it suffices to prove that the set { $(y,q) \mid T_{y}(q)(B) \le c$ } is measurable for $B \in \mathcal{T}$ and $c \in \mathbb{R}$ (cf. lemma 1.5). Hence it suffices to show that $(y,q) \xrightarrow{\rightarrow} T_{y}(q)(B)$ is measurable, $B \in \mathcal{T}$. Note that $(y,q) \xrightarrow{\rightarrow} B p(y|\theta)q(d\theta)$ and $(y,q) \xrightarrow{\rightarrow} p(y,q)$ are measurable (cf. lemma 1.6 (iii)). Hence $(y,q) \xrightarrow{\rightarrow} T_{y}(q)(B)$ is measurable since

Therefore $(y,q) \rightarrow H(T_y(q))$ is also measurable. This proves the measurability of $(y,q) \rightarrow \int H(T_y(q))p(y,q)v(dy)$.

The equation $G_1(K) = K$ is called the *Ricatti-equation*. Now we shall consider the sequence of successive approximations:

$$v_{n}(x,q) := (\overline{v}^{n}0)(x,q), \quad x \in X, q \in W$$

We define $n_1 \times n_1$ -matrices K_n and L_n and a sequence of bounded measurable functions H_n on W, for $n \in \mathbb{N}$:

5.9 (i)
$$K_0(i,j) := L_0(i,j) := 0, i,j \in \{1,...,n_1\}; H_0(q) := 0, q \in W$$
.
(ii) $K_n := G_1(K_{n-1}), L_n := G_2(L_{n-1}, K_{n-1}),$
 $H_n(q) := G_3(q, H_{n-1}, K_{n-1}, L_{n-1}), q \in W, n \in \mathbb{N}^*$

 $(G_1, G_2 \text{ and } G_3 \text{ are defined in 5.7}).$ It is a direct consequence of lemma 5.6 that

5.10
$$v_n(x,q) = x'K_n x + x'L_n m_q + H_n(q), \quad n \in \mathbb{N}$$
.

In lemma 5.7 we prove that K_n and L_n converge elementwise to matrices K^* and L^* respectively. The proof of $K_n \rightarrow K^*$ can also be found in [Kushner (1971), section 9.2.3]. In our proof we use the same arguments. In lemma 5.8 we show the pointwise convergence of H_n as n tends to infinity.

Lemma 5.7

- (i) K_n converges, elementwise, to a nonnegative definite matrix K^* satisfying the Ricatti-equation $(K^* = G_1(K^*))$.
- (ii) L_n converges, elementwise, to a matrix L^* satisfying $L^* = G_2(L^*, K^*)$.

Proof.

Since K_n and L_n do not depend on the measure \vee , to study the limit behaviour we may assume that \vee is concentrated in a point $m \in \mathbb{R}^n$. Hence $m_q = m$ for $q \in W$, and we are dealing with a deterministic system. Let us denote the value function of this system by \vee and the sequence of successive approximations by $\{v_n(x), n \in \mathbb{N}\}$ (note that we omit the dependence on $q \in W$). We first show that this value function \vee is finite. Let $x = x_0$ be the starting state. Note that

$$\mathbf{x}_{n_{i}} = \mathbf{c}^{n_{i}} \mathbf{x}_{0} + \sum_{k=0}^{n_{i}-1} \mathbf{c}^{k_{Ba}}_{n_{i}-1-k} + \sum_{k=0}^{n_{i}-1} \mathbf{c}^{k_{m}} \mathbf{x}_{m}$$

hence

$$x_{n_1} - c^{n_1} x_0 - \sum_{k=0}^{n_1-1} c^{k_m *} = \sum_{k=0}^{n_1-1} c^{k_{Ba}} x_{n_1-1-k}$$

By the controllability assumption 5.4(iv) we can find actions a_0, \dots, a_{n_1-1} such that $x_{n_1} = 0$. So there is, for the deterministic linear system, a strategy π such that $x_{kn_1} = 0$ for $k \in \mathbb{N}^*$, and such that each cycle from $x_{kn_1} = 0$ until $x_{(k+1)n_1} = 0$ passes through the same states and actions $(k \in \mathbb{N}^*)$. Hence the discounted total costs of the system under π is finite. Since the one-step costs are nonnegative, $v_n(x)$ is nondecreasing in n. Note that, by a simple dynamic programming argument, $v_n(x) \leq v(x)$ for all $n \in \mathbb{N}$ and $x \in X$. Hence $v_n(x)$ converges if n tends to infinity. Note that, by 5.7(iii) and the special form of v,

$$v_n(x) = x'K_nx + x'L_nm^* + H_n$$

where H_n is constant on W (therefore we omit the dependence on $q \in W$). Since $v_n(0)$ converges, we find that $\lim_{n \to \infty} H_n$ exists and is finite. Let $m^* = 0$. So we find that $\lim_{n \to \infty} x'K_n x$ exists for all $x \in X$, since K_n does not depend on the value of m^* . It is straightforward to show that this implies that K_n converges elementwise. Consequently, for arbitrary m^* , $x'L_n^m^*$ converges for all $x \in X$. Hence L_n converges elementwise. As K_n is nonnegative definite we have $x'K_n x \ge 0$ for all $x \in X$, hence $x'K^* x \ge 0$ for all $x \in X$. Since $K_n = K_n^*$ we have $K^* = K^*$ and therefore K^* is nonnegative definite. Finally, since $(S + \beta B'K_n B)^{-1}$ converges elementwise to $(S + \beta B'K^*B)^{-1}$ we find $K^* = G_1(K^*)$ and $L^* = G_2(L^*, K^*)$. Lemma 5.8

The sequence of functions H_n , defined in 5.9, converges pointwise to a bounded measurable function H^* (on W), such that $H^*(\cdot) = G_3(\cdot, H^*, K^*, L^*)$.

Proof.

Let

5.11 (a)
$$b_n(q) := -\frac{1}{2}\beta^2 m'_q (2K_n B + L'_n B) (S + \beta B'K_n B)^{-1} (2B'K_n + B'L_n)m_q + \beta$$

+ β trace $(K_n \Sigma_q) + \beta$ trace $(L_n M_q)$.

It follows from lemma 5.7 that $b_n(q)$ converges if n tends to infinity. Denote:

5.11 (b) $b(q) := \lim_{n \to \infty} b_n(q), \quad q \in W$.

By definition 5.7 we have $H_{m+1}(q) = b_m(q) + \beta \int H_m(T_y(q))p(y,q)v(dy)$, $m \in \mathbb{N}$. Note that $\int H_m(T_y(q))p(y,q)v(dy) = \mathbb{E}_q[H_m(Q_1)]$. Therefore, by th. 5.1 and the measurability of $(y,q) \stackrel{\sim}{\to} H_m(T_y(q))$ (cf. the proof of lemma 5.6):

(*)
$$H_{m+1}(Q_k) = b_m(Q_k) + \beta E_q [H_m(Q_{k+1}) | Y_1, \dots, Y_k], \quad k \in \mathbb{N}^*, m \in \mathbb{N}.$$

Hence we have

And by iteration, using (*) we find:

(**)
$$H_{n+1}(q) = \sum_{\ell=0}^{n} \beta^{\ell} E_{q}[b_{n-\ell}(Q_{\ell})] + \beta^{n+1} E_{q}[H_{0}(Q_{n+1})].$$

The last term vanishes since, by definition, $H_0 = 0$. Since K_n and L_n are bounded in n (elementwise; see the proof of lemma 5.7), and since $q \neq m_q$, $q \neq M_q$ and $q \neq \sum_q$ are bounded functions (elementwise), we have (cf. 5.11) the boundedness of $(n,q) \neq b_n(q)$. Hence, for all $\varepsilon > 0$ there is a $N_{\varepsilon} \in \mathbb{N}$ such that:

$$\sum_{\ell=N_{\varepsilon}}^{\infty} \beta^{\ell} \mathbf{E}_{\mathbf{q}}[\mathbf{b}_{n-\ell}(\mathbf{Q}_{\ell})] \leq \varepsilon.$$

By the dominated convergence theorem, for fixed ℓ we now have:
$$\lim_{n \to \infty} \mathbf{E}_{q} [b_{n-\ell}(Q_{\ell})] = \mathbf{E}_{q} [b(Q_{\ell})] .$$

Hence, using (**) we find:

5.12
$$H^{*}(\mathbf{q}) := \lim_{n \to \infty} H_{n}(\mathbf{q}) = \sum_{k=0}^{\infty} \beta^{k} \mathbf{E}_{\mathbf{q}}[b(\mathbf{Q}_{k})].$$

Since b is bounded, H^* is also bounded. The measurability of H^* is immediate. Finally, note that 5.12 and the Markov property of the process $\{Q_n, n \in \mathbb{N}\}$ (cf. th. 5.1) imply:

$$H^{*}(q) = b(q) + \beta \sum_{k=0}^{\infty} \beta^{k} \mathbf{E}_{q} [b(\mathbf{Q}_{k+1})] = b(q) + \beta \mathbf{E}_{q} [\sum_{k=0}^{\infty} \beta^{k} \mathbf{E}_{q} [b(\mathbf{Q}_{k+1})|\mathbf{Q}_{1}]] =$$
$$= b(q) + \beta \mathbf{E}_{q} [\sum_{k=0}^{\infty} \beta^{k} \mathbf{E}_{\mathbf{Q}_{1}} [b(\mathbf{Q}_{k})]] = b(q) + \beta \mathbf{E}_{q} [h(\mathbf{Q}_{1})].$$
$$= G(q, H^{*}, K^{*}, L^{*}).$$

Hence $H^{*}(q) = G_{3}(q, H^{*}, K^{*}, L^{*})$.

The next theorem is one of the main results of this section. It is an immediate consequence of the foregoing lemmas and a well-known argument for *ne*gative dynamic programming (cf. [Strauch (1966)]).

Theorem 5.9

(i) The value function satisfies

$$\mathbf{v}(\mathbf{x},\mathbf{q}) = \mathbf{x}^{\dagger}\mathbf{K}^{\dagger}\mathbf{x} + \mathbf{x}^{\dagger}\mathbf{L}^{\dagger}\mathbf{m}_{\mathbf{q}} + \mathbf{H}^{\dagger}(\mathbf{q}) \quad .$$

(ii) In state (x,q) the optimal strategy chooses the action:

$$a(\mathbf{x},\mathbf{q}) = -\beta(\mathbf{S} + \beta \mathbf{B'K^*B})^{-1}\mathbf{B'K^*Cx} - \beta(\mathbf{S} + \beta \mathbf{B'K^*B})^{-1}(\mathbf{B'K^*} + \frac{1}{2}\mathbf{B'L^*})\mathbf{m}_{\mathbf{q}}$$

(where K^* and L^* are defined in lemma 5.7 and $H^*(\cdot)$ in 5.12).

Proof.

From the lemmas 5.6, 5.7 and 5.8 it follows that

$$\mathbf{v}_{\infty}(\mathbf{x},\mathbf{q}) := \lim_{n \to \infty} \mathbf{v}_{n}(\mathbf{x},\mathbf{q}) = \mathbf{x}'\mathbf{K}^{*}\mathbf{x} + \mathbf{x}'\mathbf{L}^{*}\mathbf{m}_{\mathbf{q}} + \mathbf{H}^{*}(\mathbf{q}), \mathbf{x} \in \mathbf{X}, \mathbf{q} \in \mathbf{W}$$

and also that, for $x \in X$, $q \in W$:

(*)
$$v_{\infty}(x,q) = (Uv_{\infty})(x,q) = c(x,a(x,q)) + \beta E_{x,q}^{\pi} [v_{\infty}(x_{1},Q_{1})]$$

where π^* is the strategy that chooses in (x,q) action a(x,q) (defined above). Since the process { (X_n,Q_n), n $\in \mathbb{N}$ } is a Markov chain under π^* we find by iteration of (*):

(**)
$$\mathbf{v}_{\infty}(\mathbf{x},\mathbf{q}) = \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \left[\sum_{n=0}^{N-1} \beta^{n} \mathbf{c} (\mathbf{x}_{n}, \mathbf{a} (\mathbf{x}_{n}, \mathbf{Q}_{n})) \right] + \beta^{N} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \left[\mathbf{v}_{\infty} (\mathbf{x}_{N}, \mathbf{Q}_{N}) \right] .$$

Note that according to a simple dynamic programming argument $v_n(x,q) \le v(x,q)$ for all $n \in N$, since c is a nonnegative function. Hence

$$\mathbf{v}(\mathbf{x},\mathbf{q}) \geq \mathbf{v}_{\infty}(\mathbf{x},\mathbf{q}) \geq \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi^{\star}} \left[\sum_{n=0}^{\infty} \beta^{n} \mathbf{c}(\mathbf{x}_{n},\mathbf{a}(\mathbf{x}_{n},\mathbf{Q}_{n}))\right] \geq \mathbf{v}(\mathbf{x},\mathbf{q})$$

Here the second inequality follows from (**) since $v_{\infty}(x,q) \ge 0$, $(x,q) \in X \times W$.

The following theorem provides a bound for the extra costs we incur due to lack of information about the parameter value $\theta \in \Theta$.

Theorem 5.10

$$0 \leq \mathbf{v}(\mathbf{x},\mathbf{q}) - \int \mathbf{v}(\mathbf{x},\theta)\mathbf{q}(\mathrm{d}\theta) \leq \frac{1}{1-\beta} \{\mathbf{b}(\mathbf{q}) - \int \mathbf{b}(\theta)\mathbf{q}(\mathrm{d}\theta)\}, \ (\mathbf{x},\mathbf{q}) \in \mathbf{X} \times \mathbf{W}$$

(where b is defined in 5.11 (b)).

Proof.

By th. 3.16 we have $v(x,q) \ge \int v(x,\theta)q(d\theta)$. (Remember that we are minimizing here.) Note that, by th. 5.9:

$$\mathbf{v}(\mathbf{x},\boldsymbol{\theta}) = \mathbf{x}'\mathbf{K}^{*}\mathbf{x} + \mathbf{x}'\mathbf{L}^{*}\mathbf{m}_{\boldsymbol{\theta}} + \sum_{n=0}^{\infty} \beta^{n}\mathbf{b}(\boldsymbol{\theta})$$

since all posterior distributions are concentrated in θ , if the prior distribution is concentrated in $\theta \in \theta$. Since $\int q(d\theta) \mathbf{x' L}^{\star} \mathbf{m}_{\theta} = \mathbf{x' L}^{\star} \mathbf{m}_{q}$ we have

(*)
$$\mathbf{v}(\mathbf{x},\mathbf{q}) - \int \mathbf{v}(\mathbf{x},\theta)\mathbf{q}(d\theta) = \sum_{n=0}^{\infty} \beta^{n} \{\mathbf{E}_{\mathbf{q}}[\mathbf{b}(\mathbf{Q}_{n})] - \int \mathbf{b}(\theta)\mathbf{q}(d\theta)\}$$
.

Note that b(q), satisfies 5.11 (a), with K_n and L_n replaced by K^* and L^* , respectively. Note that the matrix E, defined by

$$\mathbf{E} := (2K^{*}B + L^{*}B) (S + \beta B'K^{*}B)^{-1} (2B'K^{*} + B'L^{*}) ,$$

107

is positive definite, since $(S + \beta B'K^*B)$ is. Therefore E can be written as $E = J^{\Lambda}J$, where J is an orthogonal matrix and Λ is a diagonal matrix with positive entries $\lambda_1, \dots, \lambda_{n_1}$ on the diagonal. Hence

$$\mathbf{m}_{\mathbf{q}}^{\mathsf{'Em}} = \sum_{i=1}^{n_{1}} \lambda_{i} \{\sum_{n=1}^{n_{1}} J_{ij} \mathbf{m}_{\mathbf{q}}^{(j)}\}^{2}$$

And, by Schwarz's inequality, we find

$$\mathbb{E}_{\mathbf{q}}[\mathbb{E}_{\mathbf{Q}_{\mathbf{n}}}[\mathbb{E}_{\mathbf{Q}_{\mathbf{n}}}]^{2} \cong \sum_{i=1}^{n_{1}} \lambda_{i} \{\mathbb{E}_{\mathbf{q}}[\sum_{j=1}^{n_{1}} J_{ij}\mathbb{Q}_{\mathbf{n}}(j)]\}^{2}.$$

Since $m_{Q_n}(j) = \int Q_n(d\theta) m_{\theta}(j)$ we have

$$\mathbb{E}_{q}\begin{bmatrix}\sum_{j=1}^{n_{1}} J_{ij}m_{Q_{n}}(j)\end{bmatrix} = \sum_{j=1}^{n_{1}} J_{ij}m_{q}(j) .$$

n.

Hence we have

$$\mathbf{E}_{\mathbf{q}}[\mathbf{m}_{\mathbf{Q}_{\mathbf{n}}}^{\mathbf{E}\mathbf{m}}] \geq \sum_{i=1}^{n_{1}} \lambda_{i} \{\sum_{j=1}^{n_{1}} J_{ij}\mathbf{m}_{\mathbf{q}}(j)\}^{2} = \mathbf{m}_{\mathbf{q}}^{*}\mathbf{E}\mathbf{m}_{\mathbf{q}}.$$

Note that

and

$$\operatorname{trace} (\mathbf{L}^{*}\mathbf{M}_{\mathbf{q}}) = \sum_{\substack{i=1 \ j=1}}^{n_{1}} \sum_{\substack{j=1 \ j=1}}^{n_{1}} \mathbf{L}^{*}(i,j) \int \mathbf{m}_{\theta}(i)\mathbf{m}_{\theta}(j)\mathbf{q}(d\theta)$$
$$\operatorname{trace} (\mathbf{K}^{*}\mathbf{\Sigma}_{\mathbf{q}}) = \sum_{\substack{i=1 \ j=1}}^{n_{1}} \sum_{\substack{j=1 \ j=1}}^{n_{1}} \mathbf{K}^{*}(i,j) \int \{\int \mathbf{y}_{\underline{i}}\mathbf{y}_{\underline{j}}\mathbf{p}(\mathbf{y}|\theta) \mathbf{v}(d\mathbf{y})\}\mathbf{q}(d\theta)$$

Hence we find:

$$\mathbf{E}_{\mathbf{q}}[\operatorname{trace}(\mathbf{L}^{\star}\mathbf{M}_{\mathcal{Q}_{n}})] = \operatorname{trace}(\mathbf{L}^{\star}\mathbf{M}_{q}) \text{ and } \mathbf{E}_{\mathbf{q}}[\operatorname{trace}(\mathbf{K}^{\star}\boldsymbol{\Sigma}_{\mathcal{Q}_{n}})] = \operatorname{trace}(\mathbf{K}^{\star}\boldsymbol{\Sigma}_{q}).$$

So we have

$$\mathbf{E}_{\mathbf{q}}[\mathbf{b}(\mathbf{Q}_{n})] \leq \mathbf{b}(\mathbf{q})$$

This proves the theorem.

Remarks.

The linear system with (known) transition law given by: (i) $p(y,q)v(dy), D \in X$ $\widetilde{P}(D|\mathbf{x},a) =$ $\{Cx+Ba+y\in D\}$

and the same cost structure as in model 4 has the value function $\widetilde{v}_q({}^{\ast})$ defined by:

$$\widetilde{v}_{q}(x) := x'K'x + x'L'm_{q} + \frac{b(q)}{1-\beta}$$
.

Hence, by th. 5.10, we have

$$v(x,\theta)q(d\theta) \leq v(x,q) \leq \widetilde{v}_{q}(x)$$
.

(ii) The optimal strategy we found in th. 5.9 is a Bayesian equivalent rule (cf. 4.4), since the action in state $(x,q) \in X \times W$ is the minimizer of the function (cf. the proof of lemma 5.6)

$$\mathbf{a} \rightarrow \int \mathbf{q}(\mathbf{d}\theta) \{\mathbf{a}^{\prime} (\mathbf{S} + \beta \mathbf{B}^{\prime} \mathbf{K}^{*} \mathbf{B}) \mathbf{a} + (2\beta \mathbf{x}^{\prime} \mathbf{C}^{\prime} \mathbf{K}^{*} \mathbf{B} + 2\beta \mathbf{m}_{\theta}^{\prime} \mathbf{K}^{*} \mathbf{B} + \beta \mathbf{m}_{\theta}^{\prime} \mathbf{L}^{*} \mathbf{B}) \mathbf{a} \}$$

(iii) Since $m_q(i) = \int m_{\theta}(i)q(d\theta)$, $m_q(i)$ is the Bayes estimate of $m_{\theta}(i)$ when q is the prior distribution. Hence the optimal strategy we found in th. 5.9 can also be formulated as: at each moment use the Bayes estimate for m_{θ} in the formula for the optimal action instead of m_{θ} , which should be used if θ were known. In the linear system with known transition law, i.e. with $\theta = \{\theta\}$, it turns out that the optimal strategy is the same as if ν is concentrated at m_{θ} . In that situation we are dealing with a deterministic system. This property, which we used in the proof of lemma 5.7, is called the *certainty equivalent principle* (cf. [Bertsekas (1976)]). We showed that in each state (x,q) we may act as if ν is concentrated on m_q .

5.3 A simple inventory control model

In this section we consider an inventory control model which is closely related to the model described in example 5.2: the main difference is that $D(x) \neq A$ for all $x \in X$. Further we shall specify here the functions b, c and G of the example. The model we shall deal with is extensively studied by several authors: [Scarf (1959)], [Iglehart (1964)], [Rieder (1972)], [Zacks and Fennel (1973)] and [Waldmann (1976)]. Except for Zacks and Fennel all these authors prove structural properties of the optimal strategy under various conditions. Only Zacks and Fennel considered an easy-to-handle suboptimal strategy and they studied its behaviour using Monte Carlo methods. We also study a suboptimal strategy, namely a Bayesian equivalent rule, and we give bounds on the difference of its Bayesian total discounted return and the optimal value. Further we consider conditions under which this strategy is optimal. We start with a sketch of the model. Model 5A: inventory control model

5.13 (1) $X := A := (-\infty, M], M > 0$ is called the *capacity*.

(ii) D(x) := [x,M], a is the *inventory* after ordering.

- (iii) $P(\{a-y\} \mid x,a,y) = 1$, y represents the demand.
- (iv) $Y := \{y \in \mathbb{R} \mid y \ge 0\}.$
- (v) $c(x,a) := hx^{+} + px^{-} + k(a x)$ where h is the holding cost, p the penalty cost for shortage and k the production cost; h,p,k > 0 and k < β (k + p).

(It is easy to verify that, if $k \ge \beta(k+p)$ never ordering is optimal.)

We shall compare this model with model 5B.

Model 5B.

5.14 D(x) := [0,M].

Further all specifications as in 5.13.

It is easy to verify that all assumptions of example 5.2 are fulfilled. The optimal action in (x,q) for model 5B is the minimizer in [0,M] of:

5.15
$$a \rightarrow [ka + \beta] \{h(a - y)^{+} + p(a - y)^{-} - k(a - y)\}p(y,q)v(dy)\}].$$

We shall determine the minimizer. The term between braces equals

5.16 {k-
$$\beta(p+k)$$
} a + m_q $\beta(p+k) + \beta(h+p) \int (a-y)p(y,q)v(dy) [0,a]$

where

5.17
$$m_q := \int yp(y,q)v(dy)$$

It is easy to verify that

$$f(a) := \int (a-y)p(y,q)v(dy) = \int du \{ \int p(y,q)v(dy) \} .$$
[0,a] 0 [0,u]

Hence the function f is continuous and $f(\frac{1}{2}(a+b)) \leq \frac{1}{2} \{f(a) + f(b)\}$. Therefore f is convex. So 5.16 has a minimum in

5.18 (a)
$$\widetilde{s}(q) := \inf\{a \in \mathbb{R} \mid \int p(y,q) \vee (dy) \ge \frac{p - \frac{1 - \beta}{\beta} k}{p + h} \}$$
.

According to 5.13 (v) there always is minimum of 5.16. Note that $q \neq \widetilde{s}(q)$, $q \in W$ is measurable.

Consequently

5.18 (b) $s(q) := \min\{M, \tilde{s}(q)\}, q \in W$

is a minimizer of 5.16 in the set [0,M].

It is well known that for the inventory control model 5A with known parameter value θ , the strategy:

"order up to level $s(\theta)$ each period or wait if the inventory is larger than $s(\theta)$ " is optimal (cf. [Iglehart (1964)]).

To define a Bayesian equivalent rule for model 5A, define the function F by:

$$F(x,\theta,a) := ka + \beta \int \{h(a-y)^{+} + p(a-y)^{-} - k(a-y)\}p(y|\theta)v(dy) .$$

Hence, in state $(x,q) \in X \times W$ this Bayesian equivalent rule chooses a minimizer of $a \rightarrow \int q(d\theta)F(x,\theta,a)$ on the set [x,M]:

5.18 (c) the Bayesian equivalent rule chooses the action a(x,q), $a(x,q) = \max\{x,s(q)\}$ in state $(x,q) \in X \times W$.

We proceed with a definition:

5.19 Let v be the value function of model 5A, w the value function of model 5B and let \hat{v} be the Bayesian discounted total costs under the Bayesian equivalent rule defined in 5.18 (c).

Note that the Bayesian equivalent rule 5.18 (c) defines a Bayesian stationary strategy for model 5A and also for model 5B. Note also that the Bayesian discounted total costs for both models is the same under this strategy, namely $\hat{\mathbf{v}}(\mathbf{x},\mathbf{q})$, if $(\mathbf{x},\mathbf{q}) \in \mathbf{X} \times \mathbf{W}$ is the starting state.

There is an optimal strategy for model 5A of the form "choose the action $max{x,t(q)}$ in state (x,q)", where $t: W \rightarrow A$ is a measurable function such that: 5.20 t(q) is a minimizer of

 $a \rightarrow ka + \beta \int v(a-y,T_y(q))p(y,q)v(dy) \}$

on the set [0,M].

This is proved in [Rieder (1972), th. 7.2 and th. 7.3] under the additional assumption that $\theta \rightarrow m_{\theta}$ is bounded. For practical purposes this result is only interesting if the value function v is known. Lemma 5.11 shows that

 $t(q) \leq s(q)$ for $q \in W$, which is intuitively clear, since if it is not allowed to reduce the inventory, you will order more carefully.

Lemma 5.11

Let t: W \rightarrow A be measurable such that t(q) satisfies 5.20 for $q \in W$. Then $s(q) \ge t(q)$, for $q \in W$.

Proof.

Define

 $f(x,q) := v(x,q) - {hx^+ + px^- - kx}, (x,q) \in X \times W$.

From the optimality equation, v = Uv, for model 5A it follows that

(*)
$$f(x,q) = \inf_{\substack{x \le a \le M}} \{ka + \beta \int v(a - y, T_y(q)) p(y,q) v(dy)\}$$

Hence $x \neq f(x,q)$ is nondecreasing for all $q \in W$. By (*) we have:

(**)
$$f(x,q) = \inf_{\substack{x \le a \le M}} [ka + \beta] \{h(a - y)^{+} + p(a - y)^{-} - k(a - y)\} p(y,q) \vee (dy) + \beta \int f(a - y, T_{y}(q)) p(y,q) \vee (dy)].$$

Remember that the function (cf. 5.15)

$$a \rightarrow ka + \beta \int \{h(a-y)^{+} + p(a-y)^{-} - k(a-y)\}p(y,q) \vee (dy)$$

is convex and attains a minimum in [0,M] (cf. 5.18 (a)). The last term of (**) is a nondecreasing function of a, since $x \rightarrow f(x,q)$ is nondecreasing, for all $q \in W$. Hence a minimizer t(q) of

$$a \rightarrow ka + \beta \int v(a - y, T_y(q)) p(y, q) v(dy)$$

in the set a ϵ [0,M] must satisfy t(q) \leq s(q), q ϵ W.

In th. 5.12 we give bounds for the difference $\hat{\mathbf{v}} - \mathbf{w}$ (cf. 5.19). This difference is an upper bound for $\hat{\mathbf{v}} - \mathbf{v}$, the loss due to controlling the system with the Bayesian equivalent rule. The bounds are derived by comparing two strategies for model 5B. We compare the optimal strategy for this model, where $\widetilde{\mathbf{A}}_n = \mathbf{s}(\mathbf{Q}_n)$ for all $n \in \mathbf{N}$, with the strategy where $\mathbf{A}_n = \max\{\mathbf{x}, \mathbf{s}(\mathbf{Q}_n)\}$, the Bayesian equivalent rule defined in 5.18 (c). Note that the production costs at time n for these two strategies, differ if $\mathbf{s}(\mathbf{Q}_{n-1}) - \mathbf{Y}_n - \mathbf{s}(\mathbf{Q}_n) > 0$.

D

The functions v, w and \hat{v} defined in 5.19, satisfy the following inequalities:

(i)
$$w(x,q) \leq v(x,q) \leq \hat{v}(x,q), (x,q) \in X \times W$$
.

(ii)
$$\hat{v}(\mathbf{x},\mathbf{q}) - w(\mathbf{x},\mathbf{q}) \leq (\frac{\beta}{1-\beta}h+k)\{(\mathbf{x}-\mathbf{s}(\mathbf{q}))^{+} + \sum_{n=1}^{\infty} \beta^{n} \mathbf{E}_{\mathbf{q}} [\{\mathbf{s}(\mathbf{Q}_{n-1}) - \mathbf{Y}_{n} - \mathbf{s}(\mathbf{Q}_{n})\}^{+}]\}$$

Proof.

- (i) Note that for model 5B the lower bound for the action space is not essential, since $\tilde{s}(q) \ge 0$ (see 5.18 (a)). Hence it is obvious that $w(x,q) \le v(x,q)$. The right-hand side of the inequality is trivial.
- (ii) Let X_n denote the inventory at time n when the action at time n $A_n := \max\{s(Q_n), X_n\}$ is used, and \widetilde{X}_n the inventory if $\widetilde{A}_n := s(Q_n)$ is used. Further let $X_0 = \widetilde{X}_0 = x$. Since $X_{n+1} = A_n - Y_{n+1}$ and $\widetilde{X}_{n+1} = \widetilde{A}_n - Y_{n+1}$ it is easily verified that: $\widetilde{X}_n \leq X_n$, $n \in \mathbb{N}$. Let $S_n := s(Q_n)$, $n \in \mathbb{N}$, and consider the difference in immediate costs at time n:

l

$$h(x_{n}^{T} - \widetilde{x}_{n}^{T}) + p(x_{n}^{T} - \widetilde{x}_{n}^{T}) - k(x_{n} - \widetilde{x}_{n}) + k(\max\{x_{n}, s_{n}\} - s_{n}) \le$$

$$\leq h(x_{n} - \widetilde{x}_{n}) + k(\max\{x_{n} - s_{n}, 0\} - x_{n} + \widetilde{x}_{n}), n \in \mathbb{N},$$

since $X_n \leq \tilde{X}_n$ for $n \in \mathbb{N}$. We consider the term with coefficient k first. We establish:

**)
$$k(\max\{x_n - s_n, 0\} - x_n + \tilde{x}_n) \le k(s_{n-1} - s_n - Y_n)^+, n \in \mathbb{N}^*$$

To prove (**), let $x_n > s_n$. Then (**) holds, since $\tilde{X}_n = s_{n-1} - Y_n$. And if $x_n \le s_n$, we get max $\{x_n - s_n, 0\} - x_n + \tilde{X}_n \le 0$, and so (**) holds. For n = 0 we have

$$k(\max\{x_0 - S_0, 0\} - x_0 + \widetilde{x}_0) = k(x - s(q))^+$$
.

Hence if h = 0:

$$\hat{v}(x,q) - w(x,q) \le k\{(x-s(q))^{+} + \sum_{n=1}^{\infty} \beta^{n} \mathbf{E}_{q} [(s_{n-1} - s_{n} - y_{n})^{+}]\}$$

For h > 0 we consider $h(X_n - \widetilde{X}_n)$, $n \in \mathbb{N}$. Note that $X_0 = \widetilde{X}_0 = x$ and

$$X_1 - \tilde{X}_1 = \max\{x, s(q)\} - Y_1 - s(q) + Y_1 = \{x - s(q)\}^+$$
.

We shall prove

$$(***) \qquad x_{n} - \tilde{x}_{n} \le \{x - s(q)\}^{+} + \sum_{k=1}^{n-1} \{s_{k-1} - y_{k} - s_{k}\}^{+}, n \in \mathbb{N}^{*}$$

(an empty sum vanishes).

We already verified (***) for n = 1. Assume it holds for n. Consider:

$$x_{n+1} - \tilde{x}_{n+1} = \max\{x_n, s_n\} - y_{n+1} - s_n + y_{n+1} = (x_n - s_n)^+$$

By the induction hypothesis:

$$\begin{split} x_{n} &\leq \widetilde{x}_{n} + \{x - s(q)\}^{+} + \sum_{k=1}^{n-1} \{s_{k-1} - y_{k} - s_{k}\}^{+} = \\ &= \{x - s(q)\}^{+} + \sum_{k=1}^{n-1} \{s_{k-1} - y_{k} - s_{k}\}^{+} + s_{n-1} - y_{n} \end{split}$$

Hence

$$(x_n - s_n)^+ \le \{x - s(q)\}^+ + \sum_{k=1}^n \{s_{k-1} - y_k - s_k\}^+,$$

which proves (***),

Now we add the upperbounds for the differences in holding costs:

$$h \sum_{n=1}^{\infty} \beta^{n} (x_{n} - \widetilde{x}_{n}) \le h \frac{\beta}{1 - \beta} \{x - s(q)\}^{+} + h \sum_{n=1}^{\infty} \beta^{n} \sum_{k=1}^{n-1} \{s_{k-1} - y_{k} - s_{k}\}^{+} =$$

$$= \frac{h\beta}{1 - \beta} \{x - s(q)\}^{+} + h \sum_{k=1}^{\infty} \{s_{k-1} - y_{k} - s_{k}\}^{+} \sum_{n=k+1}^{\infty} \beta^{n} =$$

$$= \frac{h\beta}{1 - \beta} \{x - s(q)\} + \frac{h\beta}{1 - \beta} \sum_{k=1}^{\infty} \beta^{k} \{s_{k-1} - y_{k} - s_{k}\}^{+}$$

which accounts for the term with h in the right-hand side of (ii). $\hfill \square$

Corollary 5.13 If for all $q \in W$: 5.21 $\int_{\{y \mid s(q) - y \leq s(T_y(q))\}} p(y,q) \lor (dy) = 1$

then, for $x \le s(q)$, we have v(x,q) = w(x,q) and therefore the Bayesian equivalent rule (defined in 5.18 (c)) is optimal.

```
Example 5.3
```

We define (cf. 5.18 (a)):

$$s_{\min} := \inf_{\substack{\theta \in \Theta}} \widetilde{s}(\theta), \ s_{\max} := \sup_{\theta \in \overline{\Theta}} \widetilde{s}(\theta) \text{ and } \delta := s_{\max} - s_{\min} \cdot$$

Since

$$\int_{[0,s_{max}]} p(y|\theta)v(dy) \ge \frac{p - \frac{1 - \beta}{\beta}h}{p + h} \ge \int_{[0,s_{min}]} p(y|\theta)v(dy)$$

for all $\theta \in \Theta$, we have

$$s_{\min} \leq \widetilde{s}(q) \leq s_{\max}$$
 for all $q \in W$.

Note that $s(q) = \min\{\widetilde{s}(q), M\}$ for $q \in W$. Further note that

5.22
$$\{\min(\widetilde{s}(Q_{n-1}), M) - \min(\widetilde{s}(Q_n), M) - Y_n\}^+ \le \{\widetilde{s}(Q_{n-1}) - \widetilde{s}(Q_n) - Y_n\}^+.$$

Hence

$$\mathbf{E}_{\mathbf{q}}[\{\mathbf{s}(\mathbf{Q}_{n-1}) - \mathbf{s}(\mathbf{Q}_{n}) - \mathbf{Y}_{n}\}^{\dagger}] \leq \mathbf{E}_{\mathbf{q}}[\qquad p(\mathbf{y}, \mathbf{Q}_{n-1}) \lor (\mathbf{d}\mathbf{y})] = \\
 \begin{bmatrix} \mathbf{0}, \delta \end{bmatrix}$$

$$= \int_{[\mathbf{0}, \delta]} \mathbf{p}(\mathbf{y}, \mathbf{q}) \lor (\mathbf{d}\mathbf{y}) \quad .$$

Therefore we have by th. 5.12

$$\hat{\mathbf{v}}(\mathbf{x},\mathbf{q}) - \mathbf{w}(\mathbf{x},\mathbf{q}) \leq \left\{\frac{\beta}{1-\beta}\mathbf{h} + \mathbf{k}\right\}\left\{\left(\mathbf{x} - \mathbf{s}(\mathbf{q})\right)^{+} + \frac{\beta}{1-\beta}\int_{[0,\delta]} \mathbf{p}(\mathbf{y},\mathbf{q}) \vee (\mathbf{d}\mathbf{y})\right\}.$$

Hence, if $x \leq s(q)$ and $\int p(y|\theta)v(dy) = 0$ for all $\theta \in \theta$, then the Bayes-[0, δ] ian equivalent rule is optimal.

Remarks.

- (i) The statement of corollary 5.13 is not new. In [Veinott (1965), section 6] a similar condition is considered for a multi-product inventory model with dependent demand, to prove an analogous statement. In [Rieder (1972)] Veinott's result has been translated to the Bayesian inventory model. However, the inequality of th. 5.12 (ii) seems to be new.
- (ii) For the problem with known parameter, i.e. when q ∈ W is degenerate at θ ∈ θ, all posterior distributions are degenerate at θ and therefore 5.21 holds. So we actually proved the optimality of the rule:
 "order up to level s(θ) at each stage" for this situation.
- (iii) Condition for lemma 5.13 can be weakened by requiring 5.21 only for all possible posterior distributions of a given $q \in W$.
- (iv) In [van Hee (1976)] a different proof of th. 5.12 is given.

We conclude this section with an extensive study of the behaviour of the Bayesian equivalent rule (5.18 (c)) for the inventory model with exponentially distributed demand, where we assume the parameter of the demand distribution to have a gamma prior distribution. We shall compute the bound given in th. 5.12 (ii).

We also consider an upper bound for the relative error if we use the Bayesian equivalent rule (5.18 (c)) instead of an optimal rule. This *relative error* is defined by:

5.23
$$\{\hat{v}(0,q) - v(0,q)\}/v(0,q)$$
.

Remember that model 5B satisfies all assumptions of example 5.2, and note that we are minimizing now.

Hence (cf. example 5.2) we have

$$w(0,q) \geq \frac{1}{1-\beta} \int q(d\theta) \widetilde{e}(\theta)$$

where

5.24
$$\widetilde{e}(\theta) := \min \left[ka + \beta \int \{ h(a-y)^{\dagger} + p(a-y)^{\dagger} - k(a-y) \} p(y|\theta) v(dy) \right].$$

Note that $v(0,q) \ge w(0,q)$. Therefore we have the following upper bound for 5.23 (cf. th. 5.12 (ii)):

116

5.25
$$\Delta(\beta,k,h,p,q) := \frac{\left(\frac{\beta}{1-\beta}h+k\right)\sum_{n=1}^{\infty}\beta^{n}\mathbf{E}_{q}\left[\left\{s\left(Q_{n+1}\right)-Y_{n}-s\left(Q_{n}\right)\right\}^{+}\right]}{\frac{1}{1-\beta}\int q\left(d\theta\right)\widetilde{e}\left(\theta\right)}$$

We first give, in lemma 5.14, conditions guaranteeing that $\lim_{n \to \infty} \widetilde{s}(Q_n) = \widetilde{s}(Z)$ \mathbb{P}_q -a.s. Under these conditions we have, by 5.22

5.26 $\lim_{\beta \neq 1} \Delta(\beta, k, 0, p, q) = 0$

since

$$\lim_{\beta \neq 1} (1 - \beta) \sum_{n=1}^{\infty} \beta^{n} \mathbb{E}_{q} [\{\widetilde{s}(Q_{n-1}) - Y_{n} - \widetilde{s}(Q_{n})\}^{+}] =$$
$$\lim_{n \to \infty} \mathbb{E}_{q} [\{\widetilde{s}(Q_{n-1}) - Y_{n} - \widetilde{s}(Q_{n})\}^{+}] = 0.$$

Hence the relative error (cf. 5.23) tends to zero in this case.

Lemma 5.14

Let for all $\theta \in \Theta$, the function $a \rightarrow \int p(y|\theta)v(dy)$ be continuous and [0,a] (strictly) increasing in a neighbourhood $(\tilde{s}(\theta) - \delta, \tilde{s}(\theta) + \delta)$ for $\delta > 0$ (cf. 5.18 (a). Then

$$\lim_{n \to \infty} \widetilde{s}(Q) = \widetilde{s}(Z) , \mathbb{P} \text{-a.s.} \text{ for all } q \in \mathbb{W} .$$

Proof.

Define for $q \in W$ and $a \in \mathbb{R}$, $F_q(a) := \int p(y,q)v(dy)$. Note that $\begin{bmatrix} 0,a \end{bmatrix}$ $F_q(a) = \int F_{\theta}(a)q(d\theta)$. Since $a \neq F_{\theta}(a)$ is continuous for all $\theta \in \theta$ the function $a \neq F_q(a)$ is continuous for all $q \in W$. According to th. 2.4 we have for each function $\theta \neq F_{\theta}(a)$ a set $\Omega_a \in H$ such that $P_q[\Omega_a] = 1$ and

(*)
$$\lim_{n\to\infty} F_{Q_n}(a) = F_{Z}(a) \text{ on } \Omega_{a}.$$

Let R be the set of rational numbers in IR. Define $\Omega^* := \cap \Omega_a$. Note that $\mathbf{P}_{\mathbf{q}}[\Omega^*] = 1$. Let $\ell := (p - \frac{1-\beta}{\beta}h)(p+h)^{-1}$. Then $\mathbf{F}_{\mathbf{z}}(\widetilde{\mathbf{s}}(\mathbf{Z})) = \ell$. Fix $\omega \in \Omega^*$. Fix $a_1, a_2 \in \mathbb{R}$ such that $a_1 < \widetilde{s}(Z(\omega)) < a_2$ and $a_2 - a_1 < \delta$. Then

$$F_{Z(\omega)} \stackrel{(a_1)}{=} \stackrel{\langle l \rangle \leq l \leq F_{Z(\omega)}}{=} \stackrel{(a_2)}{=}$$

Hence

$$\lim_{n\to\infty} F_{Q_n}(\omega) \stackrel{(a_1) \leq \ell \leq \lim_{n\to\infty} F_{Q_n}(\omega)}{\longrightarrow} \stackrel{(a_2)}{\longrightarrow}$$

Therefore we have, for n sufficiently large

$$a_1 \leq \widetilde{s}(Q_n(\omega)) \leq a_2$$
.

Hence

$$\lim_{n \to \infty} \widetilde{s}(Q_n(\omega)) = \widetilde{s}(Z(\omega)) .$$

Example 5.4 Exponential demand, gamma prior distribution

In this example we consider the inventory control model (model 5A), with an exponential demand distribution and a gamma distribution for the unknown parameter of the demand distribution.

Let $p(y|\theta) := \theta e^{-\theta y}$, $Y := \theta := [0,\infty)$ and let $q = \Gamma(\lambda, N)$, where

$$\Gamma(\lambda, N)(B) := \int_{B} \frac{\theta^{N-1}}{(N-1)!} \lambda^{N} e^{-\lambda \theta} d\theta$$

for a Borel subset B of IR. It is easy to verify that

$$p(y,q) = \frac{N\lambda^N}{(\lambda+y)^{N+1}}$$
 and $\int_0^a p(y,q) dy = 1 - (\frac{\lambda}{\lambda+a})^N$.

Let

5.27 c :=
$$(p+h) \{h + (1-\beta)\beta^{-1}k\}^{-1}$$
.

Then the minimizer a^{*} of 5.15 is a^{*} = $\lambda(c^{1/N} - 1)$. So we have here

5.28
$$\widetilde{s}(q) = \lambda(c^{1/N} - 1)$$
 for $q = \Gamma(\lambda, N)$.

Further we consider the posterior distribution $T_{\ensuremath{\begin{array}{c} y \\ y \end{array}}}(q)$. It is straightforward to verify that

5.29
$$T_{y}(q) = \Gamma(\lambda + y, N + 1) \quad \text{if } q = \Gamma(\lambda, N) .$$

Therefore the posterior distribution after n observations is

5.30
$$Q_n = \Gamma(\lambda + \sum_{i=1}^n Y_i, N+n)$$
.

Hence, using 5.28 we find:

5.31
$$\tilde{s}(Q_n) - \tilde{s}(Q_{n+1}) - Y_{n+1} = \{\lambda + \sum_{i=1}^n Y_i\}\{c^{\frac{1}{N+n}} - c^{\frac{1}{N+n+1}}\} - Y_{n+1}c^{\frac{1}{N+n+1}}$$

۰.

For a fixed $\theta \in \Theta$ we compute, for positive constants a and b:

5.32
$$\mathbb{E}_{\theta}[\{a \sum_{i=1}^{n} \mathbb{Y}_{i} + b - \mathbb{Y}_{n+1}\}^{+}] = \frac{an}{\theta} + b - \frac{1}{\theta} + \frac{e^{-\theta b}}{\theta (a+1)^{n}}$$

(remember here that, given θ , Y_i is exponentially distributed and $\sum_{i=1}^{n} Y_i$ is $\Gamma(\theta, n)$ -distributed).

Now we integrate both terms in 5.32 over θ , with respect to the $\Gamma\left(\lambda,N\right)$ -distribution:

5.33
$$\mathbf{E}_{\mathbf{q}}[\{\mathbf{a} \sum_{i=1}^{n} \mathbf{Y}_{i} + \mathbf{b} - \mathbf{Y}_{n+1}\}^{+}] = \mathbf{a}_{1} \frac{\lambda}{N-1} + \mathbf{b} - \frac{\lambda}{N-1} + \frac{1}{(a+1)^{n}} \frac{1}{N-1} \frac{\lambda^{N}}{(\lambda+b)^{N-1}}$$

Finally we substitute for a and b the appropriate values (cf. 5.31). Hence

5.34
$$E_{q}^{L(s(Q_{n}) - s(Q_{n+1}) - Y_{n+1})} = \frac{1}{N - 1} \left\{ (n + N - 1)c^{\frac{1}{N+n}} - (n + N)c^{\frac{1}{N+n+1}} + c^{\frac{2(n + N) - 1}{(n+N)(n+N+1)}} \right\}.$$

According to 5.22 we have

5.35
$$\mathbb{E}_{q}[\{s(Q_{n}) - s(Q_{n+1}) - Y_{n+1}\}^{+}] \le \mathbb{E}_{q}[\{\tilde{s}(Q_{n}) - \tilde{s}(Q_{n+1}) - Y_{n+1}\}^{+}]$$

with equality if $M = \infty$ (M is the capacity cf. 5.13).

Note that the minimum in 5.24 is nonincreasing if M tends to infinity. Hence $\Delta(\beta,k,h,p,q)$ is nondecreasing if M tends to infinity. Therefore we shall assume M = ∞ .

It is easy to verify that:

5.36
$$\widetilde{\Theta}(\theta) = \{k(1-\beta) - \beta_p\} \frac{\log c}{\theta} + \frac{\beta(p+k)}{\theta} + \frac{\beta(p+h)}{\theta} \{\frac{1}{c} + \log c - 1\}$$

Integration with respect to $q = \Gamma(\lambda, N)$ yields:

5.37
$$\int \widetilde{e}(\theta)q(d\theta) = \frac{\lambda}{N-1} \{k(1-\beta)\log c + \frac{\beta(p+h)}{c} + \beta h(\log c - 1) + \beta k\}.$$

Finally $\Delta(\beta, k, h, p, q)$ is determined by 5.34 and 5.36.

In the table below we display $\Delta(\beta,k,h,p,\Gamma(1,N))$ for various parameter values. We also display the upperbound of th. 5.12 (ii):

5.38
$$B(\beta,k,h,p,q) := \{ \frac{\beta}{1-\beta} h + k \} \sum_{n=1}^{\infty} \beta^{n} E_{q} [\{ s(Q_{n-1}) - Y_{n} - s(Q_{n}) \}^{+}],$$
$$q = \Gamma(1,N) .$$

β	k	h	p	N	с	$B(\beta,k,h,p,f(1,N))$	$\Delta(\beta,k,h,p,\Gamma(1,N))$ in %
0.90	10	0	. 2	5	1.8	2.22	8
0.90	10	0	2	15	1.8	0.33	4
0.90	10	0	10	5	9	10.39	34
0.90	10	2	10	5	3.8	15.89	46
0.90	10	2	10	15	3.8	2.21	22
0.95	10	0	1	5	1.9	3.79	7
0.95	10	2	1	5	1.2	4.61	9
0.95	10	2	5	15	2.8	4.76	26
0.95	100	2	10	5	1.6	40.29	8
0.95	100	2	10	15	1.6	6.56	4
0.99	10	0	1	5	9.9	31.85	12
0.99	10	0	10	15	99.0	12.54	16
0.99	10	2	5	15	3.3	60.28	67
0.99	100	1	2	5	1.4	95.22	4
0.99	100	1	5	5	2.9	274.41	10
0.999	10	0	1	5	99.9	130.83	5
0.999	10	0	2	5	199.8	157.38	6
0.999	10	0	5	5	499.5	196.55	7
					1		1

Remember that $M = \infty$ and $\lambda = 1$ in the table.

6. Approximations

In this chapter we give several approximations to the value function of the Bayesian control model (cf. 2.12). Special attention is paid to the situation where there is only one unknown parameter, i.e. where θ_i is a singleton for all indices $i \in I$ except one. In section 6.1 we consider upper and lower bounds on the value function and their use in successive approximations. These approximations are computable when X, Y, A and θ are finite. In chapter 7 we shall consider algorithms based on these approximations. In section 6.1 we also give a lower bound on the Bayesian discounted total return when a certain Bayesian equivalent rule is used and we also consider another easy-to-handle Bayesian Markov policy. Since in practice the set θ is seldom finite, we study the consequences of approximating of θ by a finite subset in section 6.2. Throughout this chapter we assume that r is bounded and I finite.

6.1 Bounds on the value function and successive approximations

The bounds we consider require the knowledge of the value function of the dynamic program with known parameter value θ for all $\theta \in \theta$ and of the expected discounted total return under several stationary strategies, also for all $\theta \in \theta$. First we introduce some notations:

6.1 (i) F is the set of Bayesian Markov policies (cf. 3.10).

(ii) F is the set of Markov policies (cf. 3.9 and note that $F \subset \widetilde{F}$).

We identify each Bayesian Markov policy with the Bayesian stationary strategy which is determined by it (hence we write v(x,q,f), $f \in \tilde{F}$). An important role is played by a subset \bar{F} of F satisfying:

6.2 inf sup $\{v(x,\theta) - v(x,\theta,f)\} = 0$ for all $\theta \in \theta$. $f \in \vec{F} x \in X$

We shall assume that such a set \overline{F} is given and that $v(x,\theta,f)$ is known for all $x \in X$, $\theta \in \theta$ and $f \in \overline{F}$. Note that, if there exists for all $\theta \in \theta$ a $f_{\theta} \in F$ that is optimal for all $x \in X$ for the dynamic program with known parameter value θ , then the set $\{f_{\theta}, \theta \in \theta\}$ satisfies 6.2. For each $f \in \widetilde{F}$ we define the (non-linear) operator L_{f} on the set of bounded measurable functions b on $X \times W$ as follows:

6.3 (i)
$$(L_{f}b)(x,q) := \sum_{i \in I} i_{K_{i}}(x,f(x,q)) \int v(dy)p_{i}(y,q)\{r(x,f(x,q),y) + \beta \int P(dx'|x,f(x,q),y)b(x',T_{i,y}(q))\}$$

(ii) L_{f}^{n} is the n-th iterate of L_{f} , $n \in \mathbb{N}^{+}$ and L_{f}^{0} := b.

Note that $\sup_{f \in \widetilde{F}} L_f b = Ub$ for each bounded measurable function b on X × W (cf. f f f the remark following 3.10). We further note that, for $f \in \widetilde{F}$

$$\lim_{n\to\infty} (L^n_f b)(x,q) = v(x,q,f) .$$

Although this is easily proved directly, it is an immediate consequence of th. 3.14 (iii) if we consider the model with $D(x,q) = \{f(x,q)\}, x,q \in X \times W$. For $f \in F$ and $\theta \in \Theta$ 6.3 (i) reduces to

$$(\mathbf{L}_{\mathbf{f}}\mathbf{b}) (\mathbf{x}, \theta) = \sum_{\mathbf{i} \in \mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}} (\mathbf{x}, \mathbf{f}(\mathbf{x})) \int \mathcal{V}(d\mathbf{y}) \mathbf{p}_{\mathbf{i}} (\mathbf{y} | \theta_{\mathbf{i}}) \{ \mathbf{r}(\mathbf{x}, \mathbf{f}(\mathbf{x}), \mathbf{y}) + \\ + \beta \int \mathbf{P}(d\mathbf{x}' | \mathbf{x}, \mathbf{f}(\mathbf{x}), \mathbf{y}) \mathbf{b}(\mathbf{x}', \theta) \}$$

which is the usual return operator for the discounted dynamic program with known transition law (cf. [Blackwell (1965)]).

On X \times W we define two functions:

6.4 (i)
$$w(x,q) := \int q(d\theta)v(x,\theta)$$
.
(ii) $\ell(x,q) := \sup_{f \in \overline{F}} \int q(d\theta)v(x,\theta,f)$.

Note that l depends on the choice of the subset \overline{F} of F. Further we define for $n \in \mathbb{N}^*$, $\theta \in \theta$ and $f \in F$:

6.5
$$\varphi_n(\theta, f) := \sup_{\mathbf{x}} \{ \mathbf{v}(\mathbf{x}, \theta) - (\mathbf{L}_f^n \mathbf{v})(\mathbf{x}, \theta) \}$$

$$\varphi_{\infty}(\theta, f) := \sup_{\mathbf{x}} \{ \mathbf{v}(\mathbf{x}, \theta) - \mathbf{v}(\mathbf{x}, \theta, f) \} .$$

Note that, if X is finite, $\lim_{n\to\infty} \varphi_n(\theta, f) = \varphi_{\infty}(\theta, f)$ since

$$\lim_{n \to \infty} (L_f^n v) (x, \theta) = v (x, \theta, f) .$$

For $\mathbf{x} \in \mathbf{X}$, $\mathbf{q} \in \mathbf{W}$, $\pi \in \mathbf{I}$ and $\mathbf{n} \in \mathbf{N}^*$ we have:

(i)
$$\ell(\mathbf{x},\mathbf{q}) \leq \mathbf{v}(\mathbf{x},\mathbf{q}) \leq \mathbf{w}(\mathbf{x},\mathbf{q})$$

(ii)
$$w(x,q) - \ell(x,q) \leq \frac{1}{1-\beta^N} \inf_{f \in \overline{F}} \int q(d\theta) \varphi_N(\theta,f).$$

(iii)
$$\mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \begin{bmatrix} \frac{1}{1-\beta^{N}} \inf_{\mathbf{f}\in\mathbf{F}} \end{bmatrix} Q_{n} (d\theta) \varphi_{N} (\theta, \mathbf{f}) \end{bmatrix}$$

is nonincreasing in n and if

$$\mathbf{P}^{\pi} \begin{bmatrix} 0 & 0 & \{\tau(\mathbf{i}, \mathbf{n}) < \infty\} \end{bmatrix} = 1$$
$$\mathbf{x}, \mathbf{q} \quad \mathbf{i} \in \mathbf{I} \quad \mathbf{n} \in \mathbf{N}^{*}$$

then it tends to zero.

Proof.

In th. 3.16 we proved $v(x,q) \leq w(x,q)$. Further we have

$$\ell(\mathbf{x},\mathbf{q}) = \sup_{\mathbf{f}\in\overline{\mathbf{F}}} \int \mathbf{q}(\mathrm{d}\theta) \mathbf{v}(\mathbf{x},\theta,\mathbf{f}) = \sup_{\mathbf{f}\in\overline{\mathbf{F}}} \mathbf{v}(\mathbf{x},\mathbf{q},\mathbf{f}) \leq \mathbf{v}(\mathbf{x},\mathbf{q}) \ .$$

We proceed with assertion (ii). Note that for N $\in \mathbb{N}^*$

(*)
$$\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\theta,\mathbf{f}) = \sum_{k=0}^{\infty} \{ (\mathbf{L}_{\mathbf{f}}^{kN}\mathbf{v})(\mathbf{x},\theta) - (\mathbf{L}_{\mathbf{f}}^{(k+1)N}\mathbf{v})(\mathbf{x},\theta) \},$$

since lim $(L_{f}^{n}v)(x,\theta) = v(x,\theta,f)$. For bounded measurable functions b and c $n \rightarrow \infty$ on X × θ we find in a familiar way (cf. [Denardo (1967), th. 1])

$$\sup_{\mathbf{x}} \{ (\mathbf{L}_{f}\mathbf{b})(\mathbf{x},\theta) - (\mathbf{L}_{f}\mathbf{c})(\mathbf{x},\theta) \} \le \beta \sup_{\mathbf{x}} \{ \mathbf{b}(\mathbf{x},\theta) - \mathbf{c}(\mathbf{x},\theta) \}$$

and therefore

$$\sup_{\mathbf{x}} \{ (\mathbf{L}_{\mathbf{f}}^{\mathbf{kN}}\mathbf{v})(\mathbf{x},\theta) - (\mathbf{L}_{\mathbf{f}}^{(\mathbf{k}+1)\mathbf{N}}\mathbf{v})(\mathbf{x},\theta) \} \leq \beta^{\mathbf{kN}} \sup_{\mathbf{x}} \{ \mathbf{v}(\mathbf{x},\theta) - (\mathbf{L}_{\mathbf{f}}^{\mathbf{N}}\mathbf{v})(\mathbf{x},\theta) \}.$$

Consequently, using (*), we find:

(**)
$$\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\theta,\mathbf{f}) \leq \frac{1}{1-\beta^N} \phi_N^{(\theta,\mathbf{f})}$$
.

Note that, for $N = \infty$ (**) also holds.

By 6.4 we have

$$w(x,q) - l(x,q) \leq \inf_{f \in \overline{F}} \int q(d\theta) \{v(x,\theta) - v(x,\theta,f)\}$$

and so, using (**) we find the desired result. We proceed with assertion (iii).

$$\begin{split} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\inf_{\mathbf{f}\in\overline{\mathbf{F}}} \int \mathcal{Q}_{\mathbf{n}}(\mathrm{d}\theta) \varphi_{\mathbf{N}}(\theta,\mathbf{f}) \mid \mathcal{F}_{\mathbf{n}-1}] &\leq \inf_{\mathbf{f}\in\overline{\mathbf{F}}} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} [\int \mathcal{Q}_{\mathbf{n}}(\mathrm{d}\theta) \varphi_{\mathbf{N}}(\theta,\mathbf{f}) \mid \mathcal{F}_{\mathbf{n}-1}] &= \\ &= \inf_{\mathbf{f}\in\overline{\mathbf{F}}} \int \mathcal{Q}_{\mathbf{n}-1}(\mathrm{d}\theta) \varphi_{\mathbf{N}}(\theta,\mathbf{f}), \ \mathbf{P}_{\mathbf{x},\mathbf{q}}^{\pi} - \mathrm{a.s.} \ (\mathrm{cf. th. 2.1}) \ . \end{split}$$

Hence the sequence $\{\inf_{f\in \overline{F}} \int Q_n(d\theta) \varphi_N(\theta, f) \mid n \in \mathbb{N}\}$ is a super martingale, which establishes the first part of (iii) and the existence of

$$\lim_{n\to\infty} \inf_{f\in F} \int Q_n(d\theta) \varphi_N(\theta, f) .$$

Assume that $\tau(i,n) < \infty$ for all $n \in \mathbb{N}^*$ and $i \in I$, $\mathbb{P}_{x,q}^{\pi}$ -a.s. From (**) it follows that $\varphi_{\mathbb{N}}(\theta,f) \ge 0$ and from corollary 2.5 that

$$\lim_{n\to\infty}\int Q_n(d\theta)\phi_N(\theta,f) = \phi_N(Z,f), \mathbb{P}_{x,q}^{\pi}-a.s.$$

Hence we have $\mathbf{P}_{\mathbf{x},\mathbf{q}}^{\mathrm{T}}$ -a.s.:

$$0 \leq \lim_{n \to \infty} \inf_{f \in \overline{F}} \int Q_n(d\theta) \varphi_N(\theta, f) \leq \inf_{f \in \overline{F}} \lim_{n \to \infty} \int Q_n(d\theta) \varphi_N(\theta, f) = \inf_{f \in \overline{F}} \varphi_N(Z, f).$$

Note that:

$$v(x,\theta) \ge (L_{f}v)(x,\theta) \ge (L_{f}^{N}v)(x,\theta) \ge v(x,\theta,f), x \in X, \theta \in \theta \text{ and } f \in \overline{F}.$$

Hence:

$$0 \leq \inf_{f \in F} \phi_N(\theta, f) \leq \inf_{f \in F} \sup_{x} \{v(x, \theta) - v(x, \theta, f)\} = 0 \ (cf. 6.2) \ .$$

Therefore we have

$$\lim_{n\to\infty}\inf_{\mathbf{f}\in\overline{\mathbf{F}}}\int Q_{\mathbf{n}}(\mathrm{d}\theta)\phi_{\mathbf{N}}(\theta,\mathbf{f})=0, \mathbf{P}_{\mathbf{X},\mathbf{q}}^{\pi}-\mathrm{a.s.}$$

Since $(\theta, f) \Rightarrow \phi_N^{-}(\theta, f)$ is bounded, the dominated convergence theorem yields the desired result.

Remark.

The bound given in th. 6.1 (ii) has significance only, if either $\varphi_N(\theta, f)$ itself, or an approximation of it, is known for $\theta \in \theta$ and $f \in \overline{F}$. If $v(x,\theta)$ is computed for all $x \in X$ and $\theta \in \theta$, and if optimal Markov policies $f_{\theta} \in F$, $\theta \in \theta$ are determined, then it requires more work to compute $v(x,\theta',f_{\theta})$ than to compute $(L_{f_{A}}v)(x,\theta')$ for $x \in X$, $\theta, \theta' \in \theta$. However,

$$\sup_{\mathbf{x}} \{\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\theta,\mathbf{f})\} \leq \frac{1}{1-\beta} \sup_{\mathbf{x}} \{\mathbf{v}(\mathbf{x},\theta) - (\mathbf{L}_{\mathbf{f}}\mathbf{v})(\mathbf{x},\theta)\}$$

(cf. (**) in the proof of th. 6.1). So for more work we get a better bound.

In th. 6.2 we consider successive approximations of the value function.

Theorem 6.2

For $x \in X$, $q \in W$

(i) $(U^{n}\ell)(x,q) \leq v(x,q) \leq (U^{n}w)(x,q)$.

(ii) $(U^{n}w)(x,q)$ is nonincreasing and $(U^{n}\ell)(x,q)$ is nondecreasing in n.

Proof.

Part (i) is a direct consequence of th. 6.1 (i) since U is monotone. To prove part (ii) it suffices to show Uw \leq w and U $\ell \geq \ell$. Using

$$\mathbf{p}_{i}(\mathbf{y},\mathbf{q})\mathbf{T}_{i,\mathbf{y}}(\mathbf{q}) (d\theta) = \mathbf{p}_{i}(\mathbf{y}|\theta_{i})\mathbf{q}(d\theta)$$

and 6.4 (i) we find

$$(Uw) (x,q) = \sup_{\mathbf{a} \in \mathbf{D}(\mathbf{x})} \int q(d\theta) \sum_{\mathbf{i} \in \mathbf{I}} \mathbf{1}_{K_{\mathbf{i}}} (x,a) \int v(dy) p_{\mathbf{i}}(y|\theta_{\mathbf{i}}) \{r(x,a,y) + \beta \int P(dx'|x,a,y) v(x',\theta) \} \leq \int q(d\theta) v(x,\theta) = w(x,q)$$

where the inequality follows from exchanging sup and $\int q(d\theta)$, and the opa $\in D(x)$ timality equation of the dynamic program with known parameter value, i.e. $v(x, \theta) = (Uv)(x, \theta)$. Using

$$\mathbf{v}(\mathbf{x}',\mathbf{T}_{i,\mathbf{y}}(\mathbf{q}),\mathbf{f}) = \int \mathbf{T}_{i,\mathbf{y}}(\mathbf{q}) (d\theta) \mathbf{v}(\mathbf{x}',\theta,\mathbf{f})$$

we find

$$(U\ell) (x,q) = \sup_{a \in D(x)} \sum_{i \in I} 1_{K_{i}} (x,a) \int \forall (dy) p_{i}(y,q) \{r(x,a,y) + \beta \int P(dx^{\prime} | x,a,y) \sup_{f \in \overline{F}} \forall (x^{\prime}, T_{i,y}(q), f) \} \geq$$

$$\geq \sup_{f \in \overline{F}} \sup_{a \in D(x)} \int q(d\theta) \sum_{i \in I} 1_{K_{i}} (x,a) \int \forall (dy) p_{i}(y | \theta_{i}) \{r(x,a,y) + \beta \int P(dx^{\prime} | x,a,y) \forall (x^{\prime}, \theta, f) \} \geq \sup_{f \in \overline{F}} \forall (x,q,f) = \ell(x,q) .$$

In th. 6.3 we consider for each $\varepsilon > 0$ a Bayesian stationary strategy, which is easy to handle, and which is (nearly) as good as all stationary strategies in \overline{F} . Moreover the strategy processes new information concerning the unknown parameter in the following sence. If, under the strategy f, $\tau(i,n) < \infty$, $\mathbf{P}_{\mathbf{x},\alpha}^{f}$ -a.s. for all $i \in I$ and $n \in \mathbb{N}^{*}$, then we have

$$\limsup_{n \to \infty} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} [\mathbf{v}(\mathbf{X}_{n},\mathbf{Q}_{n}) - \mathbf{v}(\mathbf{X}_{n},\mathbf{Q}_{n},\mathbf{f})] \leq \frac{\varepsilon}{1-\beta} .$$

-

Theorem 6.3 Fix ϵ > 0 and let f be a Bayesian Markov policy such that for (x,q) ϵ X × W

$$(L_{f}\ell)(x,q) \ge (U\ell)(x,q) - \varepsilon$$
.

Then

$$v(x,q,f) \geq l(x,q) - \frac{c}{1-\beta}$$

and if $\mathbf{P}_{x,q}^{f}[\cap \cap \{\tau(i,n) < \infty\}] = 1$, then
$$\lim_{n \to \infty} \sup \mathbf{E}_{x,q}^{f}[\{v(x_{n},Q_{n}) - v(x_{n},Q_{n},f)\}] \leq \frac{\varepsilon}{1-\beta}.$$

Proof.

Let 1 be the function on $X \times W$ which is identically equal to one. By the proof of th. 6.2 (ii):

$$\mathbf{L}_{\mathbf{f}}\boldsymbol{\ell} \geq \mathbf{U}\boldsymbol{\ell} - \boldsymbol{\epsilon}\mathbf{1} \geq \boldsymbol{\ell} - \boldsymbol{\epsilon}\mathbf{1}$$
 .

Assume: $L_{f}^{n} \ell \geq \ell - \epsilon \left(\frac{1-\beta^{n}}{1-\beta}\right) 1$. Then:

$$\mathbf{L}_{\mathbf{f}}^{n+1} \ell \geq \mathbf{L}_{\mathbf{f}} \ell - \epsilon\beta \left(\frac{1-\beta^n}{1-\beta}\right) 1 \geq \ell - \epsilon\left\{1 + \beta \left(\frac{1-\beta^n}{1-\beta}\right)\right\} = \ell - \epsilon \left(\frac{1-\beta^{n+1}}{1-\beta}\right) 1.$$

Proof.

Theorem 6.4

To verify that f is a Bayesian equivalent rule as considered above note that

If $\mathbf{P}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}}$ [$\cap \cap \{\tau(\mathbf{i},\mathbf{n}) < \infty\}$] = 1, then $\mathbf{i} \in \mathbf{I} \ \mathbf{n} \in \mathbb{N}^{*}$ $\limsup_{n \to \infty} \mathbb{E}_{x,q}^{f} [\{ v(x_n,Q_n) - v(x_n,Q_n,f) \}] \leq \frac{\varepsilon}{1-\beta} .$

$$v(x,q,f) \ge w(x,q) - \frac{1}{1-\beta} \{ \inf_{\overline{f} \in \overline{F}} \int q(d\theta) \varphi_1(\theta,\overline{f}) + \epsilon \}$$

Then f is a red above, and

$$(L_{\varepsilon}W)(x,q) \ge (UW)(x,q) - \varepsilon$$
.

Let $\varepsilon > 0$ and let f be a Bayesian Markov policy such that, for $(x,q) \in X \times W$:

is maximized within an E-bound. The strategy is "adaptive" in the same sence as the strategy in th. 6.3.

$$\int q(d\theta) \sum_{i \in I} \mathbf{1}_{K_{i}}(\mathbf{x}, \mathbf{a}) \int v(dy) \mathbf{p}_{i}(\mathbf{y} | \theta_{i}) \{ \mathbf{r}(\mathbf{x}, \mathbf{a}, \mathbf{y}) + \beta \int P(d\mathbf{x}' | \mathbf{x}, \mathbf{a}, \mathbf{y}) v(\mathbf{x}', \theta) \}$$

For the Bayesian equivalent rule considered in section 4.1 (cf. 4.3a) we give in th. 6.4 a lower bound on its Bayesian discounted total return. Hence we consider in th. 6.4 a Bayesian Markov policy such that

$$v(X_n,Q_n) - v(X_n,Q_n,f) \le w(X_n,Q_n) - \ell(X_n,Q_n) +$$

$$+ \frac{\varepsilon}{1-\beta} \le \inf_{\overline{f}\in\overline{F}} \int Q_n(d\theta)\varphi_{\infty}(\theta,\overline{f}) + \frac{\varepsilon}{1-\beta} .$$

This proves the first statement. The second statement is a consequence of th. 6.1 (iii), since

$$\mathbf{v}(\mathbf{x},\mathbf{q},\mathbf{f}) = \lim_{\mathbf{n}\to\infty} (\mathbf{L}_{\mathbf{f}}^{\mathbf{n}}\ell)(\mathbf{x},\mathbf{q}) \geq \ell(\mathbf{x},\mathbf{q}) - \frac{\varepsilon}{1-\beta}$$

Hence, if we let n tend to infinity, we get:

+
$$\beta \int P(dx'|x,a,y)v(x',\theta) \}$$
.

We have $P_{x,q}^{f}$ -a.s.:

$$(\mathbf{L}_{\mathbf{f}} \mathbf{w}) (\mathbf{X}_{n}, \mathbf{Q}_{n}) = \tilde{\mathbf{r}} (\mathbf{X}_{n}, \mathbf{Q}_{n}, \mathbf{A}_{n}) + \beta \mathbf{E}_{\mathbf{x}, \mathbf{q}}^{\mathbf{f}} [\mathbf{w} (\mathbf{X}_{n+1}, \mathbf{Q}_{n+1}) | \mathbf{X}_{n}, \mathbf{Q}_{n}]$$

(cf. 3.1(e) for the definition of \tilde{r}). Hence:

$$\mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} \begin{bmatrix} \sum_{n=0}^{\infty} \beta^{n} \widetilde{r}(\mathbf{x}_{n}, \mathbf{Q}_{n}, \mathbf{A}_{n}) \end{bmatrix} = \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} \begin{bmatrix} \sum_{n=0}^{\infty} \beta^{n} (\mathbf{L}_{\mathbf{f}} w) (\mathbf{x}_{n}, \mathbf{Q}_{n}) - \sum_{n=1}^{\infty} \beta^{n} w (\mathbf{x}_{n}, \mathbf{Q}_{n}) \end{bmatrix}.$$

And therefore, by the definition of f we have:

(*)
$$\mathbf{v}(\mathbf{x},\mathbf{q},\mathbf{f}) = \mathbf{w}(\mathbf{x},\mathbf{q}) + \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} \left[\sum_{n=0}^{\infty} \beta^{n} \{ (\mathbf{L}_{\mathbf{f}} \mathbf{w}) (\mathbf{x}_{n},\mathbf{Q}_{n}) - \mathbf{w}(\mathbf{x}_{n},\mathbf{Q}_{n}) \} \right] \geq$$

$$\geq w(\mathbf{x},\mathbf{q}) + \mathbb{E}_{\mathbf{x},\mathbf{q}}^{f} \left[\sum_{n=0}^{\infty} \beta^{n} \{ (\mathbf{U}w) (\mathbf{x}_{n},\mathbf{Q}_{n}) - w(\mathbf{x}_{n},\mathbf{Q}_{n}) \} \right] - \frac{\varepsilon}{1-\beta} .$$

Since

$$(Uw) (x,q) = \sup_{f \in T} \int (L_{f}v) (x,\theta)q(d\theta), (x,q) \in X \times W,$$

we have:

$$(Uw) (x,q) - w(x,q) \ge \sup_{\vec{f} \in \vec{F}} \int q(d\theta) \{ (L,v) (x,\theta) - v(x,\theta) \} \ge$$
$$\ge -\inf_{\vec{f} \in \vec{F}} \int q(d\theta) \varphi_1(\theta, \vec{f}) .$$

And therefore:

$$(**) \qquad \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} [(\mathbf{U}\mathbf{w}) (\mathbf{x}_{n},\mathbf{Q}_{n}) - \mathbf{w} (\mathbf{x}_{n},\mathbf{Q}_{n})] \geq -\mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} [\inf_{\mathbf{f}\in\mathbf{F}} \int \mathbf{Q}_{n} (\mathbf{d}\theta) \varphi_{1} (\theta,\mathbf{f})] \geq \\ \geq -\inf_{\mathbf{f}\in\mathbf{F}} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\mathbf{f}} [\int \mathbf{Q}_{n} (\mathbf{d}\theta) \varphi_{1} (\theta,\mathbf{f})] = -\inf_{\mathbf{f}\in\mathbf{F}} \int \mathbf{q} (\mathbf{d}\theta) \varphi_{1} (\theta,\mathbf{f}) .$$

Combination of (*) and (**) yields the first statement. To prove the second statement assume that $\tau(i,n) < \infty$, $\mathbf{P}_{x,q}^{f}$ -a.s. Note that, by the first statement:

$$\begin{split} & v(X_n,Q_n) - v(X_n,Q_n,f) \leq w(X_n,Q_n) - v(X_n,Q_n,f) \leq \\ & \leq \frac{1}{1-\beta} \inf_{\overline{f} \in \overline{F}} \int Q_n(d\theta) \varphi_1(\theta,\overline{f}) + \frac{\varepsilon}{1-\beta} \end{split}$$

Hence the desired result follows from th. 6.1 (iii).

128

Remark.

Note that, by th. 6.1 (ii),

$$w(\mathbf{x},\mathbf{q}) - \frac{1}{1-\beta} \inf_{\overline{\mathbf{f}}\in\overline{\mathbf{F}}} \int \mathbf{q} (\mathbf{d}\theta) \phi_1(\theta,\overline{\mathbf{f}}) \leq \ell(\mathbf{x},\mathbf{q}) .$$

Hence the lowerbound on v(x,q,f) in th. 6.4 is not better than the lowerbound we found for the strategy in th. 6.3.

In practise, when we are dealing with finite sets X, Y and A we often have to approximate the value function v. Th. 6.2 gives us the opportunity to do this as accurately as we like. However it is impossible to compute, for example, $(U^{N}w)(x,q)$ for all $(x,q) \in X \times W$, since W is not finite. Nevertheless it is possible to compute $(U^{N}w)(x,q)$ for a fixed $q \in W$, since the number of possible posterior distributions after N transitions is finite. Hence we have to determine a *horison* N \in N such that

$$|(\mathbf{U}^{N}\mathbf{w})(\mathbf{x},\mathbf{q}) - (\mathbf{U}^{N}\boldsymbol{\ell})(\mathbf{x},\mathbf{q})| \leq \varepsilon$$

where $\varepsilon > 0$ is the maximal allowed error in the approximation for v(x,q). Then we compute $\frac{1}{2}\{(U^{N}w)(x,q) + (U^{N}\ell)(x,q)\}$, which is an acceptable approximation for v(x,q) (cf. th. 6.2). To determine N we have to compute first $(U^{N}w)(x,q) - (U^{N}\ell)(x,q)$ for $n = n_0, n_0 + 1, \ldots, N$, where n_0 is a lowerbound on the horizon. In general the horizon determination in this way is very time consuming compared to the backward induction to compute $(U^{N}\{\frac{1}{2}(w+\ell)\})(x,q)$, another acceptable approximation of v(x,q). To see this, we note that in general the sets $W_n(q)$ and $W_m(q)$ of possible posterior distributions of q after n and m transitions, respectively, are disjoint if $m \neq n$ (cf. the remarks at the end of this section).

Hence, to compute $(U^n b)(x,q)$ for some bounded measurable function b on $X \times W$, we first have to compute $(Ub)(x,\tilde{q})$ for $x \in X$ and all $\tilde{q} \in W_{n-1}(q)$ and afterwards $(Ub)(x,\tilde{q})$ for $x \in X$ and $\tilde{q} \in W_{n-2}(q)$ etc. So we have computed, together with $(U^n b)(x,q)$, the set of values

$$\begin{array}{c|c} n-1 \\ \cup & \{(\underline{U}^m_b)(x,\widetilde{q}) \mid x \in X, \ \widetilde{q} \in W_{n-m}(q) \} \\ m=1 \end{array}$$

However, to compute $(U^{n+1}b)(x,q)$, we need the values:

$$\bigcup_{m=1}^{n} \{ (\underline{U}_{b}^{m}) (\mathbf{x}, \widetilde{\mathbf{q}}) \mid \mathbf{x} \in \mathbf{X}, \ \widetilde{\mathbf{q}} \in \mathbf{W}_{n+1-m}(\mathbf{q}) \}$$

and since $W_{n+1-m}(q) \cap W_{n-m}(q) = \emptyset$ in general, we cannot use the already computed values for the computation of $(U^n b)(x,q)$, to determine $(U^{n+1}b)(x,q)$. (We return to this matter in the next chapter.)

It will be clear that it would be nice to have a simpler method to determine a suitable horizon. Indeed such a procedure exists when we are dealing with the simple parameter structure that we introduce below.

Assumption on the parameter structure

6.6 Let I := {1,2,...,t} and let θ₁ be a singleton for i = 2,...,t. Further let {L₁,L₂} be a measurable partition of X and let K₁ := L₁ × A. The models 4 and 5, considered in chapter 5, satisfy 6.6 in a trivial way: there we have L₂ = Ø. In chapter 7 we consider other models satisfying 6.6 (cf. examples 7.4 and 7.5). In the rest of this section we assume that 6.6 holds.

Note that for states $x \in L_2$ the transition law is completely known and for $x \in L_1$ it is incompletely known but the chosen action does not influence the kind of information we get after the transition. It is easy to verify that $q = \bigotimes q_i$ for all $q \in W$, in this situation, since $q_i(\{\theta_i\}) = 1$ for all $i \ge 2$.

Consider the stopping time of

6.7
$$\sigma := \inf\{n > 0 \mid X_n \in L_1\}$$

We shall use the optimal reward operator U_{σ} (cf. section 3.2). Let $x \in L_2$ and let b be a bounded measurable function on $X \times W$. Recall:

$$(\mathbf{U}_{\sigma}\mathbf{b})(\mathbf{x},\mathbf{q}) = \sup_{\pi \in \Pi_{\sigma}} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \begin{bmatrix} \sigma^{-1} \\ \sum \\ n=0 \end{bmatrix} \beta^{n} \mathbf{r}(\mathbf{X}_{n},\mathbf{A}_{n},\mathbf{Y}_{n+1}) + \beta^{\sigma} \mathbf{b}(\mathbf{X}_{\sigma},\mathbf{Q}_{\sigma})] .$$

Next we discuss a nice property of this operator.

If $X_0 \in L_2$ then we have $(X_n, A_n) \in \bigcup K_i$ for $n < \sigma$. Hence the expectation $2 \le i \le t$ of the first term does not depend on $q \in W$ (however, it does depend on the known parameter values $\theta_2, \ldots, \theta_+$). Further we note that

6.8
$$\tau(1,1) = \sigma + 1$$
 if $X_0 \in L_2$

(cf. 2.17 for the definition of τ). Hence, if $X_0 \in L_2$ then $Q_\sigma = Q_0 = q$, and therefore we may write:

6.9
$$(\mathbf{U}_{\sigma}\mathbf{b})(\mathbf{x},\mathbf{q}) = \sup_{\pi \in \Pi_{\sigma}} \mathbf{E}_{\mathbf{x}}^{\pi} [\sum_{n=0}^{\sigma-1} \beta^{n} \mathbf{x}(\mathbf{X}_{n},\mathbf{A}_{n},\mathbf{Y}_{n+1}) + \beta^{\sigma} \mathbf{b}(\mathbf{X}_{\sigma},\mathbf{q})], \mathbf{x} \in \mathbf{L}_{2}$$

since the expectation does not depend on q. The computation of $(U_{\sigma}b)(x,q)$ for $x \in L_2$ is an ordinary dynamic programming problem which is feasible if X, Y and A are finite (this will be clarified in chapter 7).

In th. 6.5 we assume that the function b on X \times W is an approximation of v, such that

$$v(x,q) - b(x,q) \leq \varepsilon(q), x \in X, q \in W$$

First we introduce some notations:

6.10 (i) For
$$q \in W$$
 and $y_1, \dots, y_n \in Y$ we define the probability
 $\chi_q(y_1, \dots, y_n)$ on θ by
 $\chi_q(y_1, \dots, y_n)(B) := \int_{B} \prod_{j=1}^{n} p_1(y_j | \theta_1) q(d\theta) \{ \int_{\theta} \prod_{j=1}^{n} p_1(y_j | \theta_1) q(d\theta) \}^{-1}$
if the denominator is positive;
 $:= q(B)$ otherwise $(B \in T)$.
(ii) $E(q, \varepsilon, n) := \int q(d\theta) \{ \int_{\theta} \dots \int v(dy_1) \dots v(dy_n) \}$
 $\cdot \prod_{j=1}^{n} p_1(y_j | \theta_1) \varepsilon (\chi_q(y_1, \dots, y_n)) \}$

where $\boldsymbol{\varepsilon}$ is a real-valued, bounded measurable function on W.

It is easy to verify that, if $L_2 = \emptyset$ then $E(q,\varepsilon,n) = E_q[\varepsilon(Q_n)]$ since $\tau(1,n) = n$ for all $n \in \mathbb{N}^*$, in this situation (note that here the expectation is independent of the starting state and the strategy).

6.11 (i)
$$\varepsilon_0(\mathbf{q}) := \frac{1}{2} \inf_{\mathbf{f} \in \overline{\mathbf{F}}} \int q(d\theta) \sup_{\mathbf{x} \in \mathbf{L}_1} \{ \mathbf{v}(\mathbf{x}, \theta) - \mathbf{v}(\mathbf{x}, \theta, \mathbf{f}) \};$$

 $\mathbf{b}_0(\mathbf{x}, \mathbf{q}) := \frac{1}{2} \{ \mathbf{w}(\mathbf{x}, \mathbf{q}) + \ell(\mathbf{x}, \mathbf{q}) \}.$

(ii)
$$\varepsilon_{k}(q) := \frac{1}{1-\beta^{k}} \inf_{f \in \overline{F}} \int q(d\theta) \phi_{k}(\theta, f);$$

 $b_{k}(x,q) := w(x,q) - \varepsilon_{k}(q), k \in \overline{N}^{*}.$

Theorem 6.5

(i) Let b be a bounded measurable function on $X \times W$ and let ε be a bounded nonnegative measurable function on W such that for $x \in L_1$ and $q \in W$:

6.12
$$|v(x,q) - b(x,q)| \le \varepsilon(q)$$
.
Then, for $x \in X$, $q \in W$, $n \in \mathbb{N}^*$:

6.13
$$|v(x,q) - (U_{\sigma}^{n}b)(x,q)| \leq \beta^{n}E(q,\varepsilon,n)$$
 if $x \in L_{1}$
 $\leq \beta^{n+1}E(q,\varepsilon,n-1)$ if $x \in L_{2}$.

(ii) In particular the functions b_k and ε_k (cf. 6.11) for $k \in \overline{N}$ satisfy 6.12 and $E(q,\varepsilon_k,n)$ is nonincreasing in n with $\lim_{n\to\infty} E(q,\varepsilon_k,n) = 0$, for $q \in W$.

Proof.

Part (i). Define the operator $\widetilde{U}_{_{\rm J}}$ on the bounded measurable functions on X \times W by:

$$(\widetilde{\mathbf{U}}_{\sigma}\mathbf{f})(\mathbf{x},\mathbf{q}) := \sup_{\boldsymbol{\pi} \in \Pi_{\sigma}} \mathbf{E}_{\mathbf{x},\mathbf{q}}^{\boldsymbol{\pi}} [\beta^{\sigma}\mathbf{f}(\mathbf{x}_{\sigma},\mathbf{Q}_{\sigma})] .$$

Note that this is an optimal reward operator of the kind we studied in section 3.2, for the model with r identically zero. We define $\varepsilon^*(x,q) := \varepsilon(q)$, $x \in X$, $q \in W$. Using corollary 3.13 and th. 3.14 (ii), we find:

$$\mathbf{v} = \mathbf{U}_{\sigma}^{n} \mathbf{v} = \mathbf{U}_{\sigma_{n}} \mathbf{v} \le \mathbf{U}_{\sigma_{n}}(\mathbf{b} + \boldsymbol{\varepsilon}^{\star}) \le (\mathbf{U}_{\sigma_{n}}\mathbf{b}) + (\widetilde{\mathbf{U}}_{\sigma_{n}}\boldsymbol{\varepsilon}^{\star}),$$

where $\sigma_1 := \sigma$ and $\sigma_n := \sigma \circ \sigma_{n-1}$ (cf. 3.14 and 3.23). And similarly

$$v = U_{\sigma_n} v \ge U_{\sigma_n} (b - \varepsilon^*) \ge (U_{\sigma_n} b) - (\widetilde{U}_{\sigma_n} \varepsilon^*)$$
.

Hence

$$|\mathbf{v}(\mathbf{x},\mathbf{q}) - (\mathbf{U}_{\sigma}^{n}\mathbf{b})(\mathbf{x},\mathbf{q})| \leq (\widetilde{\mathbf{U}}_{\sigma}\epsilon^{*})(\mathbf{x},\mathbf{q}), \mathbf{x} \in \mathbf{X}, \mathbf{q} \in \mathbf{W}$$
.

Next we consider $(\widetilde{U}_{\sigma} \overset{\epsilon}{\epsilon}^{*})(x,q)$ in more detail.

Note that $\tau(1,1) = \sigma + 1$ if $X_0 \in L_2$ and $\tau(1,2) = \sigma + 1$ if $X_0 \in L_1$ (remember that $\tau(1,1) = 1$ if $X_0 \in L_1$). By induction it is easy to verify that

$$\sigma_{n} = \inf\{k > \sigma_{n-1} \mid x_{k} \in L_{1}\}, n = 2, 3, \dots$$

We show, using induction, that for $n \in \mathbb{N}^*$:

$$\tau(1,n) = \sigma_n + 1 \text{ if } X_0 \in L_2 \text{ and } \tau(1,n+1) = \sigma_n + 1 \text{ if } X_0 \in L_1$$
.

For n = 1 the statement is true, so assume it holds for n (n \geq 2). If $X_0 \in L_2$ we have

$$\tau(1,n+1) = \inf\{k > \tau(1,n) \mid X_{k-1} \in L_1\} =$$

= $\inf\{k > \sigma_n + 1 \mid X_{k-1} \in L_1\} = 1 + \inf\{k > \sigma_n \mid X_k \in L_1\} = \sigma_{n+1} + 1.$

Similarly if $X_0 \in L_1$. Remember that $\tau(1,n) \ge n$ for $n \in \mathbb{N}^*$. Hence

$$(\widetilde{U}_{\sigma_{n}} \varepsilon^{*})(\mathbf{x}, \mathbf{q}) = \sup_{\pi \in \Pi_{O}} \mathbb{I}_{\mathbf{x}, \mathbf{q}}^{\pi} [\beta^{n} \varepsilon(Q_{\sigma_{n}})] \leq \sup_{\pi \in \Pi_{O}} \beta^{k} \mathbb{I}_{\mathbf{x}, \mathbf{q}}^{\pi} [\varepsilon(Q_{\tau(1, k) - 1})]$$

with k = n if $x \in L_2$ and k = n + 1 if $x \in L_1$. Note that $Q_n = \bigotimes_{\substack{i,n \\ i \in I}} Q_{i,n}$ and $Q_{i,n}(\{\theta_i\}) = 1$ for $i \ge 2$. According to 2.26 we have for $B \in T \quad \mathbb{P}_{x,q}^{\pi}$ -a.s. (cf. the proof of th. 2.1):

$$Q_{n}(B) = \int_{B} \frac{\mu}{\{\ell \geq 0 \mid \tau(1,\ell) \leq n\}} P_{1}(Y_{\tau(1,\ell)} \mid \theta_{1})q(d\theta) + \left\{ \int_{\theta} \prod_{\{\ell \geq 0 \mid \tau(1,\ell) \leq n\}} P_{1}(Y_{\tau(1,\ell)} \mid \theta_{1})q(d\theta) \right\}^{-1}.$$

Hence for $k = 2, 3, \ldots$ we have $\mathbb{P}_{\mathbf{x}, \mathbf{q}}^{\pi}$ -a.s.:

$$Q_{\tau(1,k)-1}(B) = \int_{B}^{k-1} \prod_{j=1}^{k-1} p_{1}(Y_{\tau(1,j)} | \theta_{1})q(d\theta) + \left\{ \int_{\theta}^{k-1} \prod_{j=1}^{k-1} p_{1}(Y_{\tau(1,j)} | \theta_{1})q(d\theta) \right\}^{-1}.$$

Therefore

$$Q_{\tau(1,k)-1} = \chi_q(Y_{\tau(1,1)}, \dots, Y_{\tau(1,k-1)}), P_{x,q}^{\pi}$$
-a.s.

By lemma 2.2 we find:

$$\mathbf{E}_{\mathbf{x},\mathbf{q}}^{\pi} \left[\varepsilon \left(\mathbf{Q}_{\tau(1,k)-1} \right) \right] \leq \mathbf{E}(\mathbf{q},\varepsilon,k-1), \ k = 2,3,\ldots$$

This proves part (i). We proceed with part (ii). It is easy to verify that 6.12 holds for the functions b_k and ε_k , $k \in \overline{N}$. Hence 6.13 holds by part (i). As already noted we have

$$E(q,\varepsilon,n) = E_q[\varepsilon(Q_n)]$$

for the model with $L_2 = \emptyset$ (note that the expectation is independent of the starting state x and the strategy π , here). For this model the assumption of th. 6.3 (iii) is true, which implies that $E(q, \varepsilon, n)$ converges monotonically to zero, as n tends to infinity, in case $k \in \overline{N}^*$. For k = 0 the proof is analogous.

In chapter 7 we discuss algorithms in which the horizon determination is based on th. 6.5 (ii). It turns out that computation of $E(q,\varepsilon,n)$ is rather easy compared with the backward induction.

Corollary 6.6

If $L_2 = \emptyset$ (and 6.6 holds) we have

$$|\mathbf{v}(\mathbf{x},\mathbf{q}) - (\mathbf{U}^{n}\mathbf{b}_{k})(\mathbf{x},\mathbf{q})| \leq \beta^{n}\mathbf{E}_{\mathbf{q}}[\varepsilon_{k}(\mathbf{Q}_{n})], n \in \mathbb{N}^{*}, k \in \mathbf{\bar{N}}$$

(b_k and ε_k are defined in 6.11). This statement is already proved in the proof th. 6.5 (ii).

For the functions $\epsilon_{\rm k}$, k $\in {\bf \bar N}$ defined in 6.11, 6.10 (ii) reduces to:

6.14
$$E(q, \varepsilon_0, n) = \frac{1}{2} \int \dots \int v(dy_1) \dots v(dy_n)$$
$$\cdot \inf_{f \in \overline{F}} \int q(d\theta) \prod_{j=1}^{n} p_1(y_j | \theta_1) \sup_{x \in L_1} \{v(x, \theta_1) - v(x, \theta_1, f)\}$$

and for $k \in \overline{N}^*$:

$$E(q, \varepsilon_{k}, n) = \frac{1}{2} \frac{1}{1 - \beta^{k}} \int \dots \int \nu(dy_{1}) \dots \nu(dy_{n})$$

$$\inf_{f \in F} \int q(d\theta) \prod_{j=1}^{n} p_{1}(y_{j} | \theta_{1}) \phi_{k}(\theta_{1}, f) .$$

To verify this, note that

$$\varepsilon_{k}(\chi_{q}(y_{1},\ldots,y_{n})) = \frac{1}{2} \frac{1}{1-\beta^{k}} \inf_{f \in \overline{F}} \frac{\int_{\substack{q(d\theta) \ \Pi \ p_{1}(y_{j} \mid \theta_{1})\phi_{k}(\theta,f)}}{\int_{\substack{q(d\theta) \ \Pi \ p_{1}(y_{j} \mid \theta)}}, \ k \in \overline{N}^{\star},$$

if the denominator is positive. Further note that, for all $q \in W$:

$$\varepsilon_0^{(q)} \leq \varepsilon_{\infty}^{(q)} \leq \varepsilon_{k+1}^{(q)} \leq \varepsilon_k^{(q)}, \text{ for } k \in \mathbb{N}^2,$$

and note that $\varepsilon_0(q) = \varepsilon_{\infty}(q)$ if $L_2 = \emptyset$.

We conclude this section with some remarks. The first four remarks complement the results we derived in this section. The last three remarks concern other approaches, not treated here.

Remarks.

(i) In most situations the sets of possible posterior distributions at successive stages are disjoint. However the following example shows that this is not always the case.

Let 6.6 hold and let $L_2 = \emptyset$. Further let $\theta_1 := \{t, 1-t\}, 0 < t < \frac{1}{2}$ and let $Y := \{0, 1\}$ and $p_1(y|\theta_1) := \theta_1^Y (1-\theta_1)^{1-y}, y \in Y, \theta_1 \in \theta_1$. It is easy to verify that if $q(\{t\}) := \pi$, then the posterior distribution after n transitions is:

$$Q_{n}(\{t\}) = \{1 + (\frac{t}{1-t}), \frac{\sum_{i=1}^{n} Y_{i}}{\frac{1}{\pi}} + \frac{1}{\pi} + \frac$$

Hence if n > m and $\sum_{i=m+1}^{n} Y_i = \frac{n-m}{2}$ then $Q_n = Q_m$.

(ii)

We have already suggested a choice for the set \overline{F} (see 6.2). Now we consider:

 $\overline{F} := \{f \in F \mid \text{for some } \theta \in \theta: v(x,\theta) = v(x,\theta,f) \text{ for all } x \in X\}$. At first sight one may expect that the best Markov policy for the Bayes criterion can be found in \overline{F} . However for an example we show that

$$\sup v(x,q,f) \leq \sup v(x,q,f)$$
.
 $f \in F$ $f \in F$

Counterexample



 $\begin{array}{l} x := \{0,1,2,3,4\}, \ D(0) := A := \{1,2,3\}, \ D(x) := \{1\}, \ x \neq 0. \\ \theta := \{\frac{1}{10}, \frac{9}{10}\}. \end{array}$ Consider the transition probability P(x'|x,a) from X × A to X:

$$P(1|0,1) = 1 - P(3|0,1) = \theta, P(3|0,3) = 1 - P(1|0,3) = \theta$$

$$P(2|0,2) = P(4|1,1) = P(4|2,1) = P(4|3,1) = P(4|4,1) = 1$$

Only in the states 1, 2 and 3 a reward is obtained: 110, 70 and 10 respectively. It is easy to fit this example into our framework. In case of known parameter values actions 1 or 3 are optimal in state 0 but action 2 is never optimal. We identify the three possible Markov policies with the actions chosen in state 0. Hence $\tilde{F} = \{1,3\}$. Let $q \in W$ be such that $q(\{\theta\}) = \frac{1}{2}$ for $\theta \in \theta$ and let $\beta = \frac{1}{2}$. Then

$$v(0,q,2) = 35$$
 and $v(0,q,1) = v(0,q,3) = 30$.

- (iii) If there exists a $f^* \in F$ such that $v(x,\theta,f^*) = v(x,\theta)$ for all $x \in X$, $\theta \in \theta$ then $v(x,q,f^*) = \int q(d\theta)v(x,\theta) = w(x,q)$, for $x \in X$ and $q \in W$ and therefore f^* is optimal.
- (iv) If b is a bounded measurable function on $X \times W$ such that $b(x,q) = \int b(x,\theta)q(d\theta)$ for all $x \in X$, $q \in W$ then

$$(U^{n}b)(x,q) \leq \int q(d\theta) (U^{n}b)(x,\theta), n \in \mathbb{N}^{*}$$
.

To prove this note that using arguments of the proof of th. 6.2 we find: (Ub) $(x,q) \leq \int q(d\theta)$ (Ub) (x,θ) . Hence, by putting b' $(x,q) := \int q(d\theta)$ (Ub) (x,θ) and repeating the argument, we have

$$(U^{2}b)(x,q) \leq (Ub')(x,q) \leq \int q(d\theta) (U^{2}b)(x,\theta)$$

since $(Ub')(x,\theta) = (U^2b)(x,\theta)$. The statement follows by induction.

(v) In [Martin (1967)] the usual method of successive approximations is described for Bayesian control models with finite state and action spaces. Martin suggested the use of "scrap functions" b on $X \times W$ that are constant on W (in fact Martin specifies a function b^{*} on X and he sets $b(x,q) := b^*(x)$ for $q \in W$). Then he approximates v(x,q)by $(U^n b)(x,q)$. The difficulty of this method is the choice of the horizon such that $|v(x,q) - (U^n b)(x,q)|$ is sufficiently small. He gives the following bound for this difference (cf. [Martin (1967), th. 3.4.3]):

$$\beta^{n}\max\{\overline{b} - \frac{m}{1-\beta}, \frac{M}{1-\beta} - \underline{b}\}$$

where

 $M := \sup r(x,a,y), m := \inf r(x,a,y), \overline{b} := \sup \overline{b}^{*}(x)$ and x,a,y x,a,y x,a,y x $\underline{b} := \inf \overline{b}^{*}(x) .$

To verify this, note:

$$(\mathbf{U}^{n}\mathbf{0})(\mathbf{x},\mathbf{q}) + \beta^{n} \frac{m}{1-\beta} \leq (\mathbf{U}^{n}\mathbf{v})(\mathbf{x},\mathbf{q}) \leq (\mathbf{U}^{n}\mathbf{0})(\mathbf{x},\mathbf{q}) + \beta^{n} \frac{M}{1-\beta}$$

anđ

$$(\mathbf{U}^{n}\mathbf{0})(\mathbf{x},\mathbf{q}) + \beta^{n}\underline{\mathbf{b}} \leq (\mathbf{U}^{n}\mathbf{b})(\mathbf{x},\mathbf{q}) \leq (\mathbf{U}^{n}\mathbf{0})(\mathbf{x},\mathbf{q}) + \beta^{n}\overline{\mathbf{b}}.$$

Since $v = U^{n}v$ we have

$$\beta^{n}\left\{\frac{m}{1-\beta}-\overline{b}\right\} \leq v(\mathbf{x},\mathbf{q}) - (\overline{u}^{n}\mathbf{b})(\mathbf{x},\mathbf{q}) \leq \beta^{n}\left\{\frac{M}{1-\beta}-\underline{b}\right\} .$$

It is obvious that this bound (*) is minimized by setting $b_m := b_M := \frac{1}{2} \frac{M+m}{1-\beta}$. Then the bound becomes $\frac{1}{2} \frac{\beta^n}{1-\beta}(M-m)$ which is poor, in general. In our approach a better scrap function is suggested for the special parameter structure given in 6.6, and the convergence of the posterior distributions is used to get a smaller horizon (see th. 6.5) (see also chapter 7 for some examples). The use of upper and lower bounds is also suggested in [Satia and Lave (1973)]. The authors consider bounds of the form:

(vi)

$$ub(x,q) := \sup_{\substack{f \in F \\ \theta \in \Theta_{q}}} v(x,\theta,f)(1-\varepsilon) + \frac{M}{1-\beta} \varepsilon$$
$$f \in F \\ e \\ f \in F \\ \theta \in \Theta_{q}} v(x,\theta,f)(1-\varepsilon) + \frac{m}{1-\beta} \varepsilon$$

where $\theta_q \subset \theta$ such that $\mathbb{P}_q[\mathbb{Z} \in \theta_q] \ge 1 - \varepsilon$ for some fixed $\varepsilon > 0$ and m and M are defined in the foregoing remark. They compute their bounds with very time-consuming algorithms for Markov games. It is clear that ub(x,q) $\ge w(x,q)$ and $lb(x,q) \le l(x,q)$, if \overline{F} is defined as in remark (ii).

- (vii) In [Waldmann (1976)] the space W is approximated by a finite subset of W, i.e. a finite (measurable) partition of W is constructed, and in each set of this partition a representative is chosen. Then the transition law is modified such that the process only visits these representative points. Waldmann suggests to solve the modified dynamic program with state space $X \times \widetilde{W}$, where \widetilde{W} is the set of representative points. The value function of this dynamic program is an approximation for the value function of the original model. The idea of approximating a dynamic program with an uncountable state space by one with a finite state space is also found in [Whitt (1976)]. Whitt also provides bounds on the approximation.
- (viii) In [Van Hee (1977)] a generalization of the well-known MacQueen extrapolation is considered (see [MacQueen (1966)]) for the situation where 6.6 holds and $L_2 = \emptyset$.

6.2 Discretizations

Although most of the material presented in section 6.2 is valid if X, Y and A are noncountable, the results have practical relevance only if these sets are finite. However, we do not assume that θ is finite, but rather we study the problems caused by θ being infinite.

First we consider the determination of the upper and lower bounds given in th. 6.1. We recall that, if X, Y and A are finite, the computation of $(U_{\sigma}^{n_{1}}(w+l))(x,q)$ for fixed $q \in W$ and $n \in \mathbb{N}^{\star}$ is rather simple if w(x',q')and l(x',q') are known, for $x' \in X$, $q' \in W$. To approximate these upper and lower bound we approximate $\int v(x,\theta)q(d\theta)$ and $\int v(x,\theta,f)q(d\theta)$ using straightforward numerical integration methods.

Afterwards we shall consider another approach, namely the "finitization" of the parameter set in advance. This means that we only consider prior distributions that are concentrated in finitely many points. It is easy to verify that in that situation, all posterior distributions are also concentrated on this finite subset of θ .

We start with a result on perturbations of the function $\theta \rightarrow v(x,\theta)$. In the proof of th. 6.7 we use the same technique as used in [Whitt (1976), th. 6.3].

Let $\theta, \tilde{\theta} \in \theta$. Then:

$$\sup_{\mathbf{x}\in\mathbf{X}} \{\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta})\} \leq \frac{1}{1-\beta} \operatorname{span}(\mathbf{r}) \frac{\frac{1}{2}\Delta(\theta,\theta)}{1-\beta + \frac{1}{2}\beta\Delta(\theta,\widetilde{\theta})}$$

~

,

where

6.15
$$\Delta(\theta, \widetilde{\theta}) := \max_{i \in I} \int v(dy) |p_i(y|\theta_i) - p_i(y|\widetilde{\theta}_i)|$$

anđ

$$span(r) := sup r(x,a,y) - inf r(x,a,y)$$
.
x,a,y x,a,y

Proof.

First assume inf r(s,a,y) = 0. For each $\varepsilon > 0$ there is an action $a \in D(x)$ such that x,a,y

.

$$\mathbf{v}(\mathbf{x},\theta) \leq \varepsilon + \sum_{\mathbf{i} \in \mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}} (\mathbf{x},\mathbf{a}) \int \mathbf{v}(d\mathbf{y}) \mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}})$$

$$\cdot \{\mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y}) + \beta \int \mathbf{P}(d\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y}) \mathbf{v}(\mathbf{x}',\theta)\}.$$

Hence:

$$\begin{aligned} \mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta}) &\leq \varepsilon + \sum_{\mathbf{i}\in\mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x},\mathbf{a}) \int \mathbf{v}(d\mathbf{y}) [[\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}}) - \mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\theta}_{\mathbf{i}})] \mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y}) + \\ &+ \beta \int \mathbf{P}(d\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y}) \{\mathbf{v}(\mathbf{x}',\theta)\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}}) - \mathbf{v}(\mathbf{x}',\widetilde{\theta})\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\theta}_{\mathbf{i}})\}] \leq \\ &\leq \varepsilon + \sum_{\mathbf{i}\in\mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x},\mathbf{a}) [\int \mathbf{v}(d\mathbf{y}) \{\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}}) - \mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\theta}_{\mathbf{i}})\}^{+} \sup_{\mathbf{x},\mathbf{a},\mathbf{y}} \mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y}) + \\ &+ \beta \int \mathbf{v}(d\mathbf{y})\mathbf{P}(d\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y}) \{\mathbf{v}(\mathbf{x}',\theta) - \mathbf{v}(\mathbf{x}',\widetilde{\theta})\} \min\{\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}}),\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\theta}_{\mathbf{i}})\} \\ &+ \beta \int \mathbf{v}(d\mathbf{y})\mathbf{P}(d\mathbf{x}'|\mathbf{x},\mathbf{a},\mathbf{y}) \{\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}}) - \min(\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\theta_{\mathbf{i}}),\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\theta}_{\mathbf{i}}))\} \mathbf{v}(\mathbf{x}',\theta)]. \end{aligned}$$

Remember that $\int v(dy) \mathbf{p}_{i}(y|\theta_{i}) = 1$. Note that, for $i \in I$ $i - \int v(dy) \min\{\mathbf{p}_{i}(y|\theta_{i}), \mathbf{p}_{i}(y|\tilde{\theta}_{i})\} = \int v(dy)\{\mathbf{p}_{i}(y|\theta_{i}) - \mathbf{p}_{i}(y|\tilde{\theta}_{i})\}^{+} =$ $= \frac{1}{2} \int v(dy) |\mathbf{p}_{i}(y|\theta_{i}) - \mathbf{p}_{i}(y|\tilde{\theta}_{i})|.$ Let $\Delta := \Delta(\theta, \tilde{\theta})$ and $M := \sup r(x, a, y)$. Then

x,a,y

$$\begin{aligned} \mathbf{v}(\mathbf{x},\boldsymbol{\theta}) &- \mathbf{v}(\mathbf{x},\widetilde{\boldsymbol{\theta}}) \leq \varepsilon + \frac{1}{2}\Delta M + \beta \sum_{\mathbf{i}\in\mathbf{I}} \mathbf{1}_{\mathbf{K}_{\mathbf{i}}}(\mathbf{x},\mathbf{a}) \\ & [\sup_{\mathbf{x}'} |\mathbf{v}(\mathbf{x}',\boldsymbol{\theta}) - \mathbf{v}(\mathbf{x}',\widetilde{\boldsymbol{\theta}})| \{1 - \frac{1}{2} \int \mathbf{v}(d\mathbf{y}) |\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\boldsymbol{\theta}_{\mathbf{i}}) - \mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\boldsymbol{\theta}_{\mathbf{i}}})|\} + \\ & + \frac{M}{1 - \beta} \frac{1}{2} \int \mathbf{v}(d\mathbf{y}) |\mathbf{p}_{\mathbf{i}}(\mathbf{y}|\boldsymbol{\theta}_{\mathbf{i}}) - \mathbf{p}_{\mathbf{i}}(\mathbf{y}|\widetilde{\boldsymbol{\theta}_{\mathbf{i}}})|] . \end{aligned}$$

Note that

$$\sup_{\mathbf{x}'} \left| \mathbf{v}(\mathbf{x}', \theta) - \mathbf{v}(\mathbf{x}', \widetilde{\theta}) \right| \leq \frac{1}{1 - \beta} M.$$

Hence

$$\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta}) \leq \varepsilon + \frac{1}{2} \Delta \mathbf{M} + \beta \sup_{\mathbf{x}'} |\mathbf{v}(\mathbf{x}',\theta) - \mathbf{v}(\mathbf{x}',\widetilde{\theta})| + \mathbf{x}' + \beta \frac{1}{2} \Delta \{\frac{\mathbf{M}}{1 - \beta} - \sup_{\mathbf{x}'} |\mathbf{v}(\mathbf{x}',\theta) - \mathbf{v}(\mathbf{x}',\widetilde{\theta})|\} .$$

And therefore, by rearranging terms and omitting $\boldsymbol{\varepsilon},$ we find:

(*)
$$\sup_{\mathbf{x}} \{\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta})\} \le \frac{\frac{1}{2}\Delta}{1-\beta} \mathbf{M} + \beta(1-\frac{1}{2}\Delta) \sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta})| .$$

$$\sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta})| \leq \frac{1}{2} \frac{\Delta}{1-\beta} \operatorname{span}(\mathbf{r}) + \beta (1-\frac{1}{2}\Delta) \sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\widetilde{\theta})|$$

which proves the theorem.

Remarks.

(i) If $\{\theta \rightarrow p_i(y | \theta_i); y \in Y\}$ is equicontinuous for all $i \in I$ then the function $\{\theta \rightarrow v(x, \theta); x \in X\}$ is equicontinuous. This is an immediate consequence of th. 6.7.

(ii) If $f \in F$ then

$$\sup_{\mathbf{x}} \left| \mathbf{v}(\mathbf{x},\theta,\mathbf{f}) - \mathbf{v}(\mathbf{x},\widetilde{\theta},\mathbf{f}) \right| \leq \frac{\operatorname{span}(\mathbf{x})}{1-\beta} \frac{\frac{1}{2}\Delta(\theta,\widetilde{\theta})}{1-\beta+\frac{1}{2}\beta\Delta(\theta,\widetilde{\theta})}$$

The proof is exactly the same if we assume $D(x) = {f(x)}, x \in X$.

Assume 6.6 and identify θ_1 and θ . We shall split up the parameter space into a measurable partition $\{B_1, \ldots, B_n\}$ and we assume that in each set B_j a point of discretization b_j is fixed. Further we suppose that, for $j = 1, \ldots, n$ and $x \in X$, $v(x, b_j)$ is known and also that for $j = 1, \ldots, n$ a Markov policy $f_j \in F$ is known such that $v(x, b_j) = v(x, b_j, f_j)$ for all $x \in X$. It is easy to verify that, if $f_k \in \overline{F}$ for $k = 1, \ldots, n$, then

6.16 (i)
$$\ell(\mathbf{x},\mathbf{q}) \geq \max \sum_{1 \leq k \leq n \ j=1}^{n} \mathbf{v}(\mathbf{x},\mathbf{b}_{j},\mathbf{f}_{k})\mathbf{q}(\mathbf{B}_{j}) - \frac{\operatorname{span}(\mathbf{r})}{1-\beta} \frac{\frac{1}{2\Delta}}{1-\beta+\frac{1}{2}\beta\Delta}$$

(ii)
$$w(x,q) \leq \sum_{j=1}^{n} v(x,b_j)q(B_j) + \frac{\operatorname{span}(r)}{1-\beta} \frac{\frac{1}{2}\widetilde{\Delta}}{1-\beta+\frac{1}{2}\widetilde{\Delta}}$$

where $\widetilde{\Delta} := \max \sup_{\substack{\Delta \in B \\ j \leq n \ \theta \in B \\ j}} \Delta(\theta, b_j)$.

Hence we derived an upper and a lower bound for v(x,q) involving only the points of discretization. Statements similar to 6.16 are possible with the other bounds considered in th. 6.1. Note that the difference in the bounds of 6.16 is positive , even if q is degenerate. If we assume more structure we may derive better bounds. Let θ be an interval on the real line: $\theta := [b_0, b_n]$ and let $b_0 < b_1 < \ldots < b_n$ be the points of discretization. Further assume that $\theta \Rightarrow v(x, \theta, f_j)$ is *nondecreasing* for $x \in X$, $j = 1, \ldots, n$ (an example of this situation is considered in example 6.4). Then it is straightforward to verify that, for $(x,q) \in X \times W$:

6.17
$$\max_{\substack{j=1\\1\leq k\leq n \ j\neq 1}}^{n} v(x,b_{j-1},f_{k})q([b_{j-1},b_{j})) \leq v(x,q) \leq \sum_{j=1}^{n} v(x,b_{j})q([b_{j-1},b_{j})).$$

Even if q is degenerate, the upper and lower bound in 6.17 differ at least:

$$\min_{\substack{i \leq j \leq n}} \{v(x,b_j) - v(x,b_{j-1})\}.$$

Now we shall consider the discretization in advance. We only treat the situation where I is a singleton. We omit the dependence on i ϵ I in the notations.
Theorem 6.8

Let I be a singleton, let $b_1, \ldots, b_n \in \theta$ be the points of discretization and let $\{B_1, \ldots, B_n\}$ be a measurable partition of θ such that $b_j \in B_j$, $j = 1, \ldots, n$. Let $q \in W$ and define $\varphi \in W$ such that $\varphi \left(\{b_j\} \right) = q(B_j)$, $j = 1, \ldots, n$. Then:

$$\sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\mathbf{q})| \leq \frac{\operatorname{span}(\mathbf{r})}{1-\beta} \sum_{\mathbf{j}=1}^{n} \int_{\mathbf{B}_{\mathbf{j}}} q(d\theta) \frac{\mathbf{1}_{\mathbf{j}} (\theta, \mathbf{b}_{\mathbf{j}})}{1-\beta + \mathbf{1}_{\mathbf{j}} \beta \Delta(\theta, \mathbf{b}_{\mathbf{j}})} \leq \frac{\operatorname{span}(\mathbf{r})}{1-\beta} \frac{\mathbf{1}_{\mathbf{j}} \Delta}{1-\beta + \mathbf{1}_{\mathbf{j}} \beta \Delta}$$

where

6.18
$$\Delta := \sum_{j=1}^{n} \int_{B_{j}} q(d\theta) \Delta(\theta, b_{j})$$

v

$$(\Delta(\theta, b_4))$$
 has been defined in 6.15; note that I is a singleton).

Proof.

Fix
$$\varepsilon > 0$$
. There is a $\pi \in \Pi_0$ such that for a fixed x ϵ X

$$(x,q) - v(x,\varphi) \le \varepsilon + v(x,q,\pi) - v(x,\varphi,\pi)$$
.

Hence

$$\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\boldsymbol{\varphi}) \leq \varepsilon + \sum_{k=0}^{\infty} \beta^{k} \sum_{j=1}^{n} \left\{ \int_{B_{i}} \mathbf{q}(d\theta) \mathbf{E}_{\mathbf{x},\theta}^{\pi} [\mathbf{r}(\mathbf{x}_{k},\mathbf{A}_{k},\mathbf{x}_{k+1})] - \mathbf{E}_{\mathbf{x},b_{j}}^{\pi} [\mathbf{r}(\mathbf{x}_{k},\mathbf{A}_{k},\mathbf{x}_{k+1})] \mathbf{q}(\mathbf{B}_{i}) \right\}.$$

Let

$$f(y_1, \dots, y_{k+1}) := \int \pi_0 (da_0 | x) \int P(dx_1 | x, a_0, y_1) \dots$$
$$\cdot \int P(dx_k | x_{k-1}, a_{k-1}, y_k) \int \pi_k (da_k | x, a_0, y_1, \dots, y_k, x_k) r(x_k, a_k, y_{k+1}) \dots$$

It is easy to verify that $f(y_1, \ldots, y_{k+1})$ is a version of

$$\mathbf{E}_{\mathbf{x},\theta}^{\pi}[\mathbf{r}(\mathbf{X}_{k},\mathbf{A}_{k},\mathbf{Y}_{k+1}) \mid \mathbf{Y}_{1} = \mathbf{y}_{1},\ldots,\mathbf{Y}_{k+1} = \mathbf{y}_{k+1}] \quad \text{for all } \theta \in \Theta$$

Note that $m \le f \le M$ where $M := \sup r(x,a,y)$ and $m := \inf r(x,a,y)$. Then we have $x,a,y \qquad x,a,y$

$$\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\mathbf{q}) \leq \varepsilon + \int \dots \int \mathbf{v}(\mathrm{d}\mathbf{y}_{1}) \dots \mathbf{v}(\mathrm{d}\mathbf{y}_{k+1})$$

$$\cdot \sum_{i=1}^{n} \int_{\mathbf{B}_{i}} \mathbf{q}(\mathrm{d}\theta) \left\{ \begin{array}{c} \mathbf{H} \\ \mathbf{H} \\ \mathbf{j=1} \end{array} \mathbf{p}(\mathbf{y}_{j} \mid \theta) - \frac{\mathbf{H}}{\mathbf{I}} \mathbf{p}(\mathbf{y}_{j} \mid \mathbf{b}_{i}) \right\} \mathbf{f}(\mathbf{y}_{1}, \dots, \mathbf{y}_{k+1}) \leq$$

$$\leq \varepsilon + \int \dots \int \mathbf{v}(\mathrm{d}\mathbf{y}_{1}) \dots \mathbf{v}(\mathrm{d}\mathbf{y}_{k+1}) \sum_{i=1}^{n} \int_{\mathbf{B}_{i}} \frac{\mathbf{f}(\mathbf{H})}{\mathbf{j=1}} \mathbf{p}(\mathbf{y}_{j} \mid \theta) - \frac{\mathbf{H}}{\mathbf{j=1}} \mathbf{p}(\mathbf{y}_{j} \mid \mathbf{b}_{i}) \right\}^{+} \mathbf{M} -$$

$$- \int \dots \int \mathbf{v}(\mathrm{d}\mathbf{y}_{1}) \dots \mathbf{v}(\mathrm{d}\mathbf{y}_{k+1}) \sum_{i=1}^{n} \int_{\mathbf{B}_{i}} \mathbf{q}(\mathrm{d}\theta) \left\{ \begin{array}{c} \mathbf{H} \\ \mathbf{H} \\ \mathbf{J} = 1 \end{array} \mathbf{p}(\mathbf{y}_{j} \mid \mathbf{b}_{i}) \right\}^{-} \mathbf{m} = \varepsilon + \mathbf{k}_{i} \operatorname{span}(\mathbf{r}) \int \dots \int \mathbf{v}(\mathrm{d}\mathbf{y}_{1}) \dots \mathbf{v}(\mathrm{d}\mathbf{y}_{k+1})$$

$$\cdot \sum_{i=1}^{n} \int_{\mathbf{B}_{i}} \mathbf{q}(\mathrm{d}\theta) \left| \begin{array}{c} \mathbf{H} \\ \mathbf{H} \\ \mathbf{J} = 1 \end{array} \mathbf{p}(\mathbf{y}_{j} \mid \mathbf{b}_{i}) \right|^{-} \mathbf{m} = \varepsilon + \mathbf{k}_{i} \operatorname{span}(\mathbf{r}) \int \dots \int \mathbf{v}(\mathrm{d}\mathbf{y}_{1}) \dots \mathbf{v}(\mathrm{d}\mathbf{y}_{k+1})$$

(Here we use

$$\int \cdots \int \nu(dy_1) \cdots \nu(dy_{k+1}) \prod_{j=1}^{k+1} p(y_j | \theta) = 1, \text{ for all } \theta \in \theta) .$$

Likewise there is a $\pi^* \in \Pi_0$ such that $v(x,\varphi) - v(x,q) \le \varepsilon + v(x,\varphi,\pi^*) - v(x,q,\pi^*)$. Therefore we have:

$$(*) \qquad \sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\boldsymbol{\varphi})| \leq \frac{1}{2} \operatorname{span}(\mathbf{r}) \sum_{k=0}^{\infty} \beta^{k} \int \dots \int \nu(dy_{1}) \dots \nu(dy_{k+1}) \\ \cdot \sum_{i=1}^{n} \int_{B_{i}} q(d\theta) |\prod_{j=1}^{k+1} p(y_{j}|\theta) - \prod_{j=1}^{k+1} p(y_{j}|b_{1})| .$$

Let

$$\alpha_{k} := \frac{1}{2} \int \cdots \int \nu(dy_{1}) \cdots \nu(dy_{k+1}) \sum_{i=1}^{n} \int_{B_{i}} q(d\theta) \left| \prod_{j=1}^{k+1} p(y_{j}|\theta) - \prod_{j=1}^{k+1} p(y_{j}|b_{i}) \right| .$$

Further let $c_1, \ldots, c_{k+1}, d_1, \ldots, d_{k+1}$ be nonnegative numbers. The following inequality is immediate

(**)
$$k+1 \ k+1 \ k+1$$

 $min{ I c , I d } \ge I min{ c , d } , j=1 \ j=1 \ j=1$

144

It is easy to verify that

$$\alpha_{k} = 1 - \int \dots \int \nu(dy_{1}) \dots \nu(dy_{k+1}) \sum_{i=1}^{n} \int q(d\theta) \min\{ \frac{k+1}{I p(y_{j}|\theta)}, \frac{k+1}{j=1} p(y_{j}|b_{i}) \}.$$

Using (**) we find, after changing the order of integration

$$\alpha_{\mathbf{k}} \leq 1 - \sum_{\mathbf{i}=1}^{n} \int_{B_{\mathbf{i}}} q(d\theta) \left[\int v(dy) \min\{p(\mathbf{y}|\theta), p(\mathbf{y}|\mathbf{b}_{\mathbf{i}})\} \right]^{\mathbf{k}+1}$$

Hence (*) becomes:

$$\sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\boldsymbol{\varphi})| \leq \operatorname{span}(\mathbf{r}) \{ \frac{1}{1-\beta} - \sum_{i=1}^{n} \int_{\mathbf{B}_{i}} q(d\theta) \frac{F(\theta, \mathbf{b}_{i})}{1-\beta F(\theta, \mathbf{b}_{i})} \},$$

where $F(\theta, b_{\underline{i}}) := \int v(dy)\min\{p(y|\theta), p(y|b_{\underline{i}})\}$. Note that $\sum_{i=1}^{n} \int_{B_{\underline{i}}} q(d\theta) = 1$ and $F(\theta, b_{\underline{i}}) = 1 - \frac{1}{2}\Delta(\theta, b_{\underline{i}})$. Hence we get the first inequality:

$$\sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\boldsymbol{\varphi})| \leq \operatorname{span}(\mathbf{r}) \sum_{i=1}^{n} \int_{B_{i}} q(d\theta) \frac{\frac{1}{2}\Delta(\theta, \mathbf{b}_{i})}{(1-\beta)\{1-\beta+\frac{1}{2}\beta\Delta(\theta, \mathbf{b}_{i})\}} .$$

Since the function $s \rightarrow \frac{\frac{l_{2S}}{1-\beta+\frac{l_{2S}}{1-\beta+\frac{l_{2S}}{2}}}$ is concave on [0,1] we find using Jensen's inequality:

$$\sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\mathbf{q}) - \mathbf{v}(\mathbf{x},\varphi)| \leq \frac{\operatorname{span}(\mathbf{r})}{1-\beta} \frac{\frac{1}{2}\Delta}{1-\beta+\frac{1}{2}\beta\Delta},$$

which proves the second inequality.

Remarks.

(i) We can also use the proof of th. 6.8 to compare the original model with a slightly different Bayesian control model. Let $\tilde{\theta} := \{1, 2, \ldots, n\}$ be the parameter space of the modified model and let $\tilde{p}(\cdot | \theta)$ be a probability density with respect to the measure v, for $\theta \in \tilde{\theta}$. Further let $\varphi(\{j\}) := q(B_j), j = 1, \ldots, n$ be the prior distribution on $\tilde{\theta}$ and $\tilde{v}(x, \phi)$ be the value of the modified model. All other specifications of the modified model are as in the original model. Then the statement of th. 6.8 remains valid with $v(x, \phi)$ replaced by $\tilde{v}(x, \phi)$ and with $\Delta(\theta, b_j)$ replaced by $\int v(dy) |p(y|\theta) - \tilde{p}(y|j)|$. (ii) Note that, in case I is a singleton, th. 6.7 is a consequence of th. 6.8. To verify this let $q \in W$ be degenerate at θ and let the partition of 6.16 consist of θ only with discretization point θ . Then apply the first inequality of th. 6.8.

Corollary 6.9

Let I be a singleton and let $\tilde{v}(x,q)$ be the value of the model with known transition law, given by v(dy)P(dx'|x,a,y)p(y,q). Then we have

$$\sup_{\mathbf{x}} |\mathbf{v}(\mathbf{x},\mathbf{q}) - \widetilde{\mathbf{v}}(\mathbf{x},\mathbf{q})| \leq \frac{\operatorname{span}(\mathbf{r})}{1-\beta} \frac{\frac{1}{2} \Delta^*}{1-\beta + \frac{1}{2} \beta \Delta^*}$$

where

$$\Delta^* := \int q(d\theta) \int v(dy) \left| p(y|\theta) - p(y,q) \right|$$

Note that this is an example of the situation considered in remark (i) above, if we set $\tilde{\theta} := \{1\}$ and $\tilde{p}(\cdot | 1) := p(\cdot,q)$.

If there is a b $\epsilon \theta$ such that $p(\cdot | b) = p(\cdot,q)$ for some $q \epsilon W$, then corollary 6.9 is a special case of th. 6.8.

In practice one often considers the value $\tilde{v}(\mathbf{x}, \mathbf{q})$, defined in corollary 6.9 as an approximation to $v(\mathbf{x}, \mathbf{q})$. This is justified by the following interpretation. In the Bayesian approach the prior distribution \mathbf{q} is determined, using data from the past. Then the Bayes estimation of the density is computed: $\mathbf{p}(\mathbf{y}, \mathbf{q}) = \int \mathbf{q}(d\theta) \mathbf{p}(\mathbf{y} | \theta)$, for $\mathbf{y} \in \mathbf{Y}$, and finally this density is considered to be the true one.

We conclude this section with some examples and remarks.

Example 6.1

The first bound in th. 6.8 is tight. Consider the model with only one action in each state, and with $X := \{1,2\}$, $Y := \{0,1\}$, $A := \{1\}$, $\theta := \{0,1\}$

> P(1|1,1,1) := 1, P(2|1,1,0) := 1, P(2|2,1,0) := P(2|2,1,1) := 1P(1|0) := 0, r(1,1,1) := 1, r(1,1,0) := r(2,1,0) := r(2,1,1) := 0.



146

Let the only point of discretization be 0, hence $B_1 = \theta$. Let $q \in W$ be defined by $q(\{0\}) = q(\{1\}) = \frac{1}{2}$. It is easy to verify that $v(1,q) = \frac{1}{2} \frac{1}{1-\beta}$. The prior φ defined in th. 6.8 has all mass in the point 0. Hence $v(1,\varphi) = 0$ and $v(1,q) - v(1,\varphi) = \frac{1}{2} \frac{1}{1-\beta}$. Now we consider the first bound of th. 6.8. We have

$$\int v(\mathrm{d}\mathbf{y}) \left| \mathbf{p}(\mathbf{y}|\theta) - \mathbf{p}(\mathbf{y}|\theta) \right| = \left| \theta - 0 \right| + \left| 1 - \theta - 1 \right| = 2\theta, \ \theta \in \Theta$$

Hence the bound becomes $\frac{1}{2} \frac{1}{1-\beta}$.

Example 6.2

The bound of th. 6.7 is tight. Consider the example above and let $\theta := 0$ and $\tilde{\theta} := 1$. Then $|v(1,\theta) - v(1,\tilde{\theta})| = \frac{1}{1-\beta}$ and

$$\int v(dy) \left| p(y|\theta) - p(y|\tilde{\theta}) \right| = 2|\theta - \tilde{\theta}| = 2 .$$

Hence the bound is $\frac{1}{1 - \beta}$ also.

Example 6.3

The bounds of th. 6.8 behave badly if β tends to 1. Consider the model of example 6.1 and modify it as follows: $\theta := [0,1]$, r(1,1,1) := r(1,1,0) := 1. Let the only point of discretization be $\frac{1}{2}$ and let $q \in W$ be homogeneous. Hence $v(1,\theta) = \frac{1}{1-\beta\theta}$ and so $v(1,q) = -\frac{1}{\beta}\log(1-\beta)$ and $v(1,\phi) = \frac{1}{1-\frac{1}{2}\beta}$. The first bound of th. 6.8 becomes:

$$\frac{1}{1-\beta}\int_{0}^{1}\frac{\left|\theta-\frac{1}{2}\right|}{1-\beta+\beta\left|\theta-\frac{1}{2}\right|} d\theta = \frac{1}{1-\beta}\left\{\frac{1}{\beta}-\frac{2\left(1-\beta\right)}{\beta^{2}}\log\left(\frac{1-\frac{\beta}{2}}{1-\beta}\right)\right\} = O\left(\frac{1}{1-\beta}\right).$$

Note that $|v(1,q) - v(1,\phi)| = O(\frac{1}{1-\beta})$.

In the next example we consider a situation where $\theta \rightarrow v(x,\theta,f_i)$ is monotone (cf. 6.17).

Example 6.4

Consider the inventory model: model 5B where the demand is exponentially distributed:

$$\mathbf{p}(\mathbf{y}|\boldsymbol{\theta}) = \boldsymbol{\theta}\mathbf{e}^{-\boldsymbol{\theta}\mathbf{y}}, \ \boldsymbol{\theta} \in [\mathbf{a},\mathbf{b}], \ \boldsymbol{0} < \mathbf{a} < \mathbf{b} < \infty$$

For each $\theta \in [a,b]$ the optimal strategy is characterized by a number s_{θ} , $s_{\theta} \in [a,b]$, such that, at time zero the inventory level is brought to s_{θ} and afterwards at each stage the demand is supplied. For a fixed $s \in [a,b]$ we determine the total expected costs corresponding to the strategy as described above with s instead of s_{θ} .

$$v(x,\theta,s) = hx^{+} + px^{-} + k(s-x) + \sum_{n=1}^{\infty} \beta^{n} E_{\theta} [h(s-Y_{n})^{+} + p(s-Y_{n})^{-} + kY_{n}]$$

It is easy to verify that

$$\mathbf{E}_{\theta}[h(s-\mathbf{Y}_{n})^{+}+p(s-\mathbf{Y}_{n})^{-}+k\mathbf{Y}_{n}] = (h+p)(s+\frac{1}{\theta}e^{-\theta s}-\frac{1}{\theta}) - ps + (p+k)\frac{1}{\theta}.$$

If k > h, then this function is decreasing in θ .

Hence for each point of discretization and each $x \in X$, $\theta \Rightarrow v(x, \theta, f_i)$ is decreasing. (Remember that we considered costs instead of rewards in this example.)

Remarks.

(i) If $\theta \in \mathbb{R}$ and for some $f \in F$, $\theta \neq v(x,\theta,f)$ is convex then $\int q(d\theta)v(x,\theta,f) \geq v(x,\int q(d\theta)\theta,f) ,$

by Jensen's inequality, and if $\theta \to v(x,\theta)$ is *concave* then $w(x,q) \leq v(x, \int q(d\theta)\theta)$. These properties are sometimes useful in approximating upper and lower bounds.

(ii) In [Whitt (1976)] discretizations of the state and action spaces are considered for discounted dynamic programs. If we apply Whitt's approach here we have to discretize the set of posterior distributions W, i.e. we have to fix a finite measurable partition B_1, \ldots, B_n of W and in each set B_i a representant b_i . Then the original model is compared with the model with a perturbed transition law, which causes the process to visit the points b_i , i $\in \{1, \ldots, n\}$ only. However, if $(\tilde{X}_n, \tilde{Q}_n)$ is the state at time n of this new process, then \tilde{Q}_n is not the posterior distribution of Z, in general. So th. 6.5 is not valid anymore.

7. COMPUTATIONAL ASPECTS AND EXAMPLES

In this chapter we consider algorithms for the computation of the value function (cf. 2.12) in two special cases of the Bayesian control model. In section 7.1 we consider the model where the index set I is a singleton. Here we also consider the rate of convergence of the algorithm. In section 7.2 we consider Bayesian control models where assumption 6.6 holds and where in addition the set L_1 is a singleton (cf. 6.6). Finally, in section 7.3, we study some examples of the models considered in sections 7.1 and 7.2, and we illustrate the quality of the algorithms by numerical data. The algorithms are based on the approximations given in th. 6.5. Throughout this chapter we assume that X, Y, A and θ are finite sets. (For notational convenience we write $q(\theta)$ instead of $q(\{\theta\})$.)

7.1 Algorithm for models where I is a singleton

In this section we assume that the index set I is a singleton. We consider an algorithm, based on th. 6.5, to approximate v(x,q) for all $x \in X$ and one fixed prior distribution $q \in W$. The accuracy of the approximation has to be given in advance. In section 6.1 we already considered the set of all possible posterior distributions after n transitions

7.1 $W_n(q) := \{\chi_q(y_1, \dots, y_n) \mid y_1, \dots, y_n \in Y\}, q \in W (cf. 6.10 (i)) .$

Since I is a singleton we omit the subscript $i \in I$ in this section. We first give the algorithm and afterwards we discuss each of its steps. Let $\Delta > 0$ be given, and let $\tilde{v}(x,q)$ be the approximation to v(x,q). If $\max |v(x,q) - \tilde{v}(x,q)| \leq \Delta$ for a fixed $q \in W$ then we say that the accuracy x of the approximation is (at least) Δ . In the algorithms the symbol ":=" denotes an assignment instead of a definition.

Algorithm 1

part 1: parameter influence

(a) For all θ ∈ θ and x ∈ X determine v(x,θ) and an optimizing f_θ ∈ F (i.e. v(x,θ) = v(x,θ,f_θ) for θ ∈ θ). Let F := {f_θ, θ ∈ θ} (cf. 6.2).
(b) For all θ ∈ θ and f ∈ F determine φ_∞(θ,f) = max {v(x,θ) - v(x,θ,f)}.

At the end of the algorithm the values $\tilde{v}(x,q) := v_0(x,q)$, $x \in X$ have been computed and it follows from th. 6.5 that the accuracy is at least Δ . We proceed with a discussion of each of the steps of the algorithm.

Remarks on algorithm 1

(i) The computations of step (a) can be carried out by a standard method, such as the policy iteration algorithm or the method of successive approximations with the MacQueen extrapolation (cf. [Ross (1970), section 6.8], [MacQueen (1966)]). More sophisticated methods can be found in [Van Nunen (1976), section 7.3] and in [Hastings and Van Nunen (1977)]. Note that \overline{F} in step (a) satisfies 6.2. It often occurs that, if the differences between the parameter values are small then also the differences in the value function are small (cf. th. 6.7). Hence if $f_{\theta} \in F$ is optimal, if $\theta \in \theta$ is the true parameter and if $\tilde{\theta} \in \theta$ is near to θ , then it is wise to start the policy iteration for the parameter value $\tilde{\theta}$ with the policy f_{θ} . And likewise we recommend to start the successive approximations for $\tilde{\theta}$ with the scrap function $v(\cdot, \theta)$. (ii) In step (b) we have to determine v(x,θ,f) for all x ∈ X, θ ∈ θ and f ∈ F. If N := # θ then it requires the solution of at most N(N-1) systems of linear equations, in fact we only have to solve them all if we found for each θ ∈ θ a different optimal policy. Instead of the function φ_∞(θ,f) we may also use the function φ₁(θ,f) = max {v(x,θ) - (L_fv)(x,θ)}, if we replace ε_∞(•) by ε₁(•) in step (d).
It is easy to verify that the computation of (L_fv)(x,θ) for x ∈ X, θ ∈ θ and f ∈ F requires less effort than the computation of v(x,θ,f) for x ∈ X, θ ∈ θ and f ∈ F. However we have to do more work in part 2 of the algorithm in this case (cf. th. 6.1).

(iii) Let $Y = \{0, 1, ..., m\}$. We may compute $E(q, \varepsilon_{\infty}, n)$ in the following way:

7.2
$$E(q,\varepsilon_{\infty},n) = \frac{1}{2} \sum_{\substack{k_0,\dots,k_m \\ j=0}} \frac{n!}{m} \min_{\substack{j \in \overline{F} \\ i \in \overline{P} \\ j=0}} \sum_{\substack{m \\ j=0}} \frac{m}{m} (j|\theta)^k j \varphi_{\infty}(\theta,f)$$

with summation over all $k_0, k_1, \ldots, k_m \in \mathbb{N}$ such that $\sum_{j=0}^{m} k_j = n$. Note that we have to sum over $\binom{m+n}{n}$ terms here, so the amount of work to compute δ in this way is very large if n is large. Therefore we suggest another approach in case $\{p(\cdot | \theta), \theta \in \theta\}$ is an *exponential family* of the following form

7.3
$$p(y|\theta) = a(\theta)b(y)exp\{c(\theta)y\}, \theta \in \Theta, y \in Y = \{0,1,\ldots,m\}$$

where a and b are nonnegative functions such that

$$\sum_{\mathbf{y}} a(\theta) \mathbf{b}(\mathbf{y}) \exp\{c(\theta) \mathbf{y}\} = 1, \text{ for all } \theta \in \Theta.$$

In this case the posterior distribution of $q \in W$ after n observations y_1, \ldots, y_n becomes:

$$\frac{a(\theta)^{n} \exp\{c(\theta) \sum_{j=1}^{n} y_{j}\}q(\theta)}{\sum_{i=1}^{n} a(\theta')^{n} \exp\{c(\theta') \sum_{j=1}^{n} y_{j}\}q(\theta')}$$

(provided that the denominator does not vanish). Hence the number of different posterior distributions is:

$$# W_{n}(q) = # \{ \sum_{j=1}^{n} y_{j} \mid y_{j} \in Y, j = 1, ..., n \} = nm + 1 .$$

(This number is small compared to $\binom{m+n}{n}$, the number of terms in 7.2.) Note that there is a one-to-one correspondence between $\sum_{j=1}^{n} Y_{j}$ and Q_{n} , due to relation 7.4. Instead of computing $E(q, \varepsilon_{\infty}, n)$ as proposed in step (d) we approximate this quantity in the following way. Note that here

$$\mathbb{E}(\mathbf{q}, \boldsymbol{\varepsilon}_{\infty}, \mathbf{n}) = \mathbb{I}\!\!\mathbb{E}_{\mathbf{q}} \begin{bmatrix} \boldsymbol{\varepsilon}_{\infty}(\boldsymbol{Q}_{\mathbf{n}}) \end{bmatrix} = \sum_{\mathbf{q}' \in W_{\mathbf{n}}} \mathbb{I}\!\!\mathbb{P}_{\mathbf{q}} \begin{bmatrix} \boldsymbol{Q}_{\mathbf{n}} = \mathbf{q'} \end{bmatrix} \boldsymbol{\varepsilon}_{\infty}(\mathbf{q'}) .$$

It is relatively easy to compute $\varepsilon_{\infty}(q')$ for each $q' \in W_n(q)$. Instead of computing $\mathbb{P}_q[Q_n = q']$ directly, we approximate this probability by the normal probability in the following way. Let $q' \in W_n(q)$ correspond to all sequences $y_1, \ldots, y_n \in Y$ with $\sum_{j=1}^{n} y_j = s$, $0 \le s \le nm$. Further let

7.5
$$\mu_{\theta} := \sum_{j=0}^{m} j_{p}(j|\theta) \text{ and } \sigma_{\theta}^{2} := \sum_{j=0}^{m} (j - \mu_{\theta})^{2} p(j|\theta),$$

Then we have

$$\mathbb{P}_{\mathbf{q}}[\mathcal{Q}_{\mathbf{n}} = \mathbf{q}'] = \sum_{\theta} \mathbf{q}(\theta) \mathbb{P}_{\theta}[\sum_{j=1}^{n} \mathbf{x}_{j} = \mathbf{s}]$$

and therefore

7.6
$$\mathbb{P}_{\mathbf{q}}[\mathcal{Q}_{\mathbf{n}} = \mathbf{q}'] \approx \sum_{\theta} \mathbf{q}(\theta) \left\{ \Phi(\frac{\mathbf{s} + \mathbf{i}_{\mathbf{z}} - \mathbf{n}\mu_{\theta}}{\sigma_{\theta}\sqrt{\mathbf{n}}}) - \Phi(\frac{\mathbf{s} - \mathbf{i}_{\mathbf{z}} - \mathbf{n}\mu_{\theta}}{\sigma_{\theta}\sqrt{\mathbf{n}}}) \right\}$$

(where Φ is the standard normal distribution function). Note that μ_{θ} and σ_{θ} can be computed in advance, also in part 1. Since $\# W_n(q')$ is relatively small it is easy to approximate $E(q, \varepsilon_{\infty}, n)$ in this way.

(iv) Besides the convergence due to the discount factor β , we also use in step (e) the convergence of the posterior distributions. In fact we might replace the stop criterion by " $\delta \leq \Delta$ " without loosing convergence of the algorithm (cf. th. 6.5). Instead of an absolute stop criterion we might use a relative criterion. For instance we could use the inequality

$$\begin{split} \beta^n \delta \{ \mathbf{i} + \max_{\mathbf{x}} \big| \, \mathfrak{l} (\mathbf{x}, \mathbf{q}) \, \big| \, \}^{-1} &\leq \Delta \\ \mathbf{x} \\ \text{instead of } \beta^n \delta &\leq \Delta \text{ , in step (e).} \end{split}$$

(v) The backward induction in part 3 requires the following storage capacity for numbers {#W_n(q) + # W_{n-1}(q)} # X = { (2n - 1)m + 2} # X if Y = {0,1,...,m}, {p(·|θ), θ ∈ θ} is an exponential family of the form 7.3, and n is the horizon determined in part 2.

We note that the work in part 1 has to be carried out only once, while we have to perform part 2 and part 3 for each prior distribution $q \in W$ for which we want to approximate v(x,q), $x \in X$.

We continue with the discussion of a simple modification of the algorithm
to determine in part 3 upper and lower bounds on v(x',q'),
q' e U W_k(q). These bounds shall allow us to exclude some subk=1,...,n-1
optimal actions, during the backward induction procedure. To derive these
bounds we proceed as follows. Let n be the horizon determined in part 2
and remember that

$$v_n(x^i,q^i) = \frac{1}{2} \{ w(x^i,q^i) + \ell(x^i,q^i) \}$$
 for $q^i \in W_n(q), x^i \in X$.

According to corollary 6.6 we have

7.7
$$|\mathbf{v}(\mathbf{x}',\mathbf{q}') - (\mathbf{u}^{n-k}\mathbf{v}_n)(\mathbf{x}',\mathbf{q}')| \leq \beta^{n-k} \mathbb{E}_{\mathbf{q}'}[\varepsilon_{\infty}(\mathbf{Q}_{n-k})]$$

for $q' \in W_{L_k}(q)$, $k \in \{0, 1, \dots, n-1\}$ and $x' \in X$.

Further note that, according to the Markov property of $\{Q_k, k \in \mathbb{N}\}$ (cf. th. 5.1):

7.8
$$\mathbb{E}_{q'}[\varepsilon_{\infty}(Q_{n-k})] = \sum_{y} \mathbb{E}_{T_{y}}(q')[\varepsilon_{\infty}(Q_{n-k-1})]p(y,q')$$

for q' $\in W_k(q)$, k $\in \{0,1,\ldots,n-1\}$ and x' $\in X$. Hence the values $\mathbb{E}_{q'}[\epsilon_{\infty}(Q_{n-k})]$ for q' $\in W_k(q)$ are upper and lower bounds on v(x',q'), for x' $\in X$.

These values are easy to compute by 7.9.

7.9 (i) Let
$$\gamma_n(q^{\prime}) := \varepsilon_{\infty}(q^{\prime})$$
 for $q^{\prime} \in W_n(q)$.
(ii) For $k = n - 1$, $n - 2$,...,1 compute

$$\gamma_k(q^{\prime}) := \sum_{y} p(y,q^{\prime})\gamma_{k+1}(T_y(q^{\prime})), \text{ for } q^{\prime} \in W_k(q) .$$

Note that the computations of 7.9(ii) can be incorporated in step (h) of the algorithm.

If we use the normal approximation, as suggested in remark (iii), then we loose our exact accuracy. However if we incorporate the computations of 7.9, then we have exact bounds after the execution of step 3. It is obvious that an action $a \in D(x)$ is *sub-optimal* in state (x,q') $q' \in W_k(q'), x \in X$ and $k \in \{0,1,\ldots,n-1\}$ if

7.10
$$\sum_{y} [y,q'] [r(x,a,y) + \beta \sum_{x'} P(\{x'\} | x,a,y) \{v_{k+1}(x',T_y(q')) + \gamma_{k+1}(T_y(q'))] \le y$$
$$\le (U(v_{k+1} - \gamma_{k+1})) (x,q') .$$

We conclude this section with a qualitative statement concerning the rate of convergence of the algorithm.

We start with some preparations. Remember that X, Y, A and heta are finite sets.

7.11 A maximum likelihood estimator M_n of the parameter based on the observations Y_1, Y_2, \dots, Y_n is a θ -valued function of Y_1, \dots, Y_n such that

$$\begin{array}{l} & n \\ & \Pi p(\mathbf{Y}_{j} \mid \mathbf{M}_{n}) \geq \Pi p(\mathbf{Y}_{j} \mid \boldsymbol{\theta}) \quad \text{on } \Omega, \quad \text{for all } \boldsymbol{\theta} \in \boldsymbol{\theta} \ . \\ & j=1 \end{array}$$

Lemma 7.1

There are numbers k and a, k,a > 0 such that for all $\theta \in \Theta$ $\mathbb{P}_{\theta}[M_{n} \neq \theta] \leq k \exp\{-an\}, \quad n \in \mathbb{N}^{*}.$

Proof.

Define on Ω :

$$Z_{j}(\theta, \varphi) := \log\{\frac{p(Y_{j}|\theta)}{p(Y_{j}|\varphi)}\}, j \in \mathbb{N}^{*}, \theta, \varphi \in \theta.$$

(let log $0 = -\infty$, log $\infty = \infty$ and let $0 \cdot \infty = 0$). Note that

(*)
$$\{ \underset{n}{\mathbb{M}} \neq \theta \} \subset \{ \underset{\varphi \neq \theta}{\max} \quad \underset{j=1}{\overset{n}{\mathbb{I}}} \underset{j \neq 0}{\overset{n}{\mathbb{I}}} (\underset{j}{\mathbb{V}} | \varphi) \geq \underset{j=1}{\overset{n}{\mathbb{I}}} \underset{j \neq 0}{\overset{n}{\mathbb{I}}} (\underset{j \neq 0}{\overset{n}{\mathbb{I}}} | \theta) \} =$$

154

$$= \{ \min_{\substack{\varphi \neq \theta \\ \varphi \neq \theta}} \prod_{j=1}^{n} \frac{p(\mathbf{Y}_{j} | \theta)}{p(\mathbf{Y}_{j} | \varphi)} \le 1 \} = \bigcup_{\substack{\varphi \neq \theta \\ \varphi \neq \theta}} \{ \sum_{j=1}^{n} \mathbb{Z}_{j}(\theta, \varphi) \le 0 \} .$$

By a Chebyshev-type inequality we have for all $t \leq 0$ and $\theta \in \theta$:

$$(**) \qquad \mathbb{P}_{\theta}\left[\sum_{j=1}^{n} Z_{j}(\theta, \phi) \leq 0\right] \leq \mathbb{E}_{\theta}\left[\exp\left\{t\sum_{j=1}^{n} Z_{j}(\theta, \phi)\right\}\right] = \mathbb{E}_{\theta}\left[\exp\left\{tZ_{j}(\theta, \phi)\right\}\right]^{n}$$

Note that $f_{\theta,\varphi}(t) := \mathbb{E}_{\theta}[\exp\{tZ_{j}(\theta,\varphi)\}] = \sum_{Y} \{\frac{p(y|\varphi)}{p(y|\theta)}\}^{-t} p(y|\theta)$ is finite for $x \neq y$ Further note that $f_{\theta,\varphi}(0) = f_{\theta,\varphi}(-1) = 1$, for $\theta,\varphi \in \theta$. Since the function $t \Rightarrow x^{-t}$ (x > 0) is strictly convex, except when x = 1, we conclude that for each pair $\theta,\varphi \in \theta$ there is a number $t, -1 \le t \le 0$ such that $f_{\theta,\varphi}(t) \le 1$, except when $\frac{p(y|\theta)}{p(y|\varphi)} = 1$ for all y with $p(y|\theta) > 0$. However, if $p(y|\theta) = p(y|\varphi)$ for all $y \in Y$ then, by the separation assumption (cf. 2.1), we have $\theta = \varphi$. Hence there is for each pair $\theta,\varphi \in \theta$, $\varphi \neq \theta$ a number $t_{\theta,\varphi}$ such that $-1 \le t_{\theta,\varphi} \le 0$ and $f_{\theta,\varphi}(t_{\theta,\varphi}) \le 1$. Hence by (*) and (**) we have:

$$\mathbb{P}_{\theta}[\mathbb{M}_{n} \neq \theta] \leq \sum_{\varphi \neq \theta} \mathbb{P}_{\theta}[\sum_{i=1}^{n} \mathbb{Z}_{j}(\theta, \varphi) \leq 0] \leq \sum_{\varphi \neq \theta} \{f_{\theta, \varphi}(t_{\theta, \varphi})\}^{n}.$$

Finally let $m := \max_{\substack{\theta \neq \phi}} f_{\theta,\phi}(t_{\theta,\phi})$, $a := -\log m$ and $k := #: \theta - 1$.

Then we have

$$\mathbb{IP}_{\theta}[M_n \neq \theta] \le k \exp\{-an\}, \text{ for all } \theta \in \theta.$$

The statement of lemma 7.1 is contained in th. 5.3.1 given in [Zacks (1971)], with a proof that is incorrect but easy to repair. Since our situation is less general our proof is easier. However, the idea has been borrowed from Zacks.

In th. 7.4 we use the maximum likelihood estimator to choose a Markov policy $f_{M_n} \in \bar{F}$ such that on Ω :

$$\sum_{\substack{\theta \in \mathbf{X}}} \max\{\mathbf{v}(\mathbf{x},\theta) - \mathbf{v}(\mathbf{x},\theta,\mathbf{f}_{\mathsf{M}_{\mathsf{n}}}) Q_{\mathsf{n}}(\theta) \geq \min_{\substack{\theta \in \mathbf{F}}} \sum_{\substack{\theta \in \mathsf{G}}} \varphi_{\infty}(\theta,\mathbf{f}) Q_{\mathsf{n}}(\theta) .$$

This bound is used implicitly to show that one has exponential convergence in part 2 of the algorithm.

Theorem 7.2

Let $f_{\theta} \in F$ be an optimal policy for $\theta \in \theta$ and let $\overline{F} := \{f_{\theta}, \theta \in \theta\}$. There are positive numbers k and a such that

$$\mathbb{E}_{q} \begin{bmatrix} \min \\ f \in \mathbf{F} \\ \theta \end{bmatrix} \xrightarrow{} \max \{ \mathbf{v}(\mathbf{x}, \theta) - \mathbf{v}(\mathbf{x}, \theta, f) \}_{Q_{n}}(\theta) \end{bmatrix} \leq k \exp\{-an\}$$

Proof.

For $f \in \vec{F}$ define $A_f := \{\theta \in \theta \mid v(x,\theta) = v(x,\theta,f) \text{ for all } x \in X\}$ and $B_f := \theta \setminus A_f$. Further let $\Delta := \max \max \{v(x, \theta) - v(x, \theta, f)\}$. $f \in F \theta$ х

Note that

$$\mathbb{E}_{q} \begin{bmatrix} \min \\ f \in F \\ \theta \end{bmatrix} \xrightarrow{} \max \{ v(x, \theta) - v(x, \theta, f) \}_{Q_{n}}(\theta) \end{bmatrix} \leq \Delta \cdot \mathbb{E}_{q} \begin{bmatrix} \min \\ f \in F \\ f \in F \end{bmatrix} Q_{n}(B_{f}) \end{bmatrix} = \Delta \mathbb{E}_{q} \begin{bmatrix} \min \\ f \in F \\ f \in F \end{bmatrix} q^{\lceil Z \in B_{f} \mid Y_{1}, \dots, Y_{n} \rceil \rceil}.$$

Note further that:

$$\min_{\mathbf{f}\in\mathbf{F}} \mathbb{P}_{\mathbf{q}}[\mathbb{Z} \in \mathbb{B}_{\mathbf{f}} | \mathbb{Y}_{1}, \dots, \mathbb{Y}_{n}] \leq \sum_{\theta} \mathbb{1}_{\{\mathsf{M}_{n}=\theta\}} \mathbb{P}_{\mathbf{q}}[\mathbb{Z} \in \mathbb{B}_{\mathbf{f}_{\theta}} | \mathbb{Y}_{1}, \dots, \mathbb{Y}_{n}]$$

and, since M_n is a function of Y_1, \ldots, Y_n (cf. 7.11), we obtain:

$$\mathbb{E}_{q}[\min \sum_{f \in \overline{F}} \max\{v(x,\theta) - v(x,\theta,f)\}_{Q_{n}}(\theta)] \leq \Delta \cdot \mathbb{P}_{q}[Z \in B] = f_{M_{n}}$$

$$= \Delta \sum_{\theta} \mathbf{q}(\theta) \mathbf{I}_{\theta} \begin{bmatrix} \theta & \mathbf{A}_{\mathbf{f}} \end{bmatrix} .$$

Since $\theta \in A_f$ for all $\theta \in \Theta$, we conclude that $\theta \notin A_f$ implies $\theta \neq \varphi$. Hence $\{\theta \in \mathbf{A}_{\mathbf{f}}\} \stackrel{\mathsf{f}}{\overset{\mathsf{c}}{\overset{\mathsf{d}}}} \{\theta \neq \mathbf{M}_{n}\} \cdot \prod_{n}^{\mathsf{f}}$

Finally the desired result follows from lemma 7.1.

7.2 Algorithm for models with known transition law except for one state

In this section we assume that assumption 6.6 holds and that the set L_1 is a singleton (cf. 6.6). Hence the transition law is known except for one state.

Throughout this section $X = \{0, 1, ..., N\}$ and $L_1 = \{0\}$. Hence $L_2 = \{1, 2, ..., N\}$. The algorithm is also based on th. 6.5 and it consists of three parts again. Each part is a modification of the corresponding part of algorithm 1. We start with a discussion of these parts and afterwards we describe the algorithm.

We start with part 1.

If we want to compute $\mathbb{IE}_{x,q}^{f} [\Sigma_{n=0}^{\sigma-1} \beta^{n} r(X_{n}, A_{n}, Y_{n+1}) + \beta^{\sigma} b(X_{\sigma}, Q_{\sigma})]$ (cf. 6.7) for some $\mathbf{f} \in F$ then we only have to specify the actions in the states of L_{2} . We shall use this property and therefore we introduce some useful notations: 7.12 (i) $\mathbf{F}^{*} := \{f : L_{2} \rightarrow A \mid f(\mathbf{x}) \in D(\mathbf{x})\}$.

(ii)
$$c_f(x) := \mathbb{E}_x^f [\sum_{n=0}^{\sigma-1} \beta^n r(X_n, A_n, Y_{n+1})]$$
, $f \in F^*$, $x \in L_2$.
(iii) $d_f(x) := \mathbb{E}_x^f [\beta^\sigma]$, $f \in F^*$, $x \in L_2$.
(iv) $g(x,e) := \max_{f \in F^*} \{c_f(x) + d_f(x)e\}$, $e \in \mathbb{R}$, $x \in L_2$.

As already noted in section 6.1 (cf. 6.9), the determination of $(U_{g}b)(x,q)$ $x \in L_2$ is an ordinary dynamic programming problem. To see this, extend the state space to $\{-1,0,1,\ldots,N\}$ and let $P(\{-1\}|0,a,y) := P(\{-1\}|-1,a,y) := 1$ for all $a \in A$ and $y \in Y$. Further we define for this model r(0,a,y) := b(0,q)and r(-1,a,y) := 0 for $a \in A$ and $y \in Y$, where $q \in W$ is fixed. It is easy to verify that the value function of this model in $x \in L_2$ equals $(U_{g}b)(x,q)$ of the original model. Therefore we have

7.13
$$(U_{g}b)(x,q) = \max_{f \in F} \{c_{f}(x) + d_{f}(x)b(0,q)\} = g(x,b(0,q)), x \in L_{2}.$$

Let two numbers \underline{e} and \overline{e} be fixed such that: $\underline{e} \leq v(0,q) \leq \overline{e}$ for all $q \in W$. Note that, if $m \leq r \leq M$, $m, M \in IR$ then $\underline{e} := m(1 - \beta)^{-1}$ and $\overline{e} := M(1 - \beta)^{-1}$ will do.

It is easy to compute g(x,e) for all $e \le e \le e$ and $x \in L_2$. This is due to the following properties.

Lemma 7.3

For each x \in X the function $e \rightarrow g(x,e)$ is nondecreasing, convex and piecewise linear. The proof of this lemma is trivial.

It is a consequence of lemma 7.3 that for each $f \in F^*$ the set $\{e | e \le e \le e, g(x,e) = c_f(x) + d_f(x)e\}$ is a closed interval.

For $x \in L_2$ and $a \in D(x)$ we define

7.14 (i)
$$\widetilde{r}(\mathbf{x},\mathbf{a}) := \sum_{i \in I} \mathbf{1}_{K_i}(\mathbf{x},\mathbf{a}) \sum_{\mathbf{y}} \mathbf{p}_i(\mathbf{y}|\boldsymbol{\theta}_i) \mathbf{r}(\mathbf{x},\mathbf{a},\mathbf{y})$$
.

(ii)
$$\widetilde{P}(\mathbf{x}^{\dagger}|\mathbf{x},\mathbf{a}) := \sum_{i \in I} \mathbf{1}_{K_{i}}(\mathbf{x},\mathbf{a}) \sum_{\mathbf{y}} \mathbf{p}_{i}(\mathbf{y}|\boldsymbol{\theta}_{i}) P(\{\mathbf{x}^{\dagger}\}|\mathbf{x},\mathbf{a},\mathbf{y})$$

(note that these definitions are consistent with definitions 3.1 (d) and (e), since θ_i is a singleton for $i \neq 1$, $i \in I$).

Lemma 7.4

Let $f \in F^*$ be optimal for some e_1 , $e_2 \leq e_1 \leq e_2$, i.e. $g(x,e_1) = c_f(x) + d_f(x)e_1$, for $x \in L_2$. Then f is optimal for all $e \in [e_1,e_2]$ and non-optimal for $e > e_2$ where

7.15
$$e_{2} := \max_{x \in L_{2}} [\min_{\substack{f \in L_{2} \\ x \in L_{2}}} \frac{c_{f}(x) - \widetilde{r}(x,a) - \beta \sum_{\substack{x' \in L_{2} \\ \overline{r}(x) + \beta \sum_{\substack{x' \in L_{2} \\ x' \in L_{2}}}} \widetilde{P}(x'|x,a) c_{f}(x') + \beta \widetilde{P}(0|x,a)}],$$

where the minimum has to be taken over all $a \in D(x)$ for which the denominator is positive (the minimum over the empty set is infinite).

Proof.

Note that, for $f \in F^*$ and $x \in L_{2}$:

$$c_{f}(x) + d_{f}(x)e = \widetilde{r}(x,f(x)) + \beta \sum_{x' \in L_{2}} \widetilde{P}(x'|x,f(x)) \{c_{f}(x') + d_{f}(x')e\} + \beta \widetilde{P}(0|x,f(x))e.$$

Moreover for a = f(x) the denominator in 7.15 vanishes. By Howard's policy improvement routine (cf. [Ross (1970), corollary 6.4]) the policy $f \in F^*$ is optimal if and only if for all $a \in D(x)$ and $x \in L_2$:

$$c_{f}(\mathbf{x}) + d_{f}(\mathbf{x}) e \geq \widetilde{r}(\mathbf{x}, a) + \beta \sum_{\mathbf{x}' \in \mathbf{L}_{2}} \widetilde{P}(\mathbf{x}' | \mathbf{x}, a) \{c_{f}(\mathbf{x}') + d_{f}(\mathbf{x}')e\} + \beta \widetilde{P}(0 | \mathbf{x}, a)e.$$

Hence $f \in F^{*}$ is optimal if and only if for all $a \in D(\mathbf{x})$ and $\mathbf{x} \in \mathbf{L}_{2}$:

$$(*) c_{f}(x) - r(x,a) - \beta \sum_{x' \in L_{2}} P(x'|x,a) c_{f}(x') \ge e[-d_{f}(x) + \beta \sum_{x' \in L_{2}} \widetilde{P}(x'|x,a) d_{f}(x') + \beta \widetilde{P}(0|x,a)],$$

So, if the denominator in 7.15 is less than or equal to zero, then (*) holds for all $e > e_1$, if it holds for e_1 . On the other hand, if the denominator in 7.15 is greater than zero, then (*) holds for all $e > e_1$ such that e is less than or equal to the e_2 . This proves the lemma.

According to lemma 7.4 we now have the following procedure to determine $g(x, \cdot)$ for $x \in L_2$. Compute for $e_1 := e$ an optimal $f \in F^*$. Then determine e_2 by 7.15. If $e_1 < e_2 < \bar{e}$ then compute for e_2 a new optimal policy $f \in F^*$ and repeat this procedure. So e_1, e_2, e_3, \ldots are computed. If during this process $e_n = e_{n-1}$ then we have to determine another optimal policy for the value e_{n-1} such that $e_n > e_{n-1}$. Note that there always is such a policy (cf. lemma 7.3) and that we have to examine only finitely many Markov policies since X and A are finite.

Let us denote the set of optimal Markov policies determined in this way by \overline{F}^* . Using the values $c_f(x)$ and $d_f(x)$ for $x \in L_2$ and $f \in \overline{F}^*$ it is easy to determine $v(0,\theta)$ and optimal policies $f_{\theta} \in F$ for $\theta \in \theta$. So, $\overline{F} = \{f_{\theta}, \theta \in \theta\}$ and $v(0,\theta,f)$ for $\theta \in \theta$ and $f \in \overline{F}$ are easy to compute. This concludes the discussion of part 1.

In part 2 of the algorithm a suitable horizon is determined. Here we have to compute for $q' \in W_n(q)$ the value $\varepsilon_0(q')$ and afterwards $E(q,\varepsilon_0,n)$ (cf. 6.11 and th. 6.5). However, $\varepsilon_0(q')$ has a simple form in this case:

$$\varepsilon_0(\mathbf{q'}) = \frac{1}{2} \min_{\mathbf{f} \in \mathbf{F}} \sum_{\boldsymbol{\theta} \in \boldsymbol{\Theta}} \mathbf{q}(\boldsymbol{\theta}) \{ \mathbf{v}(\mathbf{0}, \boldsymbol{\theta}) - \mathbf{v}(\mathbf{0}, \boldsymbol{\theta}, \mathbf{f}) \} .$$

Therefore we have, in a way similar to 6.14:

7.16
$$E(q, \epsilon_0, n) = \frac{1}{2} \{ \sum_{\theta \in \Theta} q(\theta) v(0, \theta) - \sum_{\substack{y_1, \dots, y_n \ f \in F \ \theta \in \Theta}} \max_{f \in F} \sum_{\theta \in \Theta} q(\theta) \prod_{j=1}^n p_j (y_j | \theta_j) v(0, \theta, f) \}.$$

If $\{p_1(\cdot | \theta_1), \theta_1 \in \theta_1\}$ is an exponential family, then we might use the normal approximation to approximate $E(q, \varepsilon_0, n)$ (cf. remark (iii) section 7.1).

In the final part of the algorithm the backward induction is carried out. Here we use

7.17
$$(U_{g}b)(0,q) = \max_{a \in D(0)} \sum_{y} (y,q) [r(0,a,y) + \beta P(\{0\} | 0,a,y) b(0,T_{1,y}(q)) + a \in D(0)]$$

+
$$\beta \sum_{\mathbf{x} \in \mathbf{L}_2} \mathbb{P}\left\{\{\mathbf{x}\} \mid 0, \mathbf{a}, \mathbf{y}\right\} \max_{\mathbf{f} \in \mathbf{F}} \{ \mathbf{c}_{\mathbf{f}}(\mathbf{x}) + \mathbf{d}_{\mathbf{f}}(\mathbf{x}) \mathbb{D}\left\{0, \mathbf{T}_{\mathbf{1}, \mathbf{y}}(\mathbf{q})\right\} \}$$

Finally we summarize the steps of the algorithm.

Algorithm 2

part 1 : parameter influence.

- (a) Determine for all $e \in [e, \overline{e}]$ an optimal $f \in \overline{F}^*$ (cf. 7.12) and simultaneously $c_f(x)$ and $d_f(x)$ for $x \in L_2$. So \overline{F}^* and g(x, e) are determined for $x \in L_2$ and $\underline{e} \leq e \leq \overline{e}$.
- (b) For all $\theta \in \Theta$ compute $v(0, \theta)$ and an optimal $f_{\theta} \in F$ (using the results of step (a)). Hence \tilde{F} is determined.
- (c) For all $\theta \in \Theta$ and $f \in \overline{F}$ determine $v(0, \theta, f)$ part 2 : horizon determination.
- (d) Compute w(0,q) = $\Sigma_{\theta \in \Theta} q(\theta) v(0,\theta)$.
- (e) Set $n := n_0$ (n_0 is a lower estimate of the horizon).
- (f) Compute δ:

$$\delta := \sum_{\substack{\mathbf{y}_1, \dots, \mathbf{y}_n \ \mathbf{f} \in \mathbf{F}}} \min_{\boldsymbol{\theta} \in \mathbf{\theta}} \sum_{j=1}^n (\mathbf{y}_j | \boldsymbol{\theta}_1) \mathbf{v}(0, \boldsymbol{\theta}, \mathbf{f}) \} \quad (cf. 7.16).$$

- (g) If ½βⁿ{w(0,q) δ} ≤ Δ then go to step (h), otherwise set n := n + 1 and go to step (f) (Δ is the desired accuracy).
 part 3 : backward induction.
- (h) For q' $\in W_n(q)$ set : $v_n(0,q') := \frac{1}{2} \{ w(0,q') + \ell(0,q') \}$.
- (i) Set k := n 1.
- (j) For all q' \in W_k(q) compute p(y,q') and T_{1,y}(q') for all y \in Y, and then compute

$$v_{k}^{(0,q')} = \max_{a \in D(0)} \sum_{y} p_{1}^{(y,q)}[r(0,a,y) + \beta P(\{0\} | 0,a,y)v_{k+1}^{(0,T_{1,y}(q'))} + a_{\ell}D(0) y$$

$$+ \beta \sum_{\mathbf{x} \in \mathbf{L}_{2}} \mathbb{P}(\{\mathbf{x}\} \mid \mathbf{0}, \mathbf{a}, \mathbf{y}) \max_{\mathbf{f} \in \overline{\mathbf{F}}^{\star}} \{ \mathbf{c}_{\mathbf{f}}(\mathbf{x}) + \mathbf{d}_{\mathbf{f}}(\mathbf{x}) \mathbf{v}_{\mathbf{k}+1}(\mathbf{0}, \mathbf{T}_{1, \mathbf{y}}(\mathbf{q}')) \}] .$$

(k) If k > 0 then set k := k - 1 and go to (j), otherwise go to (l). (l) For x $\in L_{\gamma}$ compute

$$v_0(x,q) := \max_{f \in F^*} \{ c_f(x) + d_f(x) v_0(0,q) \}$$

Note that $\tilde{v}(x,q) := v_0(x,q)$ is an approximation of v(x,q) of the desired accuracy.

Finally we note that a modification to determine upper and lower bounds in part 3 of algorithm 2 can be incorporated in a way similar to the modification of algorithm 1. It is straightforward to modify th. 7.2 to show that we have exponential convergence in part 2 of algorithm 2.

7.3 Numerical examples

In this section we present 5 examples. The first three examples illustrate algorithm 1 (cf. section 7.1), and the last two examples illustrate algorithm 2 (cf. section 7.2).

Example 7.1 Inventory control

We consider a well-known inventory control model. The model we studied in section 5.3 (model 5A) is a special case. The cost function is

$$c(x,a) := hx^{+} + px^{-} + k(a - x) + K\{1 - \delta(a,x)\}, \quad K \ge 0$$

Here K represents the *order cost* or the cost for starting the production. Note that the cost K is incured only if the inventory is brought to a higher level. If K = 0 then we are dealing with model 5A again. In [Rieder (1972)] it is proved that there is an optimal Bayesian Markov policy f of the following form: 162

7.18
$$f(x,q) = S(q)$$
 if $x < S(q)$

= x if $x \ge s(q)$, $(x,q) \in X \times W$

where s and S are measurable functions from W to the interval [0,M]. (Note that s = S if K = 0). We specify the numerical data of the model. In this example the demand is binomially distributed.

$$P(Y|\theta) = {5 \choose Y} \theta^{Y} (1 - \theta)^{5-Y}, \qquad Y \in Y = \{0, 1, \dots, 5\},$$

and $\theta \in \theta = \{0.1, 0.2, \dots, 0.9\}.$

The prior distribution q is uniform on θ , i.e. $q(\{\theta\}) = \frac{1}{9}$ for $\theta \in \theta$. The various costs are:

$$h = 0.1$$
, $p = 5$, $k = 3$.

We consider two cases: K = 0 and K = 1. Finally the discount factor is $\beta = 0.9$.

In the tables below, we display the optimal strategies $s(\theta)$ and $(s(\theta), S(\theta))$ for the models with known parameter values and the value function, at starting-inventory level zero for the various parameter values. Further we display, for several horizons, the value $E(q, \varepsilon_{\infty}, n)$ and finally we display the optimal actions for the first 3 stages and the value function at inventory level zero for the prior distributions $q' \in \bigcup_{n=0}^{2} W_{n}(q)$.

table 1
$$(K = 0)$$

θ	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9
s(0)	1	2	3	4	4	4	5	5	5
ν(0,θ)	21	37	52	68	82	97	111	124	137

(The optimal strategy, if θ is known, is "order up to level $s(\theta)$ iff $x < s(\theta)$ ")

table	2	(K =	0)
-------	---	------	----

horizon n	0	1	2	3	4	5	6	7
$E(q, \varepsilon_{\infty}, n)$	3.6	1.7	1.1	0.8	0.6	0.5	0.4	0.3

table 3 (K = 1)

θ	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9
s(θ)	1	2	2	3	3	4	4	5	5
S(θ)	2	3	4	4	4	5	5	5	5
v(0,θ)	24	42	59	76	91	107	121	134	147

(The optimal strategy here is: "order up to $S(\theta)$ iff $x < s(\theta)$ ").

table 4 (K = 1)

horizon n	0	1	2	3	4	5	6	7
$E(q, \varepsilon_{\infty}, n)$	3.2	1.5	1.0	0.8	0.6	0.5	0.4	0.3

In table 5 we represent the posterior distributions $q' \in W_n(q)$ by $\sum_{i=1}^n y_i$, $y_i \in Y$ (cf. remark (iii) section 7.1).

table :

	$\Sigma_{i=1}^{n}$	У ₁	.0	1	2	3	4	5	6	7	8	9	10
n		K											
0	s (q)	0	5										
0	v(0,q)	0	82										
0	s (q)	1	4										
0	S (q)	1	5										
0	v(0,q)	1	90			ł							
1	s(q')	0	2	3	4	5	5	5					
1	v(q')	0	36	52	72	93	111	125					
1	s(q')	1	2	2	3	4	4	5			ĺ		
1	S(q')	1	3	3	4	5	5	5			l	×.	
1	v(0,q')	1.	41	58	80	101	120	134					
								[
2	s(q')	0	2	2	3	3	4	4	5	5	5	5	5
2	v(q')	0	27	34	45	58	70	83	94	106	117	126	132
2	s(q')	1	1	2	2	2	3	3	4	4	4	5	5
2	S(q')	1	2	3	3	4	4	5	5	5	5	5	5
2	v(0,q')	1	31	39	51	65	78	91	104	116	127	136	142

163

(The optimal strategies are, if K = 0: "order up to s(q') iff x < s(q')" and if K = 1: "order up to S(q') iff x < s(q')"). Note that $\max\{\ell(x,q) - w(x,q)\} = E(q, \varepsilon_{\omega}, 0)$. It is remarkable that, although the variations in $\theta \rightarrow v(0,\theta)$ are large, the upper and lower bounds on v(0,q) are very close.

Example 7.2 Replacement under additive damage

We consider the following replacement problem. In each time interval (n-1,n], $n \in \mathbb{N}^{*}$, there is a random shock \mathbb{Y}_{n} , which is observed at time n. The random variables $\mathbb{Y}_{1}, \mathbb{Y}_{2}, \mathbb{Y}_{3}, \dots$ are i.i.d., $\mathbb{Y}_{n} \in \mathbb{Y} = [0, \infty)$ and they act on the state of the system in the following way:

7.19
$$X_{n+1} = \min\{\delta(A_n, 0)X_n + Y_{n+1}, x^*\}$$
, $n \in \mathbb{N}$,

where X_n is the state of the system at time n, $X_n \in X = [0, x^*]$ and A_n is the action at time n. The action space is $A = \{0,1\}$. Action 0 means "do not replace" and action 1 means "replace" the machine. D(x) := A for $x \in [0, x^*)$, $D(x^*) := \{1\}$. Replacement takes place instanteneous.

If the system is in state x^* then replacement is more expensive than in the other states. The costs are:

$$c(x,a) = m(x)\delta(a,0) + R\delta(a,1)$$
, if $0 \le x < x^{*}$, $a \in A$
= R^{*} , if $x = x^{*}$, $a \in A$

where $R^* > R$. Here m(x) are the maintenance costs for one period, if the state of the system is x. We assume that $x \to m(x)$, $x \in X$ is real-valued, measurable and nondecreasing, and $0 \le m(0)$, $m(x^*) < R$. It is further assumed that the distribution of Y_n is incompletely known with density $p(\cdot | \theta)$ with respect to a σ -finite measure v on Y, with $\theta \in \theta$ where θ is a complete separable metric space called the parameter space.

It is easy to transform this model into a Bayesian control model with index set I a singleton (cf. example 2.3).

Before we consider numerical data we first establish a property concerning the form of an optimal strategy for this Bayesian control model. In lemma 7.5 we show that the optimal strategy is characterized by so-called *control limits* in the following way. There is an optimal Bayesian Markov policy f such that 7.20 f(x,q) = 0 if $x \le s(q)$

= 1 if x > s(q), $(x,q) \in X \times W$,

where s is a measurable function from W to X.

(The values s(q) are called control limits).

The proof proceeds in a familiar way (cf.[Ross (1970), th. 6.9]).

Lemma 7.5

There is an optimal strategy of the form, given in 7.20.

Proof.

We first show that the value function v is nondecreasing in the first coordinate.

Consider the sequence of successive approximations v_0, v_1, v_2, \ldots of v where $v_0 := 0, v_n := Uv_{n-1}$. By th. 3.14 we have $\lim_{n \to \infty} v_n(x,q) = v(x,q)$, for $(x,q) \in X \times W$. Note that, for $0 \le x < x^*$, $q \in W$ and $n \in \mathbb{N}^*$:

(*)
$$v_{n}(x,q) = \min\{m(x) + \beta \int v(dy)p(y,q)v_{n-1}(\min(x+y,x^{*}),T_{y}(q)), R+\beta \int v(dy)p(y,q)v_{n-1}(\min(y,x^{*}),T_{y}(q))\}$$
.

It is easily verified that for $q \in W$ and $n \in \mathbb{N}^{\times}$:

(**)
$$v_{n}(x^{*},q) = R^{*} + \beta \int v(dy)p(y,q)v_{n-1}(\min(y,x^{*}),T_{y}(q))$$

Hence $x \to v_1(x,q)$ is nondecreasing, since $x \to m(x)$ is nondecreasing for all $q \in W$. Suppose that $x \to v_{n-1}(x,q)$ is nondecreasing for all $q \in W$. Then, by (*), $x \to v_n(x,q)$ is nondecreasing for all $q \in W$. Hence, by induction, $x \to v_n(x,q)$ is nondecreasing for all $q \in W$ and $n \in \mathbb{N}$ and therefore $x \to v(x,q)$ is nondecreasing for all $q \in W$.

It is straightforward to verify that $(x,q) \rightarrow v_1(x,q)$ is measurable and therefore, by (*) and an induction argument, $(x,q) \rightarrow v(x,q)$ is measurable (cf. lemma 1.6 (iii)). Define, for $x \in X$ and $q \in W$:

$$d(\mathbf{x},\mathbf{q}) := m(\mathbf{x}) + \beta \int v(d\mathbf{y})p(\mathbf{y},\mathbf{q})v(\min(\mathbf{x} + \mathbf{y},\mathbf{x}^{\star}), T_{\mathbf{y}}(\mathbf{q}))$$

and

$$b(q) := R + \beta \int v(dy)p(y,q)v(\min(y,x^*),T_y(q)) .$$

It is easy to verify that $q \neq d(x,q)$ and $q \neq b(q)$ are measurable for $x \in X$. Note that v satisfies the functional equation:

$$(***)$$
 $v(x,q) = min\{d(x,q),b(q)\}, \quad 0 \le x \le x^*, q \in W.$

It is straightforward to prove that a strategy which chooses in each state (x,q), $0 \le x < x^*$ and $q \in W$, a minimizing action in (***) is optimal. Let

$$s(q) := \sup\{x \mid 0 \le x \le x^*, d(x,q) \le b(q)\}$$
.

If $q \rightarrow s(q)$ is measurable, then the policy f defined in 7.20 is optimal. To verify the measurability of $q \rightarrow s(q)$ note that for $0 \le a < x^*$:

$$\{q \in W | s(q) > a\} = \{q \in W | d(x,q) \le b(q) \text{ for some } x > a\} =$$
$$= \{q \in W | d(x,q) \le b(q) \text{ for some rational number } x > a\}.$$

Since $q \rightarrow d(x,q)$ and $q \rightarrow b(q)$ are measurable for all $x \in X$ we conclude that $q \rightarrow s(q), q \in W$ is measurable.

In the numerical example the following data are used: $x^* = 25$, m(x) = 0 for all $x \in X$, R = 75, $R^* = 125$ and $\beta = 0.95$. Further $p(y|\theta) = {9 \choose y} \theta^9 (1-\theta)^{9-y}$, for $y \in Y = \{0,1,\ldots,9\}$ and $\theta \in \Theta = \{0.1,0.2,\ldots,0.9\}$.

Hence, if we start the system in an integer x, $0 \le x \le 25$ then the state x_n is always an integer (cf. 7.19).

The prior distribution q on θ is the uniform distribution, i.e. $q(\{\theta\}) = \frac{1}{9}$, for $\theta \in \theta$.

In table 6 we display the optimal strategies for the models with known parameter values and the value function for a new machine, i.e. in x = 0, for the various parameter values.

table	6
-------	---

θ	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9
s(θ)	5	7	9	10	11	12	13	13	14
v(0,0)	461	546	600	639	670	693	714	731	745

(the optimal strategy, when θ is the true parameter, is "replace iff $x > s(\theta)$ ").

166

In table 7 we show for several horizons n: $E(q, \epsilon_n, n)$.

table 7

horizon n	0	1	2	3	4	5	6	7	8	9	10
$E(q, \varepsilon_{\omega}, n)$	6.7	2.9	1.9	1.5	1.2	1.0	0.9	0.8	0.7	0.6	0.5

Finally we show in table 8 for all q' $\in \bigcup_{n=0}^{2} W_{n}(q)$ the optimal control limit s(q') and the optimal value v(0,q'). As in table 5 we represent q' $\in W_{n}(q)$ by $\sum_{i=1}^{n} Y_{i}$, $Y_{i} \in Y$.

∑ y _i i=1 yi	n = 0		n = 1		n = 2		$\begin{bmatrix} n \\ \sum y_i \\ i=1 \end{bmatrix}$ $n =$		= 2
	s(q')	v(0,q')	s (q')	v(0,q')	s(q')	v(0,q')		s(q')	v(0,q')
0	10	645	6	495	5	472	10	11	679
1			7	529	6	483	11	12	691
2			8	574	6	506	12	13	702
3			9	615	7	535	13	13	712
4			10	649	8	566	14	13	721
5		1	11	675	9	593	15	13	730
6			12	698	9	615	16	14	736
7			13	716	10	634	17	14	740
8			13	730	11	651	18	14	743
9			14	738	11	666			

table 8

Example 7.3 Heads or tails

We consider a simple game with only one player, who may choose heads (action 1) or tails (action 2) of a coin with unknown probabilities: the probability of heads is θ , $0 \le \theta \le 1$. The system has two states and only in state 1 the player has a choice. In state 2, the system stays there with probability θ or it goes to state 1 with probability $1 - \theta$. If "heads" has been chosen then the system stays in state 1 with probability θ and if "tails" has been chosen then it remains in state 1 with probability $1 - \theta$. Otherwise the system moves to state 2.

Only in state 1 an immediate reward 100 is obtained, independent of the action chosen. The discount factor is $\beta = 0.9$, and the prior distribution q on $\theta = [0,1]$ is:

$$q(\{\frac{i}{10}\}) = \frac{1}{9}$$
 for $i = 1, 2, ..., 9$.

It is easy to transform this problem into a model considered in section 7.1. At first glance one might think that the optimal strategy is: "if the system is in $(1,\widetilde{q})$ then choose heads if $\int \theta \widetilde{q}(d\theta) \geq \frac{1}{2}$ and choose tails otherwise" $(\widetilde{q} \in W)$.

However, if we consider a prior distribution which is concentrated on the set {0,1}, then it is straightforward to verify that the action "heads" is optimal in state $(1,\tilde{q})$ if and only if $\tilde{q}(\{1\}) \geq \frac{1-\beta}{2-\beta}$.

In table 9 the optimal actions and the value function are displayed for the models with known parameter values. In table 10 we present, for four horizons and all possible posterior distributions, the value function in state 1, the upper and lower bound in state 1 and the optimal action. Note that there is a one-to-one correspondence between the posterior distributions and the number of "heads" for each horizon.

ta	abl	.e	9
			~

θ	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9
action (in 1)	2	2	2	2 1	l or 2	1	1	1	1
v(1,0)	910	820	730	640	550	561	578	609	678

horizon n	number of heads	l(1,•)	v(1,*)	w(1,°)	action
0	0	568	639	675	1 (heads)
1	0	589	611	631	1
	1	670	687	718	2(tails)
2	o	604	611	622	1
	1	562	616	649	1
	2	739	744	758	2
3	0	616	618	623	1
	1	575	595	617	1
	2	636	652	682	2
	3	781	783	788	2

table 10

For the horizon n = 14 we have $\mathbb{E}_{q}[\epsilon_{\infty}(Q_{n})] = 8.5$. Note that the range between the upper and lower bounds are relatively large compared with the examples 7.1 and 7.2.

Example 7.4 Taxi driver's problem

We consider a model that fits into the setting of section 7.2. At the cabrank a taxi driver is offered a run of size $Y_n \in Y \subset \mathbb{N}$, at each stage n. If he accepts this run he will be away for Y_n stages and if he refuses the run he remains in the cab-rank. The random variables Y_1, Y_2, Y_3, \ldots are i.i.d. and observable if and only if the taxi driver is in the cab-rank. The distribution of Y_n is incompletely known. Only at the cab-rank the taxi driver chooses an action a $\epsilon A = Y$. Action a means: "accept all runs larger than or equal to a and refuse the runs smaller than a". Only if he accepts a run of size $y \in Y$ he obtains a reward r(y), where $y \Rightarrow r(y)$ is nondecreasing. To transform this model into a Bayesian control model define the state space X by

$$X := \{0\} \cup \{(n,k) \mid k = 1, \dots, n - 1, n \in \mathbb{N}^{n} \}$$

and the transition law by:

whe

$$\begin{split} & \mathbb{P}(\{n, k+1\}) \mid (n,k), a, y) = 1 \quad \text{for} \quad 1 \leq k \leq n-1, \quad n, k \in \mathbb{N}^{n}, \\ & \mathbb{P}(\{0\} \mid (n, n-1), a, y) = 1 \qquad \text{for} \quad n \in \mathbb{N}^{n}, \\ & \mathbb{P}(\{(n, 1)\} \mid 0, a, n) = 1 \qquad \text{for} \quad n \in \mathbb{N}^{n}. \end{split}$$

Note that, if the system is in state (n,k), $1 \le k \le n-1$, the taxi has been away for k time units on a trip of n time units in total. Further $p(\cdot | \theta)$ is the probability density of Y_n with respect to the counting measure on \mathbb{N} , for $\theta \in \theta$, where θ is the parameter space.

We consider the operator U_{σ} (cf. 6.7) and we obtain the optimality equation for the taxi driver's problem:

$$v(0,q) = \sup_{\mathbf{a} \in \mathbf{A}} \sum_{\mathbf{y} \geq \mathbf{a}} p(\mathbf{y},q) \{ \mathbf{r}(\mathbf{y}) + \beta^{\mathbf{y}} \mathbf{v}(0,\mathbf{T}_{\mathbf{y}}(q)) \} + \beta \sum_{\mathbf{y} < \mathbf{a}} p(\mathbf{y},q) \mathbf{v}(0,\mathbf{T}_{\mathbf{y}}(q))].$$

In the numerical example we used the following data:

$$\begin{split} \mathbf{Y} &= \{1, 2, \dots, 10\}, \quad \mathbf{r}(\mathbf{y}) = \mathbf{e}^{\mathbf{Y}}, \quad \mathbf{y} \in \mathbf{Y}, \quad \beta = 0.9 \ , \\ \mathbf{p}(\mathbf{y} \mid \theta) &= \mathbf{b}(\theta) \cdot \exp\{-(\mathbf{y} - \theta)^2\}, \quad \mathbf{y} \in \mathbf{Y}, \quad \theta \in \theta = \{1, 2, \dots, 10\} \\ \mathbf{re} \ \mathbf{b}(\theta) &= (\sum_{\mathbf{y} \in \mathbf{Y}} \exp\{-(\mathbf{y} - \theta)^2\})^{-1}. \end{split}$$

The prior distribution q is uniform, i.e. $q(\theta) = 0.1$, $\theta \in \theta$. Note that $\{p(\cdot | \theta), \theta \in \theta\}$ is an exponential family and there is a one-toone correspondence between $W_n(q)$ and $\{\sum_{i=1}^n y_i | y_i \in Y\}$ (cf. 7.1). In table 11 we display optimal strategies and the value function v in state 0 for the models with known parameter values.

In table 12 we display for several horizons n the value $E(q, \varepsilon_0, n)$ (cf. 7.16). This quantity is obtained using normal approximation (cf. remark (iii), section 7.1) for $n \ge 5$.

θ	1	2	3	4	5	6	7	8	9	10
action	2	3	4	5	6	7	8	9	10	10
v(0,θ)	6	16	43	118	321	874	2376	6448	15997	21207

tab	le	1	1
	<u>_</u>	-	

table	12
-------	----

horizon n	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
$E(q, \epsilon_0, n)$	327	253	153	125	80	64	44	36	25	20	13	11	8	6	5

For all $q' \in W_1(q) \cup W_2(q)$ the value function in (0,q'), v(0,q') and the optimal actions are shown. (Note that each posterior distribution is represented by $\sum_{i=1}^{n} y_i$, $y_i \in Y$.)

table	1	3
-------	---	---

n = 0	n = 0 $v(q) = 4508$												
n = 1	Y ₁	1	2	3	4	5	6	7	8	9	10		
	v(0,q*)	8	17	49	134	366	995	2697	6992	14960	19859		
	action	3	4	5	6	7	8	9	10	10	10		
n = 2	y ₁ + y ₂	2	3	4	5	6	7	8	9	10			
	v(0,q')	6	9	15	28	44	76	120	209	326			
	action	2	3	3	4	4	5	5	6	6			
	y ₁ + y ₂	11	12	13	14	15	16	17	18	19	20		
	v(0,q')	568	887	1544	2412	4187	6493	10665	15704	19100	20798		
	action	7	7	8	8	9	9	10	10	10	10		

170

Example 7.5 Compound replacement

We consider another model that fits into the setting of section 7.2. Consider a replacement problem with two types of machines. If the controller decides to replace his machine he is not sure of what type the new machine will be. The probabilistic behaviour of both types of machines is known. However the probability of obtaining a machine of type 1 is $\theta \in (0,1)$, and a machine of type 2 is 1 - θ . The parameter θ is unknown. We first describe the machines. The life time of a machine of type i (i = 1 or 2) is geometric with parameter p_1 , $p_1 = 0.9$ and $p_2 = 0.8$. If a machine is "alive" at stage n the controller may replace it (action 1) by a new one (of type 1 with probability θ and of type 2 with probability 1- θ). The costs of such a replacement are C_i for machine i: $C_1 = 10$, $C_2 = 21$. If the controller decides to keep the machine i at stage n he has to pay maintenance costs m_i (n) with probability p, and he has to pay costs for an emergency replacement R, with probability $(1 - p_1)$. Here $m_1(n) = 3(\frac{n}{10} + 1)^2 - 3$ for $n \in \mathbb{N}^*$, $m_2(n) = 4(\frac{n}{10}+1)^2$, $n \in \mathbb{N}^+$, $R_1 = 10$ and $R_2 = 25$. The discount factor is $\beta = 0.9$. We transform this model into a Bayesian control model in the following way. Let $Y = \{0,1\}$ and $p(y|\theta) = \theta^{Y}(1-\theta)^{1-Y}$, $y \in Y$, $\theta \in (0,1)$. Let $X = \{0\} \cup \{(i,n) | i = \{1,2\}, n \in \mathbb{N}^*\}$ and $A = \{0,1\}$. State (i,n) means that we have a machine of type i of age n. State 0 means that we are replacing the machine. The actions have the same meaning as in example 7.2. The trans-

ition law is given by

 $P\{\{(i,n+1)\} | (i,n), 0, y\} = p_{i} .$ $P\{\{0\} | (i,n), 0, y\} = 1 - p_{i} .$ $P\{\{0\} | (i,n), 1, y\} = 1 , \quad i \in \{1,2\}, \quad n \in \mathbb{N}^{*} \text{ and all } y \in Y .$ $P\{\{(y,1)\} | 0, 0, y\} = P\{\{(y,1)\} | 0, 1, y\} = 1 , \quad y \in Y .$

The cost function is:

 $c((i,n),0) = p_{i}m_{i}(n) + (1 - p_{i})R_{i}.$ $c((i,n),1) = C_{i}, \quad i \in \{1,2\}, \quad n \in \mathbb{N}^{*}.$ c(0,0) = c(0,1) = 0.

We consider the operator U_{σ} (cf. 6.7). It is easy to show, in a way similar to the proof of lemma 7.5 that the "interesting" strategies in the set F^* (cf. 7.12(i)) are characterized by two control limits $n_1, n_2 \in \mathbb{N}^*$ such that the controller chooses action 0 if and only if the system is in state (i,n)

172

with $n < n_i$, i ϵ {1,2}. This means that in 7.13 the maximizing $f \in F^*$ is found in this class, for all admissible scrapfunctions b.

7.21
$$v(0,q) = \beta p(1,q) \min_{n \in \mathbb{N}} \{A_1(n) + B_1(n) v(0,T_1(q))\} +$$

n-1

$$+\beta p(0,q) \min_{n \in \mathbb{N}} \{A_2(n) + B_2(n) v(0,T_0(q))\}$$

where

$$A_{i}(n) = \sum_{\ell=1}^{n} (\beta_{P_{i}})^{\ell-1} \{ p_{i}m_{i}(\ell) + (1 - p_{i})R_{i} \} + (\beta_{P_{i}})^{n-1}C_{i}$$

$$B_{i}(n) = \beta \sum_{\ell=1}^{n-1} (\beta p_{i})^{\ell-1} (1 - p_{i}) + (\beta p_{i})^{n-1} , \quad i \in \{1, 2\} , \quad n \in \mathbb{N}^{*}$$

(here $A_i(n)$ equals $c_f(\{i,1\})$ and $B_i(n)$ equals $d_f(\{i,1\})$ where c_f and d_f are defined in 7.12 and $f \in F^*$ is the strategy that replaces only in states (i,k) with $k \ge n$).

In the numerical example we have modified the model in such a way that the state space X becomes finite: in states (i,10), i \in {1,2} we allow only the action 1, i.e. we always replace the system in these states.

The prior distribution q is given by: $q(\{\frac{j}{10}\}) = \frac{1}{9}$, $j = 1, 2, \ldots, 9$. Hence all posterior distributions are concentrated on the set $\{\frac{1}{10}, \ldots, \frac{9}{10}\}$. In table 14 we display the values $v(0,\theta)$, $\theta \in \theta$ and the optimal strategies for these parameter values. The strategies are characterized by pairs of numbers (n_1, n_2) indicating the control limits for both machines. After that, in table 15 we display the values $E(q, \varepsilon_0, n)$ (cf. 7.16) for several horizons n.

tab	le	14
-----	----	----

θ	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9
ν(0,θ)	87	79	71	64	57	51	45	39	34
ⁿ 1, ⁿ 2	10,6	10,5	10,5	9,4	9,4	8,3	7,2	7,2	6,2

table 15

horizon	1	2	3	4	5	6	7	8	9	10	11	12
$E(q, \epsilon_0, n)$ x 0.01	56	46	40	35	30	27	25	23	20	19	18	17

Note that each posterior distribution \tilde{q} of q is completely characterized by the number of times the replaced machine is of type 1, i.e. $\sum_{i=1}^{n} y_i$ determines the posterior distribution. In table 16 we display, for the first 7 stages, the value function in $(0, \tilde{q})$, $v(0, \tilde{q})$ for $\tilde{q} \in \bigcup_{n=0}^{6} W_n(q)$ and the optimal control limits n_1 and n_2 for the two types of machines (for example if n = 2 and $\sum_{i=1}^{2} y_i = 1$, then $n_1 = 7$, $n_2 = 3$ and $v(0, \tilde{q}) = 48$).

$\sum_{i=1}^{n} y_i$	o	1	2	3	4	5	6
0	49						
	7,3						
1	56	42					
	7,3	6,2					
2	60	48	38				
	8,3	7,3	6,2				
3	63	53	43	35			
	8,4	7,3	6,2	6,2			
4	65	57	48	40	34		
	8,4	8,3	7,3	6,2	6,2		
5	66	60	52	45	38	33	
	8,4	8,3	7,3	7,2	6,2	6,2	
6	67	62	55	48	42	36	32
×.	9,4	8,4	7,3	7,3	6,2	6,2	6,2

ta	b	Le	- 1	6
_	_	_	_	~

APPENDIX A. RESULTS FROM ANALYSIS

1. Analytic sets, semi-analytic functions and related subjects

We summarize some pertinent facts about analytic sets and semi-analytic functions. For analytic sets we refer to [Parthasarathy (1967)]. Similar summaries are found in [Blackwell, Freedman and Orkin (1974)] and [Shreve (1977)].

Let N be the Cartesian product of countably many copies of \mathbb{N}^* where \mathbb{N}^* is endowed with the discrete topology and N with the product topology. Let X be a complete separable metric space.

A subset A \subset X is called *analytic* if there is a continuous function from N to X with f(N) = A, moreover \emptyset is analytic. The following properties hold. The proofs are found in [Parthasarathy (1967) chapter I section 3]).

- A1 Each Borel subset of a Borel space is analytic.
- A2 Countable unions, intersections and Cartesian products of analytic sets are analytic.
- A3 If A and B are analytic subsets of Borel spaces (X,X) and (Y,Y) respectively and if f is a Borel measurable function from X to Y then f(A)and $f^{-1}(B)$ are analytic.

As a consequence of A3 we have:

A4 If A is an analytic subset of X \times Y then proj_v(A) is analytic.

Let (X, X) be a Borel space. For each $p \in P(X)$ we have the σ -field $X_p := \{B \cup A | B \in X, \text{ and there is a } C \in X \text{ such that } A \subset C \text{ and } p(C) = 0\}$. X_p is called the *completion* of X with respect to p. The *universal* σ -field U_X is defined by $U_X := \bigcap_{p \in P(X)} X_p$. A $\in U_X$ is called *universally measurable*. A5 Every analytic subset of a Borel space is universally measurable. For a proof see [Christensen (1974) th. 1.5 or th. 1.7].

- A6 For each probability $p \in P(X)$ where (X, X) is a Borel space, there is a unique extension p^* on U_X . And for each real-valued function f on X which is U_X -measurable there is for each $p \in P(X)$ an X-measurable function \tilde{f} such that $f = \tilde{f} p$ -a.s. (the proof is straightforward).
- A7 Kuratowski theorem (see [Parthasarathy (1967) chapter I corollary 3.3]). If (X, X) and (Y, Y) are Borel spaces and $f : X \rightarrow Y$ is Borel measurable and one-to-one, then f(X) is a Borel subset of Y and f^{-1} is Borel measurable.

176

Let (X,X) be a Borel-space. A real-valued function f on X is called *lower* semi-analytic (l.s.a.) if $\{x \mid f(x) < c\}$ is analytic, for $c \in \mathbb{R}$, and f is called upper semi-analytic (u.s.a.) if -f is l.s.a.

The following properties hold.

A8 If f and g are l.s.a. (u.s.a.) then f + g is l.s.a. (u.s.a.) If f_k , $k \in \mathbb{N}$ are l.s.a. then $\inf_k f_k$ is l.s.a. and if f_k , $k \in \mathbb{N}$ are u.s.a. then $\sup_k f_k$ is u.s.a.

Proof.

Note that: $\{x \mid f(x) + g(x) < c\} = \bigcup \{x \mid f(x) < y, g(x) < c - y\}$ where Q is $y \in Q$ the set of rational numbers. Further $\{x \mid \inf_{k} f_{k}(x) < c\} = \bigcup \{x \mid f_{k}(x) < c\}$. So by A2 the statement has been proved for l.s.a. functions. For u.s.a. functions the proof follows from the definition.

A9 Let (X, X) and (Y, Y) be Borel space, and let $g : X \rightarrow Y$ be a Borel function and f a l.s.a. (u.s.a.) function on Y. Then f \circ g is l.s.a. (u.s.a.).

Proof.

Let f be l.s.a. Then $\{y | f(y) < c\}$ is analytic, for $c \in \mathbb{R}$. Hence $\{x | f(g(x)) < c\} = \{x | g(x) \in \{y | f(y) < c\}\}$ is analytic, by A3. Similarly if f is u.s.a.

A10 Let (X,X) and (Y,Y) be Borel spaces and $f : X \times Y \rightarrow \mathbb{R}$ be bounded from above and measurable /1.s.a./u.s.a. Further let P be a transition probability from X to Y. Then the function $x \rightarrow \int f(x,y)P(dy|x)$ is measurable/ 1.s.a./u.s.a.

Proof.

If f is l.s.a./u.s.a. the proof can be found in [Shreve (1977) th. 2.4]. Note that a function is measurable if and only if it is both l.s.a. and u.s.a., which proves the statement if f is measurable. All If f is l.s.a. or u.s.a. then f is universally measurable. (the proof is trivial).

2. Semi-continuous functions and measurable selections

Let X be a metric space. A real-valued function f on X is called upper semicontinuous (u.s.c.) at $x_0 \in X$ if $\limsup_{n \to \infty} f(x_0) \leq f(x_0)$ for any sequence $\{x_n | x_n \in X, n \in \mathbb{N}\}$ such that $\lim_{n \to \infty} x_n = x_0$ and f is called *lower semi-contin*uous (l.s.c.) if -f is u.s.c.

- Al2 Let f be u.s.c. on the metric space X. Then there is a nonincreasing sequence of bounded continuous functions $\{f_k, k \in \mathbb{N}\}$ on X such that $\lim_{k \to \infty} f_k = f$. $k \to \infty$ For a proof see [Hausdorff (1957) section 4.2].
- A13 If f and g are u.s.c. (l.s.c.) then f + g is u.s.c. (l.s.c.). If $\{f_k, k \in \mathbb{N}\}\$ is a nonincreasing sequence of nonpositive u.s.c. functions then f := $\lim_{k \to \infty} f_k$ is u.s.c. For a proof see [Hinderer (1970) page 32].
- A14 Let g be continuous and nonnegative, and let f be u.s.c. (l.s.c.) and bounded. Then f.g is u.s.c. (l.s.c.). For a proof cf. [Hinderer (1970) lemma 5.5(ii)].
- A15 Let X and Y be metric spaces, g a continuous function from X to Y and f is u.s.c. (l.s.c.) on Y. Then $g \circ f$ is also u.s.c. (l.s.c.).

Proof.

Let $x_n \in X$ $(n \in \mathbb{N})$ with $\lim_{n \to \infty} x_n = x_0$. Then, since $\lim_{n \to \infty} g(x_n) = g(x_0)$: $\lim_{n \to \infty} \sup f(g(x_n)) \le f(g(x_0)).$

A16 Let A be an index set and let f_k , $k \in A$ be lis.c. Then $\sup_{k \in A} f_k$ is lis.c. If f_k , $k \in A$ is u.s.c. then $\inf_k f_k$ is u.s.c. $k \in A$

Proof.

Let
$$x_n \in X$$
, $n \in \mathbb{N}$ and $\lim_{n \to \infty} x_n = x_0$. Then
 $\lim_{n \to \infty} \inf_{k \in A} (x_n) \ge \liminf_{n \to \infty} f_k(x_n) \ge f_k(x_0)$ for all $k \in A$.

178

We continue with a result of Schäl on measurable selections. We first introduce some notations.

Let (X,X) and (A,A) be Borel spaces. L(X × A) := {f : X × A \rightarrow R | f is bounded Borel measurable and a \rightarrow f(x,a) is continuous}. $\hat{L}(X × A) := {f : X × A <math>\rightarrow$ R | f is Borel measurable, bounded from above and f is the limit of some nonincreasing sequence of functions f_n $\in L(X × A)$ }. Al7 Let A be compact and f $\in \hat{L}(X × A)$. Then there is a measurable mapping g : X \rightarrow A such that $f(x,g(x)) = \max f(x,a)$. For a proof see [Schäl (1975) th. 12.1].

APPENDIX B. REMARKS ON THE MINIMAX CRITERION

Consider the Bayesian control model (cf. 2.1). Instead of rating the strategies $\pi \in \Pi$ by their Bayesian discounted total returns (cf 2.12) we might say π^* is at least as good as π in state x, if

$$\inf_{\theta \in \Theta} \mathbf{v}(\mathbf{x}, \theta, \pi^{\star}) \geq \inf_{\mathbf{x}, \theta, \pi} \mathbf{v}(\mathbf{x}, \theta, \pi) \cdot \mathbf{\theta}_{\epsilon} \mathbf{\theta}$$
$$\theta \in \Theta$$
$$\geq 0. \text{ A strategy } \pi^{\star} \in \mathbb{I} \text{ is called } \epsilon - minimax \text{ in state } \mathbf{x} \in \mathbb{X}, \text{ if}$$

```
 \begin{array}{ll} \inf v(\mathbf{x}, \theta, \pi^{\star}) \geq \sup \inf v(\mathbf{x}, \theta, \pi) - \varepsilon \ . \\ \theta \in \Theta & \pi \in \Pi \ \theta \in \Theta \end{array}
```

A 0-minimax strategy is simply called minimax.

Let ϵ

A term "maximin" would be preferable, however in statistical decision theory the term "minimax" is current since one is interested in minimizing the expected loss instead of maximizing the expected return (cf. [Wald (1947)]). We shall discuss a nice property of the Bayes criterion, which the minimax criterion does not have. Let $\pi = (\pi_0, \pi_1, \ldots) \in \Pi$ be an optimal strategy for the Bayes criterion, if the process is started in state $x \in X$ and if $q \in W$ is the prior distribution. Let the history at stage 1 be $(x,a,y,x') \in H_1$ and define similar to 3.22 the "tail-strategy" $\pi^* = (\pi_0^*, \pi_1^*, \ldots)$ by

$$\pi_{k}^{\star}(\cdot | x_{0}, a_{0}, y_{1}, x_{1}, \dots, y_{k}, x_{k}) := \pi_{k+1}(\cdot | x, a, y, x_{0}, a_{0}, y_{1}, x_{1}, \dots, y_{k}, x_{k})$$

for $k \in \mathbb{N}$. Then it is easy to verify that the strategy π^* is optimal for the Bayes criterion if the process is started in x' with respect to the prior distribution $\Sigma_{i \in I} \ {}^{1}_{K_{i}}(x,a)T_{i,y}(q)$.

Hence the decision maker, who chooses a strategy that is optimal for the Bayes criterion, uses at each stage a strategy that is optimal for the Bayes criterion from that stage on, with respect to an "updated" prior distribution. In fact this property is the well-known "principle of optimality", for the equivalent dynamic program (model 2). In [Groenewegen (1978)] this principle is studied extensively. The property discussed above, enables us to compute the value function and the optimal actions by backward induction. However, we show by an example that the minimax criterion does not have this property. A decision maker, who prefers the minimax criterion might be considered as a pessimist. However in the example he seems to forget his pes-

simism after one transition.

Another unpleasant property of the minimax criterion is that we may not
restrict our attention to the nonrandomized strategies. In the example it turns out that none of the nonrandomized strategies is optimal.

Example.



We start with an informal description. Only in states 0 and 1 there are two actions. In all other states the transitions are deterministic. If the decision maker chooses action 1 in state 0 then the next state will be 1 with probability θ and 2 with probability 1 - θ . If he

chooses action 2, then the system moves to state 1 with probability 1 - θ and to state 2 with probability θ . In state 1 the two actions have the same effect with respect to the states 3 and 4. The parameter θ is unknown. In state 2 the reward is large compared with the rewards in the other states. This causes the "least favourable" parameter value for each strategy to be completely determined by the action chosen in state 0, if he starts there. We continue with a formal description of the example in terms of model 1. $X = \{0, 1, 2, 3, 4\}, X = \{0, 1\}, A = \{1, 2\}, D(0) = D(1) = A, D(3) = D(2) = 0$ = $D(4) = \{1\}, \theta = \{0.1, 0.9\}$ and I is a singleton. The transition probabilities are determined by a function $F : X \times A \times Y \rightarrow X$ in the following way $P({F(x,a,y)} | x,a,y) = 1$ (cf. example 2.3 chapter 1): $F(0,1,y) = 1\delta(1,y) + 2\delta(0,y) , F(0,2,y) = 2\delta(1,y) + 1\delta(0,y)$ $F(1,1,y) = 3\delta(1,y) + 4\delta(0,y)$, $F(1,2,y) = 4\delta(1,y) + 3\delta(0,y)$ and F(2,1,y) = 4, F(3,1,y) = 3, F(4,1,y) = 4 for all $y \in Y$. Further $p(y|\theta) = \theta^{Y}(1-\theta)^{1-Y}$ for $y \in Y$ and $\theta \in \theta$. The reward function r is given by r(0,1,y) = 8, r(0,2,y) = 1, r(1,1,y) = 25, r(1,2,y) = 20r(2,1,y) = 200, r(3,1,y) = 2, r(4,1,y) = 14 for all $y \in Y$, and the discount factor $\beta = \frac{1}{2}$. We omit y in the notation for r. First we consider a decision maker who starts in state 0. Any strategy for him can be characterized by three numbers a, b, and c. Here a is the probability of choosing action 1 in state 0, b the probability of choosing 1 in state 1 if in state 0 action 1 is chosen, and c is the probability of choosing action 1 in state 1 if in state 0 action 2 is chosen. Let $v(0, \theta, (a, b, c))$ be the expected discounted total return in state 0 for

the strategy given by a, b and c, if θ is the true parameter value. It is

straightforward to verify that

 θ	v(0,θ,(a,b,c))		
 0.1	a(0.73b - 6.57c + 83.32) + 6.57c + 22.14		
0.9	a(-2.07b + 0.23c - 65.48) - 0.23c + 98.94		

Note that, for fixed b and c, the maximum over a ϵ [0,1] of min v(0, θ , (a,b,c)) is attained for $\theta \epsilon \theta$

$$a = \frac{76.8 - 6.8c}{148.8 - 6.8c + 2.8b}$$

and

$$f(b,c) := \max \min v(0,\theta,(a,b,c)) = \frac{76.8 - 6.8c}{148.8 - 6.8c + 2.8b}(0.73b - 6.57c + 83.32) + 6.57c + 6.57c + 22.14$$

Further note that f(b,c) attains its maximum over $(b,c) \in [0,1]^2$ in a boundary point. It turns out that the optimal pair (b,c) is (0,1), and $f(0,1) = 65.54 \dots$

Hence the optimal strategy is: a = 0.49..., b = 0 and c = 1. It is easy to verify that all nonrandomized strategies are less good than this strategy.

Next we consider the minimax strategy for the situation that a second decision maker starts in state 1. Suppose this second decision maker has the same information concerning the unknown parameter as the first decision maker, i.e. he performs a Bernoulli trial with parameter θ . Hence he works with the conditional distribution, given this observation. However, since this experiment is independent of the process, it does not change the transition law for the second decision maker.

(Note that if the parameter set would be $\{0,1\}$ then the observation of the experiment would reduce the parameter set to a singleton.)

The strategies for the second decision maker are characterized by the probability d of choosing action 1. Note that

θ	v(1,0,(d))	
0.1	37.8d + 23.2(1 - d)	
0.9	28.2d + 32.8(1 - d)	

So, if the first decision maker reaches state 1, he does not randomize but he chooses action 2 if he has chosen action 1 in state 0, otherwise he chooses action 1, and the second decision maker randomizes between the two actions with probability 1. The first decision maker acts in state 1 as if he knows the true parameter in state 1.

REFERENCES

- Aoki, M., Optimization of stochastic systems. New York etc., Academic Press (1967).
- Bauer, H., Wahrscheinlichkeitstheorie und Grundzüge der Masstheorie. Berlin, Walter de Gruyter, (1968).
- Behara, M. and Nath, P., Additive and nonadditive entropies of finite measurable partitions. Probability and information theory III. Lecture Notes in Mathematics 296, Springer Verlag (1973).
- Bellman, R., Dynamic programming. Princeton (N.J.), Princeton University Press (1957).
- Bellman, R., Adaptive control processes: a guided tour. Princeton (N.J.), Princeton University Press (1961).
- Bertsekas, D.P., Dynamic programming and stochastic control. New York etc., Academic Press (1976).
- Blackwell, D., Discounted dynamic programming. Ann. Math. Statist. <u>36</u> (1965), 226-235.
- Blackwell, D., Freedman, D.A. and Orkin, M., The optimal reward operator in dynamic programming. Ann. Prob. <u>2</u> (1974), 926-941.
- Christensen, J.P.R., Topology and Borel structure. Amsterdam etc., North-Holland/American Elsevier (1974).
- Denardo, E.V., Contraction mappings in the theory underlying dynamic programming. SIAM Rev. 9 (1967), 165-177.
- Derman, C., Denumerable state Markovian decision processes-average cost criterion. Ann. Math. Statist. 37 (1966), 1545-1554.
- Derman, C., Finite state Markovian decision processes. New York etc., Academic Press (1970).
- Derman, C. and Strauch, R.E., A note on memoryless rules for controlling sequential control processes. Ann. Math. Statist. 37 (1966), 276-278.

- Doob, J.L., Application of the theory of martingales. Colloq. Internat. du CNRS (1949), 23-27.
- Dynkin, E.B., Controlled random sequences. Theor. Probability Appl. <u>10</u> (1965), 1-14.
- Fabius, J., Asymptotic behavior of Bayes' estimates. Ann. Math. Statist. 35 (1964), 846-856.
- Fox, B.L. and Rolph, J.E., Adaptive policies for Markov renewal programs. Ann. Math. Statist. <u>1</u> (1973), 334-341.
- Freedman, D.A., On the asymptotic behavior of Bayes estimates in the discrete case. Ann. Math. Statist. <u>34</u> (1963), 1386-1403.
- Freedman, D.A., On the asymptotic behavior of Bayes estimates in the discrete case II. Ann. Math. Statist. <u>36</u> (1965), 451-456.
- Furukawa, N., Fundamental theorems in a Bayes controlled process. Bull. Math. Statist. <u>14</u> (1970), 103-110.
- Groenewegen, L.P.J., Characterization of optimal strategies in dynamic games. University of Technology Eindhoven, dissertation (1978).
- Hastings, N.A.J. and van Nunen, J.A.E.E., The action elimination algorithm for Markov decision processes. Markov decision theory; ed. by H.C. Tijms and J. Wessels. Amsterdam, Math. Centre Tracts <u>93</u> (1977), 161-178.
- Hausdorff, F., Set theory. New York, Chelsea (1957).
- van Hee, K.M., Adaptive control of specially structured Markov chains. Dynamische Optimierung. Tagungsband des Sonderforschungsbereiches 72. Bonner Mathematische Schriften <u>98</u> (1976), 99-116.
- van Hee, K.M., Approximations in Bayesian controlled Markov chains. Markov decision theory; ed. by H.C. Tijms and J. Wessels. Amsterdam, Math. Centre Tracts <u>93</u> (1977), 171-182.
- van Hee, K.M., Hordijk, A. and van der Wal, J., Successive approximations for convergent dynamic programming. Markov decision theory, ed. by H.C. Tijms and J. Wessels. Amsterdam, Math. Centre Tracts 93 (1977), 183-212.

- Hinderer, K., Foundations of non-stationary dynamic programming with discrete time parameter. Lecture Notes in Operations Research and Mathematical Economics 33, Springer Verlag (1970).
- Howard, R.A., Dynamic programming and Markov processes. Cambridge (Mass.) M.I.T. Press (1960).
- Iglehart, D.L., The dynamic inventory problem with unknown demand distribution. Management Sci. <u>10</u> (1964), 429-440.
- Kushner, H., Introduction to stochastic control. New York, Holt (1971).
- MacQueen, J., A modified dynamic programming method for Markovian decision problems.

J. Math. Anal. Appl. 14 (1966), 38-43.

- Mallows, C.L. and Robbins, H., Some problems of optimal sampling strategy. J. Math. Anal. Appl. <u>8</u> (1964), 90-103.
- Mandl, P., Estimation and control in Markov chains. Adv. Appl. Prob. <u>6</u> (1974), 40-60.
- Mandl, P., On the adaptive control of countable Markov chains. Unpublished report (1976).
- Martin, J.J., Bayesian decision problems and Markov chains. New York etc., Wiley (1967).
- Neveu, J., Mathematical foundations of the calculus of probability. San Francisco etc., Holden-day (1965).
- Neveu, J., Martingales & temps discret. Paris, Masson (1972).
- van Nunen, J.A.E.E., Contracting Markov decision processes. Amsterdam, Math. Centre Tracts <u>71</u> (1976).
- van Nunen, J.A.E.E. and Wessels, J., The generation of successive approximations for Markov decision processes by using stopping times. Markov decision theory, ed. H.C. Tijms and J. Wessels. Amsterdam, Math. Centre Tracts <u>93</u> (1977), 25-38.

Parthasarathy, K., Probability measures on metric spaces. New York etc., Academic Press (1967).

Revuz, D., Markov chains. Amsterdam etc., North-Holland/American Elsevier (1976).

Rieder, U., Bayessche dynamische Entscheidungs- und Stoppmodelle. University of Hamburg (1972).

Rieder, U., Bayesian dynamic programming. Adv. Appl. Prob. 7 (1975), 330-348.

- Rose, J.S., Markov decision processes under uncertainty-average return criterion. Unpublished report (1975).
- Ross, S.M., Arbitrary state Markovian decision processes. Ann. Math. Statist. <u>39</u> (1968), 2118-2122.
- Ross, S.M., Applied probability models with optimization applications. San Francisco etc., Holden-day (1970).
- Satia, J.K. and Lave, R.E., Markovian decision processes with uncertain transition probabilities. Operations Res. <u>21</u> (1973), 728-740.
- Scarf, H., Bayes solutions of the statistical inventory problem. Ann. Math. Statist. 30 (1959), 490-508.
- Schäl, M., Conditions for optimality in dynamic programming and for the limit of n-stage optimal policies to be optimal. Z. Wahrscheinlichkeitstheorie verw. Gebiete <u>32</u> (1975), 179-196.
- Shapley, L.S., Stochastic games. Proc. Nat. Acad. Sci. 39 (1953), 1095-1100.
- Shiryaev, A.N., On the theory of decision functions and control of an observation process with incomplete data. Trans. Third Prague Confer. on Inform. Theory etc. (1964), 657-682.
- Shiryaev, A.N., Some new results in the theory of controlled random processes. Trans. Fourth Prague Confer. on Inform. Theory etc. (1967), 131-203.

- Shreve, S.E., Dynamic programming in complete separable spaces. University of Illinois (1977).
- Strauch, R.E., Negative dynamic programming. Ann. Math. Statist. 37 (1966), 871-890.
- Sworder, D.D., Optimal adaptive systems. New York etc., Academic Press (1966).
- Taylor, H.M., Optimal replacement under additive damage and other failure models. Naval Res. Logist. Quart. <u>22</u> (1975), 1-18.
- Veinott, A.F., Optimal policy for a multi-product, dynamic, nonstationary inventory problem. Management Sci. <u>12</u> (1965), 206-222.
- Wald, A., Sequential analysis. New York, Wiley (1947).
- Waldmann, K.H., Stationäre Bayessche Entscheidungsmodelle mit Anwendungen in der Lagerhaltungstheorie. University of Technology Darmstadt, dissertation (1976).
- Wessels, J., Decision rules in Markovian decision processes with incompletely known transition probabilities. University of Technology Eindhoven, dissertation (1968).
- Wessels, J., Inventory control with unknown demand distribution: a discrete time-discrete level case. Colloquia Mathematica Societatis Janos Bolyai 7 (1971)
- Wessels, J., Inventory control with unknown demand distribution: a slowmover case. Statistica Neerlandica <u>26</u> (1972).
- Wessels, J., Stopping times and Markov programming. Trans. Seventh Prague Confer. on Inform. Theory etc. (1974), 575-585.
- Wessels, J., Markov programming and successive approximations with respect to weighted supremum norms.
 - J. Math. Anal. Appl. 58 (1977), 326-335.

- Whitt, W., Approximations of dynamic programs. Manuscript (1976) (to be published).
- Yuschkevich, A.A., Reduction of a controlled Markov model with incomplete data to a problem with complete information in the case of Borel state and control spaces. Theor. Probability Appl. <u>21</u> (1976), 153-158.
- Zacks, S., The theory of statistical inference. New York, Wiley (1971).
- Zacks, S. and Fennel, J., Distribution of adjusted stock levels under statistical adaptive control procedures of inventory systems. J.A.S.A. <u>68</u> (1973).

SAMENVATTING

In dit proefschrift wordt een onderzoek naar Markov beslissingsprocessen met een discrete tijdsvariabele en een onvolledig bekend overgangsmechanisme beschreven.

Alvorens de resultaten van dit onderzoek te bespreken wordt een schets van het bestudeerde model gegeven.

Beschouw een systeem bestaande uit een toestandsruimte X, een supplementaire toestandsruimte Y en een actieruimte A. De onbekendheid van het overgangsmechanisme wordt tot uiting gebracht door een parameter $\theta \in \theta$. Voor de eenvoud wordt in deze samenvatting verondersteld dat de verzamelingen X, Y, A en θ eindig zijn. Op tijdstip n $\in \mathbb{N}$ is de toestand van het systeem X, de supplementaire toestand Y_n en de actie A_n. Het overgangsmechanisme heeft de gedaante

$$\mathbf{P}_{\theta}[\mathbf{Y}_{n+1} = \mathbf{y}, \mathbf{X}_{n+1} = \mathbf{x}' | \mathbf{X}_{n} = \mathbf{x}, \mathbf{A}_{n} = \mathbf{a}] = \mathbf{P}(\mathbf{x}' | \mathbf{x}, \mathbf{a}, \mathbf{y}) \sum_{i \in \mathbf{I}} \mathbf{1}_{K_{i}}(\mathbf{x}, \mathbf{a}) \mathbf{p}_{i}(\mathbf{y} | \boldsymbol{\theta}_{i}) ,$$

$$\mathbf{x}, \mathbf{x}' \in \mathbf{X}, \mathbf{y} \in \mathbf{Y}, \mathbf{a} \in \mathbf{A}, \ \boldsymbol{\theta}_{i} \in \boldsymbol{\theta}_{i} \text{ en } \boldsymbol{\theta} = \prod_{i \in \mathbf{I}} \boldsymbol{\theta}_{i}.$$

Hierbij is P een overgangswaarschijnlijkheid van $X \times A \times Y$ naar X, I een index verzameling, $\{K_i, i \in I\}$ een partitie van $X \times A$ en $y \Rightarrow p_i(y|\theta_i)$ een kansdichtheid op Y voor alle $i \in I$ en $\theta_i \in \theta_i$. Hoewel dit overgangsmechanisme op het eerste gezicht nogal bijzonder lijkt zijn veel bekende modellen op deze wijze te formuleren. Bijvoorbeeld Markov beslissingsprocessen met aftelbare toestands- en actieruimten en volledig onbekende overgangswaarschijnlijkheden, voorraad modellen met onbekende vraag-verdeling en lineaire systemen met stochastische storing waarvan de verdeling onbekend is, kunnen op deze wijze geformuleerd worden.

Verder is er een opbrengst functie $r : X \times A \times Y \rightarrow \mathbb{R}$. Een strategie π is een voorschrift waarmee op ieder tijdstip $n \in \mathbb{N}$ een actie $A_n \in A$ geselecteerd wordt, slechts gebruikmakend van de historie van het proces tot tijdstip n: $X_0, A_0, Y_1, X_1, A_1, \dots, Y_n, X_n$. Iedere strategie π , starttoestand $x \in X$ en parameter $\theta \in \Theta$ bepalen een kansmaat en daarmee de verwachte opbrengst op tijdstip n: $\mathbb{E}_{x,\theta}^{\pi} [r(X_n, A_n, Y_{n+1})].$

Daar deze grootheid nog van een onbekende parameter afhangt wordt er een kansverdeling q op θ gekozen. Deze kansverdeling kan volgens de Bayesiaanse methode beschouwd worden als een a priorizverdeling en volgens de beslissingstheoretische methode als een gewichtsfunctie, die het belang weergeeft dat de bestuurder van het systeem aan de verschillende parameter waarden hecht.
Om strategieën met elkaar te vergelijken worden twee criteria gehanteerd:
(i) de totale verdisconteerde opbrengst van een strategie m bij start in x ∈ X

$$\mathbf{v}(\mathbf{x},\mathbf{q},\pi) := \sum_{n=0}^{\infty} \beta^{n} \{\sum_{\theta \in \Theta} \mathbf{q}(\theta) = \mathbf{E}_{\mathbf{x},\theta}^{\pi} [\mathbf{r}(\mathbf{x}_{n},\mathbf{A}_{n},\mathbf{y}_{n+1})] \}, \quad \beta \in [0,1)$$

(ii) de gemiddelde opbrengst van een strategie π bij start in x ϵ X:

$$\liminf_{N \to \infty} \frac{1}{N} \sum_{n=0}^{N-1} \{ \sum_{\theta \in \Theta} q(\theta) \ \text{If } _{x,\theta}^{\pi} [r(x_n, A_n, Y_{n+1})] \} ,$$

De onbekende parameter kan nu beschouwd worden als een stochastische grootheid Z met kansverdeling q. Het beslissingsproces blijkt equivalent te zijn met een Markov beslissingsproces (ook wel dynamisch programma genoemd) met een bekend kansmechanisme, een toestandsruimte $X \times W$ en een actieruimte A, waarbij W de verzameling van alle kansverdelingen op θ is. Op tijdstip n ϵ IN is de toestand van dit equivalente systeem (X_n, Q_n) waarbij X_n de toestand van het oorspronkelijke systeem is en Q_n de a posteriori verdeling van Z is, d.w.z. de voorwaardelijke verdeling van Z gegeven $(X_0, A_0, Y_1, X_1, A_1, \dots, Y_n, X_n)$. In hoofdstuk 2 wordt een formele definitie van het model gegeven. Hierbij zijn X, Y, A en θ Borel deelverzamelingen van volledige separabele metrische ruimten. Verder wordt in hoofdstuk 2 het proces van a posteriori-verdelingen bestudeerd. Het blijkt dat deze a posteriori-verdelingen met kans 1 convergeren naar de verdeling op θ die geconcentreerd is in het punt Z, indien met kans 1 iedere verzameling K,, i \in I oneindig vaak bezocht wordt door het systeem (d.w.z. $\sum_{n=0}^{\infty} 1_{K_2}(X_n, A_n) = \infty$ met kans 1 voor alle $i \in I$). Dit resultaat wordt in hoofdstuk¹4 gebruikt om te bewijzen dat bepaalde strategieën optimaal zijn en in hoofdstuk 6 om de convergentie van approximatie methoden te bewijzen.

In hoofdstuk 3 worden twee nogal technische onderwerpen behandeld. In de eerste plaats wordt hier de reeds genoemde equivalentie met een Markov beslissingsproces met bekend kansmechanisme bewezen. Vervolgens wordt een klasse van successieve-approximatie operatoren voor algemene Markov beslissingsprocessen bestudeerd. Van deze operatoren worden enige nieuwe eigenschappen afgeleid die later, in hoofdstuk 6 gebruikt worden. In hoofdstuk 4 worden eenvoudig te hanteren strategieën geconstrueerd die, onder enige recurrentie voorwaarden, optimaal zijn voor het gemiddelde opbrengsten criterium. Bij deze strategieën moet de bestuurder op ieder tijdstip n $\in \mathbb{N}$ een actie kiezen die de functie a $\rightarrow \Sigma_{\theta \in \Theta} Q_n(\theta) F(X_n, \theta, a)$ maximalizeert. Hier is $F : X \times \theta \times A \rightarrow \mathbb{R}$ een functie met de eigenschap dat een strategie die in elke toestand $x \in X$ een $a(x) \in A$ kiest zodanig dat $F(x, \theta, a(x)) \geq F(x, \theta, a)$ voor alle $a \in A$, optimaal is, als de parameter $\theta \in \theta$ bekend is. Deze strategieën worden "Bayesian equivalent rules" genoemd. In de bestudeerde gevallen is zo'n functie F steeds eenvoudig te bepalen. In hoofdstuk 5 worden voor drie speciale modellen Bayesian equivalent rules geconstrueerd die optimaal zijn met betrekking tot het totale verdisconteerde opbrengsten criterium. Een van deze modellen is het lineaire systeem met kwadratische kosten en storing met onbekende verdeling. Een ander model is een eenvoudig voorraad model waarbij de verdeling van de vraag onbekend is. Hier wordt ook een schatting gegeven van het verlies dat geleden wordt als de Bayesian equivalent rule gebruikt wordt terwijl deze niet optimaal is.

In hoofdstuk 6 worden approximaties van de criterium functie V voor het totale verdisconteerde opbrengsten criterium afgeleid. Hierbij is

$$V(\mathbf{x},\mathbf{q}) := \sup_{\pi} V(\mathbf{x},\mathbf{q},\pi), \quad \mathbf{x} \in \mathbf{X}, \mathbf{q} \in \mathbf{W}.$$

Er worden onder- en bovengrenzen voor V gegeven. En voor speciaal gestructureerde modellen worden deze gebruikt om successieve approximaties van V(x,q), voor een vaste $q \in W$, te krijgen met iedere gewenste nauwkeurigheid. De convergentie van de successieve approximaties berust mede op de convergentie van de a posteriori verdelingen.

Tenslotte worden approximaties gegeven voor $|V(x,q) - V(x,\varphi)|$ waarbij q een willekeurige a priori-verdeling is en φ een discretizering van q is, d.w.z. θ wordt opgesplitst in een eindige partitie B_1, \ldots, B_k , in elke verzameling B_j wordt een representant θ_j gekozen en dan wordt $\varphi(\{\theta_j\}) := q(B_j)$ gedefinieerd, $j = 1, \ldots, k$.

In hoofdstuk 7 worden algorithmen gegeven die gebaseerd zijn op de approximaties die in hoofdstuk 6 zijn behandeld en er wordt aan de hand van een aantal numerieke voorbeelden aangetoond dat deze approximatiemethoden efficiënt zijn. Met deze methoden kunnen veel grotere problemen behandeld worden dan met de in de literatuur bekende methoden, althans voor speciaal gestructureerde modellen.

CURRICULUM VITAE

De schrijver van dit proefschrift werd op 14 december 1946 te Den Haag geboren. Van 1959 tot 1964 bezocht hij het Dalton-lyceum te Den Haag waar het diploma H.B.S.-b behaald werd. In juni 1968 werd te Leiden het kandidaatsexamen "wis- en natuurkunde met sterrekunde" behaald en in juni 1971 (met lof) het doctoraal examen wiskunde met bijvak bedrijfseconomie. In de doctoraal fase werden wiskunde-colleges gevolgd bij prof. dr. J. Fabius, dr. A. Ollongren, prof. dr. C. Visser, prof. dr. A.C. Zaanen, prof. dr.

G. Zoutendijk en prof. dr. W.R. Van Zwet.

Van december 1969 tot juni 1971 was hij als student-assistent werkzaam bij prof. dr. W.R. van Zwet en vervolgens was hij tot december 1973 als wetenschappelijk medewerker werkzaam in de informatica bij prof. dr. A.A. Verrijn Stuart (te Leiden). Vanaf december 1973 is hij als wetenschappelijk medewerker in dienst van de Technische Hogeschool te Eindhoven en werkzaam bij prof. dr. J. Wessels.

Stellingen

bij het proefschrift Bayesian control of Markov chains

van

K.M. van Hee *

1

Beschouw een dynamisch programma in de zin van sectie 3.2 van dit proefschrift, met Borel deelverzamelingen van volledige separabele metrische ruimten als toestands- en actieruimten.

Indien voor elke starttoestand het supremum over alle zuivere Markov strategieën van de verwachte som der positieve opbrengsten eindig is, dan is het supremum over de zuivere Markov strategieën van de verwachte som der opbrengsten gelijk aan het supremum over alle strategieën. ref. [5].

2

Voor het benaderen van de optimale-waarde functie van een optimaal-stop probleem voor een Markov keten, convergeert de strategie-verbeteringsmethode van Howard soms ook in situaties waar niet voldaan is aan condities die convergentie van deze methode garanderen voor dynamische programma's in het algemeen. ref. [1], [7].

(referenties zijn na stelling 14 vermeld)

Beschouw een dynamisch programma met een aftelbare toestandsruimte X, een willekeurige actieruimte A (voorzien van een σ -algebra die de één-puntsverzamelingen omvat) en een overgangswaarschijnlijkheid P van X × A naar A. Laat F := {f | f : X → A} en laat voor f ϵ F de overgangswaarschijnlijkheid P_f van X naar X gedefinieerd zijn door P_f(x' |x) := P(x' | x, f(x)), x, x' ϵ X.

Als voor een positieve functie µ op X geldt

 $\rho := \sup_{f \in F} \limsup_{n \to \infty} \sup_{x \in X} \frac{(P_f^n \mu)(x)}{\mu(x)} < 1$

en

$$\sup_{\substack{f \in F \ x \in X}} \sup \left(\frac{(P_f \mu)(x)}{\mu(x)} < \infty \right),$$

dan is er voor iedere ε > 0 een positieve functie ν op X en een positief getal M zodanig dat voor alle x ϵ X

(i)
$$(\rho + \varepsilon) \vee (\mathbf{x}) \ge \sup_{\mathbf{f} \in \mathbf{F}} (\mathbf{P}_{\mathbf{f}} \vee) (\mathbf{x})$$

(ii)
$$\mu(\mathbf{x}) \leq \nu(\mathbf{x}) \leq M\nu(\mathbf{x})$$
.

ref. [4].

4

Voor een dynamisch programma met een aftelbare toestandsruimte X en een willekeurige actieruimte (voorzien van een σ -algebra die de één-puntsverzamelingen omvat) bestaat er een sterk excessieve functie dan en slechts dan indien er een partitie $\{X_k, k \in Z\}$ van X is en er reële getallen $\alpha > 1$, $\beta \ge 1$ zijn, zodanig dat onder elke zuivere Markov strategie het verwachte aantal bezoeken aan X_k bij start X_l ten hoogste $\beta \min\{1, \alpha^{l-k}\}$ bedraagt, voor alle $k \in Z$, $l \in Z$. (Z is de verzameling der gehele getallen.) ref. [4].

5

Beschouw het dynamische programma dat gedefinieerd is in 3.1 van dit proefschrift. Laat toe dat de opbrengstfunctie r onbegrensd is en dat de verdisconteringsfactor $\beta = 1$. Veronderstel echter dat er aan een sterke convergentievoorwaarde is voldaan, d.w.z. dat er een rij niet-dalende functies $a_n: S \neq [1,\infty) n \in \mathbb{N}$ is, zodanig dat $\lim_{n \to \infty} a_n(s) = \infty$ en $\sup_{n \to \infty} \mathbb{E}_{S}^{T} \left[\sum_{n=0}^{\infty} a_n(s) |r(S_n, A_n)| \right] < \infty$, voor alle $s \in S$, $\pi \in M$ waarbij M de verzameling der zuivere Markov strategieën is. Laat v de waardefunctie zijn, laat τ een stoptijd zijn als in 3.12 en laat U_{τ} de optimalizeringsoperator zijn gedefinieerd in 3.16. Als $\tau(\omega) \geq 1$ voor alle $\omega \in \Omega$, dan geldt

$$\lim_{n\to\infty} (U^{H}_{\tau}O)(s) = v(s), \text{ voor alle } s \in S$$

ref. [3].

6

Omdat contraherende dynamische programma's aan een sterke convergentievoorwaarde (zie stelling 5) voldoen, kan convergentie der successieve approximaties van de waardefunctie eenvoudig bewezen worden zonder gebruik te maken van de contractiestelling van Banach en bovendien voor een ruimere klasse van startfuncties. ref. [3].

7

Voor dynamische programma's met een aftelbare toestandsruimte en een willekeurige actieruimte (voorzien van een σ -algebra die de één-puntsverzamelingen omvat), die aan een sterke convergentievoorwaarde voldoen (zie stelling 5), convergeert de strategie-verbeteringsmethode van Howard en dus bestaan er voor elke starttoestand &-optimale stationaire strategieën. ref. [2], [7].

8

Laat B een M \times N-matrix zijn en $\textbf{p}_1,\ldots,\textbf{p}_N$ getallen uit het interval (0,1). Dan geldt



is niet-stijgend in n (n \in IN, n $\rightarrow \infty$), met limiet

```
\sum_{i=1}^{N} \min B(j,i) .
```

9

Uit de definities van Parthasarathy en Hinderer blijkt dat de term "standaard Borel ruimte" niet standaard is. ref. [6], [8].

10

Het is noodzakelijk dat technieken voor het simuleren van stochastische systemen opgenomen worden in het onderwijs aan hen die zich later als besliskundigen wensen te beschouwen.

11

De toepasbaarheid van wiskundige resultaten neemt toe naarmate deze in meer algemene modellen worden bewezen, terwijl de toegankelijkheid en daarmee het gebruik, afneemt.

12

Het is inconsequent dat voor het geven van onderwijs op middelbaar niveau een certificaat van vak-didactische bekwaamheid nodig is, terwijl zo'n certificaat niet vereist is voor het geven van wetenschappelijk onderwijs.

13

Dat y-wetenschappers vaak slecht met wiskundige methoden omgaan, is niet in de laatste plaats te wijten aan het feit dat zij doorgaans door vakgenoten in plaats van vaklieden in deze methoden opgeleid zijn.

14

Een democratisch stelsel dient voorzien te zijn van een grondwet die uitsluit dat door democratische besluitvorming de democratie wordt opgeheven.

Referenties.

- K.M. van Hee, The policy iteration method for the optimal stopping of a Markov chain with an application.
 in: Optimization Techniques (p. 22-36). Lecture Notes in Computer Science 41. Springer Verlag (1976).
- K.M. van Hee en J. van der Wal, Strongly convergent dynamic programming: some results.
 in: Dynamische Optimierung (p. 165-172). Bonner Mathematischer Schriften 98, Bonn (1976).
- [3] K.M. van Hee, A. Hordijk en J. van der Wal, Successive approximations for convergent dynamic programming.
 in: Markov Decision Theory (p. 183-212). Mathematical Centre-Tracts 93. Amsterdam (1977).
- [4] K.M. van Hee en J. Wessels, Markov decision processes and strongly excessive functions. Memorandum COSOR 77-11 (1977).
- [5] K.M. van Hee, Markov strategies in dynamic programming. Mathematics of Operations Research, vol. 3, no. 1 February (1978).
- [6] K. Hinderer, Foundations of non-stationary dynamic programming with discrete time parameter.
 Lecture Notes in Operations Research and Mathematical Economics, 33. Springer Verlag (1970).
- [7] R.A. Howard, Dynamic programming and Markov processes. M.I.T. Press, Cambridge (Mass.) (1960).
- [8] K.R. Parthasarathy, Probability measures on metric spaces. Academic Press, New York etc. (1967).